

MESCAL

Management of End-to-end Quality of Service Across the Internet at Large

IST-2001-37961

D3.2: Final Experimental Results: Validation and Performance Assessment of Algorithms and Protocols for Inter-domain QoS through Servicedriven Traffic Engineering

Document Identifier: MESCAL/WP3/ALGO/D3.2/final		
Deliverable Type: Report	Contractual Date: 31 August 2005	
Deliverable Nature: Public	Actual Date: 4 July 2005	

Editor:	Eleni Mykoniati, Algonet S.A.		
Authors:	FTR&D: M. Boucadair, P. Morand		
	TRT: H. Asgari, R. Egan		
	UCL: J. Griem, D. Griffin, J. Spencer		
	UniS: S. Georgoulas, K. H. Ho, M. Howarth, P. Trimintzios, N. Wang		
	Algo: P. Georgatsos, I. Liabotis, E. Mykoniati		
Abstract:	The deliverable presents the tests undertaken and the results produced for validating and assessing MESCAL's solution for inter-domain QoS delivery.		
	Experimentation was carried out through simulations and in testbeds comprised of experimental Linux-based routers. It covered functional and performance testing, cost/benefit, scalability, stability, usability tests, of the following components: q-BGP, the enhanced inter-domain QoS routing protocol, regarding its employment in providing QoS in the Internet; inter- and intra-domain traffic engineering for uni- and multicast traffic, pSLS negotiations and handling functions for producing the required TE info and c/pSLS admission control. For each of these component tests, results and the conclusions drawn are presented. The deliverable also includes a scalability analysis of the MESCAL approach.		
Keywords:	Inter-domain QoS, Experimentation, Validation, Performance Assessment, Results.		

Copyright © MESCAL Consortium:

FTR&D	Co-ordinator	France
TRT	Principal Contractor	UK
UCL	Principal Contractor	UK
UniS	Principal Contractor	UK
Algo	Principal Contractor	Greece
	FTR&D TRT UCL UniS Algo	FTR&DCo-ordinatorTRTPrincipal ContractorUCLPrincipal ContractorUnisPrincipal ContractorAlgoPrincipal Contractor



Project funded by the European Community under the "Information Society Technology" Programme (1998-2002)

Executive Summary

This deliverable presents the experimentation work undertaken for validating and assessing the performance of the functionality pertaining to the MESCAL solution for inter-domain QoS delivery.

The MESCAL solution relies on interactions between adjacent providers at the service layer, for establishing agreements for QoS traffic exchange, pSLSs, and at the network (IP) layer for finding, determining and maintaining suitable inter-domain QoS routes. The commonly used inter-domain routing protocol BGP has been enhanced to convey QoS related information. In addition, the solution specified the required service management and traffic engineering functionality per provider domain. Three technical options of the general MESCAL solution have been specified to meet the QoS requirements of different service types. Solution option 1 provides for loose (qualitative) QoS guarantees across the Internet, while solution option 2 delivers statistical guarantees (i.e. not per flow but per flow aggregate) for quantitative QoS targets, in addition to qualitative QoS guarantees. Solution option 3 is suitable for services requiring hard QoS guarantees. The technical aspects and details of the MESCAL solution are included in deliverables [D1.1], [D1.2] and [D1.3].

Experimentation was carried out in either physical testbeds, comprised of Linux-based routers, or simulated networks and covered functional validation and performance assessment aspects in terms of cost/benefit, scalability and stability assessment.

The deliverable presents the tests, results and conclusions drawn regarding the following functional aspects of the MESCAL inter-domain QoS delivery solution, including:

- Behaviour of the specified q-BGP protocol and associated route selection process;
- Off-line inter-domain TE algorithms and their coupling with intra-domain TE;
- Off-line intra-domain IP-based QoS TE algorithms;
- Off-line intra- and inter-domain multicast TE algorithm;
- SLS Mgt functions -pSLS modelling, negotiation, translation and request handling- and admission control on c/pSLS invocations;
- 'In-router' deployment and operation of q-BGP and delivery of inter-domain QoS with loose guarantees according to the specified solution (option 1) in a realistic network set-up (testbed);
- Delivery of inter-domain QoS with hard QoS guarantees through the establishment of interdomain LSPs (MPLS tunnels) based on the concept of PCSs (Path Computation Systems) according to the specified solution (option3) in a realistic network set-up (testbed).

Furthermore, the deliverable includes a scalability analysis of the overall MESCAL solution approach. The scalability analysis addressed a number of aspects of the MESCAL solution, including:

- The extent and complexity of message flow/processing for pSLS set-up during the negotiation phase. In this respect, a comparison between the CADENUS model, see [CADENUS], and the MESCAL model in serving service requests is made;
- An analysis of the MESCAL QoS peering model in terms of the number of pSLSs required for large networks;
- An analysis of the number and granularity of QoS Classes required for the MESCAL solution options.

The results of the tests undertaken prove the validity and feasibility of the MESCAL inter-domain QoS delivery solution and the proposed algorithms/schemes and protocols. They show that better performing routes for carrying QoS traffic can be established through the proposed approach (q-BGP exchanges, following pSLS establishment), compared to using standard BGP. The specified traffic engineering and service handling functions performed well, giving favourable results compared to adhoc configurations/solutions or alternative schemes.

Table of Contents

E	XECUTIVE SUMMARY	2
Т	ABLE OF CONTENTS	3
L	IST OF FIGURES	7
L	IST OF TABLES	9
1	INTRODUCTION	. 10
	1.1 Background.	. 10
	1.2 Scope of the Deliverable	. 10
•		. 10
2	EXPERIMENTATION FRAMEWORK	. 12
	2.1 Experimentation Activities	. 12
	2.1.1 Experimentation Environment	. 12
	2.1.2 Experimentation Categories	. 13
	2.2 Experimentation Structure	. 13
3	OFFLINE TRAFFIC ENGINEERING TESTS AND RESULTS	. 15
	3.1 Inter-domain Traffic Engineering Tests	. 15
	3.1.1 Introduction	. 15
	3.1.2 Genetic Algorithm for Decoupled Inter-domain TE	. 15
	3.1.2.1 Overview	15
	3.1.2.2 Experiment Setup and Test Description	15
	3.1.2.5 Test Results	17
	3.1.3 Heuristic Algorithm for Integrated Inter-/Intra-domain TE	. 23
	3.1.3.1 Overview	23
	3.1.3.2 Experiment Setup and Test Description	23
	3.1.3.3 Test Results	24
	3.1.3.4 Conclusions	27
	3.2 Online frame Engineering Interactions	. 27
	3.2.1 Overview	. 27
	3.2.1.2 Performance Metrics	27
	3.2.2 Experiment Setup and Test Description	. 28
	3.2.3 Test Results	. 28
	3.2.4 Conclusions	. 31
	3.3 Intra-domain Traffic Engineering Tests	. 31
	3.3.1 Overview	. 31
	3.3.2 Experiment setup and lest description	. 52
	3 3 3 1 Algorithm Performance and Optimisation	
	3.3.3.2 Algorithm Efficiency and Optimisation	39
	3.3.4 Conclusions	. 42
	3.4 Multicast Traffic Engineering Tests	. 44
	3.3.1 Offline Dimensioned Test	. 44
	3.4.1.1 Uverview	44 11
	3.4.1.3 Test Results	44
	3.4.1.4 Conclusions	48
	3.4.2 Real-Time Test	. 49
	3.4.2.1 Overview	49
	3.4.2.2 Experiment Setup and Test Description	49
	3.4.2.4 Conclusions	55
		-

4	DYN	AMIC TRAFFIC ENGINEERING TESTS AND RESULTS	. 56
	4.1 c	-BGP Simulation Tests	56
	4.1.1	Simulation Scenarios	56
	4.1.2	<i>a-BGP Policies under test</i>	57
	4.1.	2.1 QoS NLRI QoS Attributes	57
	4.1.	2.2 Route Selection Policies	57
	4.1.3	Experimental overview	. 58
	4.1.4	Experimental results: efficacy	. 58
	4.1.5	Experimental results: comparison of q-BGP selection policies	64
	4.1.6	Experimental results: scalability	67
	4.1.7	Experimental results: stability	69
	4.1.8	Conclusions	69
	4.2 I	Data Plane Testbed Tests	70
	4.2.1	Overview	70
	4.2.2	Experiment Setup and Test Description	70
	4.2.3	Test Results	70
	4.2.4	Conclusions	. 71
	4.3 ¢	-BGP Testbed Tests	72
	4.3.1	Overview	72
	4.3.2	Experiment Setup and Test Description	72
	4.3.3	Test Results	72
	4.3.4	Conclusions	75
	4.4 J	PCS Testbed Tests	75
	4.4.1	Overview	75
	4.4.2	Experiment setup and test description	75
	4.4.3	Test Results	75
	4.4.4	Conclusions	76
3	5.1 F	SLS Ordering Tests	• 77 77 78
	5.1.2	Experiment Setup and Test Description	. 79
	5.1.3	Test Results	80
	5.1.4	Conclusions	84
	5.2 \$	SLS Order Handling Tests	85
	5.2.1	Objectives	85
	5.2.2	Controlled and Uncontrolled Variables	85
	5.2.3	Experimentation Environment	87
	5.2.4	Test Campaigns and Results	87
	5.2.5	Conclusions	88
	5.3 \$	SLS Invocation Handling Tests	88
	5.3.1	Intra-domain cSLS	88
	5.3.	1.1 Overview	88
	5.3.	1.2 Experiment Setup and Test Description	90
	5.3. 5.3	1.5 Lest Results	101
	532	Inter domain cSLS	101
	53	21 Overview	101
	5.3.	2.2 Experiment Setup and Test Description	. 102
	5.3.	2.3 Test Results	. 103
	5.3.	2.4 Conclusions	. 106
6	SYST	TEM-LEVEL SCALABILITY ANALYSIS	107
	6.1 (Comparison of CADENUS & MESCAL Scalability	107
	6.2 \$	Scalability of Inter-Provider Peering Models	111
	6.3	The Extent of pSLS Set-up	112
	6.4 1	Number and Granularity of QCs	114
	6.5 \$	Summary	115
-	CON	CL LICIONG	117
1	CUN		110

7.	1.1 Overview			
7.	7.2 Implementation of the MESCAL Solution			
7.	7.3 Scalability of the MESCAL Solution			
7.	.4 q-BGP			
7.	.5 Off-line TE			
7.	c/pSLS Management			
•		100		
8	REFERENCES			
API	PENDIX A			
9	TESTBED CONFIGURATION			
0	1 Introduction	100		
9.	Autonomous system tonology			
9.	2 Autonomous system topology			
9.	0.2.1 Hawkuma components			
	9.5.1 Hardware components			
	9.3.1.1 FCS			
	032 Software components	127		
	9 3 2 1 Operating system	128		
	9.3.2.2 Software information			
9	4 Configuration for phase 1	128		
	9.4.1 User' accounts			
	9.4.2 Remote connection			
	943 Internet access	128		
	944 Firewall rules	128		
	9.4.5 Time synchronisation			
	946 Printer	129		
	947 AS identifiers	129		
	948 LANs	129		
	949 Customer addresses	129		
	9.4.10 Network addresses announced by each AS	130		
	9.4.11 Routing configuration	131		
	94111 e-hgn	133		
	9.4.11.2 i-bgp			
	9.4.11.3 Networks			
	9.4.11.4 Static routes			
	9.4.11.5 Prefix list			
	9.4.11.6 Fast link failover detection			
	9.4.11.7 BGP timers			
	9.4.11.8 Koule selection process			
	9.4.12 Local QoS class DSCP values			
	9.4.15 Inter-aomain Meta-Qos-classes DSCP values			
	9.4.14 Dunuwiun intestious per Meia-Quo-cluss			
	9.4.15 Maximum banawiath per local-QoS-class			
	9.4.10 Dijjserv-reialea conjiguration			
	9.4.10.1 qsa			
	9.4.16.3 gse	138		
	9.4.16.4 gsdel			
	9.4.16.5 qsi-eth1			
	9.4.16.6 qsHTB-eth1			
	9.4.17 Backup			
	9.4.18 Logs			
	9.4.19 Check the sanity of the test bed			
	9.4.20 Configuration scripts			
9.	.5 Specific Configuration for phase 2			
9.	9.6 Specific Configuration for phase 3			
A D1	DENDIV B	1 40		
Arl	I ЦЛИІА D			
10		1.40		
10	DETAILED TESTBED VALIDATION TESTS			

10.1 Phase	e 1	
10.1.1	TB P1 FUNCT/ROUT	
10.1.2	TB P1 FUNCT/DSSW	
10.1.3	TB P1 FUNCT/SHAP	
10.1.4	TB_P1_FUNCT/POLI	
10.1.5	TB_P1_FUNCT/BWMA	
10.2 Phase	e 2	
10.2.1	TB_P2_FUNCT/CMES	
10.2.2	TB_P2_FUNCT/DSCP	
10.2.3	TB_P2_FUNCT/QCMP	
10.2.4	TB_P2_FUNCT/RSEL	
10.2.5	TB P2 FUNCT/INT	
10.3 Phase	e 3	
10.3.1	TB_P3_FUNCT/CMES	
10.3.1.1	Reminder	
10.3.1.2	pSLS agreement	
10.3.2	TB_P3_FUNCT/QAGG	389
10.3.3	TB_P3_FUNCT/RESAV	396

List of Figures

Figure 1: Verv Small Topology for functional tests	15
Figure 2: Simulation network topology for benefit / cost performance tests	16
Figure 3: OC Mapping	16
Figure 4: \tilde{V} alidation of full-scale tests (single e-OC)	18
Figure 5: pSLS cost plus Intra-domain TE cost $(\Omega + \Phi)$	19
Figure 6: Comparison of pSLS utilisation in random and genetic algorithms (pSLS cost Ω and Intra-TE cost Φ	Φ
only)	19
Figure 7. Link utilisations (nSLS cost Q and Intra-TE cost Φ only)	19
Figure 8: pSLS cost Intra-domain TE and Inter-domain link utilisation costs ($O+\Phi+\Theta$)	20
Figure 9: Link utilisations (hased on $O+\Phi+\Theta$)	20
Figure 10: Impact of Genetic Algorithm parameters; variation of n	20
Figure 10. Impact of Genetic Algorithm parameters, variation of p_c	$\frac{21}{24}$
Figure 17: Randwidth consumption difference between Greedy-cost and Greedy-nenalty heuristics	25
Figure 12: Bandwidth accentance ratio for Greedy-nenalty heuristic	26
Figure 13: Banawhan acceptance range for Greedy penany neurisite instruments for Greedy-penalty heuristic	26
Figure 15: Evaluation of inter-domain cost	20
Figure 16: Evaluation of intra-domain cost	29
Figure 13: Evaluation of total bandwidth consumption	30
Figure 18: Evaluation of maximum inter-domain link utilization	30
Figure 19: Evaluation of maximum inter-domain link utilisation	31
Figure 20: 10 Node Test Network	32
Figure 20, 10 House rest retirion without on a 100 node network after 500 iterations	34
Figure 22: routing plane effectiveness on a 10 node network 500 iterations	35
Figure 23: Effect of Routing Planes on Utilisation for a 50 Node Network, 1300 demands	36
Figure 24: Average hon count for 50 node 100 link network, 500 iterations, 1300, 3000 demands	37
Figure 25: average utilisation for bandwidth constrained class. 1300 demands. 500 iterations	38
Figure 27: Convergence Efficiency for the 50 node topology. 1600 demands, 5 routing planes	40
Figure 28: Algorithm efficiency measured on utilisation StDev. 60% average utilisation	41
Figure 29: Convergence Time 50 Node Topology	42
Figure 30: Total bandwidth consumption vs. Max D	45
Figure 31: Overloaded link rate vs. Max D_{a}	46
g	16
Figure 32: MLOR vs. Max D_g	40
Figure 33: GA Success rate vs. MLOR _{SPH}	47
Figure 34: Total bandwidth consumption vs. Max D_g	48
Figure 35: Highest inter-domain link utilization vs. $D_{ m g}$	48
Figure 36: ns-2 based simulation topology	50
Figure 37: Real-time performance in average network load (Max $D_g = 3000, \omega = 1$)	51
Figure 38: Real-time performance in maximum link utilisation (Max D_g =3000, ω =1)	51
Figure 39: Real-time performance in average network load (Max D_g =6000, ω =1)	52
Figure 40: Real-time performance in maximum link utilisation (Max D_g =6000, ω =1)	52
Figure 41: Join block rate vs. invocation ratio ω	53
Figure 42: Network load vs. invocation ratio ω	53
Figure 43: Transmission ratio of 2 groups	54
Figure 44: Simultaneous l-QC joins	55
Figure 45: Dynamic l-QC upgrading	55
Figure 45 Mean delivered bandwidth fraction (delivered/offered) for a range of pBW under the BWOA-only	
policy	59
Figure 46 Mean pSLS utilisation for a range of pBW equivalence margins for the BWQA-only policy	60
Figure 47 Mean delivered delay for various BW QA equivalence margins for the BWQA-only policy	61

Figure 49 The mean delivered bandwidth fraction over a range of over-provisioning coefficients for the varia a-BGP policies	ous 63
Figure 50 Mean delivered delay for a select range of policies against the over-provisioning co-efficient	64
Figure 51 Effect of a-RGP selection policy on delivered delay and bandwidth	
Figure 57 O-BGP scalability: mean one way delay versus number of ASs	67
Figure 52 \mathcal{G} DGP scalability: number of a-RGP messages sent from initialisation until it settles in a stable s	tate
with a full mesh of demands applied	68
Figure 55: nSLS Ordering Experimentation Environment	00
Figure 56 Evolution of ontimality over negotiation rounds	, 0
Figure 57 Evolution of processing time over negotiation rounds	
Figure 58 Rate of decrease of the confirmed cost	01
Figure 50 Number of negotiation rounds for successful conclusion	02
Figure 60 Processing time of the negotiation logic	05
Figure 61. SLS Order Handling Experimentation Fourienment	07
Figure 67: Simulation topology	
Figure 63: Incurred PIR for VoIP sources and target 1-OC PIR 0.01	
Figure 64: Achieved 1-OC utilization for VoIP sources and target 1-OC PLR 0.01	
Figure 65: Incurred blocking for VoIP sources and target I-OC PLR 0.01	
Figure 66: Incurred PIR for VoIP sources and target 1-OC PIR 0.001	
Figure 67: Achieved 1-OC utilization for VoIP sources and target 1-OC PI R 0.001	
Figure 68: Incurred blocking for VoIP sources and target I-OC PLR 0.001	
Figure 60: Incurred PIR for Videoconference sources and target LOC PIR 0.01	
Figure 70: Achieved 1-OC utilization for Videoconference sources and target 1-OC PIR 0.01	24
Figure 71: Incurred blocking for Videoconference sources and target LOC PLR 0.01	95
Figure 71: Incurred PIR for Videoconference sources and target 1 OC PIR 0.001	75
Figure 72: Incurrent LK for Vineoconference sources and target 1-QC PLR 0.001	90
Figure 75. Achieven 1-QC unitzation for videoconference sources and target LOC PLR 0.001	90
Figure 74. Incurred Diocking for viceoconference sources and target 1-90 TER 0.001)/
Figure 75. Incurrent LK for mixed von und vineoconference sources and unger i-QC I LK 0.01	97
Figure 70. Achieved 1-QC utilization for mixed VoII and Videoconference sources for target LOC PLR 0.01 Figure 77: Incurred blocking for mixed VoIP and Videoconference sources for target LOC PLR 0.01	90
Figure 77. Incurred Dioching for mixed VoID and Videoconference sources and target 1 OC PLP 0.001	90
Figure 70: Achieved LOC utilization for mixed VoIP and Videoconference sources for target LOC PLP 0.00	99
Figure 79. Achieved 1-QC utilization for mixed VoII and Videoconference sources for target 1 QC I LK 0.00 Figure 80: Incurred blocking for mixed VoIP and Videoconference sources for target 1 QC PLP 0.001	1 99
Figure 80. Incurred blocking for mixed von and videoconjerence sources for larger 1-QC 1 LK 0.001	100
Figure 81: Average cSUS blocking rate	100
Figure 82: Simulation topology	101
Figure 65. Simulation topology	102
Figure 64. Total incurred FLK	104
Figure 65. Inter-aomain link utilization	104
Figure 80. Incurred Diocking	104
Figure 87: Utilization comparison for the inter-aomain link	105
Figure 88: Utilization comparison for the inter-aomain link	105
Figure 89: The CADENUS architecture as a queuing network (from CAD-D8)	109
Figure 90: Message flow auring service negotiation phase (from CAD-D8).	109
Figure 91: MESCAL model in SLS negotiation.	111
Figure 92. Star topology for connectivity.	115
rigure 95: Inree-Iter Internet model pSLS-based agreements.	115
Figure 94: Ine trend of pSLS set-up in each peering model	114
Figure 95: FIK&D MESCAL testbed: hierarchical view	124
Figure 90: FIK&D MESCAL testbed: detailed architecture	123
Figure 9/: FIK&D MESCAL testbed: Network interfaces schema	126

List of Tables

Table 1: Experimentation Activities	12
Table 2: Scalability: offline Inter-domain TE runtime as function of egress link utilisation	22
Table 3: Comparison of approaches (overall egress utilisation=18%, single e-QC validation scenario)	23
Table 4: Topologies used for Simulations	33
Table 5: Demands used for Simulations	33
Table 6: Running time vs. topology size (100 groups)	47
Table 7: Running time vs. number of groups (100 nodes)	47
Table 8 Mean delivered BW fraction and delivered end-to-end delay for the MCID-only	59
Table 9 Convergence time versus q-BGP selection policy	69
Table 10: Phase 1 Test Suites	70
Table 11: Phase 1 Tests results	71
Table 12: Phase 2 Validation Test Suites	72
Table 13: Phase 2 Validation Tests results	74
Table 14: Phase 3 Validation Test Suites	75
Table 15: Phase 3 Validation Tests results	76
Table 16: pSLS Ordering Performance Metrics	77
Table 17: pSLS Ordering Variables	77
Table 18: Test Configurations	79
Table 19: SLS Order Handling Controlled Variables	85
Table 20: SLS Order Handling Uncontrolled Variables	86
Table 21: Traffic Matrices Size Test Configuration Options	86
Table 22: SLS Order Handling Test Suites	87
Table 23: SLS Order Handling Tests	88
Table 24: PC characteristics	127
Table 25: Software information	128
Table 26: AS numbers	129
Table 27: Administrative network addressing	129
Table 28: Customers IP address realms	130
Table 29: Customers IP address realms	131
Table 30: 1-QC DSCP values	135
Table 31: Inter-domain meta-QoS-class DSCP values	136
Table 32: Bandwidth threshold per meta-QoS-class and per pSLS	137
Table 33: Bandwidth threshold per local-QoS-class	137
Table 34: List of useful scripts	146
Table 35: PCE locations	147
Table 36 - Bandwidth Threshold per meta-QoS-class	193
Table 37: Local QoS Class Characteristics	352
Table 38: Local QoS Class Characteristics	397
Table 39: Maximum bandwidth allowed for MC1	397

1 INTRODUCTION

1.1 Background

MESCAL addresses the problem of IP QoS-based service delivery across different provider domains. MESCAL adopts a hop-by-hop, cascaded model for the interactions between providers, at the service and network (IP) layers. Interactions at the service layer aim at the establishment of agreements for QoS traffic exchange, pSLSs in MESCAL terminology, to allow providers to expand the topological scope of their offered QoS-based services beyond the boundaries of their domains. Interactions at the IP layer are required to enable providers to find, determine and maintain suitable QoS routes for forwarding traffic in the Internet. In addition to appropriate protocols for supporting these interactions, MESCAL has specified the required service management and traffic engineering functionalities per provider domain to the end of effectively supporting these interactions, while optimising the utilisation of the network resources.

Driven by the different levels of QoS guarantees on packet transfer performance and bandwidth that could be provided to services – loose, statistical and hard contractual QoS guarantees – three corresponding technical solution options have been specified. As such, each solution option suits the needs of a different service type, targeting different customer/user segments and requiring different levels of operational complexity and scalability. Solution option 3 is suitable for services requiring hard QoS guarantees but with the inherent limitation that it cannot scale to the mass market (size of the Internet). Following the aggregate philosophy of DiffServ networks, solution option 1 has been designed to provide for loose, qualitative QoS guarantees across the Internet, while solution option 2 delivers statistical guarantees (i.e. not per flow but per flow aggregate) for either quantitative or qualitative QoS targets. The technical targets, aspects and constraints of the three MESCAL solution options have been presented in [D1.1], while suitable protocols and algorithms are described in [D1.2] and [D1.3].

Technical work in the MESCAL project is split over 3 work packages (WPs), and follows a phased approach: a theoretical phase followed by an experimentation-driven design and implementation phase and subsequently by an experimentation and dissemination phase. WP1 – Functional Architecture and Algorithms – specifies the inter-domain solution, per-domain architecture and related protocols and algorithms. WP2 – System Design and Implementation – develops aspects of the specified functionality subject to experimentation and required testing components. WP3 – Integration, Validation and Experimentation – sets up the experimentation infrastructure, testbeds and simulators, and conducts experiments with the purpose to validate and assert on performance of the specified functionality.

1.2 Scope of the Deliverable

The deliverable presents the tests, results and conclusions drawn regarding the validity and feasibility of the proposed inter-domain QoS delivery solution and its specified functional aspects; q-BGP, intraand inter-domain traffic engineering and pSLS-aware service handling functions.

1.3 Organisation of the Deliverable

The rest of this document is structured as follows:

Chapter 2 presents the overall experimentation approach of the project.

Chapter 3 presents tests and results regarding the specified off-line TE algorithms; intra- and interdomain traffic engineering for uni- and multi-cast traffic. As inter- and intra-domain traffic engineering inter-depend, tests regarding alternative ways for their coupling are also included.

Chapter 4 focuses on the specified protocol, q-BGP, for inter-domain QoS routing. Simulation-based tests and results are presented to assess q-BGP behaviour and acquire insight into intrinsic aspects of its operation in Internet-like topologies. It also presents testbed-based tests to verify the validity of the

q-BGP implementation and operation in a realistic network environment to realise the proposed interdomain QoS delivery solution. Furthermore, it includes testbed results regarding the computation of QoS-constrained paths using the PCS-based approach.

Chapter 5 presents tests and results regarding the specified c/pSLS-aware service handling functions. It focuses on pSLS negotiations with emphasis on automated client-side logic and on cSLS admission control taking into account inter-domain considerations.

Chapter 6 analyses the proposed solution for providing QoS in the Internet from a scalability perspective.

Chapter 7 summarises the conclusions drawn from the undertaken tests.

Appendix A outlines the testbed topology and set-up used in the tests and appendix B presents the detailed validation tests undertaken in the testbed and the results yielded.

2 EXPERIMENTATION FRAMEWORK

2.1 Experimentation Activities

Experimentation is an essential aspect of MESCAL work to verify and validate the overall project objectives. Table 1 depicts the related activities undertaken by the project, in terms of the:

- Functional aspect under test (with reference to the functional architecture in [D1.1] and algorithm/protocol specifications in [D1.2]).
- Type of the environment where experimentation will be undertaken.
- Category of the experiments to be carried out.

Component/ Sub-system	Environment	Reference Section	Functional	Benefit/Cost	Scalability	Stability	Usability
Offline Inter-domain TE	Simulation	3.1	х	х	х		
Offline Intra-domain TE	Simulation	3.2.4	х	х	х	х	
Offline Intra-domain Multicast TE	Simulation	3.4	х	х	x	х	
Data Plane	Testbed	4.2	х				
Dynamic Inter-domain TE q-BGP	Simulation	4.1		х	х	х	
	Testbed	4.2.4	х	х		х	х
Dynamic Inter-domain TE PCS	Testbed	4.3.4	х	х			х
pSLS Ordering	Simulation	5.1	х	х			
SLS Order Handling	Simulation	5.2	х				
SLS Invocation Handling	Simulation	5.3	х	х		х	

Table 1: Experimentation Activities

2.1.1 Experimentation Environment

Experimentation activities were carried out both in testbed and simulated network environments, as appropriate to the aspect under test and the experimentation objectives. Specifically, experiments were undertaken:

- In a testbed comprised of Linux-based routers that enabled the incorporation of the specified interdomain QoS routing protocol, q-BGP. The project testbed is provided by FTR&D and is located in Caen, France.
- In simulators, which, depending on the aspect of the network environment they simulate, can be distinguished into:
 - Dynamic network operation simulation engines, simulating the dynamics of network behaviour at a level of abstraction appropriate to the experiment e.g. at a packet, flow or protocol or control-plane activity levels.
 - Static network environment simulation tools, simulating the static aspects of the network environment i.e. the context in which the network is to operate; such aspects include network topology, number of supported QoS-classes, established service agreements, aggregate QoS traffic demands and QoS traffic generation patterns.

Three incremental phases of testbed-based experimentation have been identified, as discussed below:

- The objective of the first phase was to deploy an operational testbed including several ASs exchanging BGP-based inter-domain routing information between them and exercise the notion of meta-QoS-classes within separate autonomous systems.
- The objectives of the second phase were to deploy q-BGP and the associated route selection algorithm, as specified by MESCAL, thus creating a prototype of the solution option 1 (loose end-to-end guarantees on multiple meta-QoS-class planes) and verify its operation.
- The objectives of the third phase were to deploy PCSs and their communication protocol on top of set-up of phase 2 and validate the machinery for computing QoS paths across domains.

2.1.2 Experimentation Categories

As for their objectives, experimentation activities fall under the following commonly recognised categories:

- Functional validation experiments, aiming at assessing feasibility of implementation and validity of specifications. Not all functional validation tests and results carried out are reported in this deliverable for reasons of document length.
- Performance assessment experiments, aiming at assessing the behaviour of the aspect under test in a variety of network operation and environment set-ups and conditions. Behaviour is assessed in terms of scalability, stability, sensitivity and yielded benefits/incurred cost. Specifically:
 - Benefit/Cost assessment experiments aim at assessing the benefits/costs that the aspect under test yields/incurs in network performance, as measured through specific metrics in a representative set of network and traffic cases.
 - Scalability assessment experiments aim at calculating and verifying the resource requirements and/or computational performance of the aspect under test as a function of various uncontrollable variables, to see if it can be used in a large scale deployment.
 - Stability assessment tests verify that the aspect under test, given its specified dynamics/responsiveness, is operating in a way that drives the network to a stable state of operation, in a representative set of network and traffic cases.
 - Usability tests demonstrate that the aspect under test can operate as expected (according to its functional objectives) in terms of policy-based and/or tuning parameters upon which it may depend.

Obviously, experimentation objectives are restricted by the capabilities of the experimentation environment. As such, performance assessment experiments were primarily performed in a simulated network environment, static or dynamic, while functional validity experiments fit better in a testbed environment.

2.2 Experimentation Structure

The identified experimentation activities were specified in a clear and concise manner using a common structure/template, along the following headings:

- *Objectives*: An answer to the question "What do we want to test?". The aspects under test (specified algorithm, protocol, mechanism) and the particular goals of experimentation are outlined. Specifically, the broad experimentation categories of functional validity and assessment of benefit/cost, scalability, stability and usability are qualified in terms of concrete objectives as appropriate to the functional aspect under test.
- Performance Metrics: The metrics inherent to the particular functional aspect under test that quantify the experimentation objectives such as processing time, overhead, throughput, size of etc. are

described. How these metrics can be obtained, e.g. through probes in the entity under test or through test tools, is also discussed where appropriate.

- *Controlled Variables*: The configuration parameters of the aspect under test. The defined performance metrics will be calculated as a function of these configuration parameters.
- *Uncontrolled Variables:* The parameters of the external environment where the aspect under test is to operate are defined affecting its behaviour and/or its performance. Such parameters are network topology, volume and symmetry of traffic, number of peers, contracts etc. Generators or models for creating a realistic and representative set of their values are described where appropriate.
- *Experimentation Environment*: The platform and the set-up upon which the envisaged experimentation is to be carried out are described in terms of: components of the functional architecture, experimentation platform and required test tools, their capabilities and interactions.
- *Test Campaigns:* The tests to be carried out in achieving the specified objectives. Each of the tests aims at verifying/assessing a particular aspect of the behaviour/performance of the functional aspect under test (quantified by appropriate performance metrics) in a variety of test cases (quantified by appropriate combinations of uncontrolled variables) as a function of its configuration parameters (quantified by appropriate controlled variables). Tests are aggregated in test suites according to the general category they fall in.

3 OFFLINE TRAFFIC ENGINEERING TESTS AND RESULTS

3.1 Inter-domain Traffic Engineering Tests

3.1.1 Introduction

In this section we describe results from two Inter-domain TE test groups:

- Genetic Algorithm, implementing both delay and bandwidth as QoS metrics. This algorithm is decoupled from any intra-domain TE algorithms. In the presented results, the algorithm is compared with random assignment and brute force approaches;
- Heuristic Algorithms, implementing bandwidth only as a QoS metric, and including intradomain route optimisation. This set of algorithms implements inter- and intra-domain TE in an integrated approach.

3.1.2 Genetic Algorithm for Decoupled Inter-domain TE

3.1.2.1 Overview

This Section describes the results of the Inter-domain traffic engineering tests specified in section 5 of [D3.1].

3.1.2.2 Experiment Set-up and Test Description

The functional tests were conducted using a Very Small Network Topology (Figure 1). This is a degenerate case where there is no intra-domain TE and hence the testing of the Inter-domain TE functions is separate from the behaviour of any Inter-domain TE software.





The network topology for the simulations is shown in Figure 2, and focuses on the inter-domain connectivity. We assume a moderate sized AS with 20 adjacent ASs. The AS under test supports two I-QCs (25ms and 50 ms delays), and as a result of its service planning wishes to offer two e-QCs for its inter-domain flows (100ms and 175ms). Each adjacent AS is connected to the AS under test by either 1 or 2 links, giving a total of 27 inter-domain links, each of whose link bandwidth is set in the range 150-300 units. A number of destination prefixes are reachable through each adjacent AS (there may be other ASs en-route to the final destination prefix. But these are not relevant to our model). Each AS is able to reach between 30 and 60 of the prefixes. This reflects the observation that a small number of destination prefixes are responsible for a large fraction of an AS' outbound traffic volume [Feam03]. Although in reality the destination prefixes will in general overlap each other, for simplicity here we assume they are disjoint.



Figure 2: Simulation network topology for benefit / cost performance tests

Each adjacent AS is assumed to support a subset of three downstream o-QC delays. For simplicity, the set of supported delays is identical in each adjacent AS, being set to any of 50, 75 and 125ms (Figure 3). QoS advertisements for each link are generated based on a random combination of downstream o-QCs and random pSLS costs; for the QoS advertisements announced by any individual adjacent AS, the cost of a higher QoS class (i.e. lower delay) is set higher than the cost of a lower QoS class. Each pSLS has a bandwidth in the range 0 to 300, and the pSLS cost is set to a value between 1 and 10 per unit bandwidth. This results in overbooked pSLSs that support a total bandwidth that is 1.9 times the inter-domain link capacity. In the evaluation described here, each QoS advertisement is assumed to have resulted in the establishment of a pSLS, resulting in a total of 47 pSLSs being available to the 20 adjacent ASs. Finally the entire system is driven by a set of eTM flows randomly generated in such a way that the destination prefix in each eTM entry can be reached through one or more pSLSs supported by at least one adjacent AS. Each flow requires either a 100ms or 175ms e-QC to one of the 100 remote destination prefixes, and has a bandwidth requirement randomly selected in the range 1 to 40.



Figure 3: QC Mapping

Three cost functions were used in the Inter-domain tests, two of which represent inter-domain parameters and the third was used to represent Intra-domain costs. The cost functions were as follows:

- pSLS cost Ω, representing the cost per unit bandwidth of all pSLSs to which the domain is subscribed. We assume a subscription cost for a pSLS to be proportional to the bandwidth used.
- Inter-domain link utilisation cost function $\Theta = \sum_{i} \theta(x_i)$, where $\theta(x_i)$ is based on the Fortz and

Thorup piecewise linear cost function [For02], and reflects the desirability of minimising the inter-domain link utilisation.

• Intra-domain TE cost Φ that reflects the cost of using Intra-domain resources. In order to decouple Intra-domain and Inter-domain effects, and to implement a decoupled algorithm (i.e.

inter-domain TE and intra-domain TE algorithms are independent) a simple illustrative model was used that reflects the higher cost of using low-delay l-QCs:

$$\Phi = K \sum_{flows} \frac{bandwidth}{delay}$$

In our tests we used a variety of cost function combinations; the Inter-domain cost defined in [D1.3] is to be read as either the pSLS cost Ω or the Inter-domain link utilisation cost function Θ or their sum, as appropriate.

3.1.2.3 Test Results

3.1.2.3.1 Functional Tests

The Binding Selection and Inter-domain Resource Optimisation function blocks were designed, coded and tested.

Test Id	Purpose	Result
InterTE/Funct/BSel	Binding Selection functions	Successfully completed.
InterTE/Funct/IDRO/1	General operation of Inter-domain Resource Optimisation	Successfully completed.
InterTE/Funct/IDRO/2	Random algorithm functions	Variation 1 successfully completed.
InterTE/Funct/IDRO/3	Brute force algorithm functions	Successfully completed.
InterTE/Funct/IDRO/4	Genetic algorithm functions	Successfully completed.
InterTE/Funct/System	System functional tests: interworking between Binding Selection / Binding Activation and Inter-domain Resource Optimisation	Successfully completed.

3.1.2.3.2 Algorithm Benefit/Cost Performance Tests

The behaviour of the algorithm was validated by considering a simplified set of QCs, in which only a single downstream o-QC is employed. The delay values in this validation were a single e-QC (150ms), three 1-QCs (25, 30, 40ms), and a single downstream o-QC per adjacent AS (all o-QCs=100ms). We assume that the intra-domain links have sufficient capacity to carry all flows. We simplify the problem by assuming that all destination addresses in the eTM can be carried by the set of lowest cost pSLSs that have a total bandwidth equal to the total bandwidth in the eTM. Finally we relax the problem constraints by allowing a single eTM flow to be partially assigned to more than one pSLS. The result of these simplifications is to decouple the task of assigning an I-OC to a given eTM flow from the task of pSLS selection, and an analytically solvable approximation to the problem can be produced. In this case, all flows are optimally carried within the AS using the cheapest I-QC (i.e. the one with the highest delay), and for the Inter-domain link the flows are all assigned to the set of lowest cost pSLSs. By considering only the two cost functions pSLS cost and Intra-domain TE, we can calculate using a spreadsheet a lower bound cost, shown in Figure 4 by the solid line. This lower bound cost is better than the brute force solution. The motivation for calculating a lower bound is to observe how close the genetic algorithm approaches this simplified approximation. We see that the genetic algorithm produces results close to this lower bound.



Figure 4: Validation of full-scale tests (single e-QC)

We now present results for the test scenario of Figure 2 with the full set of QoS classes shown in Figure 3. The Genetic Algorithm divides the population into three classes [D1.3]: we set the best to be the top 35%, the middle class to be the next 35% and the bottom class to be the bottom 30%. In producing each generation all members of the bottom class are discarded and replaced with child chromosomes one of whose parents is from the top class and the other parent from the middle class. We used a population size of N=250 chromosomes, crossover probability $p_c=0.6$ and mutation probability $p_m=0.05$.

We first consider only two cost functions: pSLS cost Ω and Intra-domain TE cost Φ . Figure 5 shows how the sum of these costs varies as the total eTM traffic increases. The x-axis is normalised by dividing the total eTM flow by the sum of the capacities of the Inter-domain links. The genetic algorithm has a lower cost than the random assignment algorithm at all values of utilisation. We note in passing that the brute force algorithm is only computationally feasible at very low utilisation, and that at this point, the genetic algorithm solution successfully matches the cost of the brute force solution.

In essence, the genetic algorithm identifies solutions where a flow can be assigned to a low-cost combination of l-QC and downstream o-QC. A destination prefix is in general reachable with a given downstream o-QC through a number of different pSLSs, and each of these pSLSs is offered by an adjacent AS at one of a number of different pSLS costs. The genetic algorithm identifies the pSLS with the lowest cost.

We can observe this behaviour by analysing the utilisation of each pSLS. In Figure 6 the 47 pSLSs are shown, arranged in ascending order of cost per unit bandwidth. For each pSLS, the assigned bandwidth is shown for the random assignment algorithm and for the genetic algorithm. We see that the random assignment algorithm has distributed the flows over all pSLSs approximately evenly. However, the genetic algorithm has weighted the flows towards the lower cost pSLSs. In fact, the random assignment algorithm has assigned only 18% of the traffic to the pSLSs with cost per unit bandwidth of 2.2 or less, whereas the genetic algorithm has assigned 85% of the traffic to these pSLSs.







Figure 6: Comparison of pSLS utilisation in random and genetic algorithms (pSLS cost Ω and Intra-TE cost Φ only)



Figure 7: Link utilisations (pSLS cost Ω and Intra-TE cost Φ only)

However, the flow assignments are made without consideration of the maximum inter-domain link utilisation and have resulted in the genetic algorithm assigning flows such that some links are heavily utilised (Figure 7). This can be corrected by introducing the third cost function Θ so that the total cost function is the sum of the pSLS cost, Intra-domain TE cost, and Inter-domain link utilisation (Figure 8). The link utilisation cost function is scaled so that all three components are given approximately equal weight. By introducing the link utilisation function, the peak link utilisations are reduced (Figure 9), with the worst link utilisation from the genetic algorithm reduced from 99% to 69%.



Figure 8: pSLS cost, Intra-domain TE, and Inter-domain link utilisation costs ($\Omega + \Phi + \Theta$)



Figure 9: Link utilisations (based on $\Omega + \Phi + \Theta$)

3.1.2.3.2.1 Impact of Controlled Variables on Performance of Genetic Algorithm

We assessed the performance of the Genetic Algorithm as a function of its controlled variables. We used the same test scenario as that described above for validation (Figure 4): i.e. a single e-QC (150ms), three l-QCs (25, 30, 40ms), and a single downstream o-QC value (100ms) per AS.

To assess the impact of the crossover probability p_c , three versions of the genetic algorithm were compared:

- Crossover probability $p_c=0.6$, child chromosomes based on one "best" parent and one "middle" parent (both randomly chosen);
- Crossover probability $p_c=0.9$, child chromosomes based on one "best" parent and one "middle" parent (both randomly chosen);
- Crossover probability *p_c*=1.0, child chromosome based on *single* parent chosen randomly from "best"/"middle" groups.

To enable fair comparison of these versions, all results were obtained for a fixed length run of 250 generations (and a population size of N=250 chromosomes). It should be noted that the GA results at high utilisation can be improved still further by increasing the number of generations.



Figure 10: Impact of Genetic Algorithm parameters: variation of *p_c*

The results are compared in Figure 10. This graph shows that at low utilisation, selecting either crossover probability of $p_c=0.6$ or 0.9 makes very little difference. However, at high utilisation (>0.75) $p_c=0.9$ give much better results than $p_c=0.6$. We believe that this is because at low utilisation there are many valid solutions that give a low cost, and mixing two solutions results in a valid child solution. However, at high utilisation the solution space is much smaller: retaining the bulk of a consistent solution (by setting $p_c=0.9$) means that the algorithm is able to converge and reduce costs much faster than mixing two possible inconsistent solutions ($p_c=0.6$). It is further to be noted that $p_c=0.9$ always gives better results then $p_c=1.0$: in this final case, there is no evolutionary / genetic component, and the algorithm is relying purely on random mutations to develop improved solutions. However, the random mutations occur slowly (we set $p_m=0.03$ here) and so convergence is slow.

We also investigated values of mutation probability p_m in the range 0.01 to 0.05 and found no general difference in the algorithm convergence or final results.

Test Id	Purpose	Result
InterTE/Perf/1	Performance assessment of genetic algorithm	Successfully completed.
		Performance assessed for GA parameters: crossover and mutation probabilities, pc and pm.
		Performance assessed for impact of different cost functions pSLS cost, Intra-domain TE cost, and inter-domain link utilisation cost.
InterTE/Perf/3	Performance comparison of algorithms	Successfully completed.
InterTE/Perf/4	Performance assessment of interactions	Successfully completed.

3.1.2.3.3 Scalability Tests

We investigated the elapsed run time for each algorithm as a function of the number of flows in the eTM (Table 2). This shows that as expected the brute force approach is not scalable and is not applicable for any realistic configuration. The random assignment algorithm runtime increases with eTM size because it only assigns flows to pSLSs that have sufficient spare capacity; if a flow cannot be assigned the solution is discarded and a further attempt at randomly assigning flows is made [D1.3].

Number of eTM rows	3	4	5	30	50	75	100
Percentage utilisation	0.8%	0.9%	1.2%	11%	18%	28%	36%
Brute Force Assignment runtime	0.1 mins	5 mins	>24 hrs	-	-	-	-
Random Assignment runtime	-	-	-	2 secs	2 secs	3 secs	4 secs
Genetic Algorithm runtime	-	-	-	5 mins	15 mins	50 mins	180 mins

Table 2: Scalability: offline Inter-domain TE runtime as function of egress link utilisation

Given the observations that the Genetic Algorithm with $p_c=1.0$ gives reasonable (but not excellent) results (Figure 10) and that the random assignment algorithm runs very quickly (Table 2) we investigated the hypothesis that an algorithm that selects the lowest of several randomly chosen configurations might provide an approach that gives a good solution with a moderately fast run time and might provide a scalable solution.

For a GA with N=250 chromosomes, running for 250 generations, with the bottom 30% of the population being replaced in each generation, a total of 19 000 chromosomes are generated. We therefore ran an algorithm which selects the lowest cost solution from 19 000 randomly generated solutions. The results are compared with the GA and random assignment algorithms in Table 3. This shows that while the lowest of 19 000 random solutions is significantly better than the random assignment algorithm, the Genetic Algorithm produces the best solutions.

By assuming that the probability density function of the costs is a normal distribution, an estimate can be made of the number of random solutions required to approach the GA solution. The random solutions can be printed to a spreadsheet, where their mean and standard deviation are found to be 10 300 and 450 respectively. The solution space contains approximately $(3*12)^{50} = 10^{78}$ solutions (since each of the 50 rows of the eTM can be assigned to any of 3 l-QCs and on average any of 12 pSLSs). To find a solution that is 6 standard deviations from the mean (i.e. 10 300 - 6 * 450 = 7 600) requires ~ 10⁹ random guesses. This algorithm is therefore not scalable, and we therefore conclude that the Genetic Algorithm provides the superior approach in a way that is scalable.

Algorithm	pSLS and Intra-TE cost (Ω + Φ)		
Genetic Algorithm, $p_c=0.6$	6 200		
Genetic Algorithm, $p_c=0.9$	6 200		
Genetic Algorithm, $p_c=1.0$	7 000		
Random Assignment	10 300		
Lowest of 19 000 random solutions	8 800		

Table 3: Comparison of approaches (overall egress utilisation=18%, single e-QC validation scenario)

3.1.2.4 Conclusions

We compared three algorithms for offline QoS-aware traffic engineering: a random assignment algorithm (effectively representing current day best-effort inter-domain traffic engineering practices applied to a QoS-aware environment); a brute force assignment algorithm; and an evolutionary Genetic Algorithm.

We have shown that in a simplified validation scenario the genetic algorithm obtains results that are close to an analytically obtainable lower bound solution. We have also demonstrated that in a more complex scenario the GA can be used to obtain offline QoS-aware traffic engineering solutions that are of significantly lower cost than a random approach; and that we can reduce the maximum interdomain link utilisation by representing this utilisation in the cost function, minimising the total of the Inter-domain pSLS costs, Intra-domain TE costs and Inter-domain link utilisation costs.

3.1.3 Heuristic Algorithm for Integrated Inter-/Intra-domain TE

3.1.3.1 Overview

In this section we describe the performance tests of the offline inter-domain TE heuristic algorithms. These algorithms provide an integrated approach between inter-domain and intra-domain TE. They implement bandwidth as the single QoS parameter, and we consequently use the total bandwidth consumption as the performance metric. The total bandwidth consumption is defined as the sum of bandwidth needed on each link in order to accommodate the projected eTM.

3.1.3.2 Experiment Setup and Test Description

We evaluate the three proposed heuristic algorithms through simulation. The simulation results are based on 100-node transit domain topologies. The topologies are randomly generated by the method described by Waxman. The set of ingress and egress routers are disjoint. We set the number of ingress routers to 30, whereas the number of egress routers is a variable, as we will evaluate some effects by changing its value between 10 and 30. Each egress router is attached to a maximum of two interdomain links. We assume that the inter-domain resource is less than that of intra-domain resource. The capacity of each link within a domain is randomly generated between 400 and 500, and the capacity of each inter-domain link is randomly generated between 250 and 300.

Feamster [Feam03] discovered that a typical default-free routing table may contain routes for more than 90,000 prefixes, but only a small fraction of prefixes are responsible for a large fraction of the traffic. Based on this finding, we consider 1000 routing prefixes. As these routing prefixes are usually popular destinations, we assume that each egress router can reach all of them. This set of routing prefixes is randomly distributed on the inter-domain link(s) of each egress router. Each routing prefix is advertised with available bandwidth randomly generated between 200 and 250.

For each customer traffic flow, the destination prefix and the ingress router are randomly generated and its bandwidth requirement is randomly generated between 10 and 40.

3.1.3.3 Test Results

Figure 11 presents the total bandwidth consumption as a function of the number of eTM customer traffic flows under the three proposed greedy-based heuristic algorithms. This simulation is based on the scenario of 30 egress routers. The Greedy-penalty heuristic consumes less bandwidth than the others because it considers the penalties of all unassigned customer traffic flows and determines which of these flows, if assigned in the first place, can avoid consuming additional bandwidth. On the contrary, the Greedy-cost heuristic does not take this into consideration and often results in a greater penalty in terms of consuming more bandwidth. As the Greedy-random heuristic randomly selects an egress router without considering any optimisation, any efficient egress router selection algorithms should always outperform it.



Figure 11: Total bandwidth consumption as function of traffic

In Figure 12, we show the difference of bandwidth consumption between the Greedy-cost and Greedypenalty heuristics for a different number of egress routers. We study the bandwidth consumption difference under three traffic loads with 100% acceptance ratio at any considered number of egress routers: 50, 100 and 150 customer traffic flows. The bandwidth consumption difference is the total bandwidth consumption using the Greedy-cost heuristic minus the total bandwidth consumption using the Greedy-penalty heuristic. It is worthwhile to determine the improvement of bandwidth consumption when using the Greedy-penalty heuristic over the Greedy-cost heuristic.



Figure 12: Bandwidth consumption difference between Greedy-cost and Greedy-penalty heuristics

When the number of traffic flows increases, the bandwidth consumption difference between the two heuristic algorithms increases. This can be explained by the case that, as traffic load to the egress routers increases, some egress routers do not have sufficient resource so that some customer traffic flows are directed to the "distance" egress router with possible great penalty in terms of consuming more bandwidth. It is the case where Greedy-penalty heuristic is used to avoid additional bandwidth consumption.

Something else that can be deduced from the figure is that as the number of egress routers increases, the bandwidth consumption difference decreases. This is the opposite effect to the previous one, with the aforementioned case occurs less frequently as more capacity is added. As a result, the two heuristic algorithms are likely to have same selection for traffic flows and the performance of them tends to become identical.

From the above, we conclude that the Greedy-penalty heuristic provides significant performance improvement over the Greedy-cost approach, under the situation where the network has a certain level of loading in order to take the advantage of penalty-based selection, and that no more than one egress router can preferentially accommodate most of the traffic flows while leaving the other egress routers barely selected. The latter situation is achievable due to the fact that resources are commonly distributed in the network for load balancing.



Figure 13: Bandwidth acceptance ratio for Greedy-penalty heuristic

For the rest of simulations, we continue to study the performance as the number of egress routers varies. As the Greedy-penalty heuristic outperforms the others, we only consider this one. Figure 13 shows the influence of the number of egress routers on the bandwidth acceptance ratio. The bandwidth acceptance ratio is the sum of bandwidths of accepted traffic flows over the sum of bandwidths of all the traffic flows. As the number of egress routers increases, the bandwidth acceptance ratio increases. This is due to the property that performance improves as more capacity, such as inter-domain link and advertised bandwidth capacity, is added by increasing the number of egress routers. It is also worthwhile to determine when the bandwidth acceptance ratio reaches a level of diminishing return.



Figure 14: Total bandwidth consumption vs. number of egress routers for Greedy-penalty heuristic

To evaluate the influence of the number of egress routers on the total bandwidth consumption, we study the bandwidth consumption under three traffic loads as they were previously used: 50, 100 and 150 customer traffic flows. Figure 14 shows the total network bandwidth consumption with a different number of egress routers. For all the traffic flows, as the number of egress routers increases, the total bandwidth consumption decreases. This is because, as the number of

egress routers increases, the traffic flow can be directed to a "closer" router which results in reduced bandwidth consumption. This effect becomes more apparent when the number of traffic flows is large since the traffic load of each egress router is high, while adding additional egress routers can significantly improve the performance. On the contrary, this effect is less apparent when the number of traffic flows is small.

3.1.3.4 Conclusions

We have developed three heuristic algorithms to solve the integrated inter-domain / intra-domain TE problem. Simulation results show that the Greedy-penalty performs better than the other two algorithms in terms of total network bandwidth consumption. We have also evaluated the influence of the number of egress routers on the total bandwidth consumption and bandwidth acceptance ratio. We found that the total bandwidth consumption decreases and the bandwidth acceptance ratio increases as the number of egress routers increases.

3.2 Offline Traffic Engineering Interactions

3.2.1 Overview

In this section we describe the objectives, performance metrics and experimentation environment for the performance tests of the interactions between offline intra- and inter-domain TE.

In [D1.3], we proposed two approaches, namely the decoupled and integrated approaches, to combine offline intra- and inter-domain TE. The objective of the performance tests is to assess the performance of the decoupled and the integrated optimisation approaches.

3.2.1.1 Assumptions

For the proposed heuristic algorithms for the decoupled and integrated approaches, there are of course many possible algorithm or solution combinations for the two approaches. However, since this paper is not intended as a comparative study of these options, we will propose a classical greedy-based heuristic algorithm as the TE algorithm for the decoupled and integrated approaches. The proposed heuristic algorithms are similar to that proposed by Xiao [Xiao00] which has been deployed in a real network system. The proposed algorithms for both approaches are very similar in order to accurately compare their TE performance. Although it might be appealing to test some more complex algorithms, the approach presented here is sufficient to illustrate the point of interest. For simplicity, but without loss of generality, we make the following assumptions for our algorithm and evaluation:

- Only outbound and transit traffic are considered.
- Bandwidth is considered as the QoS metric.
- The inter-domain resource objective to optimise is the inter-domain link utilization, and the outbound provider SLA is used as capacity constraint.
- Explicit routing is assumed and bandwidth constrained minimum cost routing algorithm is used for intra-domain route selection, where the cost is dynamically calculated for each considered traffic flow by the piece-wise linear cost function proposed in [For02]. This not only minimizes resource consumption but also attempts to achieve load balancing within the network. The granularity of explicit paths is per-prefix.
- The AS under consideration has sufficient capacity to meet the end-to-end bandwidth requirements of all inter-domain traffic flows. Thus, the traffic-oriented TE objective can be negligible.

3.2.1.2 Performance Metrics

We use the following performance metrics as the optimisation criteria to evaluate the decoupled and the integrated approaches: (i) Total network cost (the sum of intra-domain and inter-domain cost), (ii)

total bandwidth consumption, and (iii) maximum intra-domain and inter-domain link utilization. The first metric captures the overall network cost. Overall intra-domain (respectively inter-domain) cost is defined as the sum of the cost of the intra-domain (inter-domain) links. Fortz and Thorup [For02] propose a piecewise linear increasing function of link utilization which imitates the response time of M/M/1 queue to access the cost of intra-domain links. By using the piecewise linear cost function, two objectives of bandwidth usage and resource load balancing are taken into account simultaneously. These two objectives are related to our second and third performance metrics. In other words, the overall network cost is a function of both bandwidth consumption and link utilization.

In this paper, we adopt the piecewise linear function to quantify the cost of intra-domain and interdomain links. Since inter-domain links are the bottleneck in the Internet [Akam99], we assume that the cost of using them is a factor α times the cost of intra-domain links. We assume $\alpha=2$ as our initial evaluation in this paper. The impact of α on network performance will be evaluated in our future work.

The total bandwidth consumption is the amount of bandwidth needed to accommodate all traffic flows within an AS. It is calculated based on the bandwidth requirement of each traffic flow and the length of path on which the traffic flow has been assigned.

The utilization of a link is the amount of traffic on the link divided by its capacity. The maximum link utilization is the maximum utilization over all links in a network. Minimizing this objective ensures that traffic is moved away from congested to less utilized links and the distribution of traffic is balanced over the links [Wang99].

For all three metrics, the lower values are preferred.

3.2.2 Experiment Setup and Test Description

The simulation is based on 100-node topologies generated by BRITE with node degree of 4. The number of border routers is set to 30% of the total network nodes. Note that inter-domain links can be ingress or egress links, and we only consider egress links for outbound TE in this paper. Without loss of generality, we assume that each border router is attached to a maximum of three egress links and the capacity of each egress link is randomly generated between 150 and 300 units. The capacity of each intra-domain link is randomly generated between 80 and 200 units. As inter-domain links are usually the bottleneck in the Internet, the total capacity of all intra-domain links should be larger than that of all inter-domain links.

Due to the fact that only a small fraction of prefixes are responsible for a large fraction of the traffic [Feam03], we consider 100 popular remote destinations which are uniformly and randomly distributed over all the border routers. The number of remote destinations that each border router can reach, specified in outbound provider SLAs, is randomly generated between 30 and 60 units, and these remote destinations are randomly distributed among all the egress links. The contracted bandwidth for a remote destination is randomly generated between 30 and 60 units.

For each aggregated inter-domain traffic flow, the remote destination and the ingress router are randomly generated. The bandwidth demand of each aggregated inter-domain flow is randomly generated between 1 and 40 units.

To ensure confident results, each simulation point takes an average value based on 10 trial runs.

3.2.3 Test Results

Figure 15 and Figure 16 show the inter-domain and intra-domain cost as a function of number of interdomain traffic flows achieved by the decoupled and integrated approaches respectively. The inter-cost achieved by the two approaches is nearly identical. This is because the cost of using inter- AS links is higher than that of intra-domain links, so the inter-domain link utilization becomes a dominant factor in the selection decision in both approaches. It is possible that there are several inter-domain links that have very similar utilization, but the intra-domain routes connected to them may have different costs. In this case, the integrated approach can select the best combination of inter-domain links and intradomain routes. We see in Figure 16 that the performance difference between the two approaches is primarily in their intra-domain cost.



Figure 16: Evaluation of intra-domain cost

The total network cost is defined as the sum of intra-domain and inter-domain cost. Since the interdomain cost achieved by both approaches are nearly identical, the total cost will mainly depend on the intra-domain cost. Hence, the total cost achieved by the integrated approach is much lower than that achieved by the decoupled approach. This resembles the performance shown in Figure 16.



Figure 17: Evaluation of total bandwidth consumption

A major reason for the large intra-domain cost in the decoupled approach is due to the increase in bandwidth consumption and link utilization within an AS. Figure 17 shows that the integrated approach uses less bandwidth to accomplish its end-to-end QoS provisioning within the network than the decoupled approach. This is because, when choosing egress routers, the number of hops on the corresponding intra-domain routes has been considered as the selection criteria. The decoupled approach on the other hand may choose an egress router with the best inter-domain link utilization but at the expense of long intra-domain route towards the egress router, resulting in high bandwidth consumption.

Although Figure 18 shows that the integrated approach has a slightly higher maximum inter-domain link utilization than the decoupled approach, both approaches incur nearly identical inter-domain costs, as shown in Figure 15. This may result partially from the piecewise linear cost function, which gives the same penalty to links with utilizations in the same block, such as between 1/3 and 2/3. In this case, such links are considered as at the same level of congestion. Based on the fact that both approaches result in nearly identical inter-domain TE performance, Figure 19 shows that the integrated approach exhibits the advantage of significantly reducing the maximum intra-domain link utilization, compared to the decoupled approach.



Figure 18: Evaluation of maximum inter-domain link utilization



Figure 19: Evaluation of maximum intra-domain link utilisation

As the decoupled approach performs inter-domain TE prior to intra-domain TE, utilization performance on inter-domain link is good compared to that on intra-domain link. On the other hand, the integrated approach takes the balanced approach optimizing between intra-domain and interdomain resource utilization, therefore the achieved inter-domain resource utilization may not be good as that achieved by the decoupled approach. Nevertheless, significant improvement in intra-domain utilization achieved by the integrated approach compared to the decoupled approach offsets this minor degradation in inter-domain resource utilization.

To compare the overall performance achieved by the decoupled and integrated approaches, our numerical experiments reveal that the integrated approach could save a significant amount of resource cost and achieve a good overall network resource performance, compared to the decoupled approach. Hence, we attempt to answer the question posed in the introduction section by introducing the integrated approach to achieve lower cost complete TE solution.

In fact, other factors can also affect the performance of the two approaches, such as the efficiency of algorithms, the definition of link cost function (linear, concave or discrete), network size and topology, etc. Further experiments are needed to understand their impact on traffic and resource utilization performance.

3.2.4 Conclusions

We have established a direct relationship between intra-domain and inter-domain TE, and explored the interaction between them by proposing and analysing both the decoupled and integrated approaches. We have shown through simulation how the integrated approach results in lower cost TE solutions with lower total consumed bandwidth.

3.3 Intra-domain Traffic Engineering Tests

3.3.1 Overview

MESCAL's intra-domain traffic engineering approach is based on layer 3 mechanisms (rather than MPLS-TE, for example). Its purpose is to compute a set of OSPF link weights to balance network load while honouring the QoS constraints of the traffic; and to provide answers to "what if" scenarios posed by Inter-domain Traffic Engineering in order to coordinate and optimise inter-domain and intra-domain traffic engineering decisions.

The IPTE approach is built on classical OSPF routing, but additionally introduces DSCP based routing to form multiple OSPF routing planes in the network. Each DSCP plane has individual link weights and can thus route traffic independently of the other planes. Each plane may be used to route traffic of

an equivalent QoS-class to meet the performance constraints of that class. Another benefit of this approach is that multiple routing planes – even for a single QoS-class – allow for better load balancing across an AS.

The IPTE algorithm runs off-line at Resource Provisioning Cycle epochs. Given a traffic demand matrix and the network topology, the algorithm computes a set of link weights using a search heuristic. The optimisation is cost function based, so that individual QoS class constraints as well as other optimisation goals can be taken into account by factoring them into the algorithms cost function. This allows for parallel existence of hop-count-constrained, bandwidth-constrained and best effort traffic classes.

Because the solution relies on IP routing, the IPTE approach is more lightweight than MPLS-TE in terms of state-information required to be maintained in the network and the associated management configuration overhead for establishing LSPs. Since QoS information remains at the management layer in the off-line algorithms, no QoS awareness is required at layer 3. Recent MT-OSPF Internet Drafts provide the required DSCP based routing support, so potentially no major changes at the router level are required for the approach to be implemented.

This section gives a detailed description of the simulations that have been carried out for the Intradomain Traffic Engineering component described in the MESCAL functional architecture. The campaigns focus predominantly on the *Resource Optimisation* block contained within Intra-domain Traffic Engineering. *Resource Optimisation* contains the essential link weight based IP traffic engineering functionality. In contrast, functionality contained within the *Resource Reconfiguration Scheduler* block is concerned with the efficient implementation of results computed by *Resource Optimisation* and is therefore of secondary concern. The test campaigns have three overall objectives:

- 1. Functional Validation
- 2. Algorithm Performance Measurement and Optimisation
- 3. Algorithm Efficiency Measurement and Optimisation

3.3.2 Experiment setup and test description

This section contains a description of the simulation setup and the input data used for each test. The network shown in Figure 20 resembles the early NSFNet backbone. It was used as test topology for functional testing as well as for load spread simulations to determine the effectiveness of routing planes.



Figure 20: 10 Node Test Network

Larger topologies were also used ranging from 50 to 300 nodes. Topologies were generated using the BRITE [Brite] topology generator. A list of topologies used with more details is displayed in Table 4.

Number of Nodes	Number of Links	Topology Type	Min, Max Link Capacity	Link Capacity Distribution	Node Distribution	Average Degree
10	17	NSFNet	2, 5	exponential	heavy tail	3.2
50	100	Waxman	10, 1024	exponential	heavy tail	4
100	200	Waxman	10, 1024	exponential	heavy tail	4
300	400	Waxman	10, 1024	exponential	heavy tail	4
300	600	Waxman	10, 1024	exponential	heavy tail	4

Table 4: Topologies used for Simulations

As in [Fortz00] demands were generated according to

$$\alpha O_u D_v C_{uv} e^{-\frac{\lambda(u,v)}{2\Delta}}$$

Where O and $D \in [0,1]$ are random numbers chosen for each node. Similarly, $C \in [0,1]$ is chosen for each pair of nodes u, v. The parameter $\lambda(u, v)$ denotes the Euclidian distance between u and v and Δ is the maximum Euclidian distance between two nodes. This ensures that demands are greater between nodes with shorter Euclidian distance. Also, since there are three random numbers multiplied, the variation between demands is large. For the simulations, three demand matrices were generated for each topology size with small, medium and large numbers of individual demands; they are detailed in Table 5.

Nodes in Topology		Demand		
	Small (#)	Medium (#)	Large (#)	a factor
10	N/A	35	N/A	N/A
50	900	1300	3000	1
100	900	1300	3000	5
200	900	1300	3000	10
300	900	1300	3000	15

Table 5: Demands used for Simulations

The demand set for the 10 node topology was created manually; not using the method described above, but rather by applying individual demands across some of the network edges.

3.3.3 Test Results

3.3.3.1 Algorithm Performance and Optimisation

3.3.3.1.1 (Perf1) Load Balancing Performance

The load balancing performance of the IPTE system is important to give the network more flexibility towards changes in the demand pattern. The more evenly balanced the network, the less traffic engineering changes have to be made over time in order to achieve operational goals of the network operator. The plots in Figure 21 show the results of an IPTE optimisation cycle run on a 100 node topology, utilising 5 routing planes. Both mean utilisation and standard deviation of utilisation are shown. As comparison, values for inverse capacity link weights are also plotted. Inverse capacity link weights are the Cisco recommended configuration for OSPF networks.

Plotting the standard deviation of the link utilisation gives a measure of how well loads are balanced across the network. The mean utilisation is the mean of individual link utilisation values and thus gives a measure the overall network utilisation.



Figure 21: Load balancing improvement on a 100 node network, after 500 iterations

Whereas the mean utilisation increases from 30% to 80% for the IPTE case, the standard deviation decreases slightly from 30% to 40% utilisation and then increases as the mean utilisation is increased to 80%. The IPTE solution stays about 10% below the inverse capacity link weight settings. This shows that better load balancing is achieved.

3.3.3.1.2 (Perf4) Routing Plane Effectiveness

Since there are 64 routing planes available, not all of which are necessary for MESCAL inter-domain QoS, spare planes may be used to split traffic of the same class for the purpose of intra-domain load balancing. The tests in this category were thus designed to investigate how the individual routing planes can be employed for the purposes of load balancing. Figure 22 shows the results simulated on the 10 node topology.



Figure 22: routing plane effectiveness on a 10 node network, 500 iterations

From the graph it can be seen that even a single routing plane based IPTE link weight optimisation can provide large gains in terms of load balancing in this case. Overloaded links from the Inverse Capacity routing disappear almost completely. However, several links remain underutilised with IPTE 1 (single routing plane) and some links still remain at near 100% utilisation. Performing the optimisation on 5 routing planes shows that the demands are now more effectively balanced, with many of the available links utilised and maximum link loads of less than 80% in this example. Interestingly, increasing the number of routing planes to 64 does not bring further improvement (second graph). Whereas only the 64 plane IPTE utilises all links, maximum link utilisation increases slightly, compared to the IPTE solution with 5 planes. Considering that a 10 node 17 link network does not offer many alternative paths, this result seems intuitive. In fact, even larger networks with 50+ nodes of the type generated for these simulations do not appear to benefit from more than 5 routing planes as can be seen on the graphs in Figure 24 which show the results for the 50 node topology.

Each of the four graphs shows a histogram of the utilisation of individual links on the network for Inverse Capacity and IPTE with 1, 5 and 64 routing planes with additional statistics displayed in the table below the plot. Again IPTE 1 shows significant improvement over inverse capacity weights, removing the overloaded link and significantly decreasing the utilisation Standard Deviation. The IPTE with 5 routing planes shows the best performance, with a further decrease in standard deviation accompanied by an increase in utilisation mean. This shows that IPTE 5 indeed performs load balancing more effectively than a single routing plane optimisation is able to. More links are utilised and the average utilisation increases, yet the maximum utilisation decreases (to 64% compared to 84% for IPTE 1 and 123% for InverseCap). IPTE 5 also removes the peak of underutilised links that featured in both InverseCap and IPTE 1.



Utilisation statistics						
Туре	Mean (%)	StdDev (%)	Max (%)	Min (%)		
InverseCap	19.9789	18.0273	123.2366	0		
IPTE 1	19.9368	14.715	84.8083	0		
IPTE 5	23.021	11.6104	63.9776	0.8214		
IPTE 64	28.9738	17.9128	85.3099	1.145		

Figure 23: Effect of Routing Planes on Utilisation for a 50 Node Network, 1300 demands

IPTE 64 does improve on the result of InverseCap, but not as significantly as either IPTE 1 or IPTE 5, which appears to be counterintuitive. However, one possible cause for this behaviour could be that IPTE 64 has a larger probability for arriving at local minimums than solutions with less routing planes. While iterations with 5 routing planes have an effect on the routing of up to 1/5 of the traffic, with 64 routing planes it is only up to 1/64. The optimisation algorithm may therefore track towards a near local minimum with marginal per-iteration improvements, from which it cannot escape without non improving moves (Perturbations are available as non-improving moves. However, these may be too coarse for the purpose). The Convergence performance of the IPTE 64 scenario supports this theory, by quickly arriving at the final solution with no further improvement for several 1000 iterations after that. IPTE 1 on the other hand shows the longest improvement time, with improvements still being occurring after 4500 iterations without plateau effect (see section 3.3.3.2.1 for details on plateau effect). In order to get maximum benefit from more routing planes, it thus seems necessary to modify the heuristic approach with increasing numbers of planes to compensate for the effect described.

For comparison, the cost function values range from 35358.0 for InverseCap to 4014 for IPTE 1, 2955 for IPTE 5 and 9264 for IPTE 64.
3.3.3.1.3 (Perf5/Perf6) QoS Constrained Performance

This section shows how individual routing planes can be optimised with different performance goals using the cost function. Two types of this individual treatment are shown: hop count constraint and utilisation constraint. The hop count constrained optimisation can be used for delay sensitive traffic classes, whereas the utilisation constrained optimisation can be used for bandwidth constrained classes. More elaborate QoS based optimisations could be devised, the results in this section are a demonstration of the feasibility of the approach.

Hop Count Constrained Optimisation

Before performing the hop count constrained optimisation it is important to realise that whereas the optimisation algorithm is based on identification of high cost links, hop count is an end-to-end feature. Thus, in order to factor hop count constraints into the cost function it is necessary to calculate the end-to-end path of each demand on a link. Once the hop-count has been determined, a per-link cost can be calculated based on routing plane membership, which is then added to the equivalent utilisation of the link as the sum of d_h over all (delay constraint) routing planes. (For more details on the build up of the basic cost function, see [D1.3], section 10.6.2.3.9)

$$\Phi = \sum_{l \in E} \Phi_l \left(\frac{\sum_{h \in H_l} (f_{l,h}(x_{l,h}))}{c(l)} + \sum_{h \in H_l} d_h \right)$$

Since the starting link weights are unit weights, the shortest path is already configured for each demand. It is thus important to ensure that the algorithm does not make these paths longer for the purposes of load balancing and so the cost calculated above helps the optimisation heuristics to identify when a link weight modification has caused a hop count limit to be exceeded. The resulting increase in cost should lead to a discarding of this modification. This ensures that hop counts are honoured while load is balanced.



Figure 24: Average hop count for 50 node 100 link network, 500 iterations, 1300, 3000 demands

The graph in Figure 24 shows the effect of hop count constrained optimisation for a 50 node 200 link network. The two delay classes were hop count constrained, using a sharp maximum hop count cut-off of 7 for class 2 (i.e. a sudden increase in cost from 0 to 5000 between 6 and 7) and a gradual cost increase from 5 to 7 for class 1. The maximum shortest path distance across the network is 8 hops. Classes 1-5 (including delay classes 1 and 2) are operated in parallel on the same network, whereas unit weight, inverse capacity weight and random weight are computed separately for the same network topology and demand matrix. Several observations may be made from this graph. Firstly, the hop count for all classes remains approximately the same for different network load. This should be expected, for all reasonable loads, as long as the IPTE algorithm has sufficient free capacity to operate. Secondly, it can be seen from the graph that non-delay constrained classes have high average hop counts. This is also as expected, since longer average paths are an effect of the load balancing on these classes. Finally, the graph shows that the two delay classes have lower average hop counts than the non hop-count constrained classes. The delay classes perform worse than the both inverse capacity and unit weights, as a result of the costs specified in the cost function which is based on the absolute hop-count limitation of 6-7, rather than limitation based on relative path length. Maximum hop counts on both delay classes are equal to that of the shortest path on unit weights (because of the large cost applied for demands exceeding 7 hops), whereas all other optimised classes (1,2 and 5) have larger maximums (10-14).

Spare bandwidth constrained optimisation

Optimisation for higher average spare bandwidth on a routing plane is accomplished by increasing the equivalent bandwidth factor of the class. Higher equivalent bandwidth causes the utilisation to appear greater to the cost function than it is. As a result, the optimisation algorithm reduces the load on links occupied by the class more than average, leading to higher average available bandwidth.



Figure 25: average utilisation for bandwidth constrained class, 1300 demands, 500 iterations

Figure 25 shows the effect of this method. The graph shows the mean link utilisation as seen by individual classes, i.e. the mean of utilisation (of all traffic) on all links that the class utilises. With all classes co-existing on the same network, Class 1 has lower mean utilisation than all other classes. The equivalent bandwidth factor is shown on the x axis. It demonstrates how the effect first increases rapidly and then levels off when the factor reaches 5. At this point, the utilisation constrained classes'

bandwidth requirements can no longer be met as efficiently through link weight optimisation, as bandwidth requirements begin to reach 100% link bandwidth.

It is also worth noting that the average link utilisation for all other classes stays approximately constant, which might be explained through the increase in both higher utilised links as well as lower utilised links on the network. Since some of the traffic on the non delay constrained classes also travels on the links with lower utilisation, the average of link utilisation remains approximately constant. Evidence for this is the increase in link utilisation standard deviation from 15.77 to 18.54 between equivalent bandwidth factors 0 to 8.

3.3.3.2 Algorithm Efficiency and Optimisation

3.3.3.2.1 (Effic5/Mixt) Convergence Properties

Measuring the algorithms performance is important in order to improve its convergence properties. The heuristic algorithm used for the IPTE system has to search a very large solution space that is a function of the number of links and routing planes. Thus, in order to achieve improvement over the inverse capacity/ unit link weight settings, a lot of work has to be invested into tweaking the heuristics. The plots in Figure 26 provide information on convergence. The top plot shows the improvement of cost function over iteration number plotted on a log scale, whereas the second and third plots track the mean and standard deviation of network utilisation. All graphs feature two plots, the best and the running value. The running value is the value computed for the current iteration and may be worse than the best value. If the value is worse, the new solution will be discarded at the beginning of the algorithm, which perturbs the link weight set if no improvement is found to the best solution for some time. It is meant to enable the algorithm to escape from suboptimal local minimum solutions. After a perturbation begins, a new temporary best cost value is determined which causes the plateau-likeness of the large bums. Once a perturbation has completed it is discarded if unsuccessful or kept if an overall improvement was achieved.





The cost improvements are largest at the beginning of the optimisation cycle, but further improvement continues until the 1600th iteration in the depicted case. The standard deviation plot shows that these small improvements in cost have an equally large effect on load balancing than the first large drop in cost at the beginning of the optimisation. The reason for this becomes clear when recalling that the cost is based on an exponential function that assigns large values to overloaded links. These links are addressed by the optimisation in its first iterations causing large improvements on the cost plot. However, later iterations with less cost improvement nevertheless have high importance to load-balancing as can be seen on the standard deviation plot. The mean utilisation stays approximately

constant with a slight decrease in the beginning and then a slow increase over a long period. The drop is caused by reducing load on the most overloaded high cost links at the beginning of the optimisation. These links are most distant in utilisation value to the mean and thus removing them has a visible impact on the average. Load balancing over the course of the optimisation causes the slow increase in the mean, while more links are utilised and the paths that traffic takes become longer. An increase in mean utilisation should be expected when the load balancing is functioning effectively.

Further analysis of the algorithms convergence properties was performed on varying topology sizes, the results of which are shown in Figure 27. Plotted is the percentage improvement of link utilisation standard deviation from the starting value unit weight value.



Figure 27: Algorithm efficiency measured on utilisation StDev, 60% average utilisation

As before, the results show that for all topology sizes there is a sharp improvement in the first few 300 iterations. However, the effect lessens with topology size with the 50 and 100 node networks benefiting most. With increasing topology size, the overall percentage improvement per iteration becomes smaller approximately halving per doubling of number of nodes. Improvements range from almost 22% for 50 nodes to only 3% for 300 nodes. As can be seen for the 200 node case, the optimisation for larger networks takes longer and significant improvement occurs at around 3800 iterations even after a long plateau of non-improvement. However, the heuristics devised for this study were not optimised for run times longer than a few thousand iterations required for the smaller topologies and runs longer than 5000 iterations are relatively inefficient. It would be more effective to adapt the heuristic for larger networks. Another likely cause for the lack of improvement on the two larger topologies is that only 1300 demands were in the demand matrix, thus causing a lack of improvement opportunities for the heuristic.

3.3.3.2.2 (Effic1/Effic2) Execution Time

In order to measure how long it takes for a result to be computed, the algorithm execution time has to be plotted against an improvement factor, such as the cost value or the load balancing measure of utilisation standard deviation. The plot in Figure 28 shows how the standard deviation of link utilisation improves over the run time of the optimisation cycle.



Figure 28: Convergence Time 50 Node Topology

For this case, the last improvement can be observed at around 32 minutes. However, since a long plateau was traversed before the improvement at around 28 minutes run time, it is not certain that no further improvement can be achieved beyond the 50 minutes of total run time. Execution time is significant for IPTE link weight optimisation. Whereas for this plot, 1800 iterations were computed on 1GHz CPU, larger topologies require 5000 and more iterations for large improvements to take place. However, results of this type are indicative as several factors that lead to the long execution time can be remedied. On more powerful hardware, such as a 3 GHz Pentium4, 5000 iterations for the same network configuration merely take 30 minutes. Additionally, the efficiency of the software implementation can be improved to reduce the algorithm run time. Finally, since offline intra-domain TE is an "offline" process, abundant time is available for processing.

3.3.4 Conclusions

IPTE has been tested in various networks ranging from 10 - 300 nodes with up to 600 bi-directional links. For each topology, three demand matrices were generated ranging from 900 to 3000 individual source destination demands.

Link weight optimisation on a single routing plane achieves better performance in terms of load balancing and avoidance of overloaded links compared to the inverse capacity link weight rule of thumb. It can be concluded that while inverse capacity is a good rule of thumb if the demand matrix is unknown, overloaded links and packet losses may occur when the traffic demand is high. With average utilisation figures of 30-40% on a 10-100 node network, off-line link weight optimisation can distribute traffic so that overloaded links do not occur while the same demand pattern on the same physical network will cause packet losses when random or inverse capacity link weight assignment

policies are used. Use of IP-TE can therefore relieve congestion without resorting to a potentially expensive reconfiguration of the physical network, e.g. installation of additional links. IP-TE implements its traffic redistribution policy through a soft configuration of the existing network, which can be achieved more easily and furthermore can be done periodically as traffic demand matrices change significantly, also accommodating temporary demand fluctuations.

While improvements over random and inverse capacity link weight assignment policies are achieved with the IP-TE link-weight optimisation heuristic, the use of multiple routing planes on the same physical infrastructure results in further gains in load balancing. Up to 64 independent routing planes are available with DSCP-aware routing and forwarding mechanisms deployed in the routers. Simulation results comparing the utilisation and load balancing performance of assigning demands to a number of parallel routing planes show that the use of 64 link-weight-optimised planes exhibits a significant improvement in load balancing compared to a single link-weight-optimised routing plane. However it has been demonstrated that using 5 parallel link-weight-optimised routing planes achieves a comparable performance.

The conclusions so far relate to a single QoS-class, e.g. the current best effort Internet. For QoSenabled IP networks assumed by MESCAL an AS is required to implement multiple QoS-classes on the same physical infrastructure. We have demonstrated that the cost function of the link-weight optimisation heuristic can accommodate QoS-classes with different performance goals on different routing planes. Both hop-count- and bandwidth-constrained classes were considered and deployed on the same simulated network. The results show that link-weights for classes with different QoS targets can be derived and implemented on the same network through multiple routing planes. The resulting traffic distribution results show that delay-constrained classes take shorter paths and that bandwidthconstrained classes are routed over lower-utilised paths. Further investigations could study the impact of multiple equivalent routing planes for each QoS-class to determine whether the improvements in network load-balancing seen for a single QoS-class still hold. As it was seen that only 5 routing planes were required to achieve significant improvement over a single routing plane it can be seen that up to 64/5 parallel QoS-classes could be deployed within an AS, however, since traffic of all classes is shared on the same links it is unlikely that each QoS class requires 5 load balancing classes. Rather the different OoS classes should provide balancing between them if their OoS constraints are not too stringent.

With an iterative heuristic-based optimisation it is difficult to determine whether the algorithm has converged. Experimental results have shown that significant improvements in cost, utilisation and load-balancing are achieved after relatively few iterations of the algorithm (order of 100) for networks in the order of 10 - 100 nodes with 5 routing planes. Larger networks require more iterations (order of 1000) to reach significant improvement. Similarly if only a single routing plane is deployed, improvements may be observed after 1000s of iterations. However, even for smaller networks with 50-100 nodes and 5 routing planes, further significant improvements are occasionally observed at higher iterations (order of 1000) when a random perturbation of the current best solution finds an alternative set of solutions. It was seen overloaded links were removed early in the solution (order of 100 iterations), improvements in load-balancing were the main reason behind the smaller reductions in cost-function values at iterations in the order of 1000s. The reason for this is due to the high-cost associated with high link utilisation that causes the algorithm to quickly reroute traffic to remove overloaded links.

Algorithm run time was shown to scale approximately linearly with network size. Typical run times on a 3GHz Pentium 4 processor were around 30 minutes for 5000 iterations on a 50 node network with 1300 demands, however run time increases with topology size and the same run with a 300 node network can take up to 2 hours. The IP-TE optimisation heuristic is not intended as an on-line traffic control algorithm and therefore the execution time is acceptable for periodic off-line traffic engineering purposes.

3.4 Multicast Traffic Engineering Tests

3.3.1 Offline Dimensioned Test

3.4.1.1 Overview

We provide in this section the test results from the simulation software for the Offline Multicast Traffic Engineering (OMTE) specified in section 7 in [D1.3]. The objectives and controlled/uncontrolled variable settings are same as those included in [D3.1].

3.4.1.2 Experiment Setup and Test Description

We adopt the Waxman's model in GT-ITM topology generator for constructing our network models. This approach distributes the nodes randomly on the rectangular grid and nodes are connected with the probability function:

$$P(u,v) = \lambda \exp(\frac{-d(u,v)}{\rho L})$$

where d(u, v) is the distance between node u and v and L is the maximum possible distance between any pair of nodes in the network. The parameters λ and ρ ranging (0, 1) can be modified to create the desired network model. A larger value of λ gives a node with a high average degree, and a small value of ρ increases the density of shorter links in comparison to longer ones. In our simulation we set the values of λ and ρ to be 0.2 respectively, and generate a random network of 100 nodes, out of which 50 are configured as Designated Routers (DRs) with attached group sources or receivers. The scaled bandwidth capacity of each link is set to 10^5 units. Apart from the GA approach, we also implemented two non-TE based hop-by-hop routing approaches and one explicit routing approach: (1) shortest path routing with random link weight setting (Random), (2) shortest path routing in terms of hop-counts (SPH), and (3) Steiner tree approach using the TM heuristic. For this TM Steiner tree algorithm, we use hop count as the link weight, and the resulting trees are group specific, i.e., one Steiner tree is specifically constructed for each multicast group.

3.4.1.3 Test Results

3.4.1.3.1 Functional Tests

The software is functioning correctly.

3.4.1.3.2 McastTE/Perf/OMTE-GA

Figure 29 illustrates the feature of overall bandwidth conservation capability of individual schemes with the variation of maximum group traffic demand D_g . As it is expected, explicit routing with the

TM heuristic achieves the lowest overall network loading while random link weight assignment results in the poorest performance. We can also see in the figure that the GA approach exhibits the best capability in conserving bandwidth among all the hop-by-hop routing schemes. Typically, when the network is under-utilised, our proposed GA approach exhibits significantly higher performance than the conventional IP based solutions without explicit routing. For example when $D_g = 3000$, the overall bandwidth consumption of the Random and SPH solutions are higher than that of GA by 19.3% and 14.9% respectively. Compared with the TM heuristic that needs support from MPLS overlaying, the gap from GA is below 8%. However, when the external traffic demand grows, the performance of GA converges to that of the SPH approach. On the other hand, although the TM algorithm exhibits significant higher capability in bandwidth conservation when the external traffic demand grows ($D_g > 4000$), this does not mean what have been obtained are feasible solutions without introducing overloaded links.



Figure 29: Total bandwidth consumption vs. Max D_g

Figure 30 shows the relationship between the proportion of overloaded links and the maximum group traffic demand D_g in time of network congestions. From the figure we can see that there are more overloaded links as D_g increases. The most interesting result is that, through our GA optimisation, the percentage of overloaded links is significantly lower than all the other routing schemes. In the most congested situation (D_g =6000), the average rate of overloaded links computed by GA is only 1.4%, in contrast to 12.6% by random link weight setting, 8.6% by the TM heuristic, and 4.4% by SPH respectively. On the other hand, the amount of overloaded bandwidth occurred on the most congested links is another important parameter an INP is interested in. We define the Maximum Link Overload Rate (MLOR) as follows:

$$MLOR = \max_{(i,j) \in E} \left(\frac{\sum_{g=1}^{G} D_g \times y_{ij}^g - C_{ij}}{C_{ij}} \right)$$

From this definition we can see that MLOR reflects the overloading scale of the most congested link (if any, i.e., MLOR>0). An INP should avoid configuring the network resulting in hot spots with high MLOR. Through our simulations, we also find that the proposed GA approach achieves the lowest MLOR performance. In Figure 31, the overloading scale is 45% of the bandwidth capacity on the most congested link in the GA approach with D_g equal to 6000, while this value reaches 110% and 59% in random link weight setting and SPH respectively. Even by using explicit routing TM heuristic, the overloaded bandwidth is 78% of the original link capacity.



Figure 30: Overloaded link rate vs. Max D_{a}



Figure 31: MLOR vs. Max D_{g}

From Figure 30 and Figure 31 we find that shortest path routing with hop-counts (SPH) has higher capability in finding feasible solutions (i.e., no overloaded links incurred) than random link weight setting approaches. Hence, we will start from the comparison between GA and SPH in the capability of exploring feasible solutions. Figure 32 presents the ratio of successful instances obtained by GA but failed to be found in SPH. In the figure, when the value of MLOR computed by SPH is in range of (0%, 5%], GA can obtain feasible solutions (i.e. $MLOR_{GA} \leq 0$) for 65% of these instances. We can also see that, with the increase of external bandwidth demands, the capability of GA in finding feasible solutions is decreasing. When the MLOR value of SPH grows up to 25% due to the higher external traffic demand, the success rate of GA drops to 5%. From this figure, it can be inferred that, when the external group traffic demand is at the brink of causing network congestion, GA has higher capability of avoiding link overloading compared to other approaches. Obviously, it may be the case that no feasible solution exists at all, if external traffic demand exceeds a certain threshold.



Figure 32: GA Success rate vs. MLOR_{SPH}

3.4.1.3.3 McastTE/Scal/OMTE-GA

The scalability test aims at the computing time required by the proposed GA based solution, particularly when large sized network topology and a large number of subscribed groups are considered.

The related GA configuration parameters for the following test are:

- (1) *Population* size = 100
- (2) $Maximum_generation = 300$

Topology size	10	50	100	200
Running time (s)	36	196	400	828

Table 6: Running time vs. topology size (100 groups)

Number of groups	50	100	150	200	
Running time (s)	201	400	602	801	

Table 7: Running time vs. number of groups (100 nodes)

3.4.1.3.4 Inter-domain McastTE

The following are some preliminary test results for inter-domain multicast traffic engineering. We evaluate the three algorithms that are specified in [D1.3], namely (1) greedy single ingress router selection (GSIRS), (2) hop-count based Hot Potato routing (HC-HPR) and (3) GA based Hot Potato Routing (GA-HPR). We evaluate both intra-domain bandwidth consumption performances and inter-domain load balancing performances.

The configuration of Inter-domain Multicast TE is described as follows: The total number of multicast groups is set to 50 with altogether 20 DRs. We consider 100 sources and each of them can be reached via half of the border routers on average. The rest configuration is the same as the intra-domain scenario.

Figure 33 and Figure 34 shows respectively the overall intra-domain bandwidth consumption and the highest inter-domain link utilisation with the variation of D_{g} . From the perspective of bandwidth

conservation, we can find that the two hot potato based routing approaches achieve significantly stronger capability than the single ingress router selection algorithm (GSIRS). However, from Figure

34 we can find that GSIRS has the best performance in terms of inter-domain link utilization among all the three solutions. On the other hand, the proposed GA based hot potato routing has resulted in higher link utilization by up to 10% compared to GSIRS, but it exhibits the best performance in intra-domain bandwidth consumption. Typically it only consumes 67% of bandwidth resources of GSIRS. From this point of view, we can regard the proposed GA based approach as a good trade-off example between intra- and inter-domain scenarios.



Figure 33: Total bandwidth consumption vs. Max D_g



Figure 34: Highest inter-domain link utilization vs. D_{σ}

3.4.1.4 Conclusions

From the above simulation results, we can find out that, compared to existing solutions, the proposed GA based OMTE algorithm can conserve significantly network bandwidth and is also able to guarantee higher success rate in finding feasible solutions with the bandwidth constraints. The running time for large sized networks and group numbers is within several minutes, which is feasible for offline TE computations.

3.4.2 Real-Time Test

3.4.2.1 *Overview*

In the real-time simulation tests, we mainly study the metric of blocking rate of group join requests based on the originally established mSLSes. The first objective (McastTE/Perf/DMR/1) is to study whether the proposed GA based OMTE algorithm is able to increase the service capability when individual mSLSes have been invoked. In addition, we also investigate the scenario when the proposed solution is applied to the DiffServ environment (McastTE/Perf/DMR/2), and see if effective service differentiation can be achieved without incurring any fairness issue between different l-QCs.

3.4.2.2 Experiment Setup and Test Description

Two different simulation scenarios have been applied in the real-time testing campaign. The first scenario (McastTE/Perf/DMR/1) is based on the flow level, which can be regarded as the continuation of the McastTE/Perf/OMTE-GA tests. The second (McastTE/Perf/DMR/2) is based on the packet level where the simulation is run on top of ns-2.

In the first scenario, we apply the same simulation model (topology, group membership information) as the one that is used in McastTE/Perf/OMTE-GA. Apart from that, We emulate a sequence of events for group membership updates based on the static scenario, and we evaluate the real-time traffic condition with the group dynamics derived from the original static multicast traffic matrix. For each event, we first randomly select one group $g \in G$, and then use the following probability function to decide whether this event is a group join or leave:

$$P_g = \frac{\omega(|V_g| - m_g)}{\omega(|V_g| - m_g) + (1 - \omega)m_g}$$

In the function, m_g indicates the instant number of active members while $|V_g|$ identifies the maximum size of group g (i.e. total number of subscribers). ω ranging [0, 1] is known as the invocation ratio that controls the density of each group. For example, $\omega=0$ means that no group joins are invoked, while $\omega = 1$ indicates full group membership invocation. In our simulation we use this function for creating a series of events of group join/leave based on the static multicast traffic matrix. When a join request is issued for group g ($P_g > a$ randomly created float number ranging from 0.0 to 1.0), a node $v \in V_g$ but not yet on the multicast tree T_g is selected to join the group. Likewise, in case of a leave request for group g, an on-tree node is randomly selected for pruning from T_g . By introducing this group dynamics generator, we can also investigate the stability of the proposed solutions in time of inaccurate mSLS invocations (i.e., not all receivers activate their contracts by sending group join requests).

In the ns-2 based simulation test, our configuration is as follows. The network shown in Figure 35 comprises two ingress routers (S1, S2), three egress routers (R1, R2, R3) and two core routers (C1, C2). The bandwidth capacity of each link is 10Mbps. The metric of each link is set to 1 so that the join request always follows the path with the minimum number of hops back to the source. We assume that the INP is providing 4 l-QCs, i.e. l-QC1, l-QC2, l-QC3 and l-QC4. The scheduling mechanism for individual l-QC queues is based on Weighted Round Robin (WRR), and the weight for each l-QC queue is set as follows:

$$\begin{bmatrix} l - QC1 \\ l - QC2 \\ l - QC3 \\ l - QC4 \end{bmatrix} = \begin{bmatrix} 4 \\ 3 \\ 2 \\ 1 \end{bmatrix}$$



Figure 35: ns-2 based simulation topology

3.4.2.3 Test Results

3.4.2.3.1 Functional Tests

The software is functioning correctly.

3.4.2.3.2 McastTE/Perf/DMR/1

In the following simulation tests, we assume that new group join requests will be blocked once network congestion (i.e., an overloaded link) has been detected. Figure 36 and Figure 37 show respectively one typical instance of the real-time performance (5000 events in group dynamics) in terms of overall network load and maximum link utilisation respectively, with D_g equal to 3000 and

 ω equal to 1.0. In this condition the network is lightly loaded with no link congestions (overprovisioning). From Figure 36 we can see that when the group dynamics converge to a steady state, the network load resulting from random link weight setting is the highest, while using the TM algorithm for MPLS explicit routing achieves the lowest resource consumption. We also find that the proposed link weight optimisation using the GA approach results in very low network load compared to other IP based approaches, and its performance is even very close to the TM explicit routing scheme. This result is consistent with the static simulation scenario shown in Figure 29. As shown in Figure 37, the GA optimisation approach results in very high utilisation of the most heavily loaded link, which is only next to the Random link weight solution. On the other hand, both the SPH and TM algorithms exhibit good performance in load balancing. Nevertheless, it should be noted that although the performance in maximum link utilisation by the GA approach is not as good as these two schemes, there is still no network congestion as all the links are under-utilised and the overall bandwidth resources are significantly conserved.



Figure 36: Real-time performance in average network load (Max D_g=3000, ω=1)



Figure 37: Real-time performance in maximum link utilisation (Max D_g =3000, ω =1)

From Figure 38 and Figure 39 (typical instances for over-subscription scenarios) we can see that the performance of the four approaches changes significantly in time of overwhelming traffic demand when Max D_g is increased to 6000. First, both the GA and SPH approaches converge to the highest overall network load. On the other hand, explicit routing with the TM algorithm still achieves the lowest resource consumption, which remains the same with the scenario in Figure 38. From Figure 39 we see that all four schemes result in 100% utilisation in the highest loaded link due to the overwhelming traffic demand, and thus some new group joins are blocked due to the overloaded links. We can also see from this figure that the random approach first converges to the congested state while our proposed GA optimisation is the last to reach this phase. This implies that more group joins are likely to be rejected in the former while the least join requests will be blocked in the latter. In effect, group join blocks prevent the underlying multicast trees from consuming more network resources, and this explicitly explains why the overall network load of SPH and GA is higher than the random link weight approach in Figure 38, where a large number of group joins have failed due to overloaded links. Our subsequent simulation study will continue to focus on the statistics of group join blocks for the four approaches in different scenarios (e.g., with variations of ω).



Figure 38: Real-time performance in average network load (Max D_g =6000, ω =1)



Figure 39: Real-time performance in maximum link utilisation (Max D_g =6000, ω =1)

3.4.2.3.3 McastTE/Stab/OMTE-GA/1

In this test campaign, we investigate the stability of the proposed algorithm in terms of inaccurate mSLS invocations. Figure 40 illustrates the overall block rate with the variation of the invocation ratio ω with respect to the 5000 group updates, while maximum D_g is set to 6000. From the figure we can

see that more group joins are rejected as the invocation ratio grows. The reason for this is that, bandwidth consumption increases when there are more active members in each group. Once the consumed bandwidth on any link reaches its capacity, new group joins are blocked due to the detected congestion. On the other hand, we notice that through sophisticated network dimensioning using the proposed MT-IGP link weight optimisation, group join blocks are significantly lower than in the other approaches. When ω increases from 0.5 to 1.0, the total number of blocks grows very slowly with our proposed GA solution, which is in contrast to all the other conventional methods. One interesting thing is that, compared to Figure 31 in the static scenario, although the provisioning performance of the GA approach results in 45% MLOR, the actual number of blocked join requests is quite low (2.1%) even in case of full group invocation. When $\omega < 0.7$, there are no blocked group join requests at all. The reason for this is that while there are overwhelming group joins, group leaves also take place at the

same time, with used bandwidth resources returned to the network. Finally, it is also worth mentioning that the MPLS based Steiner tree approach does not exhibit strong capability in reducing the blocking rate, as the TM algorithm is solely greedy in bandwidth conservation and not in eliminating congested links.





Figure 41 shows the overall network load versus invocation ratio ω with respect to the 5000 group updates. From the figure we can see that higher invocation ratio results in higher network load. On the other hand, the TM heuristic using MPLS explicit routing always achieves the lowest network load, which is in line with Figure 36. Moreover, we also notice that the network load of the GA optimisation is very close to that of the TM approach when ω is relatively small, and this again indicates that the proposed solution exhibits strong capability in bandwidth conservation in time of light traffic loading. However, with the growth of ω , the network load by the GA approach increases more sharply than all the other approaches, and this is because more group joins are able to be accommodated successfully, while in the other approaches, especially the random link weight one, a large number of join requests are blocked due to network congestion so that the total bandwidth consumption is relatively lower.



Figure 41: Network load vs. invocation ratio ω

3.4.2.3.4 McastTE/Perf/DMR/2

In our first experiment in this test campaign, there are two active groups whose sources send data via the ingress routers S1 and S2 respectively. For simplicity we only consider two classes of service in this experiment, i.e. I-QC1 and I-QC2. The source rate from S1 is 2Mbps and that from S2 is 1Mbps (both for I-QC1 and I-QC2). We also set 3Mbps background traffic (both I-QC1 and I-QC2) from each ingress router to all the egress routers. We consider the situation that each egress router joins both channels with I-QC1 and I-QC2 simultaneously, resulting in 4 distinct multicast trees: (S1, I-QC1), (S1, 1-QC2), (S2, 1-QC1) and (S2, 1-QC2). We define the Transmission Ratio (TR) as the number of packets received by each group member over the total number of packets sent by the source. Figure 42 illustrates the TR performance of each source/receiver pair. We can see that in most cases the TR performance of l-QC1 is significantly better than that of l-QC2 (except R2). By examining the traffic load of each link, we find that all the links between S1 and R2 (i.e., S1 \rightarrow C1 and C1 \rightarrow R2) are underloaded, resulting in 100% transmission ratio for both I-QC1 and I-QC2. On the other hand, the performance of transmission ratio also depends on the location of the egress router through which group members are attached to the distribution tree. For example, both egress routers R1 and R3 have 1-QC2 group members for S1. Our simulation results show that the TR value for R1 is 59.4% while that for R2 is significantly higher (77.6%). This is caused by the more overloaded link $C2 \rightarrow R1$.





Next we investigate the performance of individual group members attached to the same egress router. The objective of this experiment is to examine the inter-class fairness in more detail without considering the receivers' physical location. The simulation scenario is described as follows: The source rate of S2 is fixed at 1Mbps and 4 receivers attached to egress router R1 join the session by subscribing to 4 different channels, i.e. (S2, 1-QC1), (S2, 1-QC2), (S2, 1-QC3) and (S2, 1-QC4). The grey column of Figure 43 indicates that if none of the links on the tree branch S2 \rightarrow C1 \rightarrow C2 \rightarrow R1 are congested, the transmission rate of all the four classes is 100%. In order to evaluate the performance in time of congestion, we impose 3Mbps background traffic for each of the four 1-QCs. From the figure we can observe the significant differentiation of the four 1-QCs when the network cannot handle all the traffic. The group member subscribing to the 1-QC1 channel achieves virtually no packet loss, whereas the one subscribing to 1-QC4 channel only receives 24.6% of the packets from S2. The "Good Neighbour Effect" does not happen if we construct this type of QoS specific trees for each channel. This scenario demonstrates the benefit of building per QC trees for end users with different QoS requirements.



Figure 43: Simultaneous I-QC joins

Figure 44 illustrates the scenario when a particular group member dynamically upgrades its service level by joining higher QoS channels. Let's assume that a receiver attached to R1 subscribes to the (S2, 1-QC4) channel in time t_0 . Due to its capability to upgrade, this group member upgrades to the next higher QoS channel a number of times, i.e., at t_1 to 1-QC3, t_2 to 1-QC2 and t_3 to 1-QC1. From the figure we can see that this upgrading makes no improvement when there is no congestion along the tree from S2 to R1; in fact, the group member can achieve 100% packet transmission ratio at time t_0 by subscribing to (S2, 1-QC4). On the other hand, when we impose additional background traffic (in the same fashion to the last experiment), the performance differentiation of individual channels becomes obvious: the transmission ratio at t_1 goes up to 58.0%, to 73.3% at t_2 , and finally achieves 85.1% at t_3 when the user finally subscribes to the (S2, 1-QC1) channel.



Figure 44: Dynamic I-QC upgrading

3.4.2.4 Conclusions

From the real-time performance tests we found that both bandwidth conservation capability and service capacity achieved by the proposed GA approach is constantly higher compared with existing paradigms even when the mSLSes are implicitly invoked. These results have proved the high stability of the proposed algorithm. Moreover, the service capacity in terms of admitting group join requests has also been drastically increased compared with conventional approaches. Finally, we also indicated in our simulation that inter-l-QC fairness problems are avoided by applying per l-QC trees.

4 DYNAMIC TRAFFIC ENGINEERING TESTS AND RESULTS

4.1 q-BGP Simulation Tests

This section focuses on the results of a macro-scale simulation of inter-AS topologies aimed at evaluating q-BGP's large-scale behaviour – tests that would be unfeasible on the relatively small-scale MESCAL testbed. The simulation models aggregate flows rather than individual packet behaviour as it implies an enormous simulation overhead when considering large inter-AS topologies and furthermore that level of simulation detail is unnecessary for the macroscopic behaviour under investigation.

It should be noted that these experiments are aimed at evaluating the use of q-BGP in the context of MESCAL's loose guarantees solution option (LGSO) and that since each meta-QoS-class runs a separate instance of q-BGP, will limit our simulations to a single QoS-class plane. This can be justified by the assumption that there is a partitioning between meta-QoS-classes at the pSLS level and therefore no interaction or interference between meta-QoS-class planes. The experiments and results obtained cover the following aspects:

- Scalability, which examines how the number of q-BGP messages depends on variables such as network size, topology, and traffic demand patterns.
- Stability, which considers the sensitivity of the q-BGP routing algorithms and protocol to changes in the inter-domain network and their ability to settle in a stable state.
- Efficacy, which considers the ability of q-BGP routing algorithms to find the optimal routes for a given demand matrix. Optimal is considered to be an inter-domain routing configuration that will accommodate demands with an acceptable level of QoS with minimal resource usage (e.g. inter-domain link usage).

4.1.1 Simulation Scenarios

When simulating an inter-domain network with inspection of QoS performance, a number of experimental variables play a large role in the resulting performance. These were discussed in Deliverable D3.1 and include:

- **Inter-domain Topology:** A representative inter-AS topology is required which we obtain from the BRITE topology generator. This creates power-law compliant topologies [bu02] when its preferential attachment option is used, and it has been shown that the Internet is also a power-law compliant topology at the AS level [fal99]. Parameters include network size (number of ASs) and average connectivity (the number of inter-domain links per AS).
- **Demand Matrix:** This is the traffic to be applied to our network. The demand matrix comprises a full mesh of demands between all AS pairs and whose offered bandwidth is uniformly randomly distributed across all demands. The parameter for the demand matrix generator is total network demand, so therefore the average demand bandwidth is this total bandwidth divided by ½.N.(N-1), where N is the network size in number of ASs. The absolute values chosen for the total demand bandwidth aren't important (as long as they are high enough to prevent rounding errors) as the pSLS capacities are directly derived from this matrix.
- **pSLS Capacity Matrix:** This is the scarce resource for which q-BGP is to optimise routing for. The pSLS generator described in Deliverable D3.1 provides a "base" pSLS capacity matrix which is capable of satisfying the input demand matrix but only with a single very specific routing configuration. So as not to favour shortest-path routing strategies the available capacity is placed on paths away from the shortest path. It would be very difficult for any routing policy to find the exact correct routing configuration that was used to generate the pSLS capacities so a pSLS scaling co-efficient is used to scale the capacities. Scaling the

capacities and therefore over-provisioning pSLSs, the number of alternative paths with available resources increases and it becomes easier for the routing policy to find a suitable routing configuration. This pSLS over-provisioning co-efficient is therefore a parameter to influence the number of suitable routing configurations. q-BGP strategies which require less over-provisioning to achieve good end-to-end QoS characteristics for demands are therefore the better solutions. As the pSLS capacities are in "useful" locations in the network an examination of pSLS utilisation is a meaningful measure of network resource utilisation. For the 100 AS experiments here the average shortest path between every AS pair was 2.8, however pSLS capacities were allocated on paths which had an average length of 4.6.

- Aggregate flow treatment model: as part of the calculation of end-to-end delay and delivered bandwidth we must emulate the effect of network congestion on packet flows through the network. To this end we use a simple M/M/1 queue to approximate queuing delay at pSLSs. Since router buffers are finite the delay experienced is capped at 100 ms. Demands have been implemented to perform as if they were inelastic and if along the path of the demand there is not enough capacity available then the demand will experience a degradation in throughput for the successive hops. The division of available pSLS capacity between demands is performed to the ratio of incoming offered flow bandwidth.
- I-QC Generator: For simplicity it is assumed that within each AS there are pre-defined I-QCs between all ASBRs (AS Border Routers). The ASs are assumed to have sufficient bandwidth for accommodating demands for the offered I-QCs (since the scarce resource which we are optimising for is the pSLS capacity and I-QCs should be matched anyway to pSLSs), and a fixed QoS transfer characteristic (i.e. a fixed delay). For these experiments the delay parameters are generated with a uniformly random number generator between the bounds of 5 and 50 ms and remains constant for all the I-QCs within a single AS, but differ between ASs.

4.1.2 q-BGP Policies under test

4.1.2.1 QoS_NLRI QoS Attributes

In these experiments we'll be concentrating on two QoS Attributes (QAs):

4.1.2.1.1 One-Way Delay (OWD) QA:

This is the expected time for a packet to reach the prefix advertised. When traversing pSLSs and ASs this value is formed through the concatenation of the various delay contributors:

Advertised OWD QA = incoming advertisement OWD QA + 1-QC delay + pSLS queuing delay;

When calculating the actual *delivered* end-to-end delay the value calculated from the aggregate flow treatment model is used instead.

4.1.2.1.2 Bandwidth (BW) QA:

This is a value for available bandwidth to the prefix specified in the NLRI field. l-QCs are assumed to have sufficient bandwidth so the only restriction is the pSLS capacities, thus the value advertised becomes:

```
Advertised BW QA = min (incoming advertisement BW QA, offered pSLS capacity);
```

As there is no injection of dynamically monitored QoS attributes in this set of experiments the "offered pSLS capacity" is specified as the capacity of the pSLS on which the q-BGP advertisement came in, divided by the number of neighbours to which this message is to be re-advertised.

4.1.2.2 Route Selection Policies

Throughout the experiments we examine a number of route selection policies which make use of various combinations of the QoS attributes. For added variability of policies we also use a QoS

attribute equivalence margin. This margin effectively introduces a comparison granularity to QoS attributes. This approach is similar, but different to the precision parameter described in section 10.5.1.5.5.2.2.2.4 of deliverable D1.3.

In the following simulations QA equivalence is calculated by:

if(floor(MessageA_QA / QAmargin) = floor(MessageB_QA / QAmargin))

then the messages are equivalent and the decision must be performed on the next metric.

The route selection processes examined here are:

4.1.2.2.1 Meta-QoS-Class Identifier Only (MCID-only)

Routing decisions are based purely on AS Path length, and use ASN (AS number) as a tie-breaker.

4.1.2.2.2 Delay QoS Attribute only (DELAYQA-only)

The routing decision is performed based on a One Way Delay (OWD) QoS attribute first, and then on AS path length and ASN. The value of the OWD is static throughout the simulation and calculated as described in section 4.1.2.1.1. A range of equivalence margins, pOWD, for OWD are also examined.

4.1.2.2.3 Bandwidth QoS Attribute only (BWQA-only)

The routing decision is performed based on the bandwidth QoS attribute. The advertised BW QA is as described in section 4.1.2.1.2. i.e. if an AS has eight neighbours and one of these sends an incoming message advertising 49 bandwidth units to the prefix in the NLRI, and the pSLS capacity is 100 units then the AS will send seven messages, each advertising 7 bandwidth units to the prefix in the NLRI. A range of equivalence margins, pBW, for BW are also examined.

4.1.2.2.4 Delay and Bandwidth Priority scheme (DELAYBWPRIO)

A two level priority scheme where depending on the priorities specified in the policy either one of OWD QA or BW QA is checked first, and then if found equivalent (depending on the pBW and pOWD parameters) the other QA is checked. If that too is equivalent the decision is the based on AS path length and the ASN.

4.1.3 Experimental overview

We examine three aspects of q-BGP policies, as described in the introduction to section 4.1: Scalability, Stability and Efficacy. To limit the range of parameters for the various policies under test we must first find regions of parameter space which perform well in efficacy tests to continue are analysis, otherwise the graphs become cluttered and have little additional value. Any additional parameters are described as part of the experiment groups below. As mentioned before all experiments are for a single meta-QoS-class in MESCAL LGSO (loose guarantee solution option).

4.1.4 Experimental results: efficacy

The experiments here were all performed on network topologies of 100 ASs with an average connectivity degree of four unidirectional links. Each set of parameters was repeated 16 times and the results averaged. Error bars are derived from the standard deviation of the mean for each simulation run and not each individual metric. i.e. the error bars on pSLS utilisation are the standard deviation of the mean pSLS utilisation for each network and not the mean of the standard deviations for all pSLS within each network.

Our first inspection of efficacy is to examine the effect of QA equivalence margins and to find a range of useful values for pBW and pOWD.



Figure 45 Mean delivered bandwidth fraction (delivered/offered) for a range of pBW under the BWQA-only policy

In Figure 45 we can see that when resources are scarce (pSLS SF = 1.0) the delivered bandwidth is low for no margin (pBW = 1.0) but as the margin increases the fraction of delivered (to offered) bandwidth improves. This continues until the margin is so large that the majority of route selections then fall to AS path length where the delivered fraction becomes worse again. For reference the delivered BW fraction (and for later reference the mean delivered end-to-end delay) for MCID-only is:

pSLS SF	Mean Delivered BW fraction	Mean delivered end-to-end delay (ms)
1	0.4713	201.7
1.5	0.5446	193.6
1.75	0.5733	190.2
2	0.5986	189.7
2.25	0.6216	187.9
2.5	0.6422	183.6
3	0.6777	177.0
4	0.7347	171.4

Table 8 Mean delivered BW fraction and delivered end-to-end delay for the MCID-only

We hypothesise that the cause of the poor initial performance of BWQA-only with pBW = 1 is the convergence of routing paths towards the areas of high capacity and therefore the saturation of those links. As the QA values are static and administratively set they won't change to reflect this saturation

and the overall throughput for demands suffers. We refer to this phenomenon as the "QA rush". The effect can be also seen in the average utilisation of pSLSs in Figure 46:



Figure 46 Mean pSLS utilisation for a range of pBW equivalence margins for the BWQA-only policy

As the pBW margin increases route selection is no longer performed purely on the BW QA, but also on AS path length, then there is less of a rush towards the high capacity links. This can be seen here as an increase in the average pSLS utilisation as more of the demand gets through the bottlenecks and the network load is better distributed across the network.

A second performance metric is mean delivered end-to-end delay. In our simulations the per-domain delay is chosen at random from the range 5 to 50 ms, while the administratively set pSLS queuing delay contribution is considered to be 7 ms. The figure of 7 ms is really a forecast of the very worst case scenario and corresponds to a pSLS utilisation of 0.875, assuming the queue behaves as an M/M/1 process. The sum of these values is the OWD QA metric advertised in the qBGP messages. However, when calculating the delivered end-to-end delay the per-domain delay is used alongside the *value calculated by the* M/M/1 queue equation (D3.1) (which is capped to 100 ms). This will provide a feedback into delivered delay from highly utilised pSLSs. This is so as to model the benefit to delivered delay from BW QA based policies.



Figure 47 Mean delivered delay for various BW QA equivalence margins for the BWQA-only policy

In Figure 47 we see that for all over-provisioning factors pBW margin values below 125000 results in a much worse delivered delay, and higher values yield a much less significant improvement. This would suggest that a pBW value of 125000 is beneficial to end-to-end delivered delay. This improvement in delivered delay is the result of better load distribution, more available capacity, and therefore faster queuing service rates and a lower queuing delay.

The graph in Figure 48 examines the effect of the OWD equivalence margin on the actual delivered delay. When inter-domain resources are scarce (pSLS SF = 1.0) and there is no equivalence, all decisions are made purely on OWD QA and any advertisements of low delay routes cause "QA rush" causing poorer delivered performance. As the equivalence margin increases a number of alternative routes appear across which load is distributed, leading to less congestion and better delivered delay. As the margin increases further the delay gets worse (more significantly for the high pSLS SF cases) and the routes approach the shortest-path and the resulting delivered delay becomes more like the MCID only case (see Table 8 for comparison). A general purpose value for pOWD margin is therefore 50 ms as this is approximately a point where performance changed for the entire range of pSLS SFs.



Figure 48 Mean delivered delay for various OWD QA equivalence values for the DELAYQAonly policy

In Figure 49 we can see the fraction of the offered load that is actually delivered for a range of route selection policies. This time we compare a range of policies against the amount of over-provisioning. The pBW chosen (125000) for the BWQA-only policy is seen as one of the better values from the previous graph, similarly a pOWD (the OWD equivalence margin) of 50 is found to perform better than other values, e.g. Figure 48.



Figure 49 The mean delivered bandwidth fraction over a range of over-provisioning coefficients for the various q-BGP policies

For the delivered bandwidth fraction performance metric MCID initially outperforms BWQA-only with no equivalence margin (pBW = 1), but with an increase in over-provisioning the BW QA based policies outperform all other policies. MCID is outperformed by all policies, including the policies based on OWD QA which follow a non-shortest path dictated by the per-domain delays, adding heterogeneity and a certain level of load balancing.



Figure 50 Mean delivered delay for a select range of policies against the over-provisioning coefficient

In Figure 50 we can see the mean delivered delay as experienced by all demands. The plot demonstrates that BWQA-only with no equivalence margin results in very poor delays in the network. This is caused again by the "QA rush", and as over provisioning is increased the delays drop as the bottlenecks are lessened. With an equivalence margin the BWQA-only actually performs a bit better because of the load distribution afforded by the shortest path decisions. What is interesting also from this graph is that DELAYQA-only doesn't perform significantly better than MCID-only. This is probably caused by the random distribution of 1-QCs and because of the homogeneity of pSLS administratively set delay contributions (the 7 ms) effectively denature the path into a shortest-path equivalent. Any benefit from choosing a lower delay path by DELAYQA-only may also be eroded by a "QA rush" effect on that low delay path and forcing queuing delays higher until the result isn't much better than shortest-path.

4.1.5 Experimental results: comparison of q-BGP selection policies

This section analyses the relative performance of a range of different q-BGP selection policies in terms of their impact on the delivered delay and delivered bandwidth of end-to-end flows. The q-BGP route selection policies subject to test and comparison are:

- MCID-only. Given that the simulations focus on a single meta-QoS-class plane these tests are effectively without any additional QoS information injected into q-BGP and are therefore equivalent to classical BGP.
- Single QoS attributes of DELAYQA-only and BWQA-only. The performance of the q-BGP selection process where there is no margin of equivalence is compared to the delivered QoS when a margin is used. The values selected for the equivalence margin under test are those that exhibited good performance in the tests described in the previous subsection.
- Priority-based selection on both DELAYQA and BWQA attributes comparing the performance of DELAYQA having priority over BWQA and vice versa. A range of equivalence margin values were used for these tests based on the results obtained for single QoS attribute tests as described in the previous subsection.

Figure 51 shows a scatter plot of mean delivered delay against mean delivered bandwidth for a range of q-BGP selection policies. The results are shown for three pSLS over provisioning factors -1.5, 2.0 and 2.5 – shown from left to right on each of the curves. Although results were obtained for many cases of equivalence margin value, for clarity the comparison graph concentrates on results from a more limited set of values, selected to highlight the major differences between the selection policies.



Figure 51 Effect of q-BGP selection policy on delivered delay and bandwidth

The policy of selection based on BWQA-only with no equivalence margin delivers higher bandwidth fractions than MCID-only for higher pSLS over provisioning factors, but performs worse than MCID-only in congested networks. The reason for the latter is due to the phenomenon of QA-rush as described earlier in the section. In all cases the adoption of the BWQA-only policy shows worse delivered delay than MCID-only, due to it selecting the largest capacity route at any cost. By adding a margin of equivalence, e.g. of 125000 bandwidth units as shown for the BWQAONLY-125000 curve, the performance is improved in terms of delivered delay and bandwidth when compared to selection based on the absolute widest path. This also beats MCID-only in terms of delivered bandwidth fraction but not one way delay. The policy of using an equivalence margin improves performance because the QA-rush has been avoided by increasing the number of equivalent bandwidth paths and allowing route selection within the set of best bandwidth paths to be done on the basis of AS path-length, thereby adding diversity to the overall routing behaviour.

The policy of selection based on DELAYQA-only shows some improvement over selection based on shortest AS path (MCID-only) in terms of both delay and delivered bandwidth. However the improvement is marginal. One of the reasons for this is that in the simulation scenarios – as in the real world – the shortest AS path is often the one with shortest delay. If the simulated inter-AS topology is

selected carefully so that the ASs along shortest path routes have large l-QC delays then a more marked improvement in performance of the DELAYQA-only selection policy may be observed.

The best performing route selection policies are those that select paths according to advertised delay and bandwidth. PRI DEL-100 BW-75000 is first of all selecting paths on the grounds of smallest advertised delay, with a margin of equivalence of 100 ms, and subsequently selecting between these on the basis of widest advertised bandwidth with a margin of equivalence of 75000 bandwidth units, falling back on AS path length and finally AS number if a tie breaker is required. This policy delivers the best overall performance in terms of bandwidth and delay at all three pSLS over provisioning factors. It is interesting to compare this to PRI BW-75000 DEL-100 - i.e. the same bandwidth and delay margins, but with the priority reversed (first select based on bandwidth then on delay). In the latter case delivered bandwidth and delay is worse than the former and worse than selection based on BWQA-only with a wider margin of equivalence. However it can be seen that with different margins of equivalence, a selection policy with the same priority order of QoS attributes can deliver significantly improved delay/bandwidth performance. This can be seen by comparing PRI BW-75000 DEL-100 with PRI BW-175000 DEL-50. It appears, therefore, that it is better for the path selection process not be too narrow in its choice of the set of best paths on the highest priority QoS attribute so that more potential paths are passed to the selection step based on the 2^{nd} priority attribute and therefore a greater chance of finding a good path according to the 2nd priority QoS attribute.

One notable conclusion to be drawn from these comparisons is that a range of different performances can be achieved through applying different q-BGP selection policies. This means that different meta-QoS-classes may require different selection policies to implement desired end-to-end behaviour. It is important to state that this is in addition to any service differentiation implemented by utilising different PHBs/packet forwarding priorities within the routers of each AS. On the other hand it also indicates that end-to-end QoS differentiation is achievable even with homogenous forwarding behaviour for all traffic classes, e.g. BE only as in the current Internet.

4.1.6 Experimental results: scalability

The improvement in delay can be seen in Figure 52 as a function of AS topology size. It can be seen that the benefit of additional QoS info (delay QA only with no margin in the shown tests) in q-BGP messages is increasing with topology size. This is due to an increased number of alternative AS paths between a given source-destination pair (other than the default shortest AS-path length) as the topology grows, and therefore the chances of finding an improved path on one-way delay grounds is increased.



Figure 52 Q-BGP scalability: mean one way delay versus number of ASs

The use of additional QoS info in q-BGP brings an additional overhead in terms of an increased number of q-BGP UPDATE messages. Figure 53 shows the total number of q-BGP messages sent from the first set of bootstrap messages through to a stable routing configuration comparing MCID-only (equivalent to classical BGP) with DELAYQA-only selection policies. It should be noted that no equivalence margin was set for the DELAYQA-only test, and that this represents the worst case in terms of quantities of UPDATE messages generated for this class of q-BGP selection policy. Further, it should be noted that no message aggregation is performed in these simulations, either on network prefixes or QoS attributes.



Figure 53 Q-BGP scalability: number of q-BGP messages sent from initialisation until it settles in a stable state with a full mesh of demands applied

When the two plots are extrapolated to a topology size of 18,000 ASs the q-BGP category two routing scheme produces only approximately three times as many messages as q-BGP messages conveying MCID only. The inclusion of additional QoS info in q-BGP therefore scales, in terms of number of q-BGP messages, in a similar way to q-BGP UPDATES and route selection based on MCID only. By this we mean that the number of messages forms a power law with topology size, which is equivalent to the scaling of BGP today.

The main reason for the increased number of messages required for convergence is that, on QoS grounds, the preferred AS path may not always be the shortest one. Imagine, from the perspective of the AS receiving q-BGP UPDATEs that the shortest AS path to a particular destination prefix has three AS hops, but the total one-way packet delay (in the data plane) as reported in q-BGP is significantly greater than an alternative five-hop AS path. According to the q-BGP message received via the neighbouring AS announcing the 3-hop path is likely to arrive earlier than the one from the other neighbouring AS announcing the 5-hop path, due to the accumulation of processing time and propagation delay of the q-BGP route selection process at each intermediate AS. In the absence of the 5-hop shorter-delay announcement, q-BGP will select the first route and announce this to its peers. On receipt of the subsequent announcement of the shorter-delay path, q-BGP messages and introducing a transient routing instability. One could imagine a scheme where an AS would not make immediate decisions, but wait for some period to be sure that it has received all likely updates. This would improve the transient stability of the solution but at the cost of longer convergence times.

4.1.7 Experimental results: stability

Table 9 shows convergence time for a range of q-BGP path selection policies for a topology of 100 ASs with a pSLS provisioning factor of 2.0. Convergence time is measured as the number of simulation epochs required for all ASs to stabilise in terms of their path selection. Convergence is identified when no further q-BGP UPDATE messages are transmitted.

q-BGP Selection Policy	Average number of simulator epochs until convergence
MCID-only	9.5
DELAYQA-only (pOWD=50)	10.4
DELAYQA-only (no margin)	10.8
PRI_DEL_BW (pOWD=100, pBW=75000)	13.2
BWQA-only ($pBW = 125000$)	16.1
PRI_BW_DEL (pBW=175000, pOWD=50)	16.4
BWQA-only (no margin)	17.6

Table 9 Convergence time versus q-BGP selection policy

One reason for longer convergence times for some selection policies, e.g. BWQA-only with no margin of equivalence, is that they will determine that a newly arriving q-BGP UPDATE is better that the currently implemented path even if the new path outperforms the current one by only a tiny fraction. This will cause the AS to advertise its new path, which in turn will cause its neighbours to select the marginally better path, causing more q-BGP messages to be generated, and so on.

4.1.8 Conclusions

The results show that performance in terms of delivered end-to-end delay or bandwidth is improved when q-BGP selection policies are employed to select paths based on QoS attributes injected into BGP messages. However, if the equivalence margin of QoS attributes on competing paths is set too small then a degradation of performance compared to that offered by classical BGP selection policies may be observed due to the observed phenomenon of "*QA rush*", where the best routes are quickly overloaded. This can be mitigated by increasing the margin of equivalence so that, while the worst paths are excluded, sufficient quantities of "good" paths are retained so that the subsequent selection between these, based on shortest AS path, results in sufficient routing diversity which alleviates congestion.

It has been demonstrated that different route selection policies result in different delivered performance. Appropriate policies should, therefore, be selected to implement different meta-QoS-classes – e.g. delay or bandwidth constrained qualitative classes. It is important to state that this is in addition to any service differentiation implemented by utilising different PHBs/packet forwarding priorities within the routers of each AS. On the other hand this result indicates that end-to-end QoS differentiation is achievable even with homogenous forwarding behaviour.

While the performance benefits of QoS-based path selection have been demonstrated it has also been shown that the cost of the solution is not prohibitive in terms of the overhead caused by additional q-BGP UPDATE messages. Simulation results of the worst-case value of equivalence margin for the DELAYQA-only q-BGP path selection policy, i.e. no margin, show that the number of q-BGP messages required for stable inter-domain routing scales with AS-topology size in a similar way to classical BGP. When scaled to current Internet topologies the results indicate that only three times the

number of UPDATE messages is needed for convergence compared to classical BGP. With larger equivalence margins the total number of messages is reduced.

Stability tests show that convergence times are worst when q-BGP selection policies are most stringent. The adoption of these policies also delivers worse end-to-end performance and it is desirable on the counts of both convergence time and delivered QoS to adopt more moderate equivalence margin values. The results show that when the best performing q-BGP selection policies (in terms of delivered QoS) are adopted, convergence time is in the mid-range of observed values.

4.2 Data Plane Testbed Tests

4.2.1 Overview

The data plane testbed experiments were carried out in order to verify that the network is set-up and operates correctly for conducting the tests in further phases. Especially, the objectives of experimentations conducted within this phase are to verify that routing and QoS configuration detailed in [D3.1] are correctly deployed and implemented. In addition, these experiments aim at verifying that policing and shaping policies are correctly configured in all testbed ASs.

4.2.2 Experiment Setup and Test Description

The environment to execute these tests is the MESCAL testbed deployed in FTR&D premises. The testbed is composed of eight ASs and ten Linux-based routers. All ASs are composed of a single Linux-based router except AS4 which is composed of three routers. Four local QoS Classes are configured in each AS. Shaping and policing are configured in testbed routers. BGP is configured to run between two neighbouring ASs. For more detailed information about the configuration of the testbed for this phase refer to section 9. This configuration will be used as it is for executing these tests except when there are explicit recommendations in the procedure tag.

4.2.3 Test Results

In this section, we provide a list of tests that have been carried for this phase. Detailed results are provided in section 10. The experiments carried out during this phase are composed of several test groups (referred to as TB_P1_FUNCT group) that contain the following test suites:

Test Suite Id	Objective
TB_P1_FUNCT/ROUT	This group of test aims at verifying the routing features, especially the activation of BGP and reachability aspects.
TB_P1_FUNCT/DSSW	This group of tests aims at verifying the DSCP swapping operations in ingress and egress of a domain.
TB_P1_FUNCT/SHAP	This group test aims at verifying shaping operation.
TB_P1_FUNCT/POLI	This group of test aims at verifying policing issues.
TB_P1_FUNCT/BWMA	This group of test aims at examining the bandwidth management.

Table 10: Phase 1 Test Suites

The table Table 11 hereafter gives the status of sub group test results:

Test Id	Purpose	Status
TB_P1_FUNCT/ROUT/1	Validate inter-domain link connectivity.	SUCCESSFULLY PASSED
TB_P1_FUNCT/ROUT/2-12	Validate connectivity between two neighbours when BGP process is activated.	SUCCESSFULLY PASSED
TB_P1_FUNCT/ROUT/13	Check the route propagation in a simple scenario.	SUCCESSFULLY PASSED
TB_P1_FUNCT/ROUT/14	Check the reachability of all interfaces.	SUCCESSFULLY PASSED
TB_P1_FUNCT/ROUT/15	Verify reachability status when link failure occurs.	SUCCESSFULLY PASSED
TB_P1_FUNCT/ROUT/16	Verify reachability status when a link failure is re- established.	SUCCESSFULLY PASSED
TB_P1_FUNCT/ROUT/17	Verify intra-domain routing in AS4 domain.	SUCCESSFULLY PASSED
TB_P1_FUNCT/DSSW/1-10	Verify DSCP swapping at egress of MESCAL11, MESCAL21, MESCAL31, MESCAL41, MESCAL42, MESCAL43, MESCAL51, MESCAL61, MESCAL71 and MESCAL81.	SUCCESSFULLY PASSED
TB_P1_FUNCT/DSSW/11- 20	Verify DSCP swapping at ingress of MESCAL11, MESCAL21, MESCAL31, MESCAL41, MESCAL42, MESCAL43, MESCAL51, MESCAL61, MESCAL71 and MESCAL81.	SUCCESSFULLY PASSED
TB_P1_FUNCT/DSSW/21	Verify QoS configuration of the whole testbed.	SUCCESSFULLY PASSED
TB_P1_FUNCT/SHAP/1-11	Verify shaping configuration in MESCAL11, MESCAL71, MESCAL81, MESCAL51, MESCAL43, MESCAL41, MESCAL42, MESCAL21, MESCAL31 and MESCAL61.	SUCCESSFULLY PASSED
TB_P1_FUNCT/POLI/1-11	Verify policing configuration in MESCAL11, MESCAL71, MESCAL81, MESCAL51, MESCAL43, MESCAL41, MESCAL42, MESCAL21, MESCAL31 and MESCAL61.	SUCCESSFULLY PASSED
TB_P1_FUNCT/BWMA/1- 11	Verify bandwidth management configuration in MESCAL11, MESCAL71, MESCAL81, MESCAL51, MESCAL43, MESCAL41, MESCAL42, MESCAL21, MESCAL31 and MESCAL61.	SUCCESSFULLY PASSED

Table 11: Phase 1 Tests results

4.2.4 Conclusions

The results obtained during this phase certify that the configuration of the testbed is aligned with its objectives especially the following features:

- Configuration of local QoS classes: DSCP, prioritisation, bandwidth pre-emption
- DSCP marking and remarking at ingress and egress interfaces
- Shaping and policing at the boundary of domains
- Bandwidth pre-emption between a meta-QoS-class and BE configured in a given inter domain link
- Routing aspects.

4.3 q-BGP Testbed Tests

4.3.1 Overview

The objectives of experimentations of this phase are mainly: to (1) test the q-BGP messages conformance specifications enclosed in [D1.2], (2) to validate QoS computation as implemented by q-BGP machinery, (3) to validate the route selection process and finally to (4) validate DSCP swapping operations as implemented in q-BGP, especially validate the QoS route-map introduced in ZeboS.

4.3.2 Experiment Setup and Test Description

The environment to execute these tests is the MESCAL testbed that is deployed in FTR&D premises. The configuration of the testbed for this phase is detailed in section 9. This configuration will be used as it is for executing these tests except when there are explicit recommendations in the procedure of the test.

4.3.3 Test Results

In this section, we provide a list of tests that have been carried for this phase. Detailed results are provided in section 10. The experiments carried out during this phase are composed of several test groups (referred to as TB_P2_FUNCT group) that contain the following test suites:

Test Suite Id	Objective
TB_P2_FUNCT/CMES	This group of test aims at verifying the conformance of q-BGP messages such as defined in [D1.2].
TB_P2_FUNCT/DSCP	This group of tests aims at verifying DSCP swapping operations for ingress and egress.
TB_P2_FUNCT/QCMP	This group tests aims at verifying basic computation of QoS information between two peering ASs.
TB_P2_FUNCT/RSEL	This group of tests aims at verifying the behaviour of q-BGP route selection algorithm such as defined in [D1.2]. It also verifies more complex computation of QoS information.
TB_P2_FUNCT/QFIB	This group of tests aims at verifying that QoS-enabled entries are correctly installed in FIB via q-BGP.
TB_P2_FUNCT/INT	This group of tests aims at verifying the interoperability of q-BGP and BGP.

Table 12: Phase 2 Validation Test Suites

TB_P2_FUNCT group contains the tests in Table 13.

Test Id	Purpose	Status
TB_P2_FUNCT/CMES/1	Verify the capability length.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/2	Verify the QoS service capability field length.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/3	Verify that Group 1 QoS service capability is supported.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/4	Verify that Group 2 QoS service capability is supported.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/5	Validate the conformance of QoS information length.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/6	Verify that "Packet Rate QoS Code" and its associated Sub- codes are supported.	SUCCESSFULLY PASSED
----------------------	---	------------------------
TB_P2_FUNCT/CMES/7	Validate that "One Way Delay QoS Code" and its associated Sub-codes are supported.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/8	Validate that "Inter-Packet Delay Variation QoS Code" and its associated Sub-codes are supported.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/9	Validate the QoS information value.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/10	Verify that QoS class identifier can be set to a value that is between 0 and 63.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/11	Validate the QoS Origin field.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/12	Check the validity of Address Family Identifier (AFI).	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/13	Check the validity of Subsequent Address Family Identifier (SAFI).	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/14	Check the validity of Network Address of Next Hop.	SUCCESSFULLY PASSED
TB_P2_FUNCT/CMES/15	Verify the conformance of NLRI field.	SUCCESSFULLY PASSED
TB_P2_FUNCT/DSCP/1-2	Validate that egress DSCP swapping operation is correctly achieved when receiving BGP UPDATE messages.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/1	Check that the <i>reserved-rate</i> QoS parameter is correctly computed by the receiving ASBR.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/2	Check that invalid <i>reserved-rate</i> values are rejected by the command-line interface.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/3	Check that the <i>available-rate</i> QoS parameter is correctly computed by the receiving ASBR.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/4	Check that invalid <i>available-rate</i> values are rejected by the command-line interface.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/5	Check that the <i>min-owd</i> (minimum one-way-delay) QoS parameter is correctly computed by the receiving ASBR.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/6	Check that invalid <i>min-owd</i> values are rejected but the command-line interface.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/7	Check that the <i>max-owd</i> (maximum one-way-delay) QoS parameter is correctly computed by the receiving ASBR.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/8	Check that invalid <i>max-owd</i> values are rejected by the command-line interface.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/9	Check that the <i>average-owd</i> (average one-way-delay) QoS parameter is correctly computed by the receiving ASBR	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/10	Check that invalid <i>average-owd</i> values are rejected by the command-line interface.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/11	Check that the <i>loss-rate</i> QoS parameter is correctly computed by the receiving ASBR.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/12	Check that invalid <i>loss-rate</i> values are rejected by the command-line interface.	SUCCESSFULLY PASSED

TB_P2_FUNCT/QCMP/13	Check that the <i>jitter</i> QoS parameter is correctly computed by the receiving ASBR.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/14	Check that invalid <i>jitter</i> values are rejected by the command-line interface.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/15	Check that the receiving ASBR is able to compute multiple QoS parameters contained in an announcement.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QCMP/16	Check that the receiving ASBR is able to compute multiple QoS parameters for a same prefix announced within different meta-QoS-planes.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/1	Check that several ASs involved in the loose service option are able to exchange route updates containing correctly computed QoS information.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/2	Check, in simple Scenarios, that the route selection process takes into account the priority level of each QoS attribute.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/3	Check that the route selection process takes into account the QoS attributes which have a lower priority when the previous attributes (with higher priority) are equivalent.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/4	Check that the precision command line parameter is correctly handled for the reserved-rate QoS attribute.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/5	Check that the precision command line parameter is correctly handled for the available-rate QoS attribute.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/6	Check that the precision command line parameter is correctly handled for the loss-rate QoS attribute.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/7	Check that the precision command line parameter is correctly handled for the min-owd QoS attribute.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/8	Check that the precision command line parameter is correctly handled for the max-owd QoS attribute.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/9	Check that the precision command line parameter is correctly handled for the average-owd QoS attribute.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/10	Check that the precision command line parameter is correctly handled for the jitter QoS attribute.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/11- 17	Validate the behaviour of q-BGP when mandatory parameters aren't received.	SUCCESSFULLY PASSED
TB_P2_FUNCT/RSEL/18- 24	Validate the behaviour of q-BGP when optional parameters aren't received.	SUCCESSFULLY PASSED
TB_P2_FUNCT/QFIB	A series of elementary tests will be carried out in order to verify the correct installation of QoS-based routes in the q-FIB table.	SUCCESSFULLY PASSED
TB_P2_FUNCT/INT/1	Validate the behaviour of a BGP speaker when receiving unrecognised optional parameters.	SUCCESSFULLY PASSED
TB_P2_FUNCT/INT/2	Validate the behaviour of a q-BGP speaker when receiving notification set to unsupported capabilities from BGP speaker.	SUCCESSFULLY PASSED
TB_P2_FUNCT/INT/3	Validate the q-BGP router installs routes received from BGP speaker in best effort plane.	SUCCESSFULLY PASSED
TB_P2_FUNCT/INT/4	Validate the BGP router installs routes received from q-BGP speaker.	SUCCESSFULLY PASSED

Table 13: Phase 2 Validation Tests results

4.3.4 Conclusions

The obtained results of this phase experiments certify that:

- q-BGP implementation is aligned with specifications
- q-BGP QC-Id swapping operation are correctly implemented
- QoS computation as achieved by q-BGP conforms with specification
- q-BGP route selection process conforms to what has been specified in [D1.3]
- q-BGP interoperates with classical BGP
- QoS-enabled routes could be installed in q-FIB and that q-BGP can successfully install QoSenabled entries in q-FIB.

4.4 PCS Testbed Tests

4.4.1 Overview

The objectives of the experimentations of this phase are as follows:

- Test the PCP message conformance with what has been specified in [D1.2]
- Validate the QoS computation as implemented by PCS machinery
- Validate the route selection process
- Validate the interface between q-BGP and PCS
- Validate resource reservation and release

4.4.2 Experiment setup and test description

The environment to execute these tests is the MESCAL testbed that is deployed in FTR&D premises. The configuration of the testbed for this phase is detailed in section 9. This configuration will be used as it is for executing these tests except when there are explicit recommendations in the procedure of the test.

4.4.3 Test Results

TB_P3_FUNCT group contains the following test suites:

Test Suite Id	Objective
TB_P3_FUNCT/CMES	This group of test aims at verifying message conformance of PCP to what is specified in [D1.2].
TB_P3_FUNCT/QAGG	This group of tests aims at verifying QoS aggregation operations as achieved by PCE entities.
TB_P3_FUNCT/RESAV	This group if tests aims at verifying the reservation operations when a path has been computed by a PCE

Table 14: Phase 3 Validation Test Suites

TB_P3_FUNCT group contains the tests in Table 15.

Test Id	Purpose	Status
TB_P3_FUNCT/CMES/1	Check the format of OPEN, CLOSE and ACCEPT messages.	SUCCESSFULLY PASSED
TB_P3_FUNCT/CMES/2	Check the format of REQUEST, RESPONSE PATH- ERROR and ACKNOWLEDGE messages.	SUCCESSFULLY PASSED
TB_P3_FUNCT/CMES/3	Validate the REQ-REFERENCE-ID and PATH-COMPUTATION-ID.	SUCCESSFULLY PASSED
TB_P3_FUNCT/CMES/4	Validate QoS information contained in REQUEST-PATH message.	SUCCESSFULLY PASSED
TB_P3_FUNCT/CMES/5	Validate QoS information contained in RESPONSE-PATH message.	SUCCESSFULLY PASSED
TB_P3_FUNCT/CMES/6	Check the format of PATH-ERROR and messages.	SUCCESSFULLY PASSED
TB_P3_FUNCT/CMES/7	Check the format of CANCEL and messages.	SUCCESSFULLY PASSED
TB_P3_FUNCT/CMES/8	Check operational behaviours when receiving REQUEST messages.	SUCCESSFULLY PASSED
TB_P3_FUNCT/QAGG/1	Check QoS aggregation operation.	SUCCESSFULLY PASSED
TB_P3_FUNCT/RESAV/1	Check resource reservation	SUCCESSFULLY PASSED
TB_P3_FUNCT/RESAV/2-3	Check resource release when the order is cancelled by user and when the validity date expires.	SUCCESSFULLY PASSED

Table 15: Phase 3 Validation Tests results

4.4.4 Conclusions

In this phase, we have tested the MESCAL PathCompSys implementation, especially the following features:

- Configuration of path computation orders;
- Interface between PCEs and routing;
- Computation of inter-domain paths satisfying a set of QoS performance characteristics;
- The conformance of the PCP implementation.

These features are aligned with the specification.

5 SERVICE MANAGEMENT TESTS AND RESULTS

5.1 pSLS Ordering Tests

The role of the *pSLS Ordering* functional block (see section 4.5 of [D1.3]) is to establish the set of pSLS agreements, the most advantageous to the AS with respect to Traffic Engineering and business objectives. The high-level experimentation objectives for the *pSLS Ordering* functional block are:

- Functional validation of the prototype implementation;
- Verification of the convergence of the collective agreement optimisation logic;
- Assessment of the impact of environment complexity upon the scalability of the approach;
- Gaining insight on inherent benefit/cost tradeoffs of the collective agreement optimisation.

The performance metrics, the controlled and uncontrolled variables considered for the *pSLS Ordering* experimentation are listed in Table 16 and Table 17.

Performance Metrics		
negotiation logic processing time (ProcT.NLogic)	The processing time of the pSLS Ordering negotiation logic per round.	
SrNP processing time (ProcT.SrNP)	The processing time of the SrNP engine for handling send message requests.	
order execution time (OrdExecT)	The time elapsed between the reception of a <i>Negotiation Plan</i> and the completion of negotiations.	
agreement optimality (AgrOptimality)	The confirmed cost at a round of the established order over the minimum possible cost, assuming a full knowledge of the providers' cost function.	

Table 16: pSLS Ordering Performance Metrics

Controlled Variables		
maximum number of rounds (MaxRounds)	The maximum number of negotiation rounds the negotiation logic is permitted to undertake before concluding the execution of an order.	
	Uncontrolled Variables	
target limit (TrgL)	The limit each pSLS is bound to contribute to the order, takes values in $(0,1]$. A target limit of 1/3 for instance, signifies to the logic that each pSLS cannot exceed 1/3 of the total bandwidth required for a particular order; hence only combinations of three or more pSLSs may implement this order. We consider the distinct values of 1/3, 2/3 and 3/3.	
number of cost areas (CAreas)	Assuming a linear step-wise decreasing cost function, the number of discontinuities on the cost function of a provider for a service type. In other words, the number of different values of cost per unit of a provider for a service type. We consider just one area, 4 areas and 20 areas.	
delta between cost areas (CDAreas)	The difference factor of the cost per unit between adjacent cost areas; we consider 1.5 and 3 as values.	
delta between providers (CDProviders)	The maximum difference factor of the cost per unit between providers; we consider 2 and 10 as values.	

Table 17: pSLS Ordering Variables

5.1.1.1 Experimentation Environment

The test platform is composed by the *pSLS Ordering* prototype, a reduced version of the *SLS Order Handling* prototype, the *Negotiation Plan Generator* acting on behalf of *Binding Selection* block and the *SLS Order Handling Response Generator* testing tools (see Figure 54).



Figure 54: pSLS Ordering Experimentation Environment

An experiment corresponds to the execution of one order, expressed in the *Negotiation Plan*. However, multiple dynamic interactions with a number of *SLS Order Handling* servers may take place for the completion of the order execution and the conclusion of the experiment.

The *Negotiation Plan Generator* produces the negotiation plans to be fed to the *pSLS Ordering*. It can be configured to produce negotiation plans targeting a configurable number of pSLSs and providers. The acceptance criteria restrict the total bandwidth to be a fixed value (100 units) and the total cost less than a maximum value, aligned with the problem definition of *pSLS Ordering* negotiation logic (see [D1.3]). Target tolerance criteria automatically restrict the bandwidth per pSLS to be at maximum the *Target Limit* (see Table 17) times the order bandwidth of 100 units.

The pSLSs to be negotiated at each negotiation round are generated by the *pSLS Generator* function and can be parameterised based 1) on the pSLS type (Provider Loose QoS and Provider Loose QoS Tunnels, Peer Loose QoS and Peer Loose QoS Tunnels, Proxy Statistical QoS and Proxy Statistical QoS Tunnels), 2) the boundary link, 3) the QoS-class, 4) the destination labels or IP prefixes, 5) the bandwidth and 6) the cost. A simplified version used in the bulk of the experimentation activities generates pSLSs of Provider Loose QoS type for premium meta-QoS-class based on the provider identifier, the bandwidth and the cost. *pSLS Generator* is also used in functional validation of the *SLS Order Handling* function (see section 5.2).

The *SLS Order Handling* prototype is reduced and contains only its *SrNP server* engine which, instead of the *Admission Logic*, it is now controlled by the *SLS Order Handling Response Generator*. The *SLS Order Handling Response Generator* operates on the basis of the service cost function generated by the *Service Cost Function Generator*. A cost function for i-th service is of the form

$$fc_{i}(x) = \begin{cases} c_{i1} & 0 < x \le x_{i1} \\ c_{i2} & x_{i1} < x \le x_{i2} \\ \vdots & \vdots \\ c_{ik} & x_{i(k-1)} < x \le x_{ik} \end{cases}$$

where x is the bandwidth and $fc_i(x)$ the cost per bandwidth unit. The Service Cost Function Generator generates cost functions with definite number of cost function areas k equal to the value of the CAreas variable with $c_{ij}/c_{i(j+1)}$ always equal to CDAreas (see Table 17). Note that, as long as CDAreas > 1, the lesser the bandwidth the bigger the cost per unit, i.e. the function of cost per bandwidth unit monotonically decreases. Finally, the base cost c_{i1} for the cost function of each of the configured pSLSs is determined randomly based on a uniform distribution in the range of [1, CDProviders * C_{base}].

The SLS Order Handling Response Generator will reply to any Proposal received from the pSLS Ordering for bandwidth in $(x_{ij}, x_{i(j+1)}]$ and cost unspecified or other than $c_{i(j+1)}$, with a Revision SrNP message containing a pSLS generated by the pSLS Generator with the requested bandwidth and cost set to $c_{i(j+1)}$. On a Proposal or BindProposal SrNP message with bandwidth in $(x_{ij}, x_{i(j+1)}]$ and cost set to $c_{i(j+1)}$, it will reply with SrNP Accept or AgreeProposal message respectively. An SLS Order Handling Response Generator runs per provider among the configured Providers.

In order to test the negotiation logic, appropriate *Transaction Logging* is added, so that the calculated costs and the processing time can be tracked down per negotiation round.

5.1.2 Experiment Setup and Test Description

To facilitate experimentation we focus to a representative set of test configuration options (see Table 18). The maximum number of rounds *MaxRounds* is set to infinite. The number of pSLSs to concurrently pursue is fixed to three.

The undertaken tests are described in detail in Table 18. Controlled or uncontrolled variables left unspecified in a test description are set to appropriate fixed values so that they have no impact upon the subject under testing.

	id	provider delta	cost areas	cost area delta	target limit
TestSetup#1	2A[1,2,3]	2	1		
TestSetup#2	10A[1,2,3]	10	1		
TestSetup#3	2B[1,2,3]	2		1.5	
TestSetup#4	10B[1,2,3]	10	4	1.3	$(1/2 \ 2/2 \ 2/2)$
TestSetup#5	2C[1,2,3]	2	4	2	$\{1/3, 2/3, 3/3\}$
TestSetup#6	10C[1,2,3]	10		5	
TestSetup#7	2D[1,2,3]	2	20	2	
TestSetup#8	10D[1,2,3]	10	20	5	

Table 18: Test Configurations

5.1.3 Test Results

First, we assess the evolution of optimality and processing time through the negotiation rounds.

For test 10C2, Figure 55 presents per negotiation round the estimated and the confirmed cost, which are normalised over the actual minimum, theoretically calculated, cost. The estimated cost is the cost that the negotiation logic computes at each round based on the knowledge of the cost gained from the providers from the requests made at previous rounds; note that the logic builds on the assumption that the cost-rate function of the providers is decreasing. The confirmed cost at a negotiation round is the cost already agreed with the providers –the estimated cost accepted by the providers at a previous round.

As we can see from Figure 55, and actually is the case in all other conducted tests, the confirmed cost decreases, while the estimated cost increases as we progress the negotiation procedure, until they both converge to the same value, which is the optimum, minimum possible, cost value. The decrease of the confirmed cost is logical as this constitutes the mere objective of the logic. The increase of the estimated cost can be explained because of the optimistic nature of the logic and the fact the providers' cost-rate functions are decreasing. Departing from an initial valid solution, the logic asks for more bandwidth from what thought to be the cheaper providers, however this might not be the case –the providers might respond with higher than the estimated values. Should no estimates on feasible combinations yielding less than the confirmed cost can be found, the logic concludes with the last confirmed cost, which proves to be the minimum possible cost –assuming decreasing linear step-wise cost functions per provider. This proves the validity of the specified algorithm as well its stability - convergence.



Figure 55 Evolution of optimality over negotiation rounds

For the above test, 10C2, Figure 56 depicts the processing time of the negotiation logic and the time consumed in SrNP-interactions as a function of the negotiation rounds. As we can observe, the time required for the logic to execute increases as negotiations progress, whereas the time spent on protocol interactions decreases. The same behaviour is noticed throughout all conducted tests. This behaviour was anticipated as the number of possible valid combinations to yield an estimated cost increases with the growing of knowledge of actual cost the providers can offer, gained from the previous rounds.



Figure 56 Evolution of processing time over negotiation rounds

Next, we try to assess the rate of decrease of the confirmed cost, therefore the acceleration in dropping down the cost of valid solutions found, over the negotiation rounds in relation to the parameters identified to influence the complexity of the negotiation logic.

Figure 57 depicts the normalised confirmed cost over its maximum value established at the first negotiation round, per round over 6 different test configurations; from top to down the 2B2 and 2B10, 2C2 and 2C10 and 2D2 and 2D10 test configurations. The three sets of these test configurations differ amongst them in the number of cost areas and the diversity of the cost-rate values per area; the two tests configurations within each of the three sets differ in the diversity of the cost-rate values amongst the providers.

As we can see, the more the diversity in the cost values per provider, the greater the drop in the confirmed cost. This is a reasonable result, considering that in cases where there is not significant difference in the offers made by a provider, the first agreed cost will be close to the minimum possible that can be found. From this result, we can also notice that the proposed algorithmic logic indeed pursues to the end of finding the minimum possible cost and indeed achieves that –assuming a liner step-wise decrease cost function per provider. Last, an observation to be made is that the diversity in the cost amongst the providers does not impact the dropping of cost from round to round, compared to the effect of cost diversity of a provider.



Figure 57 Rate of decrease of the confirmed cost

Following, we examine the effect of the parameters identified to affect the behaviour of the negotiation logic on the 'speed of convergence' represented by the number of negotiation rounds required to conclude successfully –achieve the minimum cost solution.

Figure 58 shows the number of negotiation rounds to successful conclusion for the conducted tests, which were executed in three cases, 1, 2 and 3 corresponding to the three different values for the target limit considered, 3/3, 2/3 and 1/3. These percentages mean that the logic can buy from each of the three providers only up to these percentages of the totally required bandwidth. Considering that the logic has been built assuming a linear step-wise cost function per provider, which indeed is the case in the conducted tests, we see that in the cases of 1/3 and 3/3 the logic concludes very fast. In these cases, the number of valid combinations of the amounts of bandwidth to purchase from each provider is limited by the constraint underlying the test case, case of 1/3, or the first to try to purchase indicates the cheapest solution, case of 3/3 –buy all from the cheapest. However, in cases, as in the case of 2/3, that the number of valid combinations is not limited or the cheapest solution cannot be identified, the logic needs to go through more steps for asking/identifying possible better combinations.



Figure 58 Number of negotiation rounds for successful conclusion

Last, we try to examine how the processing time of the logic is affected by the complexity of the external environment.

Figure 59 depicts the processing time of the negotiation logic as a function of the negotiation rounds in a number of test configurations representing different cases of complexity regarding the providers' cost functions. The following observations can be made. The processing time increases with the number of negotiations rounds, as we have seen before. The more the diversity in the cost functions of the providers the faster the increase of the processing time. The processing time increases from round to round not in a linear fashion; it is affected by both the diversity of the cost values within a provider as well as by the diversity of the cost values amongst providers. Note that as we have seen in a previous test, the diversity of the cost function within providers and not between providers mainly affects the number of negotiation rounds required to conclude the negotiations.



Figure 59 Processing time of the negotiation logic

5.1.4 Conclusions

The tests carried out show that it is possible to conduct negotiations in an automated fashion, proving the validity and feasibility of the proposed ordering and negotiation framework.

We showed that an algorithmic negotiation logic can be built and operated on top of the specified negotiations protocol SrNP, which has been designed to support any kind of negotiation logic. The negotiation logic we built addresses a particular case for negotiations and it has been demonstrated that can indeed achieve the optimum bargain, provided that its assumptions regarding the providers' cost functions hold. In this set-up, a number of tests were carried out for assessing its scalability and stability in relation to various parameters representing the complexity of the external environment in which is to operate –diversity of cost functions in terms of their values per and amongst providers. The tests yielded reasonable and justifiable results, further advocating the validity of automating the negotiation logic.

From the results of the particular negotiation case we implemented, we can draw the following:

The optimality of negotiation logic depends on the amount of prior knowledge one can have regarding the decision making logic of the parties to negotiate. As demonstrated, assuming that all providers employ a specific type of cost function, step-wise linear decreasing, an algorithm for conducting negotiations to find the best solution/bargain can be built and conclude successfully in a scalable and stable fashion.

Assuming that the 'trend' of the decision-making logic of the other negotiating parties is known, then negotiation logic can be built to yield the optimum solution. Given a known trend, the larger the diversity in the values of the benefit-metrics (e.g. cost) of the issues under negotiations, the more the chances in achieving a better beneficial agreement, however at the expense of increased number of negotiation rounds and processing time.

The complexity of negotiation logic depends on the complexity of the decision making logic of the other negotiating parties and grows from negotiation round to negotiation round in a not linear fashion.

5.2 SLS Order Handling Tests

5.2.1 Objectives

The *SLS Order Handling* functional block (see section 4.4 of [D1.2]) conducts negotiations with *pSLS Ordering* so that the best matching between service requests and available resources is achieved. The *SLS Order Handling* is decomposed into four major functions (see Figure 60): the *Negotiations Server*, the *SLS Translation*, the *Admission Logic* and the *SLS Establishment* functions.

The *Negotiations Server* function conducts negotiations for all pending SLS orders from different customers in parallel, using the underlying negotiation protocol. The *SLS Translation* function translates and maps the SLSs contained in an SLS order into their network view; further it performs validity checks against already established SLSs, primarily for ensuring uniqueness of customer/users identification. The *Admission Logic* compares the anticipated demand of the requested pSLS and of the already established pSLSs against the available resources provided in the *Resource Availability Matrices* and decides accordingly the acceptance of the requested pSLS. The accepted pSLSs are established via the *SLS Establishment* function responsible for maintaining the *pSLSs* repository and for communicating the relevant information to other system components that need to be updated.

5.2.2 Controlled and Uncontrolled Variables

Controlled Variables		
Satisfaction Level (SL)	The <i>Satisfaction Level</i> operational parameter of the <i>Admission Control</i> function (see section 4.4.3.1 of [D1.2]). Its permissible value range is [-1,1], however, for experimentation purpose we consider only the following three values: #1 <i>NoGuarantees</i> (-1) #2 <i>AlmostSatisfied</i> (0) #3 <i>FullySatisfied</i> (+1)	

Table 19: SLS Order Handling Controlled Variables

	Uncontrolled Variables		
Resource Availability (Availability)		The availability of network resources as provided in the <i>Resource Availability Matrices</i> . We consider <i>low</i> and <i>unlimited</i> settings.	
Traffic Matrices size1number of lQCs (lQCs)(TMSize)number of external interfaces (ExtInterfaces)	The number of IQCs is one dimension of the internal Traffic Matrix. We only consider the fixed set of three IQCs: <i>Premium</i> , <i>Better-Than-Best-Effort</i> and <i>Best-Effort</i> IQCs.		
	number of external interfaces (ExtInterfaces)	The number of external interfaces is the second dimension of the internal Traffic Matrix.	

¹ The Resource Availability Matrices (see section 5.1.2.3 of [D1.2]) have the same size as the Traffic Matrices (see section 5.1.2.2 of [D1.2]).

	number of oQCs (oQCs)	The number of oQCs is one dimension of the external Traffic Matrix. We only consider the fixed set of three oQCs: <i>Premium</i> , <i>Better-Than-Best-Effort</i> and <i>Best-Effort</i> oQCs.	
	number of external	The number of destination prefixes outside the AS is the second dimension of the external Traffic Matrix.	
	(ExtDestPrefixes)	The external destination prefixes result from the established pSLSs, hence their number is analogous to the number of the established pSLSs.	
Traffic		Traffic forecast parameters such as <i>Multiplexing Factor</i> (MF) and <i>Aggregation Weight</i> (AW) refer to a <i>service class</i> . A service class groups a homogeneous set of traffic flows allowing for aggregation under the assumed service usage and traffic source patterns.	
Forecast Parameters	classes (SrvClasses)	The greater the number of service classes, the more granular the classification of traffic sources, hence the more homogenous the set of traffic flows and the more accurate the traffic forecast result.	
		We consider just one or many service classes.	
SLS Orders	number of ordered SLSs (SLSsOrdered)	The total number of non alternative SLSs contained in every SLS order placed during the experiment. Assuming only valid SLSs and <i>Availability</i> set to <i>infinite</i> then at the end of experiment there will be <i>SLSsOrdered</i> number of established SLSs.	
		The type of the service the SLS order belongs to. The supported service types are:	
	service types (SrvTypes)	 #1 Internet access at loose QoS #2 Loose QoS tunnels in the Internet #3 Traffic inter-exchange at a loose QoS 	
	(SrvTypes)	 #4 Loose QoS tunnel extension #5 Internet access at a statistically guaranteed QoS #6 Statistically guaranteed QoS tunnels in the Internet 	

Table 20: SLS Order Handling Uncontrolled Variables

To facilitate experimentation we focus to a representative set of test configuration options for *TMSize* variable (see Table 21).

Traffic Matrices size (<i>TMsize</i>) test configuration option	ExtInterfaces	ExtDestPrefixes
#1 Small	small	Small
#2 Medium	medium	Medium
#3 LargeAtDestinations	medium	Large
#4 LargetAtEdges	large	Medium
#5 LargeAll	large	Large

Table 21: Traffic Matrices Size Test Configuration Options

5.2.3 Experimentation Environment

The test platform used is depicted in Figure 60.



Figure 60: SLS Order Handling Experimentation Environment

5.2.4 Test Campaigns and Results

The tests undertaken are organised under the test suites outlined in Table 22 and described in detail in Table 23. All these tests have been successfully undertaken.

Test Suite Id	Objective
SLSOrderH/Funct/NServer	Functional validation of the implementation of the Negotiations Server function.
SLSOrderH/Funct/Translation	Functional validation of the service type dependent functions, namely <i>pSLS Translation</i> and <i>Establishment</i> functions.
SLSOrderH/Funct/Admission	Functional validation of the Admission Control function.

Test Id	Purpose	Description
SLSOrderH/Funct/NServer	Verify handling of multiple parallel pSLS orders. Verify the FSM implementation is according to specifications.	Runtime parameters:Diversity fixed to significant covering all protocolmessage combinations, invalid ones tooCustomers fixed to manyTest:Check the negotiation protocol embedded inNegotiations Server function works as specified.

SLSOrderH/Funct/Translation/1	Verify the implementation of pSLS translation and validation.	Runtime parameters:SrvTypes fixed to allOrdered pSLSs flow identification clauses configuredto overlap and cause validation failureTest:Check pSLSs are translated correctly to thecorresponding network view data structures.Ensure validity checks indeed fail.
SLSOrderH/Funct/Translation/2	Verify the implementation of pSLS validation and SLS establishment. Verify derivation of TE information required for q- BGP configuration in the testbed.	Static parameters: SrvClasses fixed to many Runtime parameters: SrvTypes fixed to all Test: Ensure validity checks indeed succeed. Check Demand Aggregation and Derivation calculations are in accordance to the provided traffic parameters. Check SLS establishment produces information as expected by the SLS Invocation Handling and the Dynamic Inter-domain TE components.
SLSOrderH/Funct/Admission	Verify the implementation of the admission control algorithm is according to specifications.	Static parameters: Availability varying between low and unlimited Runtime parameters: SL varying between NoGuarantees, AlmostSatisfied and FullySatisfied values Test: Check requested SLSs are admitted within the availability buffer as resized by the SL, while SLSs exceeding that buffer are rejected.

Table 23: SLS Order Handling Tests

5.2.5 Conclusions

The tests undertaken prove the validity and feasibility of implementation of the specified pSLShandling functions; modelling, translation, information exchange to/from TE functions and admission control at pSLS request epochs

5.3 SLS Invocation Handling Tests

5.3.1 Intra-domain cSLS

5.3.1.1 Overview

In this section we will describe the objectives, controlled/uncontrolled variables, performance metrics and experimentation environment for the performance and stability tests of the intra-domain cSLS Invocation Handling Component with reference to [D3.1]. The functionality of this component, named MTAC, and the details of our implementation are described in [D1.3]. We will also briefly describe the other algorithms we implemented for comparison reasons.

5.3.1.1.1 Objectives

The objective of the performance and stability tests is to assess the performance of MTAC for intradomain real-time traffic cSLSs under a variety of traffic scenarios and loading conditions and to compare it with the performance achieved by other algorithms in the literature for the same traffic scenarios and loading conditions.

5.3.1.1.2 Controlled/Uncontrolled Variables

The controlled variables are as specified in [D3.1]. The uncontrolled variables, with reference to [D3.1] are the packet loss rate of the l-QC employed for carrying the traffic of the intra-domain real-time cSLSs and the volume and characteristics of the traffic flows.

5.3.1.1.3 **Performance Metrics**

The performance metric is the trade-off between packet loss rate (PLR) and utilization/cSLS blocking rate achieved by MTAC for intra-domain real-time traffic cSLSs. The primary goal is to guarantee that the requested packet loss rate is achieved and the secondary goal is to maximise the resource utilization/minimise the cSLS blocking rate, subject to the PLR constraints.

5.3.1.1.4 Experimentation Environment

The experimental environment, with reference to [D3.1], is the intra-domain cSLS Invocation Handling software developed by UniS using the Network Simulator (ns-2). The algorithms are implemented in oTCL, which is the interface language of the simulator. The topology is a standard dumbbell topology (see Figure 61). We assume that the sources (cSLSs) connect to the ingress node through links with negligible congestion (zero losses) and that the ingress router first hop link is the bottleneck link.



Figure 61: Simulation topology

5.3.1.1.5 Comparison Algorithms

In order to compare the performance of MTAC with other algorithms from the literature we implement two other algorithms.

The first algorithm is a measurement-based admission control scheme –we refer to it as MBACdescribed by Zukerman et al in [ZUK] as Rate Envelope Multiplexing (REM), with adaptive weight factor and no histogram update. The reasons for the selection of the specific MBAC scheme for comparison with our scheme are that: (a) REM also makes the zero buffer approximation with respect to statistical multiplexing and (b) implementation-wise, in a similar fashion to our scheme, it requires only aggregate bandwidth measurements and the peak rate of the sources requesting admission in order to derive the admission control decision. The parameters involved in the implementation of MBAC are set to the values used in [ZUK].

The second algorithm is an endpoint admission control scheme –we refer to it as EAC- described by Karlsson et al in [KAR]. In order to test this scheme we implement an additional lower priority queue for the probing packets (out-of-band probing) that can store, as proposed in [KAR], a single probe

packet and which is only served when the higher priority real-time traffic queue is empty. As in [KAR], we set the probing rate equal to the peak rate of the source requesting admission, we consider probe durations of 0.5sec up to 5sec, and we also assume that there is no latency involved between the probing phase completion and the admission control decision.

5.3.1.2 Experiment Setup and Test Description

The algorithms are tested for two target l-QC PLR values: 0.01 and 0.001.

5.3.1.2.1 Topology

The details of the topology used for the intra-domain cSLS Invocation Handling tests are:

- Shared output buffer size: 5 packets
- Bandwidth assigned to 1-QC: 3.33Mbps for target 1-QC PLR value 0.01 and 3.56Mbps for target 1-QC PLR 0.001.

5.3.1.2.2 Simulated Traffic

We consider two types of traffic sources for the simulations:

- 1. Voice-over-IP (VoIP) traffic sources: VoIP traffic sources are modelled as exponential ON-OFF sources with peak rate 64kbps, average ON duration 0.350sec, average OFF duration 0.650sec and average rate 22.4kbps [HAB].
- 2. Videoconference traffic sources: Videoconference traffic sources are modelled as H.263 encoded sources with peak rate 332.8kbps and average rate 64kbps [TKN].

The durations of VoIP and Videoconference traffic sources follows two independent exponential distributions with average durations h_{VoIP} and $h_{H_{263}}$ respectively.

For the simulations we consider the traffic scenarios:

- 1. VoIP traffic sources only ($h_{VoIP} = 300$ sec)
- 2. Videoconference traffic sources only ($h_{H.263}$ =300sec)
- 3. Mixed VoIP and Videoconference traffic sources (h_{VoIP} =300sec and $h_{H,263}$ =180sec)

5.3.1.2.3 Traffic Volume

The invocation processes of both VoIP and Videoconference traffic sources are modelled as two independent Poisson arrival processes with different mean arrival rates l_{VoIP} and $l_{H,263}$ respectively.

The arrival rates are varied in order to produce various traffic loading conditions and examine the behaviour of the algorithms for these loading conditions. For the cases where both VoIP and Videoconference sources are employed (mixed traffic), the averages of their activation rates followed a ratio of 2:1

The value Load=1 corresponds to the average traffic load that that would be incurred by a VoIP source invocation rate equal to 1000 sources/hour. Given the average rates and durations of the VoIP and Videoconference traffic sources, Load=1 for the three simulated traffic scenarios corresponds to:

- 1. $l_{VoIP} = 1000$ sources/hour (VoIP traffic sources only ($h_{VoIP} = 300$ sec))
- 2. $l_{H,263}$ =350 sources/hour (Videoconference traffic sources only ($h_{H,263}$ =300sec))
- 3. $l_{VoIP} = 270$ sources/hour and $l_{H.263} = 135$ sources/hour (Mixed VoIP and Videoconference traffic sources ($h_{VoIP} = 300$ sec and $h_{H.263} = 180$ sec))

The simulated traffic loading conditions are: 0.5, 1, 2, 3, 4 and 5.

5.3.1.2.4 Algorithms Parameters

For the implementation of MTAC, as described in [D1.3] we use an exponential ON-OFF source with peak rate 64kbps, average ON duration 1.004sec and average OFF duration 1.587sec [CHU] as a *reference source* model and we fix the reference PLR level (e_{ref}) to the value 0.01.

For the implementation of MBAC, as already mentioned, we use the default values of [ZUK].

For the implementation of EAC we try probing durations 0.5sec, 1sec, 2sec, 3sec, 4sec and 5sec. The results that are presented regarding EAC are the ones for the probing duration giving the best trade-off between packet loss and utilization/blocking for each simulated traffic scenario.

5.3.1.3 Test Results

Each simulated scenario is run for 20 different randomly chosen seeds and for 4100sec, using the first 500sec as warming up period.

5.3.1.3.1 Functional Tests

MTAC is functioning properly.

5.3.1.3.2 Performance and Stability Tests

5.3.1.3.2.1 VoIP Sources

5.3.1.3.2.1.1 Target I-QC PLR 0.01



Figure 62: Incurred PLR for VoIP sources and target I-QC PLR 0.01



Figure 63: Achieved I-QC utilization for VoIP sources and target I-QC PLR 0.01



Figure 64: Incurred blocking for VoIP sources and target I-QC PLR 0.01

From the above figures, it can be seen that for VoIP sources and target I-QC PLR 0.01, the PLR achieved by MTAC always stays below the target PLR and, furthermore, MTAC is less conservative than MBAC and EAC, achieving therefore, on average, higher I-QC utilization and a lower blocking rate. For MTAC and MBAC, we observe an increase in the incurred PLR for increasing loading conditions. This is anticipated [GROS] because they both rely on measurements, so every new admission request has the potential of being a wrong decision. This means that a high source invocation rate is expected to have a negative effect on performance. For EAC we observe an increase in the incurred PLR and then a decrease. This happens for increasing loading conditions because simultaneous probing by many sources leads to a situation known as thrashing [BRES]. That is, even though the number of admitted flows is small, the cumulative level of probing packets prevents further admissions, driving therefore the utilization and the PLR to lower values.

5.3.1.3.2.1.2 Target I-QC PLR 0.001



Figure 65: Incurred PLR for VoIP sources and target I-QC PLR 0.001



Figure 66: Achieved I-QC utilization for VoIP sources and target I-QC PLR 0.001



Figure 67: Incurred blocking for VoIP sources and target I-QC PLR 0.001

From the above figures, it can be seen that for VoIP sources and target I-QC PLR 0.001, MTAC is the more conservative of the three algorithms. MBAC violates the target I-QC PLR for increasing loading conditions and by a big margin. That means that in order for MBAC to be able to keep the incurred PLR below the target PLR, its tuning parameters should be reconfigured in an ad-hoc fashion until the desired result is achieved. For EAC we observe a similar thrashing situation as with the previous case.

5.3.1.3.2.2 Videoconference Sources

5.3.1.3.2.2.1 Target I-QC PLR 0.01



Figure 68: Incurred PLR for Videoconference sources and target I-QC PLR 0.01



Figure 69: Achieved I-QC utilization for Videoconference sources and target I-QC PLR 0.01



Figure 70: Incurred blocking for Videoconference sources and target I-QC PLR 0.01

For Videoconference sources and target l-QC PLR 0.01, all three algorithms are conservative. This can be partly attributed to the stringent admission control criterion (all algorithms make the worst case assumption that the new source will be transmitting at its peak rate) and the high peak rate of the videoconference sources compared to their average rate.

5.3.1.3.2.2.2 Target I-QC PLR 0.001



Figure 71: Incurred PLR for Videoconference sources and target I-QC PLR 0.001



Figure 72: Achieved I-QC utilization for Videoconference sources and target I-QC PLR 0.001



Figure 73: Incurred blocking for Videoconference sources and target I-QC PLR 0.001

For Videoconference sources and target l-QC PLR 0.001, MTAC is the less conservative algorithm achieving therefore better utilization and lower blocking. It needs to be mentioned that the objective of an admission control algorithm is not to achieve the lowest PLR possible, but to keep the achieved PLR within the limits of the target PLR, while maximizing the utilization and minimizing the blocking.

5.3.1.3.2.3 Mixed VoIP and Videoconference Sources

5.3.1.3.2.3.1 Target I-QC PLR 0.01



Figure 74: Incurred PLR for mixed VoIP and Videoconference sources and target I-QC PLR 0.01



Figure 75: Achieved I-QC utilization for mixed VoIP and Videoconference sources for target I-QC PLR 0.01



Figure 76: Incurred blocking for mixed VoIP and Videoconference sources for target I-QC PLR 0.01

For mixed traffic, all three algorithms satisfy the target l-QC PLR 0.01. MBAC is more conservative than MTAC and EAC, achieving therefore lower utilization and higher blocking.

5.3.1.3.2.3.2 Target I-QC PLR 0.001



Figure 77: Incurred PLR for mixed VoIP and Videoconference sources and target l-QC PLR 0.001



Figure 78: Achieved I-QC utilization for mixed VoIP and Videoconference sources for target I-QC PLR 0.001



Figure 79: Incurred blocking for mixed VoIP and Videoconference sources for target l-QC PLR 0.001

For target l-QC PLR 0.001, MTAC and EAC achieve this PLR for all loading conditions with MTAC being less conservative, achieving a higher utilization. MBAC violates this PLR for very high loading conditions.

5.3.1.3.2.4 Aggregate Results

Taking into account all the simulated scenarios and traffic loading conditions, the average utilization and average blocking of the three tested algorithms are as follows:



Figure 80: Average I-QC utilization



Figure 81: Average cSLS blocking rate

As it can be seen, MTAC achieves the highest average utilization and the lower blocking rate, while, as shown in the figures so far, keeping the incurred PLR below the target I-QC PLR in all simulated scenarios and for all traffic loading conditions. MBAC achieves the second highest average utilization and the second lower blocking rate, but, as already shown, fails to satisfy the target I-QC PLR without further ad-hoc adjustment of its tuning parameters. EAC achieves the worst performance, regarding average utilization and blocking rate, but manages to keep the incurred PLR below the target I-QC PLR. We need to state here though, that the results of EAC presented in the above figures are achieved by trying various values of its tuning parameter (probing period), according to the authors recommendations [KAR], and that for certain values of probing periods -not shown-, are worse than the results for MBAC.

5.3.1.4 Conclusions

The simulation results show that MTAC can perform reasonably well for a variety of traffic scenarios -for both short-range dependent (VoIP) and long range dependent (Videoconference) sources- and loading conditions without requiring any reconfiguration of its parameters and that it compares favourably against other algorithms existing in the literature for the same simulation setup.

While satisfying the target l-QC PLR, MTAC achieves on average 3.6% higher utilization than MBAC and 7.3% higher utilization than EAC. Regarding the cSLS blocking rate, MTAC achieves on average 4.2% lower blocking than MBAC and 6.5% lower blocking than EAC.

5.3.2 Inter-domain cSLS

5.3.2.1 Overview

In this section we will describe the objectives, controlled/uncontrolled variables, performance metrics and experimentation environment for the performance and stability tests of the inter-domain cSLS Invocation Handling Component with reference to [D3.1]. The functionality of this component, named e-MTAC, and the details of our implementation are described in [D1.3].

5.3.2.1.1 Objectives

The objective of the performance and stability tests is to assess the performance of e-MTAC for interdomain real-time traffic cSLSs under a variety of traffic scenarios and loading conditions. Also in order to demonstrate the performance gains introduced by deploying status information from the interdomain link, a comparison with the conventional MTAC scheme that does not take into account status information from the inter-domain link will be made. For the conventional MTAC scheme, the interdomain real-time traffic cSLSs are treated as in the case of intra-domain real-time traffic cSLSs, with the difference that the minimum available bandwidth (see [D1.3]) is not guaranteed edge-to edge, but end-to-end, taking into account the available inter-domain link capacity. That means that for the conventional MTAC scheme, the bandwidth value that will be used as a threshold for admission control at each ingress node, will be set equal to the minimum between the first-hop link capacity and the part of the inter-domain link capacity that can be logically allocated to the inter-domain real-time traffic cSLSs entering through that ingress node and exiting through that inter-domain link.

5.3.2.1.2 Controlled/Uncontrolled Variables

The controlled variables are as specified in [D3.1]. The uncontrolled variables, with reference to [D3.1] are the packet loss rate of the l-QC employed for carrying the traffic of the inter-domain real-time cSLSs and the volume and characteristics of the traffic flows.

5.3.2.1.3 **Performance Metrics**

As in section 5.3.1.1.3.

5.3.2.1.4 Experimentation Environment

The experimental environment, with reference to [D3.1], is the inter-domain cSLS Invocation Handling software developed by UniS using the Network Simulator (ns-2). The algorithms are implemented in oTCL, which is the interface language of the simulator. The topology used is shown in Figure 82.





5.3.2.2 Experiment Setup and Test Description

The algorithms are tested for target l-QC PLR value 0.01, setting $PLR_{ingress}$ equal to 0.005 for links 1, 2 and 3 and PLR_{egress} equal to 0.005 for link 5 (inter-domain link) for e-MTAC. For the conventional MTAC scheme, since it does not use any feedback information from the inter-domain link, the allowed target loss rates for links 1, 2 and 3 are set equal to the target l-QC PLR value; that is 0.01.

5.3.2.2.1 Topology

The details of the topology used for the inter-domain cSLS Invocation Handling tests were:

- Shared output buffer size: 5 packets for links 1, 2 and 3 and 2 packets for links 4 and 5.
- Bandwidth assigned to 1-QC: 3.33Mbps for links 1, 2 and 3, 6.66Mbps for link 4 (no losses allowed in core network) and 8Mbps for link 5. That means that link 5 is "over-booked" with respect to the aggregate 1-QC capacity reserved in links 1, 2 and 3 (it is 80% the aggregate capacities assigned to the 1-QC at links 1, 2 and 3). For the conventional MTAC scheme, even though the bandwidth assigned in links 1, 2 and 3 is 3.33Mbps, the capacity that the sources can use in each one of these links is limited to 2.66Mbps so that the aggregate allocated capacity in these three links (3×2.66Mbps) does not exceed the total capacity allocated in link 5 for the 1-QC and, therefore, link 5 does not introduce any additional losses to the losses incurred by the first hop links.

5.3.2.2.2 Simulated Traffic

We consider mixed VoIP and Videoconference sources, as described in section 5.3.1.2.2.

5.3.2.2.3 Traffic Volume

In order to examine the performance of e-MTAC for various loading conditions, we simulate loading conditions for links 1 (L1), 2 (L2) and 3(L3): 0.5, 1, 2, 3, 4 and 5 (see section 5.3.1.2.3 for the definition of loading condition). We will refer to this simulated scenario as symmetrical loading

In order to demonstrate the performance gains introduced by deploying status information from the inter-domain link we simulate for e-MTAC and MTAC loading conditions for links 1, 2 and 3 as follows:

- 1. L1=0.5, L2=1, L3=0.5-1-2-3-4-5 (we fix load 1 to 0.5, load 2 to 1 and we vary load 3 from 0.5 to 5). We will refer to this simulated scenario as asymmetrical loading I.
- 2. L1=0.5, L2=0.5-1-2-3-4-5, L3=0.5-1-2-3-4-5 (we fix load 1 to 0.5 and we vary load 2 and load 3 from 0.5 to 5). We will refer to this simulated scenario as asymmetrical loading II.

5.3.2.2.4 Algorithms Parameters

For the implementation of e-MTAC, as described in [D1.3] we use an exponential ON-OFF source with peak rate 64kbps, average ON duration 1.004sec and average OFF duration 1.587sec [CHU] as a *reference source* model and we fix the reference PLR level (e_{ref}) to the value 0.01 for the first hop

links and to 0.1 for the inter-domain link.

For MTAC, the algorithm parameters are as in section 5.3.1.2.4.

5.3.2.3 Test Results

Each simulated scenario was run for 20 different randomly chosen seeds and for 4100sec, using the first 500sec as warming up period.

5.3.2.3.1 Functional Tests

e-MTAC is functioning properly.

5.3.2.3.2 Performance and Stability Tests

5.3.2.3.2.1 Symmetrical Loading

The total incurred PLR, the achieved inter-domain link utilization and the incurred blocking for e-MTAC as a function of L1, L2 and L3 are shown in the following figures.











Figure 85: Incurred blocking

Copyright © MESCAL Consortium, July 2005

As it can be seen, e-MTAC keeps the incurred PLR below the target l-QC PLR for all simulated loading conditions and achieves satisfactory inter-domain link utilization. The performance gains, regarding inter-domain link utilization, by using status information from the inter-domain link, are demonstrated in the following figures where the achieved inter-domain link utilization of e-MTAC is compared to the inter-domain link utilization of MTAC.

5.3.2.3.2.2 Asymmetrical Loading I



Figure 86: Utilization comparison for the inter-domain link

As it can be seen, e-MTAC achieves slightly better utilization than MTAC. The performance gain from using status information from the inter-domain link is relatively small, because of the low values of load in links 1 and 2 (0.5 and 1 respectively). In this case, because of the low values of load at links 1 and 2, the utilization gain is only a result of the fact that e-MTAC allows more traffic originating from router 3 to be admitted (taking advantage of the low contribution of traffic from routers 1 and 2 at the inter-domain link).

5.3.2.3.2.3 Asymmetrical Loading II



Figure 87: Utilization comparison for the inter-domain link

The utilization gain in this case is higher compared to the previous case because only load at link 1 is fixed to a low value. In this case, e-MTAC allows more traffic from both routers 2 and 3 to be admitted.

It needs to be mentioned that if the inter-domain link capacity was more "over-booked" with respect to the aggregate l-QC capacity reserved in links 1, 2 and 3 (e.g. if it was set to 50% of the aggregate l-QC capacity reserved in links 1, 2 and 3) then the utilization gains would be even higher.

5.3.2.4 Conclusions

The simulations show that e-MTAC can perform reasonably well for the simulated cases. The simulation results also illustrate the inter-domain link utilization gains that are achieved by incorporating status information from the inter-domain link in the admission control scheme.

6 SYSTEM-LEVEL SCALABILITY ANALYSIS

The MESCAL system has been built on a number of design principles and features to contribute to the system's scalability, summarized in the following:

- The MESCAL architecture is designed to function independently in each IP Network Provider (INP) domain. Each INP² only interacts with adjacent INPs. The interaction between INPs occurs at the service plane for pSLS ordering and at control plane for receiving/announcing reachability information. No protocols for resource reservation or for initiating the operations and actions of INPs are required.
- Service-layer co-operation between providers is achieved through scalable QoS peering models namely cascaded and bilateral peering models.
- pSLSs are established off-line between two adjacent domains for transporting traffic at aggregate levels in order to satisfy a large population of users at both customer and provider levels.
- No explicit signalling is propagated at inter-domain level. Any resource reservation request in term of pSLSs are carried out off-line at aggregate level between two providers. Only QoS-based routing information is propagated at inter-domain level through q-BGP, which may optionally carry performance information.
- A number of off-line and independent processes are devised to function including QoS Class (QC) discovery, pSLS ordering, off-line traffic engineering.
- Lightweight dynamic traffic engineering functions are applied at aggregate levels.
- A two-level service admission control scheme is adopted. Subscription negotiation operating offline, combined with light-weight admission control operating at service invocation instances, mainly relying on local information and coarse local network state indications.

The scalability of a solution/system is the ability for the system to function effectively and keep its performance at desired levels as the value of parameters influencing its behaviour increase. A scalable solution/system should be capable of being deployed at the scale of large networks offering a number of services to a large number of customers. Scalability in QoS-enabled IP networks has a number of distinct aspects at resource and service management levels, including network size, number and granularity of classes of service supported, the extent and complexity of service requests (c/pSLS) to manage, etc.

The pertinent parameters to be taken into account as scalability factors are as follows:

- The extent and complexity of message flow/processing for a new c/pSLS set-up during the c/pSLS negotiation.
- The extent of pSLS set-up per INP for offering inter-domain services
- The number of customer requests (cSLS) to be managed per INP
- The number and granularity of classes of service (QCs) to be offered
- The amount of routing announcements, size of routing tables, etc.

Generally, it is expected that a "no more than linear" dependency to the arrival rate of requests/messages indicates the system is prone to scale.

6.1 Comparison of CADENUS & MESCAL Scalability

In this section, we compare the CADENUS solution with MESCAL approach in terms of message flow in handling a new service request.

² In this document the terms INP, ISP, domain, and AS (Autonomous System) are used interchangeably.

The CADENUS architecture uses a business model that takes into account the stakeholders including service and network providers. The scope of the CADENUS business model is broader than MESCAL, with additional stakeholder and roles to be played in providing value-added services. MESCAL is only concerned with QoS-based IP connectivity services. However, the CADENUS architecture does not go into the details of how static and dynamic resource management and traffic engineering is achieved at the network level or how a bi-directional service is constructed, as MESCAL does.

The CADENUS project developed an architecture, which includes functional blocks at the userprovider interface within the service provider domain, and between the service provider and the network provider [CAD-D2.3]. CADENUS defined three key components: *Access Mediator*, *Service Mediator* and *Resource Mediator*. The overall mediation procedure includes the mapping of userrequested QoS to the appropriate network resources, taking into account existing business processes.

Access Mediator: It presents the current service offer to the user. The Access Mediator is responsible for selecting the appropriate service provider, according to the user's request. After authentication, the user requirements are captured, and the Access Mediator sends the information to the service provider who then employs the Service Mediators and Resource Mediators to map the requested and subsequently selected service into the deployed physical network.

Service Mediator: It is responsible for finding and in some cases building from individual elements, the service, and selecting the appropriate Resource Mediator. The Service Mediator is an off-line process, which supervises the incorporation of new services and the management of the physical access to these services via the appropriate underlying network, using the Resource Mediators. It is the task of the Service Mediator to prepare the service level agreements, and subsequently to authenticate the user and map service requests into appropriate network configuration information required by the Resource Mediators.

Resource Mediator: It is associated with the underlying network and its capabilities. There will be one Resource Mediator per administrative domain, and one *Network Controller* for each network technology within that domain. The Resource Mediator receives SLSs from network clients (i.e., Service Mediator). During the negotiation of an SLS spanning multiple domains, a certain number of Resource Mediators- those belonging to the crossed domains – must be involved in the negotiation phase. Each Resource Mediator in the chain is in charge of assuring that part of the service pertaining to its domain. This follows a forward cascaded model where a multi-domain SLS is split into two parts: a single domain SLS plus a remaining part that has to be enforced over one/more downstream domains until the end-to-end path is completed.

It should be noted that the main functionality of the *Network Controller* is intra-domain Admission Control to verify a service arising from an SLS can be accepted without jeopardising the allocated resources.

In CADENUS scalability study [Antonio04], [CAD-D8], the negotiation of a new SLA has been selected, since it exercises the scenario in terms of message-passing and processing, as well as number of entities involved. Figure 88 shows the CADENUS architecture as a cascaded queuing network. In this figure various mediators, together with the Network Controller and service directory modules are modelled as service centres.


Figure 88: The CADENUS architecture as a queuing network (from CAD-D8).

Figure 89 shows the message flow in the negotiation phase in which an Access Mediator, the Service Directory, a Service Mediator, and all Resource Mediators and Network Controllers in the end-to-end service chain are involved.



Figure 89: Message flow during service negotiation phase (from CAD-D8).

As shown in Figure 88 and Figure 89, the CADENUS solution involves a significant amount of signalling and processing to deal with a customer's (end-user) request (cSLA). An initial dialogue is performed between the end user and Access Mediator in order to select a service from a list offered by Access Mediator. Following the selection, a new set of parallel dialogues have to be carried out between Access Mediator and one or several candidate Service Mediator(s). Each Service Mediator in turn, has to contact the first Resource Mediator in the chain, (and the latter may have to contact others in the inter-domain chain based on the scope of SLSs), to allow Resource Mediators to make an evaluation of the impact the service is going to have on the network resources at each AS hop and derive a cost to be paid for the enforcement of service. The cSLA subscription operation is further performed by means of an admission control process at intra-domain level (a Network Controller function). This also implies the participation of all the network management entities in the end-to-end chain. Note that some network resources are already pre-reserved in each domain for which a Resource Mediator may reply to a request without performing any network re-configuration/redimensioning process. The computed cost is also returned to each Service Mediator by its corresponding Resource Mediator. The Service Mediator(s) return the final results to the Access Mediator, which provides the service list to the end user. Following end user selection and response, only one Service Mediator is eventually chosen and a transaction rollback should be performed by the Access Mediator for those Service Mediator(s) which have not been selected to provide the service.

The process of (automated) service definition and service offering by the Service Providers is outside the scope of MESCAL. MESCAL focuses on the business relationships between customers and INPs and between INPs in order to provide QoS across multiple domains driven by agreed SLSs. As such, the primarily concern of MESCAL work is QoS-based IP connectivity services. QoS-based IP connectivity services offered by INPs are divided into elementary and complex connectivity services. Elementary connectivity services are strictly point-to-point and uni-directional. In this section, only elementary connectivity services are considered.

For a provider domain wishing to provide QCs from its domain to destinations outside its domain, a number of QC-operations are performed off-line to build end-to-end QCs. The QC-advertisement operation enables a provider domain to inform other providers of its QoS-class capabilities (*l*-QCs, *e*-QCs, or *m*-QCs). The QC-discovery operation enables a provider to find the QCs offered by other provider domains. Following QC-discovery, the QC-mapping operation enables a provider domain to either build *e*-QCs by determining suitable combinations of the domain's own capabilities (*l*-QCs) with the QC capabilities offered by other downstream provider domains or to map its *l*-QCs to *m*-QCs. The QC-binding operation enables a provider domain to decide which of the possible QoS-mappings will be used for actually negotiating corresponding *pSLSs* with appropriate downstream providers. The QC-implementation operation enables a provider domain to implement a QoS-binding at the network (IP) layer.

In MESCAL, prior to offering any connectivity service to its customers, an INP creates the logical infrastructure in order to provide these services across multiple domains. Therefore, a provider domain independently performs QC-operations and resource provisioning for implementing *e-QC/m-QC* in advance. This type of off-line process is repeated recursively to enable other upstream domains to offer QoS-based services. In each step of the cascade, the upstream provider acts in the consumer role to the provider immediately downstream. It is each provider's responsibility to make appropriate *pSLSs* with the immediate downstream provider making it possible for individual customer IP QoS services to be created and managed along the entire route. The pSLSs are established between two adjacent INPs for exchanging traffic in the Internet, with the purpose of expanding the geographical span of their offered services. pSLSs support aggregated traffic (i.e. serving many customers), and it is assumed that they are in place prior to any cSLS agreements with end customers or pSLS agreements upstream peers. After the appropriate pSLSs are put in place, an INP can accept service requests (c/pSLSs) and offer connectivity services to its customers or peers.

In the cascaded model, each INP makes *pSLS* contracts with the adjacent interconnected INPs but not with providers more than "one hop away". A service dialogue is only performed between the end user and INP or between two neighbouring INPs in order to establish a connectivity service. Figure 90 shows the MESCAL model in servicing the SLS requests and the message flow in the negotiation

phase between two adjacent domains, i.e., SLS ordering of a source domain and SLS order handling of downstream domain.



a): SLS orders queue and order handling server

b): Message flow during SLS negotiation phase.

Figure 90: MESCAL model in SLS negotiation.

Therefore, acceptance/rejection of any service requests at subscription epoch and admission control decisions at invocation epoch are performed locally at the INP level. There is no need to contact other INPs in the inter-domain chain to fulfil a service request. This approach creates a scalable solution in terms of message flow because it avoids propagation of the service request to downstream INPs in forward direction towards the end-to-end chain for every requested service.

The complicated set of signalling/message flow transactions used in CADENUS is as follows:

- $\{\text{User}\}_K \rightarrow \{\text{Access Mediator}\} \rightarrow \{\text{Service Mediator}\}_L \rightarrow \{\text{Resource Mediator}\}_N \text{ where } K, L, N \ge 1$
- Pre-allocation/rollback actions in different Resource Mediator(s).

While, the MESCAL design philosophy, as briefly explained above and specified in more detail in [D1.1], [D1.2], avoids the complicated set of signalling/message flow transactions used in CADENUS:

• {Customer/Provider: SLS Ordering}_K \rightarrow {SLS order Handling @ INP2} where K \geq 1

6.2 Scalability of Inter-Provider Peering Models

A number of QoS peering models can be used for the interconnection and service-layer interactions between providers' for offering QoS services across multiple domains. The type of inter-domain peering impacts the service negotiation procedures, the required signalling protocols, the path discovery through QoS binding, and path selection. Any solution for QoS peering should function effectively and in a scalable manner. Mescal studied three peering models (source-based, cascaded, and bilateral) that are explained briefly below.

In the *source-based model*³, an IP Network Provider (INP) negotiates *pSLSs* directly with a number of downstream providers to construct an end-to-end QoS service. With this model, service peers are not necessarily BGP peers. The source point requires an up-to-date topology of the Internet to discover domains to negotiate with and to select end-to-end routes. In addition it needs to know the domains' advertised *l-QCs* in order to perform mapping and binding of these *l-QCs* to form *e-QCs*. The source INP directly establishes *pSLSs* with a set of potential domains (neighbour, transit, and distant ASs) in

³ Source-based model is referred to as Centralised model in D1.4.

order to reach a set of destinations and offer an end-to-end QoS-based service. Although it is possible to find and set-up optimal routes to the destinations since the source point has access to the overall QoS-based topology, the need for accurate topological and QoS related information of the Internet is a major drawback of this model. It may be feasible for a relatively small number of domains, but it raises scalability concerns when a large number of networks are involved. The source INP will end up with many *pSLSs* to manage.

In the *cascaded model*, an INP only negotiates *pSLSs* with its immediate neighbouring provider/s to construct an end-to-end QoS service. Thus, the QoS peering agreements are between adjacent neighbours, but not between providers more than "one hop away". There is no need for complete topology related information, except routing information. This type of peering agreement provides the QoS connectivity from a customer to reachable destinations that may be several domains away. Setting-up *pSLSs* with defined scope and distinct performance characteristics between adjacent INPs is the compelling feature of this model. For QoS-Class discovery and selection, each INP in the chain needs to know its adjacent neighbours and the status of related interconnection links. In addition, each INP needs to know the *e-QCs* advertised by its neighbouring domains for binding with its own *l-QCs* in order to implement its own *e-QCs*, which may subsequently be advertised to its customers and upstream domains. This is true for every INP involved in the chain in order to implement its *e-QCs*. Each INP has only a limited number of *pSLSs* to manage (see next section) making the cascaded model more scalable than source-based model.

The *bilateral model* relies on the cascaded model and the use of the *m*-QC concept. Setting-up *pSLSs* with open scope (i.e., no explicit reachability information) and no distinct performance characteristics but simple compliance with well-known m-QC behaviours between adjacent INPs is the compelling feature of this model. In this model, there is no end-to-end QoS guarantees defined and consequently there is no need to build *e*-QCs, which are the fundamental differences between this model and cascaded model. The bilateral model does not provide any end-to-end bandwidth guarantees because it enables any destination to be reached, without prior explicit indication in the *pSLS*. Each domain is engineered to support a number of local QoS classes (i.e. *l*-QCs). These *l*-QCs are mapped to globally well-known *m*-QCs. Each AS advertises the *m*-QCs that it supports in its administrative domain. Other domains can make *pSLS* arrangement in cascaded fashion with this domain to make use of offered *m*-QCs. Although, inter-domain routing is *pSLS* constrained, each domain can find out whether it can reach certain destinations in an *m*-QC plane through a BGP-like protocol (q-BGP) [Bouca05]. The basic requirement for a domain is to have one pSLS agreement with its adjacent domain to join an m-QC plane. This makes the bi-lateral model even more scalable than the other models.

6.3 The Extent of pSLS Set-up

Here, we study the scalability of pSLSs to manage when a specific QoS peering model is employed. Two different connectivity topologies are considered: a star topology (Figure 91) and a multi-tiered hierarchy topology (Figure 92).

In a simple star topology, the number of pSLSs to establish for all three models is an order of $O(N_d)$ as

 $N_p = N_{qc} * (N_d - 1)$ where:

 N_{qc} = Number of QCs offered by AS1 to its customers to reach customers in AS2, AS3, or ASn (specified as a constant value).

 N_d = Number of domains (INPs)

 N_p = Number of *pSLSs* from a source domain (i.e., AS1) to reach all domains for all e-QCs in either source-based or cascaded models. N_p is regarded as the number of *pSLSs* to join all m-QC planes in bilateral model.



Figure 91: Star topology for connectivity.

A three-tiered hierarchy topology shows a model of the global Internet, organised as a collection of independently operated networks, shown in Figure 92. Here, we are mainly concerned about the customer-provider relationship.





Figure 92: Three-Tier Internet model pSLS-based agreements.

Considering the above connectivity, the number of pSLSs to establish for all three models is as follows. It should be noted that:

 N_p = Number of *pSLSs* from a source domain to reach all domains for all e-QCs in either source-based or cascaded models. N_p is regarded as the number of *pSLSs* to join all m-QC planes in bilateral model.

 N_{qc} = Number of QCs (*e-QC* or *m-QC*) offered by a source INP (ISP-S) to its customers to reach customers in ISP-T, ISP-U, or ISP-V. Here, the number of peering points (*K*) shared between two domains is set to one. Hence, N_{qc} is regarded as a constant value that is studied in more detail in the next section.

Source-based Model:
$$N_p = N_{qc} * \left[\frac{N_d * (N_d - 1)}{2} \right]$$
, in the order of $O(N_d^2)$

Cascaded model: $N_p = N_{qc} * (N_d - 1)$, in the order of $O(N_d)$

Bilateral model: $N_p = N_{qc}$

The bilateral scalability figure (as in cascaded model) depends on the number of peering points (*K*) a domain share with its adjacent domains. With a single peering point, a domain can join the *m-QC* based parallel Internet. With more peering points, more pSLSs can be established for the same *m-QC* to improve the reachability, resiliency, etc. Figure 93 shows the scalability of different peering models in terms of pSLS set-up, where N_{qc} and K are set to one.



Figure 93: The trend of pSLS set-up in each peering model.

6.4 Number and Granularity of QCs

In MESCAL, three Solution Options are proposed to support a diversity of services. These service options are aimed at customers requiring different performance levels; Loose Guarantee (LG) for mass market with better-than-best-effort service levels, Hard Guarantee (HG) for a small number of value-added services with guaranteed bandwidth and performance and Statistical Guarantee (SG) for a range of services between the two extremes. The SG solution option is implemented using cascaded model while LG and HG solution options are implemented using bilateral model.

The scalability figures, shown in the previous section, are dependent on the number of QCs offered. Due to the definition of the LG and HG solution options, only a limited number of well-known m-QCs are used globally, whereas in SG the QC binding is more flexible, increasing the number of offered QCs and increasing the number of pSLSs to set-up.

In SG, the use of the DiffServ DSCP to distinguish QCs means that the maximum number of offered QCs for a given destination can be no more than the maximum allowable number of DSCPs, i.e. 64 in the IPv4 realm. In the worst case, the number of combinations for offering a single QC is $\{64* N_c\}$ where ' N_c ' is the number of peer ASs, and thus $\{(64^2)*N_c\}$ for offering all the possible QCs. If two ASs have on average K peering points then the worst-case scalability factor for supporting and offering the maximum number of QCs becomes:

 $N_{qc} = \{(64^2) * N_c * K\}$

We can see that the SG scales linearly with the number of peer ASs and the number of peering points for each peer AS.

The proposed Solution Options also introduce an increase to the number of BGP announcements proportional to the number of QCs supported. In LG, this is limited to the number m-QC planes and in SG, where q-BGP is optionally used, it depends on the number of e-QCs offered between the two peer domains having local significance.

In addition, the number of m-QCs in LG and e-QCs in SG have direct impact on the size of routing tables, but the size increase will depend on the level of address aggregation.

The scalability of HG as it uses MPLS-TE could be a concern. In order to establish a full mesh logical network, an order of $O(N_e^2)$ unidirectional LSP tunnels needs to be established where ' N_e ' is the number of edge nodes, which can be very large. In fact, it is even worse than $O(N_e^2)$ since multiple LSP paths are used for load sharing and different QoS services may use different LSPs between two node pairs. HG is scalable and feasible only if a limited number of LSPs are to be established. Therefore, HG has scalability concerns for large deployments in terms of end-to-end inter-domain tunnel set-up but as stated above HG Solution Option is not aimed at the mass-market deployment but at specialised services where its scalability is less of a concern.

6.5 Summary

This study of the scalability of MESCAL has

- identified the key scalability attributes of the MESCAL solution.
- compared the message flows associated with a service request in both the CADENUS and MESCAL architectures and shows how the cascaded approach in the MESCAL solution leads to a significant reduction in the amount of inter-domain signalling relative to the CADENUS approach.
- analyses various peering models (source, cascaded and bilateral) and shows that the models used in the MESCAL solution (cascaded and bilateral) are scalable in terms of the number of pSLSs required for large networks.
- shown that the three solution options (Loose, Statistical and Hard Guarantee) have scalability in terms of QoS Classes that is matched to their intended usage i.e. the mass deployment option (Loose) has excellent scalability, the Statistical option scales linearly while the Hard Guarantee option does not scale well for large deployments, but it is targeted at specialised high value services and therefore will not be mass deployed.

7 CONCLUSIONS

7.1 Overview

The overall MESCAL solution for Internet QoS provisioning can be summarised as follows: provider domains determine QoS routes by means of q-BGP, which is activated between adjacent provider domains following establishment of peering agreements, pSLSs, to exchange aggregate QoS traffic. MESCAL has proposed three instances of solutions, which we have called solution options, for delivering the type of QoS required across the spectrum of services/applications, namely loose, statistical and hard QoS guarantees.

This deliverable has presented the tests undertaken to assess the validity and performance of the MESCAL functionality for inter-domain QoS delivery and has described the yielded results. In addition to the scalability results obtained through experimentation. further aspects of the scalability of the MESCAL solution have been analysed at a theoretical level.

Overall, based on the yielded experimentation results, the following conclusions can be drawn:

- Prototype implementation and the tests undertaken in the testbed prove the overall validity of the MESCAL solution for delivering QoS across multiple domains and the proposed algorithms, schemes and protocols; and show that the specified functionality can be feasibly implemented.
- Simulation results indicate that better performing routes for carrying QoS traffic can be established through the proposed solution compared to classical BGP. The specified intra- and inter-domain traffic engineering and c/pSLS admission control algorithms are of reasonable performance, yielding favourable results compared to ad-hoc configurations or alternative schemes.
- Theoretical analysis shows that the proposed solutions, for providing QoS with loose, hard or statistical guarantees in the Internet, scale in terms of the QoS-classes to be used/offered.
- By virtue of its design, the MESCAL inter-domain QoS solution is inherently scalable and realistic; it relies on interactions between adjacencies, it does not rely on per flow end-to-end bandwidth reservations and QoS signalling means.

Finally, it should be noted that the MESCAL solution for providing QoS in the Internet supplements the current best-effort Internet and does not distort current business relationships between providers.

The following summarise the main points and conclusions drawn from experimentation for each of the major functional aspects of the MESCAL solution.

7.2 Implementation of the MESCAL Solution

MESCAL implemented a testbed of experimental Linux-based routers for 'proving the concept' of the specified QoS solution in a realistic network environment. The primary purpose of the testbed was to assess the feasibility of the enhancements and interactions required at the IP level for employing q-BGP as the inter-domain routing protocol between adjacent ASs and the computation of QoS-constrained paths.

Within the limitations on the number of emulated domains in the testbed, *it was proved that the specified q-BGP protocol achieved its functional objectives and the MESCAL solution for inter-domain QoS delivery is valid and feasible.* In particular:

- QoS-aware inter-domain routes can be constructed by using q-BGP; different routes, in terms of AS paths and end-to-end QoS performance, are determined per QoS-class and the traffic following a QoS route receives the appropriate QoS-class treatment within each domain.
- The specification of the q-BGP protocol and the associated QoS-aware route selection process can lead to feasible implementations.

- It is feasible to implement the necessary data plane interactions (DSCP swapping, q-FIB, etc.) to provide end-to-end QoS as specified.
- q-BGP can interoperate with classical BGP, due to the use of capability negotiation procedure.
- No significant impairment of the performance of q-BGP was observed, compared to classical BGP, as a result of the addition of QoS-related information.
- q-BGP is able to select distinct routes per QoS class;
- q-BGP reacts to the change of QoS class configuration that can occur on the AS chain.

Furthermore, through testbed tests it was *proved the validity and feasibility of the specified PCS-based approach for building inter-domain MPLS LSPs was proven* – for providing hard QoS guarantees. In particular:

- It is possible to compute inter-domain QoS constrained paths.
- The specification of the communication protocol between PCSs (PCP) is a basis for feasible implementation of the protocol.
- It is feasible to integrate the operation of q-BGP with the PCS to discover QoS-aware path candidates.
- The speed of path computation indicates that the PCS-based approach is a viable solution for larger networks.
- The activation of the solution option to provide hard QoS guarantees does not impact the size of inter-domain routing tables one entry per QoS-class is required.

7.3 Scalability of the MESCAL Solution

The scalability analysis has shown that *the MESCAL solution is scalable*. The design of the solution approach is based on a set of principles that contribute to overall scalability. In particular:

- Provider domains interact only with adjacent domains with distinct interfaces at the service and IP planes.
- Inter-domain service layer interactions are achieved through cascaded and bilateral QoS peering arrangements, which are shown to be scalable in terms of the number of pSLSs required for large networks.
- pSLSs address aggregate traffic flows between domains; per-flow end-to-end QoS signalling and bandwidth reservations for QoS are not required.
- As QoS routes are constrained between those provider-domains, that have established pSLSs, providers have increased levels of control in handling the volumes of QoS traffic transported through their domains.
- QoS peering is simplified through the use of Meta-QoS-Classes, reducing the complexity of QoS bindings between adjacent domains, and making the inclusion of QoS-related information in BGP scalable at the scale of the Internet.
- The MESCAL solution requires the set-up of marking/remarking mechanisms at the domains' edges and the employment of QoS-aware routing, resulting in expansion of the routing table space; both these aspects, scale with the number of pSLSs, which in turn scales with the number of QoS-classes and provider domains participating in the QoS solution.
- The three solution options proposed by MESCAL to address the diversity of customer requirements for QoS services have scalability properties that are appropriate to their expected deployment i.e. mass market solutions relying on loose QoS are highly scalable, solutions for more specialized QoS needs are less scalable.

7.4 q-BGP

In addition to proving the validity and feasibility of q-BGP through testbed implementation, simulations have been conducted for assessing and getting insight into its behaviour and impact on network performance in larger Internet-like topologies. The conclusions of this work can be summarised as follows:

Injecting QoS information into BGP when coupled with a QoS-aware route selection process can result in better performing routes across meta-QoS-class planes compared to classical BGP. However, poor selection of q-BGP policy parameters may degrade performance compared to standard BGP.

It has been demonstrated that, in addition to any service differentiation implemented by utilising different PHBs/packet forwarding priorities within the routers of each AS, the application of appropriate route selection policies on advertised QoS attributes can also deliver QoS differentiation.

While the quantity of q-BGP messages is greater than for plain BGP the results indicate that when scaled to Internet-sized AS topologies there is only a three-fold increase in q-BGP-update messages.

Stability tests show that convergence times are worst when q-BGP selection policies are most stringent. The adoption of these policies also delivers worse end-to-end performance and it is desirable on the counts of both convergence time and delivered QoS to adopt broader selection criteria.

In summary, the addition of administratively set QoS attributes to BGP can be achieved in a scalable and incremental manner that maintains the scalability of classical BGP while yielding better performance in terms of end-to-end delivered QoS. This maintains a certain level of predictability for INPs while improving performance.

7.5 Off-line TE

For realizing the off-line decision-making processes required by the MESCAL solution for dimensioning provider domains with the appropriate amount of intra- and inter-domain resources, suitable traffic engineering algorithms for uni- and multi-cast traffic have been developed and tested in computer/simulation environments. Experimentation showed that the *developed algorithms achieved their functional objectives with reasonable performance, yielding favourable results when compared – where possible - to ad-hoc configurations and alternative schemes.* In particular, simulation results show that:

Inter-domain TE

- The specified genetic-based algorithm for off-line inter-domain traffic engineering can find nearoptimum solutions/allocations for accommodating QoS-sensitive traffic demands assigning them to intra-domain resources (represented by l-QCs) and inter-domain resources (represented by pSLSs).
- The genetic-based algorithm outperforms ad-hoc random-based assignment approaches in the sense that the determined bindings (combinations of l-QCs and pSLSs) result in significant lower cost (represented by the sum of the pSLS cost, intra-domain TE cost and inter-domain link utilisation); it has been shown that under simplified conditions the cost of the genetic algorithm is close to a theoretical lower bound cost.
- The algorithm scales with the number of traffic demands, I-QCs and pSLSs, yielding processing times in the order of minutes to hours, which is acceptable for its prescribed time-scale of operation (in the order of days to weeks).

Intra-domain TE

• DSCP-aware routing can successfully be used for providing individual routing to different traffic classes. Bandwidth- as well as hop-count-optimised routing per QoS class can be configured to run in parallel on the same physical network, using the proposed link weight optimisation techniques.

- The load is balanced more evenly across the network than with standard all-class shortest path routing on inverse capacity link weights.
- The approach is scalable to large ASs with more than 500 nodes.

Inter- and intra-domain TE

• An integrated approach to off-line TE results in lower cost TE solutions with lower total consumed bandwidth than a decoupled approach in which inter-domain and intra-domain solutions are treated separately.

Multicast TE

- The specified scheme for intra- and inter-domain multicast TE scheme, which is based on a genetic algorithm approach, can produce effective resource optimisation solutions with constrained bandwidth capacity.
- The algorithm outperforms alternative, shortest-path-based, schemes, resulting in improved network design -savings in consumed intra-domain bandwidth and also balanced inter-domain link utilisation- as well as in reduced blocking rate for group join requests.
- The proposed TE scheme scales with the number of multicast groups and network nodes, yielding processing times in the order of minutes, which is acceptable for off-line TE computations.
- By applying per l-QC trees (engineered through the aforementioned scheme) for each QoS class per group, fairness problems amongst QoS-classes that appear in DiffServ aware multicast can be avoided.

7.6 c/pSLS Management

The pSLS-centric interactions between providers for negotiating pSLSs as well as the interfaces between the service handling and the TE functions within a provider domain, required by the MESCAL solution, have been developed and tested for their validity and performance. The results prove that the process of pSLS establishment between providers can be feasibly realised in a highly-comprehensive manner, hiding underlying complexity; also, that the specified pSLS handling functions can safely and efficiently –with increased levels of automation and flexibility- realise the decisions and/or provide the input required to the inter-domain TE functions, off-line and q-BGP. In particular, it was shown that:

- It is possible to describe in a highly-abstracted form the essential aspects of pSLSs, as appropriate to the type of QoS exchange and the underlying business relationships between providers, hiding underlying complexity and realisation details.
- The concepts and notions required by the operation of the pSLS-aware service layer functions of the MESCAL QoS solution are consistent and can lead to implementation; on pSLS establishment, it is feasible and scalable to derive all information required by the inter-domain TE functions traffic matrix and q-BGP configuration information.
- It is feasible to carry out pSLS ordering and negotiations in an automated fashion, facilitating therefore the process of pSLS establishment between providers; it is possible to fully automate even the logic of negotiating pSLSs, proving the validity of the proposed ordering and negotiation framework.
- On pSLS request epochs, admission control may be exerted for managing the trade-off between long-term performance of the engineered domain and accepted subscriptions/contracts; the proposed algorithm is of polynomial complexity with respect to the number of established pSLSs.
- The developed cSLS invocation handling algorithms perform reasonably well for a variety of traffic scenarios, satisfying the target packet loss rate while achieving satisfactory resource utilization; in addition, inter-domain link utilization gains can be achieved by utilising inter-domain link status information in the admission control scheme.

8 REFERENCES

- [Akam99] Akamai, "Internet Bottlenecks: the Case for Edge Delivery Services," Akamai White Paper, 1999.
- [Antonio04] S.D'Antonio, M.Esposito, S.P.Romano, G.Ventre, "Assessing the scalability of component-based frameworks: the CADENUS case study", ACM SIGMETRICS Performance Evaluation Review, Special Issue of E-commerce, Volume 32, Issue 3, pp. 34-43, December 2004.
- [Bouca05] M.Boucadair Ed., "QoS-Enhanced Border Gateway Protocol", Internet Draft, <draftboucadair-qos-bgp-spec-00.txt>, June 2005.
- [BRES] L.Breslau, E.Knightly, S.Shenker, I.Stoica and Z.Zhang "Endpoint Admission Control: Architectural Issues and Performance", SIGCOMM 2000, Stockholm 2000.
- [Brite] http://www.cs.bu.edu/BRITE
- [bu02] T. Bu and D. Towsley, "On Distinguishing between Internet Power Law Topology Generators", IEEE Infocom 2002
- [CADENUS] European IST (Information Society Technologies) research project, for more on IST-CADENUS visit: http://www.cadenus.fokus.fraunhofer.de/old_entry.html.
- [CAD-D2.3] S.P.Romano, ed., "Resource Management in SLA Networks", D2.3 CADENUS Deliverable, May 2003.
- [CAD-D8] S.P.Romano, ed., "Cadenus Scalability Analysis", D8 CADENUS Deliverable, June 2003.
- [CHU] C.Chuah, L.Subramarian and R.Katz "Furies: A Scalable Framework for Traffic Policing and Admission Control", May 2001, U.C Berkeley Technical Report No. UCB/CSD-01-1144.
- [D1.1] P.Flegkas, et al., D1.1 MESCAL Deliverable, "Specification of Business Models and a Functional Architecture for Inter-domain QoS Delivery", <u>http://www.mescal.org/</u>, June 2003.
- [D1.2] M.Howarth, et al., D1.2 MESCAL Deliverable, "Initial specification of protocols and algorithms for inter-domain SLS management and traffic engineering for QoS-based IP service delivery and their test requirements", <u>http://www.mescal.org/</u>, January 2004.
- [D1.3] N.Wang, et al., D1.3 MESCAL Deliverable, "Final specification of protocols and algorithms for inter-domain SLS management and traffic engineering for QoS-based IP service delivery", <u>http://www.mescal.org/</u>, June 2005.
- [D3.1] E.Mykoniati, et al., D3.1 MESCAL Deliverable, "Specification of test campaigns and experimentation plans for performance analysis and prototype validation", <u>http://www.mescal.org/</u>, September 2004.
- [fal99] M. Faloutsos, P. Faloutsos and C. Faloutsos, "On Power-Law Relationships of the Internet Topology", SIGCOMM 1999.
- [Feam03] N.Feamster, J.Borkenhagen and J.Rexford, "Guidelines for Interdomain traffic engineering," *ACM Computer Communications Review*, Oct 2003.
- [Fortz00] Fortz, B., and M. Thorup, "Internet traffic engineering by optimizing OSPF weights," paper presented at INFOCOM 2000. *Nineteenth Annual Joint Conference of the IEEE Computer and Communications Societies. Proceedings.* IEEE, Vol.2, 2000.
- [For02] B.Fortz and M.Thorup, "Optimizing OSPF/IS-IS weights in a changing world," *IEEE Journal on Selected Areas in Communications*, Vol. 20, pp.756-767, 2002.

- [GROS] M.Grossglauser and D.Tse "A Framework for Robust Measurement-Based Admission Control", IEEE/ACM Transactions on Networking, June 1999.
- [HAB] I.Habib and T.Saadawi "Multimedia Traffic Characteristics in Broadband Networks", IEEE Communications Magazine, July 1992.
- [KAR] V.Elek, G.Karlsson and R.Ronngren "Admission Control based on End-to End Measurements", IEEE INFOCOM 2000.
- [TKN] <u>http://www-tkn.ee.tu-berlin.de/research/trace/trace.html</u>.
- [Wang99] Y.Wang and Z.Wang, "Explicit Routing Algorithms for Internet Traffic Engineering," *Proceedings IEEE ICCCN* 1999, pp. 582-588.
- [Xiao00] X.Xiao et al., "Traffic Engineering with MPLS in the Internet," *IEEE Network*, Vol. 14, No. 2, pp. 28-33, 2000.
- [ZUK] T.Lee and M.Zukerman, "Admission Control for Bursty Multimedia Traffic", IEEE INFOCOM 2001.

Appendix A

9 TESTBED CONFIGURATION

9.1 Introduction

This document describes the MESCAL testbed that has been used during the development and validation phases 1, 2 and 3 defined in [D2.1]. This testbed is located in FTR&D premises, in Caen, France.

The same platform is used for carrying out these 3 phases. From phase 1 to phase 3 the core of the platform (AS topology, links, addressing, core technologies) remains the same but evolves, mainly in terms of configuration in order to support the specific constraints and requirements of each of these phases.

Phase 1 aims at validating the overall testing environment including: hardware, software, traffic generators, BGP and QoS configuration. This phase needs a large number of Linux and non-Linux features to be activated but does not include any of the new features MESCAL has to develop. This phase allows verifying that the selected environment is suitable for supporting phases 2 and 3 and definitively confirm the pertinence of MESCAL technical choices for this testbed. Additionally, the experience gained during this phase will help to identify and/or improve all appropriate tools for maintaining the testbed in an efficient way.

In particular, this phase will allow:

- To validate the inter-domain routing with ZebOS running on Linux PCs by setting-up several ASs and by configuring BGP between those domains.
- To validate, at the data plane level, the DSCP swapping (marking/remarking) between ASs in order to signal a given meta-QoS-class.
- To validate the implementation of l-QCs in each domain using Linux traffic control features.

Phase 2 aims at validating the loose service option. The same testbed will be used for validating q-BGP implementation and the enhanced Linux IP forwarding. Depending on the tests that will be performed the number and the definition of local-QoS-classes in each AS will be adapted together with the pSLS definitions. This will be on per test basis.

Phase 3 aims at validating the Path Computation System (PCS) function. For this purpose, the PCS implementation will be uploaded in an appropriate set of Linux routers but the overall infrastructure will remain the same as for the phase 1. No MPLS features will be deployed since the project will focus only on the computation of inter-AS paths for establishing inter-domain LSPs.

More precisely, this document provides information on:

- The AS topology
- The deployed hardware,
- The links set-up between routers,
- The IP addressing scheme,
- The BGP and q-BGP configurations,
- The QoS policies defined for each domain, including a description of the local-QoS-classes deployed in each AS,
- The tools developed for maintaining, configuring and monitoring the platform,
- The tools used for generating and analysing IP traffic.

9.2 Autonomous system topology

The testbed supports 8 autonomous systems. A single router implements each AS except AS4, which is composed of 3 routers. AS4 and AS6 represent Tier-one providers and have more network resources allocated. AS1 represents a Tier-3 provider. The remaining ASs are Tier-two providers.

In phase 1, BGP is activated at the boundaries of each domain and iBGP is activated within AS4 domain. In phase 2 q-BGP and q-iBGP will be activated instead.

These ASs are interconnected as roughly shown in Figure 94 and depicted in more details in Figure 95.

Some inter-AS interconnections have been doubled in order to allow more sophisticated inter-domain routing tests and to evaluate eventual load balancing features especially between:

- AS4 and AS6
- AS3 and AS4

Even if this topology is far from the real Internet it will nevertheless allow learning different QoS routes thanks to ad-hoc activation of pSLS. If necessary, a maximum of 6 crossed AS can be reached provided the appropriate configuration.

Figure 95 gives a more accurate view of the testbed. It groups on the same picture different level of information:

- AS numbers
- Links between ASs
- IP address of the interfaces, with their interface number on each machine
- DS code point values used to signal local-QoS-classes
- Agreed DS code point values used to signal meta-QoS-classes between domains together with the capacity provisioned for each class.
- Role of the remaining interface: management interface or customer interface

Smartbit interfaces are connected to each router via a dedicated interface. These interfaces are mainly used for injecting load traffic in the testbed. Some of the routers (AS1 and AS8) have additional Smartbit interfaces connected in order to inject customer traffic used for measurement purposes.



Figure 94: FTR&D MESCAL testbed: hierarchical view



Figure 95: FTR&D MESCAL testbed: detailed architecture

Copyright © MESCAL Consortium, July 2005



Figure 96: FTR&D MESCAL testbed: Network interfaces schema

9.3 Testbed components

The testbed deployed by France Telecom R&D in its premises contains many devices including network elements (routers, hubs...), workstations and test equipments. In this current section, we list and describe both the hardware and software components used for building the FTR&D MESCAL testbed.

9.3.1 Hardware components

9.3.1.1 PCs

PCs used in the testbed are of four types:

- Type 1: Intel® Xeo, CPU 1.7 GHz, 17 GB hard disk drive, 512 MB RAM
- Type 2: Intel® PIII, CPU 1.0 GHz, 20 GB hard disk drive, 512 MB RAM
- Type 3: Intel® P4, CPU 1.7 GHz, 20 GB hard disk drive, 512 MB RAM
- Type 4: Intel® PIII, CPU 500 MHz, 8 GB hard disk drive, 256 MB RAM

Name	Туре	Available PCI Slots
Mescal11	1	4
Mescal21	3	3
Mescal31	3	3
Mescal41	2	3
Mescal42	1	4
Mescal43	2	3
Mescal51	1	4
Mescal61	1	4
Mescal71	1	4
Mescal81	3	3
PC-admin	4	4

 Table 24: PC characteristics

Except PC-admin, all these PCs are used as PC-based routers.

Additional PCs could be added to the testbed and would be used to emulate customer premises.

9.3.1.2 Traffic Generators

9.3.1.2.1 Smartbits

Two *SmartBits* chassis (SMB 2000 and SMB600) are available in the lab. These equipments are used to inject traffic in the network and to carry out measurements.

The SMB 2000 is equipped with 20 Ethernet cards:

- 10 of them are 10/100 Mbit/s full duplex cards,
- The others are 10 Mbit/s half duplex cards.

The SMB 6000 is equipped with 2 cards of 2 x 10/100 Mbit/s Ethernet cards. Those cards support additional features the SMB 2000 card does not support and should be preferred when QoS measurements need to be carried out.

9.3.1.2.2 QARobots

In addition, *QARobot* will be used to generate BGP messages in order to test routing features. This tool will be used to validate the behaviour of q-BGP and more especially the conformity of BGP messages.

9.3.2 Software components

9.3.2.1 Operating system

Linux Red Hat version 9 (kernel version of 2.4.20-8) is installed on all PC-based routers and PC-admin.

9.3.2.2 Software information

Software	Version	Company/Package	Description
GCC	3.2.2	GNU	C++ Compiler
ZebOS	5.3.1	IPInfusion	Routing stack including BGP
Ethereal	0.9.8		Traffic analyzer
MGEN	4.2	Naval Research Laboratory	Traffic generator
Jnettop	0.9	GNU	Real-time interface bandwidth
			measurement
NTP	4.2		Time server

Table 25: Software information

During phase 1, the ZebOS routing stack is used without any modification. During phase 2, it is enhanced to support QoS related messages and information. Phase 3 relies on the q-BGP implementation realised during phase 2.

MGEN and *TG* are traffic generators that could be used as a complement to the *Smartbits*. These tools can generate UDP and TCP traffic (especially TG). They can set the DS code point on a per flow basis.

9.4 Configuration for phase 1

9.4.1 User' accounts

Only two accounts have been created in all PCs: "root" and "mescal".

9.4.2 Remote connection

Remote connections to the routers are achieved via SSH. There are no restrictions between testbed components. From outside the testbed, connections are filtered by an external firewall. Only **PC-admin** has been made accessible. The firewall ensures a network address translation for this latter machine. The corresponding public IP address will be provided by FTRD on request. Firewall rules will be updated to allow external connexions from MESCAL partners.

In addition, the **ftpd** service has been enabled in **PC-admin** and in the routers.

9.4.3 Internet access

An Internet access is configured in all machines involved in the testbed. All Internet connexions go through **PC-admin**, which acts as a HTTP/FTP proxy.

9.4.4 Firewall rules

Firewall rules have been added in order to control communications from/to testbed machines.

9.4.5 Time synchronisation

PC-admin is configured as a NTP server. It synchronizes its clock with a public NTP server. All routers in the testbed synchronise their clock with PC-admin.

9.4.6 Printer

A postscript printer, called **Gutenberg**, is available for all the machines of the testbed.

9.4.7 AS identifiers

Hereafter is listed the AS number affected to each AS. This AS number will be used when configuring BGP instances.

AS	AS number
AS1	1
AS2	2
AS3	3
AS4	4
AS5	5
AS6	6
AS7	7
AS8	8

Table 26: AS numbers

9.4.8 LANs

This table summarizes the addressing plan for each local area network.

Subnet	Address/Mask	Device/Interface	Address	Description
Administrative Network	192.168.66.0/24			
		PC Admin /Eth0	.73	AN2
		Firewall	.1	AN1

Table 27: Administrative network addressing

9.4.9 Customer addresses

The following table lists the IP network addresses used by customers connected to each AS (via a direct physical connection to each AS)

AS	Customers	IP addresses realm
AS1	Customer11	11.0.0.0/10
	Customer12	12.0.0.0/10
	Customer13	13.0.0/10
	Customer14	14.0.0.0/10
AS2	Customer21	21.0.0.0/10
	Customer22	22.0.0.0/10
	Customer23	23.0.0.0/10
	Customer24	24.0.0.0/10
AS3	Customer31	31.0.0.0/10
	Customer32	32.0.0/10
	Customer33	33.0.0/10
	Customer34	34.0.0.0/10
AS4	Customer41	41.0.0.0/10
	Customer42	42.0.0.0/10
	Customer43	43.0.0/10

	Customer44	44.0.0.0/10
AS5	Customer51	51.0.0.0/10
	Customer52	52.0.0.0/10
	Customer53	53.0.0/10
	Customer54	54.0.0.0/10
AS6	Customer61	61.0.0/10
	Customer62	62.0.0/10
	Customer63	63.0.0/10
	Customer64	64.0.0/10
AS7	Customer71	71.0.0.0/10
	Customer72	72.0.0.0/10
	Customer73	73.0.0/10
	Customer74	74.0.0/10
AS8	Customer81	81.0.0/10
	Customer82	82.0.0/10
	Customer83	83.0.0/10
	Customer84	84.0.0/10

Table 28: Customers IP address realms

9.4.10 Network addresses announced by each AS

This table lists the IP network addresses announced by each AS.

AS	Customers	IP addresses realm
AS1	Network11	193.251.128.0/19
	Network12	212.167.0.0/21
	Network13	62.42.0.0/16
	Network14	193.251.240.0/20
AS2	Network21	194.52.168.0/21
	Network22	202.167.22.0/24
	Network23	194.199.98.0/24
	Network24	128.223.60.102/20
	Network25	193.41.227.0/24
	Network26	192.176.123.0/24
	Network27	192.94.149.0/24
AS3	Network31	216.191.64.0/20
	Network32	154.101.11.10/24
	Network33	132.150.224.0/24
	Network34	155.234.165.12/19
	Network35	199.79.131.0/24
	Network36	198.26.215.0/24
	Network37	196.11.196.0/24
AS4	Network41	62.177.143.254/20
	Network42	62.216.31.254/19
	Network43	146.188.61.109/22
	Network44	195.69.144.12/18
	Network45	192.70.132.0/24
	Network46	216.169.114.0/24
	Network47	216.116.175.0/24
	Network48	214.3.214.0/24
	Network49	205.237.35.0/24
	Network410	204.222.17.0/24
	Network411	204.116.187.0/24
	Network412	203.116.188.0/24
	Network413	216.103.190.0/24
	Network414	216.84.141.0/24
	Network415	213.239.59.0/24
	Network416	213.205.25.0/30

	Network417	198.205.10.0/24
AS5	Network51	198.32.247.85/20
	Network52	12.129.192.134/20
	Network53	170.170.0.0/16
	Network54	203.34.233.0/24
AS6	Network61	205.204.1.0/24
	Network62	216.255.39.0/24
	Network63	193.100.167.0/24
	Network64	193.19.208.0/24
	Network65	198.129.78.23/20
	Network66	193.111.167.0/24
	Network67	192.197.12.0/24
	Network68	192.58.243.0/24
	Network69	192.43.226.0/24
	Network610	200.108.174.0/24
	Network611	200.10.200.0/24
	Network612	194.215.31.0/24
	Network613	195.177.120.0/24
	Network614	193.108.167.0/24
	Network615	170.163.0.0/16
	Network616	217.108.191.0/24
	Network617	217.10.217.0/24
	Network618	216.95.141.0/24
	Network619	212.78.144.0/24
AS7	Network71	209.215.12.0/24
	Network72	208.147.73.0/24
	Network73	207.245.48.0/27
	Network74	192.228.46.0/24
AS8	Network81	198.69.130.0/24
	Network82	193.16.208.0/24
	Network83	192.35.226.0/24
	Network84	192.236.46.0/24

Table 29: Customers IP address realms

9.4.11 Routing configuration

!

This section provides routing information, as currently configured in the testbed. The routing protocol used in the network is BGP (Border Gateway Protocol). Each AS has at least one e-BGP session with its adjacent ASs. i-BGP sessions have been established between all AS4 border routers.

Hereafter is an example of a BGP configuration (example of **MESCAL420**)

```
!
! Config for ZebOS version 5.3.1:03312003-Main (i686-pc-linux-gnu)
! Written 2004/05/19 17:59:55
!
!
banner motd Welcome to MESCAL42 Router :))
if-arbiter
!
interface lo
ip address 42.1.1.1/32
!
interface eth0
ip address 192.168.66.69/24
!
interface eth1
ip address 3.3.3.6/30
```

```
interface eth2
ip address 41.41.41.10/30
L
interface eth3
ip address 42.42.42.1/30
interface eth4
ip address 2.2.2.2/30
interface eth5
ip address 42.42.42.5/30
interface eth6
router bgp 4
bgp router-id 42.42.42.1
bqp log-neighbor-changes
bqp scan-time 10
network 41.0.0/10
network 42.0.0/10
network 43.0.0/10
network 44.0.0.0/10
network 62.177.128.0/20
network 62.216.0.0/19
network 146.188.60.0/22
network 192.70.132.0/24
network 195.69.128.0/18
network 198.205.10.0/24
network 203.116.188.0/24
network 204.116.187.0/24
network 204.222.17.0/24
network 205.237.35.0/24
network 213.205.25.0/30
network 213.239.59.0/24
network 214.3.214.0/24
network 216.84.141.0/24
network 216.103.190.0/24
network 216.116.175.0/24
network 216.169.114.0/24
redistribute connected
redistribute static
 timers bgp 10 15
neighbor 2.2.2.1 remote-as 2
neighbor 2.2.2.1 advertisement-interval 5
neighbor 2.2.2.1 prefix-list FLAN out
neighbor 3.3.3.5 remote-as 3
neighbor 3.3.3.5 advertisement-interval 5
neighbor 3.3.3.5 prefix-list FLAN out
neighbor 41.41.41.9 remote-as 4
neighbor 41.41.41.9 advertisement-interval 5
neighbor 41.41.41.9 prefix-list FLAN1 out
neighbor 42.42.42.2 remote-as 4
neighbor 42.42.42.2 advertisement-interval 5
neighbor 42.42.42.2 prefix-list FLAN2 out
neighbor 42.42.42.6 remote-as 5
neighbor 42.42.42.6 advertisement-interval 5
neighbor 42.42.42.6 prefix-list FLAN out
ip route 3.3.3.0/30 41.41.41.9 2
ip route 3.3.3.0/30 42.42.42.2 3
```

```
ip route 41.0.0.0/30 41.41.41.9 2
ip route 41.0.0.0/30 42.42.42.2 3
ip route 41.41.41.0/30 41.41.41.9 2
ip route 41.41.41.0/30 42.42.42.2 3
ip route 41.41.41.4/30 41.41.41.9 2
ip route 41.41.41.4/30 42.42.42.2 2
ip route 43.0.0.0/30 42.42.42.2 2
ip route 43.0.0.0/30 41.41.41.9 3
ip route 43.43.43.0/30 42.42.42.2 2
ip route 43.43.43.0/30 41.41.41.9 3
ip route 192.168.1.0/24 192.168.66.1
ip prefix-list FLAN seq 5 deny 192.168.66.0/24
ip prefix-list FLAN seq 10 deny 192.168.1.0/24
ip prefix-list FLAN seq 15 permit any
ip prefix-list FLAN1 seq 5 deny 3.3.3.0/30
ip prefix-list FLAN1 seq 10 deny 41.41.41.0/30
ip prefix-list FLAN1 seg 15 deny 192.168.66.0/24
ip prefix-list FLAN1 seg 20 deny 192.168.1.0/24
ip prefix-list FLAN1 seq 25 permit any
ip prefix-list FLAN2 seq 5 deny 43.43.43.0/30
ip prefix-list FLAN2 seg 10 deny 192.168.66.0/24
ip prefix-list FLAN2 seq 15 deny 192.168.1.0/24
ip prefix-list FLAN2 seq 20 permit any
```

9.4.11.1 e-bgp

An e-BGP session is configured in ZebOS routers as follows (example of AS4 declaring AS2 as neighbour):

router bgp 4 bgp router-id 42.42.42.1 neighbor 2.2.2.1 remote-as 2

9.4.11.2 i-bgp

In the testbed, AS4 is made of three routers. Each router has to declare the other two routers as i-BGP neighbours, to do so the following configuration has to be added (example of **MESCAL420**):

```
router bgp 4
bgp router-id 42.42.42.1
neighbor 41.41.41.9 remote-as 4
neighbor 42.42.42.2 remote-as 4
```

9.4.11.3 Networks

In order to advertise networks prefixes, the command "network" is used as listed below:

```
router bgp 4
network 214.3.214.0/24
network 216.84.141.0/24
network 216.103.190.0/24
network 216.116.175.0/24
network 216.169.114.0/24
```

9.4.11.4 Static routes

We made the decision not to activate an IGP protocol within AS4 domain. This is motivated by the fact that we prefer not having to modify IGP related processes feeding the FIBs when q-BGP is deployed and enabled. Thus, we will make use of static routes in order to simplify the development phase.

Static routes can be configured with the command "ip route". The example below illustrates the static routes that have been configured in **MESCAL420** for joining other AS4 intra-domain destination.

```
ip route 41.41.41.0/30 42.42.42.2 3
ip route 41.41.41.0/30 41.41.41.9 2
ip route 41.41.41.4/30 42.42.42.2 2
ip route 43.0.0.0/30 42.42.42.2 2
ip route 43.0.0.0/30 41.41.41.9 3
ip route 43.43.43.0/30 42.42.42.2 2
ip route 43.43.43.0/30 41.41.41.9 3
```

9.4.11.5 Prefix list

The ZebOS implementation can filter network prefixes announcements on a peer-by-peer basis thanks to the use of the "**prefix-list**" command. The configuration bellow allows to send all configured network prefixes except the 43.43.43.0/30, 192.168.66.0/24 and 192.168.1.0/24 Addresses.

```
ip prefix-list FLAN2 seq 5 deny 43.43.43.0/30
ip prefix-list FLAN2 seq 10 deny 192.168.66.0/24
ip prefix-list FLAN2 seq 15 deny 192.168.1.0/24
ip prefix-list FLAN2 seq 20 permit any
```

9.4.11.6 Fast link failover detection

ZebOS has been configured to support fast link failover detection.

9.4.11.7 BGP timers

The BGP timers that can be configured are: "keepalive", "holdtime" and "connect". We configured the two first timers to 10s and 15s respectively. This is achieved with the following command:

Bgp router 4 timers bgp 10 15

The motivation behind these values is to decrease the time needed for the detection of invalid routes.

9.4.11.8 Route selection process

ZebOS allows selecting a type of route selection process. We selected the "rfc-1771-path-selection" for all routers.

9.4.12 Local QoS class DSCP values

This table summarizes the l-QC DSCP values that are used in each AS.

AS	Local QC	DSCP	b-DSCP	x-DSCP	d-DSCP	b-DS	x-DS	d-DS
AS1	1-QC11	001010	0000-1010	0xa	10	0010-1000	0x28	40
	1-QC12	001100	0000-1100	0xc	12	0011-0000	0x30	48
	1-QC13	001110	0000-1110	0xe	14	0011-1000	0x38	56
	1-QC14	000000	0000-0000	0x0	0	0000-0000	0x0	0
AS2	1-QC21	010010	0001-0010	0x12	18	0100-0100	0x48	72
	1-QC22	010100	0001-0100	0x14	20	0101-0000	0x50	80
	1-QC23	010110	0001-0110	0x16	22	0101-1000	0x58	88
	1-QC24	000000	0000-0000	0x0	0	0000-0000	0x0	0
AS3	1-QC31	011010	0001-1010	0x1a	26	0110-1000	0x68	104
	1-QC32	011100	0001-1100	0x1c	28	0111-0000	0x70	112
	1-QC33	011110	0001-1110	0x1e	30	0111-1000	0x78	120
	1-QC34	000000	0000-0000	0x0	0	0000-0000	0x0	0

AS4	1-QC41	100010	0010-0010	0x22	34	1000-1000	0x88	136
	1-QC42	100100	0010-0100	0x24	36	1001-0000	0x90	144
	1-QC43	100110	0010-0110	0x26	38	1001-1000	0x98	152
	1-QC44	000000	0000-0000	0x0	0	0000-0000	0x0	0
AS5	1-QC51	101010	0010-1010	0x2a	42	1010-1000	0xa8	168
	1-QC52	101100	0010-1100	0x2c	44	1011-0000	0xb0	176
	1-QC53	101110	0010-1110	0x2e	46	1011-1000	0xb8	184
	1-QC54	000000	0000-0000	0x0	0	0000-0000	0x0	0
AS6	1-QC61	110010	0011-0010	0x32	50	1100-1000	0xc8	200
	1-QC62	110100	0011-0100	0x34	52	1101-0000	0xd0	208
	1-QC63	110110	0011-0110	0x36	54	1101-1000	0xd8	216
	1-QC64	000000	0000-0000	0x0	0	0000-0000	0x0	0
AS7	1-QC71	111010	0011-1010	0x3a	58	1110-1000	0xe8	232
	1-QC72	111100	0011-1100	0x3c	60	1111-0000	0xf0	240
	1-QC73	111110	0011-1110	0x3e	62	1111-1000	0xf8	248
	1-QC74	000000	0000-0000	0x0	0	0000-0000	0x0	0
AS8	1-QC11	001010	0000-1010	0xa	10	0010-1000	0x28	40
	1-QC12	001100	0000-1100	0xc	12	0011-0000	0x30	48
	1-QC13	001110	0000-1110	0xe	14	0011-1000	0x38	56
	1-QC14	000000	0000-0000	0x0	0	0000-0000	0x0	0

Table 30: I-QC DSCP values

9.4.13 Inter-domain Meta-QoS-classes DSCP values

This table summarizes DSCP values used between ASs in order to signal meta-QoS-classes. These values are used in both directions (i.e. upstream and downstream).

AS-AS	MC	b-DSCP	b-DSCP	x-DSCP	d-DSCP	b-DS	x-DS	d-DS
AS1-AS2	MC1	011010	0001-1010	0x1a	26	01101000	0x68	104
	MC2	011100	0001-1100	0x1c	28	01110000	0x70	112
	MC3	011110	0001-1110	0x1e	30	01111000	0x78	120
	MC4	000000	0000-0000	0x0	0	00000000	0x0	0
AS1-AS3	MC1	100010	0010-0010	0x22	34	10001000	0x88	136
	MC2	100100	0010-0100	0x24	36	10010000	0x90	144
	MC3	100110	0010-0110	0x26	38	10011000	0x98	152
	MC4	000000	0000-0000	0x0	0	00000000	0x0	0
AS2-AS4	MC1	110010	0011-0010	0x32	50	11001000	0xc8	200
	MC2	110100	0011-0100	0x34	52	11010000	0xd0	208
	MC3	110110	0011-0110	0x36	54	11011000	0xd8	216
	MC4	000000	0000-0000	0x0	0	00000000	0x0	0
AS7-AS5	MC1	111010	0011-1010	0x3a	58	11101000	0xe8	232
	MC2	111100	0011-1100	0x3c	60	11110000	0xf0	240
	MC3	111110	0011-1110	0x3e	62	11111000	0xf8	248
	MC4	000000	0000-0000	0x0	0	00000000	0x0	0
AS3-AS4	MC1	111010	0011-1010	0x3a	58	11101000	0xe8	232
	MC2	111100	0011-1100	0x3c	60	11110000	0xf0	240
	MC3	111110	0011-1110	0x3e	62	11111000	0xf8	248
	MC4	000000	0000-0000	0x0	0	00000000	0x0	0
AS4-AS5	MC1	001010	0000-1010	0xa	10	00101000	0x28	40
	MC2	001100	0000-1100	0xc	12	00110000	0x30	48
	MC3	001110	0000-1110	0xe	14	00111000	0x38	56
	MC4	000000	0000-0000	0x0	0	00000000	0x0	0
AS4-AS6	MC1	010010	0001-0010	0x12	18	01001000	0x48	72
	MC2	010100	0001-0100	0x14	20	01010000	0x50	80
	MC3	010110	0001-0110	0x16	22	01011000	0x58	88
	MC4	000000	0000-0000	0x0	0	00000000	0x0	0
AS6-AS7	MC1	101010	0010-1010	0x2a	42	10101000	0xa8	168
	MC2	101100	0010-1100	0x2c	44	10110000	0xb0	176

	MC3	101110	0010-1110	0x2e	46	10111000	0xb8	184
	MC4	000000	0000-0000	0x0	0	00000000	0x0	0
AS6-AS8	MC1	111010	0011-1010	0x3a	58	11101000	0xe8	232
	MC2	111100	0011-1100	0x3c	60	11110000	0xf0	240
	MC3	111110	0011-1110	0x3e	62	11111000	0xf8	248
	MC4	000000	0000-0000	0×0	0	00000000	0x0	0

Table 31: Inter-domain meta-QoS-class DSCP values

9.4.14 Bandwidth thresholds per Meta-QoS-class

This table illustrates the amount of bandwidth that is negotiated between two adjacent ASs and per direction. This amount is expressed in Mbit/s.

AS-AS	MC	BW (Mbit/s) ➔	BW (Mbit/s)
AS1-AS2	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 5	Maximum is 5
		Min is 2	Min is 2
AS1-AS3	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 5	Maximum is 5
		Min is 2	Min is 2
AS2-AS4	MC1	1	1
	MC2	1	1
	MC3		
	MC4	Maximum is 7	Maximum is 7
	NG1	Min is 4	Min is 4
AS7-AS5	MCL	1	1
	MC2	1	1
	MC3		
	MC4	Maximum 18 5	Maximum 18 5
	MC1	MIN IS Z	MIN IS Z
A53-A54	MC2	1	1
	MC2	1	1
	MC4	⊥ Mavimum ig 7	⊥ Mavimum is 7
	ine i	Min is 4	Min is 4
AS4-AS5	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 7	Maximum is 7
		Min is 4	Min is 4
AS4-AS6	MC1	3	3
	MC2	3	3
	MC3	3	3
	MC4	Max is 10	Max is 10
		Min is 1	Min is 1
AS6-AS7	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 5	Maximum is 5
		Min is 2	Min is 2
AS6-AS8	MC1	1	1
	MC2	1	1
	MC3	1	1

MC4	Maximum is 5	Maximum is 5
	Min is 2	Min is 2

Table 32: Bandwidth threshold per meta-QoS-class and per pSLS

9.4.15 Maximum bandwidth per local-QoS-class

This table summarizes the amount of bandwidth allocated per local QoS class. AS4 only is concerned since it is the sole AS in the platform to have intra-domain links.

AS	1-QC	BW in Mbit/s
AS4	l-QC1	5
	1-QC2	5
	1-QC3	5
	l-QC4	Maximum is 20
		Min is 5

Table 33: Bandwidth threshold per local-QoS-class

9.4.16 DiffServ-related configuration

The implementation of the classes of service in the testbed will be achieved thanks to the activation of the HTB (Hierarchical Token Bucket), or the priority queuing discipline coupled with a HTB.

Preliminary tests we achieved shown that the Linux CBQ implementation had difficulties to handle more than 1.5Mbit/s of IP traffic. In addition, the TBF Linux implementation has a 1Mbit/s limitation.

In order to ease the configuration and the debugging operations, the following structure is followed for all routers present in the testbed.

Two files are created for each interface: qsi-ethx and qsHTB-ethx.

- **qsi-ethx**: this file contains the ingress related DiffServ policy configuration
- **qsHTB-ethx**: this file contains the egress related DiffServ policy configuration

In addition, the following files are used to execute the configuration of all interfaces.

- **qsi**: This script launches the ingress related DiffServ policy configuration for all relevant interfaces.
- **qse**: This script launches the egress related DiffServ policy configuration for all relevant interfaces.
- **qsa**: This script launches the ingress and the egress related DiffServ policy configuration for all relevant interfaces.
- **qsdel**: This script deletes all ingress and egress policies.

Below are listed some of these files that are used to configure DiffServ policies on MESCAL110:

9.4.16.1 qsa

```
#!/bin/bash
HOME_DSMARK=/home/mescal/scripts/DSmarking
$HOME_DSMARK/qsdel
$HOME_DSMARK/qsi
$HOME_DSMARK/qse
```

9.4.16.2 qsi

#!/bin/bash

```
HOME_DSMARK=/home/mescal/scripts/DSmarking
$HOME_DSMARK/qsi-eth1
$HOME_DSMARK/qsi-eth2
```

9.4.16.3 qse

#!/bin/bash

HOME_DSMARK=/home/mescal/scripts/DSmarking \$HOME_DSMARK/qshtb-eth1 \$HOME_DSMARK/qshtb-eth2

9.4.16.4 qsdel

#!/bin/bash
tc qdisc del dev ethl root
tc qdisc del dev eth2 root
tc qdisc del dev eth1 ingress
tc qdisc del dev eth2 ingress

9.4.16.5 qsi-eth1

#!/bin/bash #### #Interfaces AS1-AS2 #_ INGRESS=eth1 #### # Masks #_ MASK1=0xfc MASK2=0x03 MASK3=0xff #### # Local QoS classes #_ 1QC1=0x28 1QC2=0x30 1QC3=0x38 1QC0=0x00 ### # Meta-QoS-classes as signaled by peers #_ ICP210=0x00 ICP211=0x68 ICP212=0x70 ICP213=0x78 #### # Meta-QoS-classes as should be signaled if similar codes are # used in both ways #_ ICP120=0x00 ICP121=0x68 ICP122=0x70 ICP123=0x78

Rates for policing purposes # RATE_Total=10Mbit RATE_ICP210=2.5Mbit RATE_ICP211=2.5Mbit RATE_ICP212=2.5Mbit RATE_ICP213=2.5Mbit #### # Attach an ingress qdisc to the \$INGRESS interfaces #_ echo \$INGRESS "ingress policies configuration started......" tc qdisc add dev \$INGRESS handle ffff: ingress ##### # Definition of filter that will put MC1 traffic in the relevant # class # tc filter add dev \$INGRESS parent ffff: protocol ip prio 1 u32 \setminus match ip tos \$ICP211 \$MASK1 \ police rate \$RATE_ICP211 buffer 10k \setminus drop flowid :1 ##### # Definition of filter that will put MC2 traffic in the relevant # class. This class is dedicated to TCP traffic # tc filter add dev \$INGRESS parent ffff: protocol ip prio 1 u32 \ match ip tos \$ICP212 \$MASK1 \ police rate \$RATE_ICP212 buffer 10k \ drop flowid :2 #match ip protocol 6 0xff \setminus ##### # Definition of filter that will put MC3 traffic in the relevant # class. This class is dedicated to UDP traffic #_ tc filter add dev \$INGRESS parent ffff: protocol ip prio 1 u32 \setminus match ip tos \$ICP213 \$MASK1 \ police rate \$RATE_ICP213 buffer 10k \ drop flowid :3 ##### # Definition of filter that will put MC0 traffic in the relevant # class. # tc filter add dev \$INGRESS parent ffff: protocol ip prio 1 u32 \ match ip tos \$ICP210 \$MASK1 \ flowid :4 ##### # Definition of filter that will drop all other types of traffic tc filter add dev \$INGRESS parent ffff: protocol ip prio 2 u32 match ip tos 0x0 0x0 police mtu 1 drop flowid :4 echo \$INGRESS ".....finished"

9.4.16.6 qsHTB-eth1

```
#!/bin/bash
####
#Interfaces AS1-AS2
#
EGRESS=eth1
####
# Masks
#_
MASK1=0xfc
MASK2=0x03
MASK3=0xff
####
# Local QoS classes
#
1QC1=0x28
1QC2=0x30
1QC3=0x38
1QC0=0x00
###
# Meta-QoS-classes as signaled by peers
#
ICP210=0x00
ICP211=0x68
ICP212=0x70
ICP213=0x78
####
# Meta-QoS-classes as should be signaled to peers if similar codes are
# used in both ways
#_
ICP120=0x00
ICP121=0x68
ICP122=0x70
ICP123=0x78
ICP130=0x00
ICP131=0x88
ICP132=0x90
ICP133=0x98
####
# Rates for policing purposes
#
RATE Total=10Mbit
RATE ICP120=2500Kbit
RATE ICP121=2500Kbit
RATE_ICP122=2500Kbit
RATE_ICP123=2500Kbit
echo $EGRESS "egress policies configuration started....."
#####
# Attach a dsmark to the eth1 interface
#_
tc qdisc add dev $EGRESS handle 1:0 root dsmark indices 8
```

Definition of four classes: MC1, MC2, MC3 and MC0 #_ tc class change dev \$EGRESS classid 1:1 dsmark mask \$MASK2 value \$ICP121 tc class change dev \$EGRESS classid 1:2 dsmark mask \$MASK2 value \$ICP122 tc class change dev \$EGRESS classid 1:3 dsmark mask \$MASK2 value \$ICP123 tc class change dev \$EGRESS classid 1:4 dsmark mask \$MASK2 value \$ICP120 #### # Definition of filters that will be invoked in order to put the ingress # traffic in the right class # tc filter add dev \$EGRESS parent 1:0 prio 1 prot ip tcindex pass_on tc filter add dev \$EGRESS parent 1:0 prio 1 prot ip handle 1 tcindex classid 1:1 tc filter add dev \$EGRESS parent 1:0 prio 1 prot ip handle 2 tcindex classid 1:2 tc filter add dev \$EGRESS parent 1:0 prio 1 prot ip handle 3 tcindex classid 1:3 tc filter add dev \$EGRESS parent 1:0 prio 1 prot ip handle 4 tcindex classid 1:4 #### # Definition of filters that will be invoked in order to put the local # traffic in the right class #_ tc filter add dev \$EGRESS parent 1:0 protocol ip prio 3 u32 match ip tos \$10C1 \$MASK1 classid 1:1 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 3 u32 match ip tos \$1QC2 \$MASK1 classid 1:2 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 3 u32 match ip tos \$1QC3 \$MASK1 classid 1:3 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 3 u32 match ip tos \$1QC0 \$MASK1 classid 1:4 ##### # Definition of filters that will be used in order to put the traffic # generated by local interfaces in the relevant egress classes # Must add other realms if used to identify local interfaces # Case of 1.1.1.0/30 and 1.1.1.4/30 realm # tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP121 \$MASK1 match ip src 1.1.1.1/30 classid 1:1 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP122 \$MASK1 match ip src 1.1.1.1/30 classid 1:2 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP123 \$MASK1 match ip src 1.1.1.1/30 classid 1:3 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP120 \$MASK1 match ip src 1.1.1.1/30 classid 1:4 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP121 \$MASK1 match ip src 1.1.1.5/30 classid 1:1 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP122 \$MASK1 match ip src 1.1.1.5/30 classid 1:2 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP123 \$MASK1 match ip src 1.1.1.5/30 classid 1:3 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP120 \$MASK1 match ip src 1.1.1.5/30 classid 1:4

tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP131 \$MASK1 match ip src 1.1.1.1/30 classid 1:1 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP132 \$MASK1 match ip src 1.1.1.1/30 classid 1:2 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP133 \$MASK1 match ip src 1.1.1.1/30 classid 1:3 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP130 \$MASK1 match ip src 1.1.1.1/30 classid 1:4 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP131 \$MASK1 match ip src 1.1.1.5/30 classid 1:1 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP132 \$MASK1 match ip src 1.1.1.5/30 classid 1:2 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP133 \$MASK1 match ip src 1.1.1.5/30 classid 1:3 tc filter add dev \$EGRESS parent 1:0 protocol ip prio 4 u32 match ip tos \$ICP130 \$MASK1 match ip src 1.1.1.5/30 classid 1:4 #### # A generic filter that will put other traffic in the BE class # tc filter add dev \$EGRESS parent 1:0 prio 5 prot ip handle 0 tcindex classid 1:4 #### # Definition of an HTB qdisc that is used to simulate a virtual link #_ tc qdisc add dev \$EGRESS parent 1:0 handle 2:0 htb default 1 tc class add dev \$EGRESS parent 2:0 classid 2:1 htb rate \$RATE_Total burst 15kb #### # Definition of an HTB qdisc that will be used to share bw between # classes # tc qdisc add dev \$EGRESS parent 2:1 handle 3:0 htb default 1 tc class add dev \$EGRESS parent 3:0 classid 3:1 htb rate \$RATE_Total burst 10kb #### # Definition of an HTB class in order to shape the MC1 traffic. # A pfifo/bfifo/sfq is attached to this class # tc class add dev \$EGRESS parent 3:1 classid 3:10 htb rate \$RATE_ICP121 burst 1500b prio 0 tc qdisc add dev \$EGRESS parent 3:10 handle a:0 pfifo limit 128 ##### # Definition of an HTB class in order to shape the MC2 traffic. # A pfifo/bfifo/sfg is attached to this class # tc class add dev \$EGRESS parent 3:1 classid 3:11 htb rate \$RATE_ICP122 burst 1500b prio 1 tc qdisc add dev \$EGRESS parent 3:11 handle b:0 pfifo limit 128 #### # Definition of an HTB class in order to shape the MC3 traffic. # A pfifo/bfifo/sfq is attached to this class # tc class add dev \$EGRESS parent 3:1 classid 3:12 htb rate \$RATE_ICP123 burst 1500b prio 2 tc qdisc add dev \$EGRESS parent 3:12 handle c:0 pfifo limit 128

```
####
# Definition of an HTB class in order to shape the MCO traffic.
# A pfifo/bfifo/sfq is attached to this class
#
tc class add dev $EGRESS parent 3:1 classid 3:13 htb rate $RATE_ICP120 ceil
$RATE_Total burst 1500b prio 3
tc qdisc add dev $EGRESS parent 3:13 handle d:0 pfifo limit 128
#####
# Definition of filter that will put the traffic in the relevant class
#_
tc filter add dev $EGRESS parent 3:0 protocol ip prio 1 handle 1 tcindex
classid 3:10
tc filter add dev $EGRESS parent 3:0 protocol ip prio 1 handle 2 tcindex
classid 3:11
tc filter add dev $EGRESS parent 3:0 protocol ip prio 1 handle 3 tcindex
classid 3:12
tc filter add dev $EGRESS parent 3:0 protocol ip prio 1 handle 4 tcindex
classid 3:13
echo $EGRESS ".....finished"
```

9.4.17 Backup

In order to limit the impact of a system failure or a possible configuration error, we put in place a twolevel backup procedure, which save most of the configuration files of the testbed.

The shared directory /home/mescal/backup has been created in PC-admin for this purpose. This folder is mounted in every router under: /mnt/backup. Routers data are saved using the backup command, which can be executed from each router.

In other hand, the **PC Admin** data is saved in **MESCAL110**.

On **PC-admin** side the "**backupall**" command will achieve a backup operation of all routers.

The **/home/mescal/backup** folder contains one sub-folder per router identified with the hostname of the router.

This folder contains again sub-folders containing data saved during a single backup operation. The name of each of these latter folders respects the following structure: **xxxx-DATE-TIME**. "**xxxx**" is set to "local" or "global" depending on the type of backup operation invoked. When backups are achieved with the **backup** command from a single router, **xxxx** takes the value "local" otherwise it is a general backup and it takes the value "global".

- The **backup** command saves:
- The DSmarking folder
- The **Zebos.conf** file
- The hosts file

The second backup level consists in saving all **PC-admin** data in **MESCAL110** router. The following data are saved:

- The **common** folder
- The scripts folder
- The hosts file
- The backup folder

9.4.18 Logs

In order to verify quickly that all routers are correctly running (interface configured and services launched), every router is configured to report its BGP (bgpd and nsm daemons) and TC status.

A crontab list has been configured in each router that triggers a reporting every 10 min.

An example of the log file is provided below (example of **MESCAL110** router):

Tue May 25 17:21:00 CEST 2004-MESCAL110: bgpd running... Tue May 25 17:21:00 CEST 2004-MESCAL110: nsm running... Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning An HTB root hasn't been configured in interface lo Tue May 25 17:21:01 CEST 2004-MESCAL110: a Warning An HTB class hasn't been configured for the interface lo Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning A DSMARK gdisc hasn't been configured for the interface lo Tue May 25 17:21:01 CEST 2004-MESCAL110: Number of configured HTB root classes is OK for interface eth0 Tue May 25 17:21:01 CEST 2004-MESCAL110: Number of configured HTB classes is OK for the interface eth0 Tue May 25 17:21:01 CEST 2004-MESCAL110: Number of configured DSMARK classes is OK for the interface eth0 Tue May 25 17:21:01 CEST 2004-MESCAL110: Number of configured HTB root classes is OK for interface eth1 Tue May 25 17:21:01 CEST 2004-MESCAL110: Number of configured HTB classes is OK for the interface eth1 Tue May 25 17:21:01 CEST 2004-MESCAL110: Number of configured DSMARK classes is OK for the interface ethl Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning An HTB root hasn't been configured in interface eth2 Tue May 25 17:21:01 CEST 2004-MESCAL110: a Warning An HTB class hasn't been configured for the interface eth2 Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning A DSMARK qdisc hasn't been configured for the interface eth2 Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning An HTB root hasn't been configured in interface eth3 Tue May 25 17:21:01 CEST 2004-MESCAL110: a Warning An HTB class hasn't been configured for the interface eth3 Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning A DSMARK qdisc hasn't been configured for the interface eth3 Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning An HTB root hasn't been configured in interface eth4 Tue May 25 17:21:01 CEST 2004-MESCAL110: a Warning An HTB class hasn't been configured for the interface eth4 Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning A DSMARK qdisc hasn't been configured for the interface eth4 Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning An HTB root hasn't been configured in interface eth5 Tue May 25 17:21:01 CEST 2004-MESCAL110: a Warning An HTB class hasn't been configured for the interface eth5 Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning A DSMARK qdisc hasn't been configured for the interface eth5 Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning An HTB root hasn't been configured in interface eth6 Tue May 25 17:21:01 CEST 2004-MESCAL110: a Warning An HTB class hasn't been configured for the interface eth6 Tue May 25 17:21:01 CEST 2004-MESCAL110: Warning A DSMARK qdisc hasn't been configured for the interface eth6
9.4.19 Check the sanity of the test bed

In order to verify that all routers in the testbed are configured as expected, two scripts that check the status of the configuration have been developed: **pfc** and **pfcheck**. These scripts must be executed from **PC-admin**.

9.4.20 Configuration scripts

The table below lists useful scripts that are used for configuring the testbed. A description of the service they provide is also given.

Location	Script	description
Routers	pcm	This script allows mounting the log, transfer,
		common and backup folders.
	pcu	This script allows un-mounting the log, transfer,
		common and backup folders.
	qsa	This script configures all ingress and egress
		policies for a given router.
	qsdel	This script deletes all ingress and egress
		policies for a given router.
	qse	This script configures egress policies for a
		given router.
	qsi	This script configures ingress policies for a
		given router.
	qsi-ethj	This script configures ingress policies of the
		interface ethj of a given router.
	qshtb-ethj	This script configures egress policies of the
		interface ethj of a given router.
	qsstat	This script displays real-time egress TC related
		statistics
	qsstati	This script displays real-time ingress TC related
		statistics
	ifstat	This script displays the interface's statistics
		as reported by the kernel. If the argument "-s"
		is given the script sets to zero all statistics.
	bgpstart	This script starts the bgpd and nsm daemons
	bgpstop	This script stops the bgpd and nsm daemons
	isalive	This scripts returns the status of the bgpd and
		nsm daemons
	svty	This script launches the VTY. If the argument "-
		l" is given, the scripts loads the configuration
		<pre>stored in /usr/local/etc/ZeboS.conf</pre>
	pfgping	This scripts displays reachability information of
		all operational interfaces
	cgping	This scripts displays reachability information
		for all customers
	confstatus	This scripts returns the status of the
		configuration of a router
	backup	This script achieved a backup operation.
	pcabackup	This script can only be executed from MESCAL110.
		It achieved a PC-admin backup operation
	clean	This script recursively deletes all files ending
		with "~" in the /home/mescal folder.
PC-admin	lup	This script activates a given interface of a
		router
	ldown	This script deactivates a given interface of a
		router
	lstatus	This script returns the status of all routers
		interfaces

rtrupdate	This script achieves an automatic update for a remote router
updateall	This script achieves an automatic update for all routers
rtrbackup	This script achieves an automatic backup for a remote router
backupall	This script achieves an automatic backup for all routers
pbackup	This script achieves a PC-admin backup
pfc	This script displays the status of the testbed configuration without details and based on differed log files.
pfcheck	This script displays the detailed status of the testbed configuration.
initqos	This script initializes the QoS configuration of all routers
delqos	This script deletes the QoS configuration of all routers
egqos	This script initializes the egress QoS configuration of all routers
ingqos	This script initializes the ingress QoS configuration of all routers
iifstat	This script initializes the interface statistics of all routers
lcap	This script configures the capabilities of Ethernet cards: lcap -s: restarts a negotiation between two back to back interfaces lcap -10: sets an interface to 10Mbit full duplex.
	duplex.

Table 34: List of useful scripts

9.5 Specific Configuration for phase 2

In the phase 2, the testbed architecture will be the same. The major difference will be the q-BGP activation.

The following scripts have been created for phase 2 purposes:

- initqbgp: this scripts initialises all q-BGP routers.
- stopqbgp: this scripts stops all q-BGP routers.
- qbgpstart: this scripts initialises a q-BGP router.
- qbgpstop: this scripts stops a q-BGP router.
- qisalive: this scripts tests if a q-BGP process is active.
- qsvty: this scripts launches an VTTY terminal for configuring q-BGP router.
- createPSLS: this scripts manages pSLSs.
- psls2qbgp: this scripts configures q-BGP router according to a given pSLS.

9.6 Specific Configuration for phase 3

In this phase, the configuration will be the same as for the phase 2. Nevertheless, we will install a PCE (Path Computation Element) in every AS. The table below shows the locations of these PCEs:

PCE identifier	Router ID	IP address	PCSId
PCE11	MESCAL110	14.0.0.2	11.11.11.11
PCE21	MESCAL210	24.0.0.2	21.21.21.21
PCE31	MESCAL310	34.0.0.2	31.31.31.31
PCE41	MESCAL410	44.0.0.2	41.41.41.41
PCE51	MESCAL510	54.0.0.2	51.51.51.51
PCE61	MESCAL610	64.0.0.2	61.61.61.61
PCE71	MESCAL710	74.0.0.2	71.71.71.71
PCE81	MESCAL810	84.0.0.2	81.81.81.81

Table 35: PCE locations

Appendix B

10 DETAILED TESTBED VALIDATION TESTS

10.1 Phase 1

10.1.1TB_P1_FUNCT/ROUT

Test Reference	: TB_P1_FUNCT/ROUT/1
Test Purpose	: Validate inter-domain link connectivity.
Procedure	: Log into MESCAL11 and verify that BGP process is disabled. Ping the following addresses:
	Scenario 1: 1 1 1 2
	 Scenario 2: 1 1 1 6
	5 Sochurio 2. 1.1.1.0
	Log into MESCAL21 and verify that BGP process is disabled. Ping the following addresses:
	• Scenario 3: 1.1.1.5
	• Scenario 4: 2.2.2.2
	• Scenario 5: 2.2.2.6
	Log into MESCAL31 and verify that BGP process is disabled. Ping the following addresses:
	• Scenario 6: 1.1.1.1
	• Scenario 7: 3.3.3.2
	• Scenario 8: 3.3.3.6
	Log into MESCAL41 and verify that BGP process is disabled. Ping the following addresses:
	• Scenario 9: 3.3.3.1
	• Scenario 10: 41.41.41.10
	• Scenario 11: 41.41.41.6
	• Scenario 12: 41.41.41.2
	Log into MESCAL42 and verify that BGP process is disabled. Ping the following addresses:
	• Scenario 13: 3.3.3.5
	• Scenario 14: 2.2.2.1
	• Scenario 15: 42.42.42.6
	• Scenario 16: 42.42.42.2
	• Scenario 17: 41.41.41.9
	Log into MESCAL43 and verify that BGP process is disabled. Ping the following addresses:
	• Scenario 18: 42.42.42.1
	• Scenario 19: 41.41.41.5
	• Scenario 20: 43.43.43.2

Log into MESCAL51 and verify that BGP process is disabled. Ping the following addresses:

- Scenario 21: 42.42.42.5
- Scenario 22: 2.2.2.5

Log into MESCAL61 and verify that BGP process is disabled. Ping the following addresses:

- Scenario 23: 6.6.6.6
- Scenario 24: 41.41.41.1
- Scenario 25: 43.43.43.1
- Scenario 26: 6.6.6.2

Log into MESCAL71 and verify that BGP process is disabled. Ping the following addresses:

• Scenario 27: 6.6.6.1

Log into MESCAL81 and verify that BGP process is disabled. Ping the following addresses:

• Scenario 28: 6.6.6.5

Expected result : Results of ping requests must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success
- Scenario 12: success
- Scenario 13: success
- Scenario 14: success
- Scenario 15: success
- Scenario 16: success
- Scenario 17: success
- Scenario 18: success
- Scenario 19: success
- Scenario 20: success
- Scenario 21: success
- Scenario 22: success
- Scenario 23: success
- Scenario 24: success
- Scenario 25: success
- Scenario 26: success
- Scenario 27: success
- Scenario 28: success

Execution date : 15/09/04

Result : Obtained results are as follows:

- Scenario 1: success
- Scenario 2: success

ess

- Scenario 4: success
- Scenario 5: Cancelled
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success
- Scenario 12: success
- Scenario 13: success
- Scenario 14: success
- Scenario 15: success
- Scenario 16: success
- Scenario 17: success
- Scenario 18: success
- Scenario 19: success
- Scenario 20: success
- Scenario 21: success
- Scenario 22: Cancelled
- Scenario 23: success
- Scenario 24: success
- Scenario 25: success
- Scenario 26: success
- Scenario 27: success
- Scenario 28: success

Failure level : None

Remarks : None

Test Reference: TB_P1_FUNCT/ROUT/2Test Purpose: Validate connectivity between two neighbours when BGPD process is activated.Procedure: Activate BGPD and NSM processes in both MESCAL11 and MESCAL21. Log to
MESCAL 21 and configure MESCAL11 as a neighbor of MESCAL21. And log to
MESCAL 11 and configure MESCAL21 as a neighbor of MESCAL11.

From MESCAL11 Ping the following addresses:

- Scenario 1: 1.1.1.6
- Scenario 2: 21.0.0.1
- Scenario 3: 2.2.2.5
- Scenario 4: 2.2.2.1
- Scenario 5: 2.2.2.2

From MESCAL21 Ping the following addresses:

- Scenario 6: 1.1.1.5
- Scenario 7: 1.1.1.1
- Scenario 8: 11.0.0.1
- Scenario 9: 11.0.0.2
- Scenario 10: 12.0.0.1
- Scenario 11: 12.0.0.2

Expected result : Results of ping requests must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success

Result : Obtained results are as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: Cancelled
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success

Failure level : None *Remarks* : None

Test Reference : TB_P1_FUNCT/ROUT/3

Test Purpose : Validate connectivity between two neighbours when BGPD process is activated.

Procedure : Activate BGPD and NSM processes in both MESCAL11 and MESCAL31. Log to MESCAL 31 and configure MESCAL11 as a neighbor. Log to MESCAL11 and configure MESCAL31 as a neighbor of MESCAL11.

From MESCAL11 Ping the following addresses:

- Scenario 1: 31.0.0.1
- Scenario 2: 3.3.3.1
- Scenario 3: 3.3.3.5
- Scenario 4: 1.1.1.2
- Scenario 5: 31.0.0.2

From MESCAL31 Ping the following addresses:

- Scenario 6: 1.1.1.5
- Scenario 7: 1.1.1.1
- Scenario 8: 11.0.0.1
- Scenario 9: 11.0.0.2
- Scenario 10: 12.0.0.1
- Scenario 11: 12.0.0.2

Expected result : Results of ping requests must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success

Result : Obtained results are as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success

Failure level: NoneRemarks: None

Test Reference: TB_P1_FUNCT/ROUT/4Test Purpose: Validate connectivity between two neighbours when BGPD process is activated.Procedure: Activate BGPD and NSM processes in both MESCAL21 and MESCAL42. Log to
MESCAL 21 and configure MESCAL42 as a neighbor. And log to MESCAL42 and
configure MESCAL21 as a neighbor.From MESCAL42 Ping the following addresses:• Scenario 1: 1.1.1.6

- Scenario 2: 21.0.0.1
- Scenario 3: 2.2.2.5
- Scenario 4: 2.2.2.1
- Scenario 5: 2.2.2.2

From MESCAL21 Ping the following addresses:

- Scenario 6: 3.3.3.6
- Scenario 7: 41.41.41.10
- Scenario 8: 42.42.42.1
- Scenario 9: 42.42.42.5
- Scenario 10: 2.2.2.2
- Scenario 11: 42.0.0.1
- Scenario 12: 42.0.0.2

Expected result : Results of ping request must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success
- Scenario 12: success

```
Result : Obtained results are as follows:
```

- Scenario 1: success
- Scenario 2: success
- Scenario 3: Cancelled
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success
- Scenario 12: success

Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/ROUT/5

Test Purpose : Validate connectivity between two neighbours when BGPD process is activated.

Procedure : Activate BGPD and NSM processes in both MESCAL21 and MESCAL51. Log to MESCAL 21 and configure MESCAL51 as a neighbor. And log to MESCAL51 and configure MESCAL21 as a neighbor.

From MESCAL51 Ping the following addresses:

- Scenario 1: 1.1.1.6
- Scenario 2: 21.0.0.1
- Scenario 3: 2.2.2.5
- Scenario 4: 2.2.2.1
- Scenario 5: 2.2.2.2

From MESCAL21 Ping the following addresses:

- Scenario 6: 2.2.2.6
- Scenario 7: 42.42.42.6
- Scenario 8: 51.0.0.1
- Scenario 9: 51.0.0.2

Expected result : Results of ping requests must be as follows:

Scenario 1: success

	 Scenario 2: success
	 Scenario 3: success
	 Scenario 4: success
	 Scenario 5: success
	Scenario 6: success
	 Scenario 7: success
	 Scenario 8: success
	 Scenario 9: success
Execution date	: 15/09/04
Result	: Obtained results are as follows:
	 Scenario 1: Cancelled
	 Scenario 2: Cancelled
	 Scenario 3: Cancelled
	 Scenario 4: Cancelled
	 Scenario 5: Cancelled
	 Scenario 6: Cancelled
	 Scenario 7: Cancelled
	 Scenario 8: Cancelled
	 Scenario 9: Cancelled
Failure level	: None
Remarks	: This test has been cancelled since the topology of the platform has been changed.
	This is not a critical test since there is some tests that check the overall testbed reachability.

Test Reference	: TB	P1	FUNCT/ROUT/6
	_		

Test Purpose : Validate connectivity between two neighbours when BGP process is activated.

Procedure : Activate BGPD and NSM processes in both MESCAL31 and MESCAL42. Log to MESCAL 31 and configure MESCAL42 as a neighbor. And log to MESCAL42 and configure MESCAL31 as a neighbor.

From MESCAL42 Ping the following addresses:

- Scenario 1: 31.0.0.1
- Scenario 2: 3.3.3.1
- Scenario 3: 3.3.3.5
- Scenario 4: 1.1.1.2
- Scenario 5: 31.0.0.2

From MESCAL31 Ping the following addresses:

- Scenario 6: 3.3.3.6
- Scenario 7: 41.41.41.10
- Scenario 8: 42.42.42.1
- Scenario 9: 42.42.42.5
- Scenario 10: 2.2.2.2
- Scenario 11: 42.0.0.1
- Scenario 12: 42.0.0.2

Expected result : Results of ping requests must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success

•	Scenario 4:	success
---	-------------	---------

- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success
- Scenario 12: success

Result	: Obtained resu	ilts are as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success
- Scenario 12: success

Failure level: NoneRemarks: None

Test Reference : **TB_P1_FUNCT/ROUT/7**

Test Purpose : Validate connectivity between two neighbours when BGP process is activated.

Procedure : Activate BGPD and NSM processes in both MESCAL31 and MESCAL41. Log to MESCAL 31 and configure MESCAL41 as a neighbor. And log to MESCAL41 and configure MESCAL31 as a neighbor.

From MESCAL41 Ping the following addresses:

- Scenario 1: 31.0.0.1
- Scenario 2: 3.3.3.1
- Scenario 3: 3.3.3.5
- Scenario 4: 1.1.1.2
- Scenario 5: 31.0.0.2

From MESCAL31 Ping the following addresses:

- Scenario 6: 3.3.3.2
- Scenario 7: 41.41.41.9
- Scenario 8: 41.41.41.5
- Scenario 9: 41.41.41.1
- Scenario 10: 41.0.0.1
- Scenario 11: 41.0.0.2

Expected result : Results of ping operation must be as follows:

- Scenario 1: success
- Scenario 2: success

 Scenario 	3:	success
------------------------------	----	---------

- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success

Result	: Obtained	results	are as	follows:	

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success

Failure level: NoneRemarks: None

Test Reference	: TB_P1_FUNCT/ROUT/8
Test Purpose	: Validate connectivity between two neighbours when BGP process is activated.
Procedure	: Activate BGPD and NSM processes in both MESCAL61 and MESCAL41. Log to MESCAL 61 and configure MESCAL41 as a neighbor. And log to MESCAL41 and configure MESCAL61 as a neighbor.
	From MESCAL41 Ping the following addresses:
	• Scenario 1: 61.0.0.1

- Scenario 2: 6.6.6.5
- Scenario 2: 0.0.0.3
 Scenario 3: 6.6.6.1
- Scenario 3: 0.0.0.1
 Scenario 4: 41.41.41.2
- Scenario 5: 43.43.43.2

From MESCAL61 Ping the following addresses:

- Scenario 6: 3.3.3.2
- Scenario 7: 41.41.41.9
- Scenario 8: 41.41.41.5
- Scenario 9: 41.41.41.1
- Scenario 10: 41.0.0.1
- Scenario 11: 41.0.0.2

Expected result : Results of ping requests must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success

- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success

Result : Results of ping requests must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success
- Scenario 11: success

Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/ROUT/9

Test Purpose : Validate connectivity between two neighbours when BGPD process is activated.

Procedure : Activate BGPD and NSM processes in both MESCAL61 and MESCAL43. Log to MESCAL 61 and configure MESCAL43 as a neighbor. And log to MESCAL43 and configure MESCAL61 as a neighbor.

From MESCAL43 Ping the following addresses:

- Scenario 1: 61.0.0.1
- Scenario 2: 6.6.6.5
- Scenario 3: 6.6.6.1
- Scenario 4: 41.41.41.2
- Scenario 5: 43.43.43.2

From MESCAL61 Ping the following addresses:

- Scenario 6: 43.43.43.1
- Scenario 7: 41.41.41.6
- Scenario 8: 42.42.42.2
- Scenario 9: 43.0.0.1
- Scenario 10: 43.0.0.2

Expected result : Results of ping requests must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success

 Scenario 7: success Scenario 8: success Scenario 9: success Scenario 10: success
: 15/09/04 : Obtained results are as follows:
 Scenario 1: success Scenario 2: success Scenario 3: success Scenario 4: success Scenario 5: success Scenario 6: success Scenario 7: success Scenario 8: success Scenario 9: success Scenario 10: success
: None
: TB_P1_FUNCT/ROUT/10
: Validate connectivity between two neighbours when BGPD process is activated.
: Activate BGPD and NSM processes in both MESCAL51 and MESCAL42. Log to MESCAL 51 and configure MESCAL42 as a neighbor. And log to MESCAL42 and configure MESCAL51 as a neighbor.
From MESCAL42 Ping the following addresses:
 Scenario 1: 2.2.2.6 Scenario 2: 42.42.4
 Scenario 2: 42.42.6 Scenario 3: 51 0 0 1
 Scenario 4: 51.0.0.2
From MESCAL51 Ping the following addresses:
 Scenario 5: 3.3.3.6 Scenario 6: 41 41 41 10
 Scenario 0: 41.41.40 Scenario 7: 42.42.42.1 Scenario 8: 42.42.42.5 Scenario 9: 2.2.2.2 Scenario 10: 42.0.0.1 Scenario 11: 42.0.0.2
 Scenario 0: 41.41.40 Scenario 7: 42.42.42.1 Scenario 8: 42.42.42.5 Scenario 9: 2.2.2.2 Scenario 10: 42.0.0.1 Scenario 11: 42.0.0.2

- Scenario 5: success
 Scenario 6: success
- Scenario 7: success
- Scenario 8: success

- Scenario 9: success
- Scenario 10: success
- Scenario 11: success

Execution date	e : 15/09/	/04
Result	: Obtain	ned results are as follows:
	•	Scenario 1: Cancelled
	•	Scenario 2: success
	•	Scenario 3: success
	•	Scenario 4: success
	•	Scenario 5: success
	•	Scenario 6: success
	•	Scenario 7: success
	•	Scenario 8: success
	•	Scenario 9: success
	•	Scenario 10: success
	•	Scenario 11: success
Failure level	: None	
Remarks	: None	

Test Reference : **TB_P1_FUNCT/ROUT/11**

Test Purpose : Validate connectivity between two neighbours when BGPD process is activated.

Procedure : Activate BGPD and NSM processes in both MESCAL61 and MESCAL71. Log to MESCAL 61 and configure MESCAL71 as a neighbor. And log to MESCAL71 and configure MESCAL61 as a neighbor.

From MESCAL71 Ping the following addresses:

- Scenario 1: 61.0.0.1
- Scenario 2: 6.6.6.5
- Scenario 3: 6.6.6.1
- Scenario 4: 41.41.41.2
- Scenario 5: 43.43.43.2

From MESCAL61 Ping the following addresses:

- Scenario 6: 6.6.6.2
- Scenario 7: 71.0.0.1
- Scenario 8: 71.0.0.2

Expected result : Results of ping operation must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success

Execution date : 15/09/04

Result : Obtained results are as follows:

Scenario 1: success

- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success

Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/ROUT/12

Test Purpose : Validate connectivity between two neighbours when BGPD process is activated.

Procedure : Activate BGPD and NSM processes in both MESCAL61 and MESCAL81. Log to MESCAL 61 and configure MESCAL81 as a neighbor. And log to MESCAL81 and configure MESCAL61 as a neighbor.

From MESCAL81 Ping the following addresses:

- Scenario 1: 61.0.0.1
- Scenario 2: 6.6.6.5
- Scenario 3: 6.6.6.1
- Scenario 4: 41.41.41.2
- Scenario 5: 43.43.43.2

From MESCAL61 Ping the following addresses:

- Scenario 6: 6.6.6.6
- Scenario 7: 81.0.0.1
- Scenario 8: 81.0.0.2

Expected result : Results of ping requests must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success

Execution date : 15/09/04

Result : Obtained results are as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success

Failure level : None

Remarks : None

Test Reference : TB_P1_FUNCT/ROUT/13

Test Purpose : Check the route propagation in a simple Scenario.

Procedure : Configure MESCAL11 to be a neighbor of MESCAL21 and MESCAL31. Configure MESCAL31 and MESCAL21 to be neighbours of MESCAL11. Activate BGPD and NSM processes in MESCAL11, MESCAL21 and MESCAL31.

From MESCAL31 Ping the following addresses:

- Scenario 1: 1.1.1.6
- Scenario 2: 21.0.0.1
- Scenario 3: 2.2.2.5
- Scenario 4: 2.2.2.1
- Scenario 5: 2.2.2.2

From MESCAL21 Ping the following addresses:

- Scenario 6: 1.1.1.2
- Scenario 7: 3.3.3.5
- Scenario 8: 3.3.3.1
- Scenario 9: 31.0.0.1
- Scenario 10: 31.0.0.2

Expected result : Results of ping requests must be as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: success
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success

Execution date : 15/09/04

Result : Obtained results are as follows:

- Scenario 1: success
- Scenario 2: success
- Scenario 3: Cancelled
- Scenario 4: success
- Scenario 5: success
- Scenario 6: success
- Scenario 7: success
- Scenario 8: success
- Scenario 9: success
- Scenario 10: success

Failure level: NoneRemarks: None

Test Reference : **TB_P1_FUNCT/ROUT/14**

Test Purpose: Check the reachability of all interfaces.Procedure: Activate BGPD and NSM in all routers present in the testbed. BGP configuration is
as described in Appendix A.

Log into the following routers and execute from bash prompt "pfgping":

- Scenario 1: MESCAL11
- Scenario 2: MESCAL21
- Scenario 3: MESCAL31
- Scenario 4: MESCAL41
- Scenario 5: MESCAL42
- Scenario 6: MESCAL43
- Scenario 7: MESCAL51
- Scenario 8: MESCAL61
- Scenario 9: MESCAL71
- Scenario 10: MESCAL81

Expected result : Results of "pfgping" must be as follows:

- Scenario 1: All destinations are successfully reached.
- Scenario 2: All destinations are successfully reached.
- Scenario 3: All destinations are successfully reached.
- Scenario 4: All destinations are successfully reached.
- Scenario 5: All destinations are successfully reached.
- Scenario 6: All destinations are successfully reached.
- Scenario 7: All destinations are successfully reached.
- Scenario 8: All destinations are successfully reached.
- Scenario 9: All destinations are successfully reached.
- Scenario 10: All destinations are successfully reached.

Execution date : 15/09/04

Result : Obtained results are as follows:

- Scenario 1: All destinations are successfully reached.
- Scenario 2: All destinations are successfully reached.
- Scenario 3: All destinations are successfully reached.
- Scenario 4: All destinations are successfully reached.
- Scenario 5: All destinations are successfully reached.
- Scenario 6: All destinations are successfully reached.
- Scenario 7: All destinations are successfully reached.
- Scenario 8: All destinations are successfully reached.
- Scenario 9: All destinations are successfully reached.
- Scenario 10: All destinations are successfully reached.

Failure level: NoneRemarks: None

Test Reference : **TB_P1_FUNCT/ROUT/15**

Test Purpose : Verify the reachability when link failure occurs.

Procedure : Activate BGPD and NSM in all routers present in the testbed. BGP configuration is as described in Appendix A.

Execute the following operations:

- Scenario 1: Disable interface eth6 of MESCAL21, and from bash prompt of MESCAL11 execute "pfgping"
- Scenario 2: Disable interface eth1 of MESCAL31, and from bash prompt of MESCAL11 execute "pfgping"

- Scenario 3: Disable interface eth1 of MESCAL42, and from bash prompt of MESCAL11 execute "pfgping"
- Scenario 4: Disable interface eth4 of MESCAL42, and from bash prompt of MESCAL11 execute "pfgping"
- Scenario 5: Disable interfaces eth4 and eth1 of MESCAL42, and from bash prompt of MESCAL11 execute "pfgping"
- Scenario 6: Disable interfaces eth2 and eth3 of MESCAL42, and from bash prompt of MESCAL11 execute "pfgping"
- Scenario 7: Disable interfaces eth5 of MESCAL41, and from bash prompt of MESCAL11 execute "**pfgping**"
- Scenario 8: Disable interfaces eth5 of MESCAL41 and eth0 of MESCAL43, and from bash prompt of MESCAL11 execute "pfgping"

Expected result : Results of "**pfgping**" must be as follows:

- Scenario 1: All destinations are successfully reached except interface eth6 of MESCAL21.
- Scenario 2: All destinations are successfully reached except interface eth1 of MESCAL31.
- Scenario 3: All destinations are successfully reached except interface eth1 of MESCAL42.
- Scenario 4: All destinations are successfully reached except interface eth4 of MESCAL42.
- Scenario 5: All destinations are successfully reached except interface eth1 and eth4 of MESCAL42.
- Scenario 6: All destinations are successfully reached except interface eth2 and eth3 of MESCAL42.
- Scenario 7: All destinations are successfully reached except interface eth5 of MESCAL41.
- Scenario 8: All destinations are successfully reached except:
 - Interface eth5 of MESCAL41 Interface eth0 of MESCAL43 All interfaces of MESCAL61 All interfaces of MESCAL71 All interfaces of MESCAL81

Execution date : 15/09/04

Result

: Obtained results are as follows:

- Scenario 1: All destinations are successfully reached except interface eth6 of MESCAL21.
- Scenario 2: All destinations are successfully reached except interface eth1 of MESCAL31.
- Scenario 3: All destinations are successfully reached except interface eth1 of MESCAL42.
- Scenario 4: All destinations are successfully reached except interface eth4 of MESCAL42.
- Scenario 5: All destinations are successfully reached except interface eth1 and eth4 of MESCAL42.
- Scenario 6: All destinations are successfully reached except interface eth2 and eth3 of MESCAL42.
- Scenario 7: All destinations are successfully reached except interface eth5 of MESCAL41.
- Scenario 8: All destinations are successfully reached except: Interface eth5 of MESCAL41

		Interface eth0 of MESCAL43
		All interfaces of MESCAL61
		All interfaces of MESCAL71
		All interfaces of MESCAL81
Failure level	: None	
Remarks	: None	

Test Reference: TB_P1_FUNCT/ROUT/16Test Purpose: Verify the reachability when a link failure is re-established.Procedure: Activate BGPD and NSM in all routers present in the testbed. BGP configuration is as described in Appendix A.
 Execute the following operations: Scenario 1: Disable interface eth6 of MESCAL21. In few minutes activate this interface, and from bash prompt of MESCAL11 execute "pfgping" Scenario 2: Disable interface eth1 of MESCAL31. In few minutes activate this interface, and from bash prompt of MESCAL11 execute "pfgping" Scenario 3: Disable interface eth1 of MESCAL42. In few minutes activate this interface, and from bash prompt of MESCAL42. In few minutes activate this interface, and from bash prompt of MESCAL42. In few minutes activate this interface, and from bash prompt of MESCAL42. In few minutes activate this interface, and from bash prompt of MESCAL42. In few minutes activate this interface, and from bash prompt of MESCAL42. In few minutes activate these interfaces, and from bash prompt of MESCAL41 execute "pfgping" Scenario 5: Disable interfaces eth2 and eth3 of MESCAL41. In few minutes activate these interfaces, and from bash prompt of MESCAL41. In few minutes activate these interfaces, and from bash prompt of MESCAL41. In few minutes activate these interfaces eth5 of MESCAL41. In few minutes activate these interfaces eth5 of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interface, and from bash prompt of MESCAL41. In few minutes activate this interfa
<i>Expected result</i> : Results of " pfgping " must be as follows:
 Scenario 1: All destinations are successfully reached. Scenario 2: All destinations are successfully reached. Scenario 3: All destinations are successfully reached. Scenario 4: All destinations are successfully reached. Scenario 5: All destinations are successfully reached. Scenario 6: All destinations are successfully reached. Scenario 7: All destinations are successfully reached. Scenario 8: All destinations are successfully reached. Scenario 9: All destinations are successfully reached. Scenario 10: All destinations are successfully reached.
Execution date: 15/09/04Result: Obtained results are as follows:

- Scenario 1: All destinations are successfully reached.
- Scenario 2: All destinations are successfully reached.
- Scenario 3: All destinations are successfully reached.
- Scenario 4: All destinations are successfully reached.

- Scenario 5: All destinations are successfully reached.
- Scenario 6: All destinations are successfully reached.
- Scenario 7: All destinations are successfully reached.
- Scenario 8: All destinations are successfully reached.
- Scenario 9: All destinations are successfully reached.
- Scenario 10: All destinations are successfully reached.

Failure level: NoneRemarks: None

Test Reference Test Purpose	: TB_P1_FUNCT/ROUT/17 : Verify intra-domain routing in AS4.
Procedure	: Configure the following static routes in MESCAL42: • 41.41.41.1/30 via 41.41.41.9 metric 1 • 41.41.41.1/30 via 42.42.42.2 metric 2 • 43.43.43.1/30 via 41.41.41.9 metric 2 • 43.43.43.1/30 via 42.42.42.9 metric 1
	Configure the following static routes in MESCAL41: • 43.43.43.1/30 via 41.41.41.10 metric 2 • 43.43.43.1/30 via 41.41.41.6 metric 1
	 Configure the following static routes in MESCAL43: 41.41.41.1/30 via 41.41.41.5 metric 1 41.41.41.1/30 via 42.42.42.1 metric 2
	 Log into MESCAL42. From a bash prompt, ping the following addresses: Scenario 1: 41.41.41.1 Scenario 2: 43.43.43.1
	 Disable interface eth2 of MESCAL43, and then ping the following addresses: Scenario 3: 41.41.41.1 Scenario 4: 43.43.43.1
	 Disable interface eth4 of MESCAL41, and then ping the following addresses: Scenario 5: 41.41.41.1 Scenario 6: 43.43.43.1
	 Enable interface eth2 of MESCAL43, and then ping the following addresses: Scenario 7: 41.41.41.1 Scenario 8: 43.43.43.1
	 Disable interface eth3 of MESCAL41, and then ping the following addresses: Scenario 9: 41.41.41.1 Scenario 10: 43.43.43.1
Expected result	: Results of ping requests must be as follows:
	Scenario 1: Success.Scenario 2: Success.

- Scenario 3: Success.
- Scenario 4: Success.
- Scenario 5: Unreachable destination.

- Scenario 6: Unreachable destination.
- Scenario 7: Success.
- Scenario 8: Success.
- Scenario 9: Success.
- Scenario 10: Unreachable destination.

Result : Obtained results are as follows:

- Scenario 1: Success.
- Scenario 2: Success.
- Scenario 3: Success.
- Scenario 4: Success.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: Success.
- Scenario 8: Success.
- Scenario 9: Success.
- Scenario 10: Unreachable destination.

Failure level: NoneRemarks: None

10.1.2TB P1 FUNCT/DSSW

In this section, we list tests that aim at verifying DSCP swapping operations in both ingress and egress of ASs. Before starting executing these tests, it is recommended to verify that the DSCP values that have been assigned in the testbed configuration are correctly settled in **qsi** and **qse** scripts in all routers of the testbed.

Test Reference : TB_P1_FUNCT/DSSW/1

Test Purpose : Verify DSCP swapping at egress of MESCAL11.

Procedure : Configure local-QoS-classes of MESCAL11 as defined in testbed Configuration section. Practically, execute **qsa** script in MESCAL11.

Launch a traffic analyzer in:

- MESCAL31 that captures traffic going through eth1.
- MESCAL21 that captures traffic going through eth6.

Execute successively the following commands from a bash terminal of MESCAL11:

- Scenario 1: ping –Q 0x28 1.1.1.2
- Scenario 2: ping -Q 0x30 1.1.1.2
- Scenario 3: ping –Q 0x38 1.1.1.2
- Scenario 4: ping 1.1.1.2
- Scenario 5: ping –Q 0x45 1.1.1.2
- Scenario 6: ping -Q 0x144 1.1.1.2
- Scenario 7: ping –O 0x28 1.1.1.6
- Scenario 8: ping -Q 0x30 1.1.1.6
- Scenario 9: ping –Q 0x38 1.1.1.6
- Scenario 10: ping 1.1.1.6
- Scenario 11: ping –Q 0x98 1.1.1.6
- Scenario 12: ping –Q 0x144 1.1.1.6

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0x88.
- Scenario 2: ping requests are received with a TOS value of 0x90.
- Scenario 3: ping requests are received with a TOS value of 0x98.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0x68.
- Scenario 8: ping requests are received with a TOS value of 0x70.
- Scenario 9: ping requests are received with a TOS value of 0x78.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.

- Scenario 1: ping requests are received with a TOS value of 0x88.
- Scenario 2: ping requests are received with a TOS value of 0x90.
- Scenario 3: ping requests are received with a TOS value of 0x98.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0x68.
- Scenario 8: ping requests are received with a TOS value of 0x70.
- Scenario 9: ping requests are received with a TOS value of 0x78.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.

Failure level: NoneRemarks: None

Test Reference : TB P1 FUNCT/DSSW/2

Test Purpose : Verify DSCP swapping at egress of MESCAL21.

Procedure : Configure local-QoS-classes of MESCAL21 as defined in testbed Configuration section. Practically, execute **qsa** script in MESCAL21.

Launch a traffic analyzer in:

- MESCAL11 that captures traffic going through eth1.
- MESCAL42 that captures traffic going through eth4.
- MESCAL51 that captures traffic going through eth2.

Execute successively the following commands from a bash terminal of MESCAL21:

- Scenario 1: ping –Q 0x48 1.1.1.5
- Scenario 2: ping –Q 0x50 1.1.1.5
- Scenario 3: ping –Q 0x58 1.1.1.5
- Scenario 4: ping 1.1.1.5
- Scenario 5: ping –Q 0x98 1.1.1.5
- Scenario 6: ping –Q 0x144 1.1.1.5
- Scenario 7: ping -Q 0x48 2.2.2.2
- Scenario 8: ping –Q 0x50 2.2.2.2
- Scenario 9: ping –Q 0x58 2.2.2.2
- Scenario 10: ping 2.2.2.2

- Scenario 11: ping –Q 0x98 2.2.2.2
- Scenario 12: ping –Q 0x144 2.2.2.2
- Scenario 13: ping –Q 0x48 2.2.2.6
- Scenario 14: ping –Q 0x50 2.2.2.6
 Scenario 15: ping –Q 0x58 2.2.2.6
- Scenario 15: ping –Q 0x38.
 Scenario 16: ping 2.2.2.6
- Scenario 17: ping -Q 0x98 2.2.2.6
- Scenario 18: ping –Q 0x144 2.2.2.6

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0x68.
- Scenario 2: ping requests are received with a TOS value of 0x70.
- Scenario 3: ping requests are received with a TOS value of 0x78.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0xc8.
- Scenario 8: ping requests are received with a TOS value of 0xd0.
- Scenario 9: ping requests are received with a TOS value of 0xd8.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.
- Scenario 13: ping requests are received with a TOS value of 0xe8.
- Scenario 14: ping requests are received with a TOS value of 0xf0.
- Scenario 15: ping requests are received with a TOS value of 0xf8.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: ping requests are received with a TOS value of 0x00.
- Scenario 18: ping requests are received with a TOS value of 0x00.

Execution date : 15/09/04

- *Result* : Obtained results are as follows:
 - Scenario 1: ping requests are received with a TOS value of 0x68.
 - Scenario 2: ping requests are received with a TOS value of 0x70.
 - Scenario 3: ping requests are received with a TOS value of 0x78.
 - Scenario 4: ping requests are received with a TOS value of 0x00.
 - Scenario 5: ping requests are received with a TOS value of 0x00.
 - Scenario 6: ping requests are received with a TOS value of 0x00.
 - Scenario 7: ping requests are received with a TOS value of 0xc8.
 - Scenario 8: ping requests are received with a TOS value of 0xd0.
 - Scenario 9: ping requests are received with a TOS value of 0xd8.
 - Scenario 10: ping requests are received with a TOS value of 0x00.
 - Scenario 11: ping requests are received with a TOS value of 0x00.
 - Scenario 12: ping requests are received with a TOS value of 0x00.
 - Scenario 13: Cancelled
 - Scenario 14: Cancelled
 - Scenario 15: Cancelled
 - Scenario 16: Cancelled
 - Scenario 17: Cancelled
 - Scenario 18: Cancelled

Failure level : None

Remarks : The scenarios cancelled are obsolete since the inter-domain link between AS2 and AS5 is dropped.

Test Reference Test Purpose Procedure	 : TB_P1_FUNCT/DSSW/3 : Verify DSCP swapping at egress of MESCAL31. : Configure local-QoS-classes of MESCAL31 as defined in testbed Configuration section. Practically, execute qsa script in MESCAL31.
	 Launch a traffic analyzer in: MESCAL11 that captures traffic going through eth0. MESCAL42 that captures traffic going through eth1. MESCAL41 that captures traffic going through eth0.
	Execute successively the following commands from a bash terminal of MESCAL31: • Scenario 1: ping $-Q$ 0x68 1.1.1.1 • Scenario 2: ping $-Q$ 0x70 1.1.1.1 • Scenario 3: ping $-Q$ 0x78 1.1.1.1 • Scenario 4: ping 1.1.1.1 • Scenario 5: ping $-Q$ 0x98 1.1.1.1 • Scenario 6: ping $-Q$ 0x144 1.1.1.1 • Scenario 7: ping $-Q$ 0x68 3.3.3.6 • Scenario 8: ping $-Q$ 0x70 3.3.3.6 • Scenario 9: ping $-Q$ 0x78 3.3.3.6 • Scenario 10: ping 3.3.3.6 • Scenario 11: ping $-Q$ 0x68 3.3.3.6 • Scenario 12: ping $-Q$ 0x68 3.3.3.2 • Scenario 13: ping $-Q$ 0x78 3.3.3.2 • Scenario 14: ping $-Q$ 0x78 3.3.3.2 • Scenario 15: ping $-Q$ 0x78 3.3.3.2 • Scenario 16: ping 3.3.3.2 • Scenario 17: ping $-Q$ 0x98 3.3.3.2 • Scenario 18: ping $-Q$ 0x98 3.3.3.2
Expected result	: Within the traffic analyzer, following results must be obtained:
	 Scenario 1: ping requests are received with a TOS value of 0x88. Scenario 2: ping requests are received with a TOS value of 0x90. Scenario 3: ping requests are received with a TOS value of 0x98. Scenario 4: ping requests are received with a TOS value of 0x00. Scenario 5: ping requests are received with a TOS value of 0x00. Scenario 6: ping requests are received with a TOS value of 0x00. Scenario 7: ping requests are received with a TOS value of 0x68. Scenario 8: ping requests are received with a TOS value of 0x60. Scenario 9: ping requests are received with a TOS value of 0x68. Scenario 10: ping requests are received with a TOS value of 0x00. Scenario 11: ping requests are received with a TOS value of 0x00. Scenario 12: ping requests are received with a TOS value of 0x00. Scenario 13: ping requests are received with a TOS value of 0x00.

- Scenario 15: ping requests are received with a TOS value of 0xf8.
- Scenario 16: ping requests are received with a TOS value of 0x00.
 Scenario 17: ping requests are received with a TOS value of 0x00.

Copyright © MESCAL Consortium, July 2005

• Scenario 18: ping requests are received with a TOS value of 0x00.

Execution date : 15/09/04

Result : Obtained results are as follows:

Failure level Remarks Test Reference Test Purpose	 Scenario 8: ping requests are received with a TOS value of 0xf0. Scenario 9: ping requests are received with a TOS value of 0x00. Scenario 10: ping requests are received with a TOS value of 0x00. Scenario 11: ping requests are received with a TOS value of 0x00. Scenario 12: ping requests are received with a TOS value of 0x00. Scenario 13: ping requests are received with a TOS value of 0x68. Scenario 14: ping requests are received with a TOS value of 0xf0. Scenario 15: ping requests are received with a TOS value of 0xf8. Scenario 15: ping requests are received with a TOS value of 0xf0. Scenario 16: ping requests are received with a TOS value of 0x00. Scenario 17: ping requests are received with a TOS value of 0x00. Scenario 18: ping requests are received with a TOS value of 0x00. Scenario 18: ping requests are received with a TOS value of 0x00. Scenario 18: ping requests are received with a TOS value of 0x00. Scenario 18: ping requests are received with a TOS value of 0x00. Scenario 18: ping requests are received with a TOS value of 0x00. Scenario 18: ping requests are received with a TOS value of 0x00.
Procedure	: Configure local-QoS-classes of MESCAL41 as defined in testbed Configuration section. Practically, execute qsa script in MESCAL41. Launch a traffic analyzer in:

- MESCAL42 that captures traffic going through eth2.
- MESCAL43 that captures traffic going through eth1.
- MESCAL61 that captures traffic going through eth4.

Execute successively the following commands from a bash terminal of MESCAL41:

- Scenario 1: ping –Q 0x88 3.3.3.1
- Scenario 2: ping –Q 0x90 3.3.3.1
- Scenario 3: ping –Q 0x98 3.3.3.1
- Scenario 4: ping 3.3.3.1
- Scenario 5: ping –Q 0x55 3.3.3.1
- Scenario 6: ping –Q 0x144 3.3.3.1
- Scenario 7: ping –Q 0x88 41.41.41.10
- Scenario 8: ping –Q 0x90 41.41.41.10
- Scenario 9: ping –Q 0x98 41.41.41.10
- Scenario 10: ping 41.41.41.10
- Scenario 11: ping –Q 0x55 41.41.41.10
- Scenario 12: ping –Q 0x144 41.41.41.10
- Scenario 13: ping –Q 0x88 41.41.41.6
- Scenario 14: ping –Q 0x90 41.41.41.6

- Scenario 15: ping –Q 0x98 41.41.41.6
- Scenario 16: ping 41.41.41.6
- Scenario 17: ping –Q 0x38 41.41.41.6
- Scenario 18: ping –Q 0x144 41.41.41.6
- Scenario 19: ping –Q 0x88 41.41.41.2
- Scenario 20: ping –Q 0x90 41.41.41.2
- Scenario 21: ping –Q 0x98 41.41.41.2
- Scenario 22: ping 41.41.41.2
- Scenario 23: ping –Q 0x38 41.41.41.2
- Scenario 24: ping –Q 0x144 41.41.41.2

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0x88.
- Scenario 8: ping requests are received with a TOS value of 0x90.
- Scenario 9: ping requests are received with a TOS value of 0x98.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.
- Scenario 13: ping requests are received with a TOS value of 0x88.
- Scenario 14: ping requests are received with a TOS value of 0x90.
- Scenario 15: ping requests are received with a TOS value of 0x98.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: ping requests are received with a TOS value of 0x00.
- Scenario 18: ping requests are received with a TOS value of 0x00.
- Scenario 19: ping requests are received with a TOS value of 0x48.
- Scenario 20: ping requests are received with a TOS value of 0x10.
- Scenario 21: ping requests are received with a TOS value of 0x58.
- Scenario 22: ping requests are received with a TOS value of 0x00.
- Scenario 22: ping requests are received with a TOS value of 0x00.
- Scenario 24: ping requests are received with a TOS value of 0x00.

Execution date : 15/09/04

Result : Obtained results are as follows:

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0x88.
- Scenario 8: ping requests are received with a TOS value of 0x90.
- Scenario 9: ping requests are received with a TOS value of 0x98.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.

Failure level Remarks	 Scenario 13: ping requests are received with a TOS value of 0x88. Scenario 14: ping requests are received with a TOS value of 0x90. Scenario 15: ping requests are received with a TOS value of 0x98. Scenario 16: ping requests are received with a TOS value of 0x00. Scenario 17: ping requests are received with a TOS value of 0x00. Scenario 18: ping requests are received with a TOS value of 0x00. Scenario 19: ping requests are received with a TOS value of 0x48. Scenario 20: ping requests are received with a TOS value of 0x50. Scenario 21: ping requests are received with a TOS value of 0x00. Scenario 22: ping requests are received with a TOS value of 0x00. Scenario 23: ping requests are received with a TOS value of 0x00. Scenario 24: ping requests are received with a TOS value of 0x00.
Test Reference	: TB_P1_FUNCT/DSSW/5
Test Purpose	: Verify DSCP swapping at egress of MESCAL42.
Procedure	: Configure local-QoS-classes of MESCAL42 as defined in testbed Configuration
	section. Practically, execute qsa script in MESCAL42.
	Launch a traffic analyzer in:
	• MESCAL 21 that captures traffic going through eth0
	 MESCAL31 that captures traffic going through eth0.
	• MESCAL41 that captures traffic going through eth4.
	• MESCAL43 that captures traffic going through eth2.
	• MESCAL51 that captures traffic going through eth1.
	Execute successively the following commands from a bash terminal of MESCAL42:
	• Scenario 1: ping –Q 0x88 2.2.2.1
	• Scenario 2: ping –Q 0x90 2.2.2.1
	• Scenario 3: ping –Q 0x98 2.2.2.1
	• Scenario 4: ping 2.2.2.1
	• Scenario 5: ping $-Q$ 0x55 2.2.2.1
	• Scenario 6: ping $-Q$ 0x144 2.2.2.1
	• Scenario /: ping $-Q$ 0x88 3.3.3.5 Sequence Q ping $-Q$ 0x88 3.3.3.5
	• Scenario 8: ping $-Q$ 0x90 3.3.3.5 • Scenario 9: ping $-Q$ 0x98 3.3.3.5
	 Scenario 10: ping -Q 0x98 5.5.5.5 Scenario 10: ping 3 3 3 5
	 Scenario 10: ping 5.5.5.5 Scenario 11: ping -O 0x55 3 3 3 5
	• Scenario 12: ping $-0.0x1443.3.3.5$
	• Scenario 13: ping –Q 0x88 41.41.41.9
	• Scenario 14: ping –Q 0x90 41.41.41.9
	• Scenario 15: ping –Q 0x98 41.41.41.9
	• Scenario 16: ping 41.41.41.9
	• Scenario 17: ping –Q 0x38 41.41.41.9
	• Scenario 18: ping –Q 0x144 41.41.41.9
	• Scenario 19: ping $-Q$ 0x88 42.42.42.1
	• Scenario 20: ping $-Q$ 0x90 42.42.42.1 • Scenario 21: ping $-Q$ 0x98 42.42.42.1
	• Scenario 21. ping $-Q$ 0x98 42.42.42.1 • Scenario 22: ping 42.42.42.1
	• Scenario 22. ping $+2.+2.42.1$ • Scenario 23: ping $-0.0x38 A2 A2 A2 1$
	• Scenario 24: ping $-0.0x14442.42.42$

- Scenario 25: ping –Q 0x88 42.42.42.6
- Scenario 26: ping –Q 0x90 42.42.42.6
- Scenario 27: ping –Q 0x98 42.42.42.6
- Scenario 28: ping 42.42.42.6
- Scenario 29: ping –Q 0x38 42.42.42.6
- Scenario 30: ping –Q 0x144 42.42.42.6

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xc8.
- Scenario 2: ping requests are received with a TOS value of 0xd0.
- Scenario 3: ping requests are received with a TOS value of 0xd8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0xe8.
- Scenario 8: ping requests are received with a TOS value of 0xf0.
- Scenario 9: ping requests are received with a TOS value of 0xf8.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.
- Scenario 13: ping requests are received with a TOS value of 0x88.
- Scenario 14: ping requests are received with a TOS value of 0x90.
- Scenario 15: ping requests are received with a TOS value of 0x98.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: ping requests are received with a TOS value of 0x00.
- Scenario 18: ping requests are received with a TOS value of 0x00.
- Scenario 19: ping requests are received with a TOS value of 0x88.
- Scenario 20: ping requests are received with a TOS value of 0x80.
- Scenario 21: ping requests are received with a TOS value of 0x98.
- Scenario 22: ping requests are received with a TOS value of 0x00.
- Scenario 23: ping requests are received with a TOS value of 0x00.
- Scenario 24: ping requests are received with a TOS value of 0x00.
- Scenario 25: ping requests are received with a TOS value of 0x28.
- Scenario 26: ping requests are received with a TOS value of 0x30.
- Scenario 27: ping requests are received with a TOS value of 0x38.
- Scenario 28: ping requests are received with a TOS value of 0x00.
- Scenario 29: ping requests are received with a TOS value of 0x00.
- Scenario 30: ping requests are received with a TOS value of 0x00.

Execution date : 15/09/04

Result : Obtained results are as follows:

- Scenario 1: ping requests are received with a TOS value of 0xc8.
- Scenario 2: ping requests are received with a TOS value of 0xd0.
- Scenario 3: ping requests are received with a TOS value of 0xd8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0xe8.
- Scenario 8: ping requests are received with a TOS value of 0xf0.
- Scenario 9: ping requests are received with a TOS value of 0xf8.
- Scenario 10: ping requests are received with a TOS value of 0x00.

- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.
- Scenario 13: ping requests are received with a TOS value of 0x88.
- Scenario 14: ping requests are received with a TOS value of 0x90.
- Scenario 15: ping requests are received with a TOS value of 0x98.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: ping requests are received with a TOS value of 0x00.
- Scenario 18: ping requests are received with a TOS value of 0x00.
- Scenario 19: ping requests are received with a TOS value of 0x88.
- Scenario 20: ping requests are received with a TOS value of 0x80.
- Scenario 21: ping requests are received with a TOS value of 0x98.
- Scenario 22: ping requests are received with a TOS value of 0x00.
- Scenario 23: ping requests are received with a TOS value of 0x00.
- Scenario 24: ping requests are received with a TOS value of 0x00.
- Scenario 25: ping requests are received with a TOS value of 0x28.
- Scenario 26: ping requests are received with a TOS value of 0x30.
- Scenario 27: ping requests are received with a TOS value of 0x38.
- Scenario 28: ping requests are received with a TOS value of 0x00.
- Scenario 29: ping requests are received with a TOS value of 0x00.
- Scenario 30: ping requests are received with a TOS value of 0x00.

Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/DSSW/6

Test Purpose : Verify DSCP swapping at egress of MESCAL43.

Procedure : Configure local-QoS-classes of MESCAL43 as defined in testbed Configuration section. Practically, execute **qsa** script in MESCAL43.

Launch a traffic analyzer in:

- MESCAL41 that captures traffic going through eth3.
- MESCAL42 that captures traffic going through eth3.
- MESCAL61 that captures traffic going through eth1.

Execute successively the following commands from a bash terminal of MESCAL43:

- Scenario 1: ping –Q 0x88 41.41.41.5
- Scenario 2: ping –Q 0x90 41.41.41.5
- Scenario 3: ping –Q 0x98 41.41.41.5
- Scenario 4: ping 41.41.41.5
- Scenario 5: ping –Q 0x55 41.41.41.5
- Scenario 6: ping –Q 0x144 41.41.41.5
- Scenario 7: ping -Q 0x88 42.42.42.1
- Scenario 8: ping –Q 0x90 42.42.42.1
- Scenario 9: ping –Q 0x98 42.42.42.1
- Scenario 10: ping 42.42.42.1
- Scenario 11: ping –Q 0x55 42.42.42.1
- Scenario 12: ping –Q 0x144 42.42.42.1
- Scenario 13: ping –Q 0x88 43.43.43.2
- Scenario 14: ping –Q 0x90 43.43.43.2
- Scenario 15: ping –Q 0x98 43.43.43.2
- Scenario 16: ping 43.43.43.2
- Scenario 17: ping –Q 0x38 43.43.43.2

• Scenario 18: ping –Q 0x144 43.43.43.2

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0x88.
- Scenario 2: ping requests are received with a TOS value of 0x90.
- Scenario 3: ping requests are received with a TOS value of 0x98.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0x88.
- Scenario 8: ping requests are received with a TOS value of 0x90.
- Scenario 9: ping requests are received with a TOS value of 0x98.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.
- Scenario 13: ping requests are received with a TOS value of 0x48.
- Scenario 14: ping requests are received with a TOS value of 0x50.
- Scenario 15: ping requests are received with a TOS value of 0x58.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: ping requests are received with a TOS value of 0x00.
- Scenario 18: ping requests are received with a TOS value of 0x00.

Execution date : 15/09/04

Result : Obtained results are as follows:

•	Scenario	1: ping requests are	received with a	TOS value of 0x88.
---	----------	----------------------	-----------------	--------------------

- Scenario 2: ping requests are received with a TOS value of 0x90.
- Scenario 3: ping requests are received with a TOS value of 0x98.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0x88.
- Scenario 8: ping requests are received with a TOS value of 0x90.
- Scenario 9: ping requests are received with a TOS value of 0x98.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.
- Scenario 13: ping requests are received with a TOS value of 0x48.
- Scenario 14: ping requests are received with a TOS value of 0x50.
- Scenario 15: ping requests are received with a TOS value of 0x58.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: ping requests are received with a TOS value of 0x00.
- Scenario 18: ping requests are received with a TOS value of 0x00.
- Failure level: NoneRemarks: None

Test Reference	: TB_P1_FUNCT/DSSW/7
Test Purpose	: Verify DSCP swapping at egress of MESCAL51.
Procedure	: Configure local-QoS-classes of MESCAL51 as defined in testbed Configuration section. Practically, execute gsa script in MESCAL51.

Launch a traffic analyzer in:

- MESCAL21 that captures traffic going through eth5.
- MESCAL42 that captures traffic going through eth5.

Execute successively the following commands from a bash terminal of MESCAL51:

- Scenario 1: ping –Q 0xa8 2.2.2.5
- Scenario 2: ping –Q 0xb0 2.2.2.5
- Scenario 3: ping –Q 0xb8 2.2.2.5
- Scenario 4: ping 2.2.2.5
- Scenario 5: ping –Q 0x55 2.2.2.5
- Scenario 6: ping –Q 0x144 2.2.2.5
- Scenario 7: ping –Q 0xa8 42.42.42.5
- Scenario 8: ping –Q 0xb0 42.42.42.5
- Scenario 9: ping –Q 0xb8 42.42.42.5
- Scenario 10: ping 42.42.42.5
- Scenario 11: ping –Q 0x55 42.42.42.5
- Scenario 12: ping –Q 0x144 42.42.42.5

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0x28.
- Scenario 8: ping requests are received with a TOS value of 0x30.
- Scenario 9: ping requests are received with a TOS value of 0x38.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.

Execution date : 15/09/04

- *Result* : Obtained results are as follows:
 - Scenario 1: Cancelled
 - Scenario 2: Cancelled
 - Scenario 3: Cancelled
 - Scenario 4: Cancelled
 - Scenario 5: Cancelled
 - Scenario 6: Cancelled
 - Scenario 7: ping requests are received with a TOS value of 0x28.
 - Scenario 8: ping requests are received with a TOS value of 0x30.
 - Scenario 9: ping requests are received with a TOS value of 0x38.
 - Scenario 10: ping requests are received with a TOS value of 0x00.
 - Scenario 11: ping requests are received with a TOS value of 0x00.
 - Scenario 12: ping requests are received with a TOS value of 0x00.

Failure level : None

Remarks : The scenarios cancelled above are obsolete since the link between MESCAL51 and MESCAL21 was broken.

Test Reference : TB_P1_FUNCT/DSSW/8

Test Purpose : Verify DSCP swapping at egress of MESCAL61.

Procedure : Configure local-QoS-classes of MESCAL61 as defined in testbed Configuration section. Practically, execute **qsa** script in MESCAL61.

Launch a traffic analyzer in:

- MESCAL41 that captures traffic going through eth5.
- MESCAL43 that captures traffic going through eth0.
- MESCAL71 that captures traffic going through eth1.
- MESCAL81 that captures traffic going through eth2.

Execute successively the following commands from a bash terminal of MESCAL61:

- Scenario 1: ping –Q 0xc8 41.41.41.1
- Scenario 2: ping –Q 0xd0 41.41.41.1
- Scenario 3: ping –Q 0xd8 41.41.41.1
- Scenario 4: ping 41.41.41.1
- Scenario 5: ping –Q 0x55 41.41.41.1
- Scenario 6: ping –Q 0x144 41.41.41.1
- Scenario 7: ping –Q 0xc8 43.43.43.1
- Scenario 8: ping –Q 0xd0 43.43.43.1
- Scenario 9: ping –Q 0xd8 43.43.43.1
- Scenario 10: ping 43.43.43.1
- Scenario 11: ping –Q 0x55 43.43.43.1
- Scenario 12: ping –Q 0x144 43.43.43.1
- Scenario 13: ping –Q 0xc8 6.6.6.2
- Scenario 14: ping –Q 0xd0 6.6.6.2
- Scenario 15: ping –Q 0xd8 6.6.6.2
- Scenario 16: ping 6.6.6.2
- Scenario 17: ping –Q 0x55 6.6.6.2
- Scenario 18: ping –Q 0x144 6.6.6.2
- Scenario 19: ping –Q 0xc8 6.6.6.6
- Scenario 20: ping –Q 0xd0 6.6.6.6
- Scenario 21: ping -Q 0xd8 6.6.6.6
- Scenario 22: ping 6.6.6.6
- Scenario 23: ping –Q 0x55 6.6.6.6
- Scenario 24: ping –Q 0x144 6.6.6.6

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0x48.
- Scenario 2: ping requests are received with a TOS value of 0x50.
- Scenario 3: ping requests are received with a TOS value of 0x58.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0x48.
- Scenario 8: ping requests are received with a TOS value of 0x10.
- Scenario 9: ping requests are received with a TOS value of 0x58.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 11. ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.
- Scenario 13: ping requests are received with a TOS value of 0xa8.
- Scenario 14: ping requests are received with a TOS value of 0xb0.
 Scenario 15: ping requests are received with a TOS value of 0xb8.
- Scenario 15: ping requests are received with a TOS value of 0x00.
 Scenario 16: ping requests are received with a TOS value of 0x00.

- Scenario 17: ping requests are received with a TOS value of 0x00.
- Scenario 18: ping requests are received with a TOS value of 0x00.
- Scenario 19: ping requests are received with a TOS value of 0xe8.
- Scenario 20: ping requests are received with a TOS value of 0xf0.
 - Scenario 21: ping requests are received with a TOS value of 0xf8.
- Scenario 22: ping requests are received with a TOS value of 0x00.
- Scenario 23: ping requests are received with a TOS value of 0x00.
- Scenario 24: ping requests are received with a TOS value of 0x00.

Result	: Obtained results are	as follows:
Кезин	. Obtained results are	as follows.

- Scenario 1: ping requests are received with a TOS value of 0x48.
- Scenario 2: ping requests are received with a TOS value of 0x50.
- Scenario 3: ping requests are received with a TOS value of 0x58.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.
- Scenario 7: ping requests are received with a TOS value of 0x48.
- Scenario 8: ping requests are received with a TOS value of 0x50.
- Scenario 9: ping requests are received with a TOS value of 0x58.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: ping requests are received with a TOS value of 0x00.
- Scenario 12: ping requests are received with a TOS value of 0x00.
- Scenario 13: ping requests are received with a TOS value of 0xa8.
- Scenario 14: ping requests are received with a TOS value of 0xb0.
- Scenario 15: ping requests are received with a TOS value of 0xb8.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: ping requests are received with a TOS value of 0x00.
- Scenario 18: ping requests are received with a TOS value of 0x00.
- Scenario 19: ping requests are received with a TOS value of 0xe8.
- Scenario 20: ping requests are received with a TOS value of 0xf0.
- Scenario 21: ping requests are received with a TOS value of 0xf8.
- Scenario 22: ping requests are received with a TOS value of 0x00.
- Scenario 23: ping requests are received with a TOS value of 0x00.
- Scenario 24: ping requests are received with a TOS value of 0x00.

Failure level : None Remarks : None

Remarks : None

Test Reference: TB_P1_FUNCT/DSSW/9Test Purpose: Verify DSCP swapping at egress of MESCAL71.Procedure: Configure local-QoS-classes of MESCAL71 as defined in testbed Configuration
section. Practically, execute qsa script in MESCAL71.

Launch a traffic analyzer in:

• MESCAL61 that captures traffic going through eth2.

Execute successively the following commands from a bash terminal of MESCAL71:

- Scenario 1: ping –Q 0xe8 6.6.6.1
- Scenario 2: ping –Q 0xf0 6.6.6.1
- Scenario 3: ping –Q 0xf8 6.6.6.1
- Scenario 4: ping 6.6.6.1

- Scenario 5: ping –Q 0x55 6.6.6.1
- Scenario 6: ping –Q 0x144 6.6.6.1

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xa8.
- Scenario 2: ping requests are received with a TOS value of 0xb0.
- Scenario 3: ping requests are received with a TOS value of 0xb8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.

Execution date : 15/09/04

Result : Obtained results are as follows:

- Scenario 1: ping requests are received with a TOS value of 0xa8.
- Scenario 2: ping requests are received with a TOS value of 0xb0.
- Scenario 3: ping requests are received with a TOS value of 0xb8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
 - Scenario 6: ping requests are received with a TOS value of 0x00.

Failure level: NoneRemarks: None

Test Reference : TB P1 FUNCT/DSSW/10

Test Purpose : Verify DSCP swapping at egress of MESCAL81.

Procedure : Configure local-QoS-classes of MESCAL81 as defined in testbed Configuration section. Practically, execute **qsa** script in MESCAL81.

Launch a traffic analyzer in:

• MESCAL61 that captures traffic going through eth3.

Execute successively the following commands from a bash terminal of MESCAL81:

- Scenario 1: ping –Q 0xe8 6.6.6.5
- Scenario 2: ping –Q 0xf0 6.6.6.5
- Scenario 3: ping –Q 0xf8 6.6.6.5
- Scenario 4: ping 6.6.6.5
- Scenario 5: ping –Q 0x55 6.6.6.5
- Scenario 6: ping –Q 0x144 6.6.6.5

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.

Execution date : 15/09/04

Result : Obtained results are as follows:

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.

- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: ping requests are received with a TOS value of 0x00.
- Scenario 6: ping requests are received with a TOS value of 0x00.

Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/DSSW/11

Test Purpose : Verify DSCP swapping at ingress of MESCAL11.

Procedure : execute **qsdel** in both MESCAL31 and MESCAL21 and **qsi** in MESCAL11. Launch a traffic analyzer in MESCAL21 that captures traffic going through eth6. Also launch a traffic analyzer in MESCAL31 that captures traffic going through eth1. Ensure that BGPD is started between AS1 and AS2, AS1 and AS3.

Execute successively the following commands in MESCAL31; examine traffic going through interface eth0 of MESCAL11:

- Scenario 1: ping –Q 0x88 1.1.1.6
- Scenario 2: ping –Q 0x90 1.1.1.6
- Scenario 3: ping –Q 0x98 1.1.1.6
- Scenario 4: ping 1.1.1.6
- Scenario 5: ping –Q 0x53 1.1.1.6
- Scenario 6: ping –Q 0x144 1.1.1.6

Execute successively the following commands in MESCAL21; examine traffic going through interface eth1 of MESCAL11:

- Scenario 7: ping –Q 0x68 1.1.1.2
- Scenario 8: ping –Q 0x70 1.1.1.2
- Scenario 9: ping –Q 0x78 1.1.1.2
- Scenario 10: ping 1.1.1.6
- Scenario 11: ping –Q 0x98 1.1.1.2
- Scenario 12: ping –Q 0x144 1.1.1.2

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0x28.
- Scenario 2: ping requests are received with a TOS value of 0x30.
- Scenario 3: ping requests are received with a TOS value of 0x38.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0x28.
- Scenario 8: ping requests are received with a TOS value of 0x30.
- Scenario 9: ping requests are received with a TOS value of 0x38.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.

Execution date : 15/09/04

Result : Obtained results are as follows:

• Scenario 1: ping requests are received with a TOS value of 0x28.
- Scenario 2: ping requests are received with a TOS value of 0x30.
- Scenario 3: ping requests are received with a TOS value of 0x38.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0x28.
- Scenario 8: ping requests are received with a TOS value of 0x30.
- Scenario 9: ping requests are received with a TOS value of 0x38.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.

Failure level	: None
Remarks	: None

Test Reference	: TB_P1_FUNCT/DSSW/12
Test Purpose	: Verify DSCP swapping at ingress of MESCAL21.
Procedure	: Ensure that BGP sessions are activated between AS2 and AS1, AS2 and AS4 and AS2 and AS5. Execute gsa in MESCAL21 and gsdel script in:

- MESCAL11
- MESCAL42
- MESCAL51

Launch a traffic analyzer in the following interfaces:

- Eth1 of MESCAL11
- Eth4 of MESCAL42

Execute successively the following commands from a bash terminal of MESCAL11; examine traffic going through eth4 of MESCAL42:

- Scenario 1: ping –Q 0x68 2.2.2.2
- Scenario 2: ping –Q 0x70 2.2.2.2
- Scenario 3: ping –Q 0x78 2.2.2.2
- Scenario 4: ping 2.2.2.2
- Scenario 5: ping –Q 0x98 2.2.2.2
- Scenario 6: ping –Q 0x144 2.2.2.2

Execute successively the following commands from a bash terminal of MESCAL42; examine traffic going through eth1 of MESCAL1:

- Scenario 7: ping –Q 0xc8 1.1.1.5
- Scenario 8: ping –Q 0xd0 1.1.1.5
- Scenario 9: ping –Q 0xd8 1.1.1.5
- Scenario 10: ping 2.2.2.1
- Scenario 11: ping –Q 0x98 1.1.1.5
- Scenario 12: ping –Q 0x144 1.1.1.5

Execute successively the following commands from a bash terminal of MESCAL51; examine traffic going through eth1 of MESCAL11:

- Scenario 13: ping –Q 0xe8 1.1.1.5
- Scenario 14: ping –Q 0xf0 1.1.1.5
- Scenario 15: ping –Q 0xf8 1.1.1.5
- Scenario 16: ping 1.1.1.5

- Scenario 17: ping –Q 0x98 1.1.1.5
- Scenario 18: ping –Q 0x144 1.1.1.5

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xc8.
- Scenario 2: ping requests are received with a TOS value of 0xd0.
- Scenario 3: ping requests are received with a TOS value of 0xd8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0x68.
- Scenario 8: ping requests are received with a TOS value of 0x70.
- Scenario 9: ping requests are received with a TOS value of 0x78.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.
- Scenario 13: ping requests are received with a TOS value of 0x68.
- Scenario 14: ping requests are received with a TOS value of 0x70.
- Scenario 15: ping requests are received with a TOS value of 0x78.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: Unreachable destination.
- Scenario 18: Unreachable destination.

Execution date : 15/09/04

Result : Obtained results are as follows:

•	Scenario 1: ping requests are received with a TOS value of 0xc8.	
•	Scenario 2: ping requests are received with a TOS value of 0x40	

- Scenario 2: ping requests are received with a TOS value of 0xd0.
 Scenario 3: ping requests are received with a TOS value of 0xd8.
- Scenario 5. ping requests are received with a TOS value of 0x08.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0x68.
- Scenario 8: ping requests are received with a TOS value of 0x70.
- Scenario 9: ping requests are received with a TOS value of 0x78.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.
- Scenario 13: Cancelled
- Scenario 14: Cancelled
- Scenario 15: Cancelled
- Scenario 16: Cancelled
- Scenario 17: Cancelled
- Scenario 18: Cancelled

Failure level : None

Remarks : The link between MESCAL51 and MESCAL21 was broken.

Test Reference : **TB_P1_FUNCT/DSSW/13**

Test Purpose : Verify DSCP swapping at ingress of MESCAL31.

Procedure : Ensure that BGP sessions are activated between AS3 and AS1 and AS3 and AS4. Execute **qsa** in MESCAL21 and **qsdel** script in:

- MESCAL11
- MESCAL42
- MESCAL41

Launch a traffic analyzer in the following interfaces:

- Eth0 of MESCAL11
- Eth1 of MESCAL42

Execute successively the following commands from a bash terminal of MESCAL11; examine traffic going through eth1 of MESCAL42:

- Scenario 1: ping –Q 0x88 3.3.3.6
- Scenario 2: ping –Q 0x90 3.3.3.6
- Scenario 3: ping –Q 0x98 3.3.3.6
- Scenario 4: ping 3.3.3.6
- Scenario 5: ping –Q 0x55 3.3.3.6
- Scenario 6: ping –Q 0x144 3.3.3.6

Execute successively the following commands from a bash terminal of MESCAL41; examine traffic going through eth0 of MESCAL11:

- Scenario 7: ping –Q 0xe8 1.1.1.1
- Scenario 8: ping –Q 0xf0 1.1.1.1
- Scenario 9: ping –Q 0xf8 1.1.1.1
- Scenario 10: ping 1.1.1.1
- Scenario 11: ping –Q 0x98 1.1.1.1
- Scenario 12: ping –Q 0x144 1.1.1.1

Execute successively the following commands from a bash terminal of MESCAL42; examine traffic going through eth0 of MESCAL11:

- Scenario 13: ping –Q 0xe8 1.1.1.1
- Scenario 14: ping –Q 0xf0 1.1.1.1
- Scenario 15: ping –Q 0xf8 1.1.1.1
- Scenario 16: ping 1.1.1.1
- Scenario 17: ping –Q 0x98 1.1.1.1
- Scenario 18: ping –Q 0x144 1.1.1.1

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0x88.
- Scenario 8: ping requests are received with a TOS value of 0x90.
- Scenario 9: ping requests are received with a TOS value of 0x98.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.
- Scenario 13: ping requests are received with a TOS value of 0x88.
- Scenario 14: ping requests are received with a TOS value of 0x90.
- Scenario 15: ping requests are received with a TOS value of 0x98.
- Scenario 16: ping requests are received with a TOS value of 0x00.

- Scenario 17: Unreachable destination.
- Scenario 18: Unreachable destination.

<i>Result</i> : Obtained results are as follows	52
---	----

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0x88.
- Scenario 8: ping requests are received with a TOS value of 0x90.
- Scenario 9: ping requests are received with a TOS value of 0x98.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.
- Scenario 13: ping requests are received with a TOS value of 0x88.
- Scenario 14: ping requests are received with a TOS value of 0x90.
- Scenario 15: ping requests are received with a TOS value of 0x98.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: Unreachable destination.
- Scenario 18: Unreachable destination.

Failure level: NoneRemarks: None

Test Reference	: TB_P1_FUNCT/DSSW/14				
Test Purpose	<i>e</i> : Verify DSCP swapping at ingress of MESCAL41.				
Procedure	: Ensure that BGP is activated between AS4 and AS AS4 and AS6. Execute qsdel in MESCAL31 and MESCAL61. And execute qsa in MESCAL41.				
	 Execute successively the following commands from a bash terminal of MESCAL61, check then what is received in eth6 of MESCAL31: Scenario 1: ping -Q 0x48 3.3.3.1 Scenario 2: ping -Q 0x50 3.3.3.1 Scenario 3: ping -Q 0x58 3.3.3.1 Scenario 4: ping 3.3.3.1 Scenario 5: ping -Q 0x70 3.3.3.1 Scenario 6: ping -Q 0x144 3.3.3.1 				
	 Execute successively the following commands from a bash terminal of MESCAL31, check then what is received in eth4 of MESCAL61: Scenario 7: ping -Q 0xe8 41.41.41.2 Scenario 8: ping -Q 0xf0 41.41.41.2 Scenario 9: ping -Q 0xf8 41.41.41.2 Scenario 10: ping 41.41.41.2 Scenario 11: ping -Q 0x55 41.41.41.2 Scenario 12: ping -Q 0x144 41.41.41.2 				

Expected result : Within the traffic analyzer, following results must be obtained:

• Scenario 1: ping requests are received with a TOS value of 0xe8.

- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0x48.
- Scenario 8: ping requests are received with a TOS value of 0x50.
- Scenario 9: ping requests are received with a TOS value of 0x58.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.

```
Result : Obtained results are as follows:
```

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0x48.
- Scenario 8: ping requests are received with a TOS value of 0x50.
- Scenario 9: ping requests are received with a TOS value of 0x58.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.

Failure level : None

Remarks : None

Test Reference : **TB_P1_FUNCT/DSSW/15**

Test Purpose : Verify DSCP swapping at ingress of MESCAL42.

Procedure : Ensure that BGP is activated between AS4 and AS3, AS4 and AS2 and AS4 and AS5. Execute **qsdel** in MESCAL21, MESCAL31 and MESCAL51. And execute **qsa** in MESCAL42.

Execute successively the following commands from a bash terminal of MESCAL31, check then what is received in eth0 of MESCAL21:

- Scenario 1: ping –Q 0xe8 2.2.2.1
- Scenario 2: ping –Q 0xf0 2.2.2.1
- Scenario 3: ping –Q 0xf8 2.2.2.1
- Scenario 4: ping 2.2.2.1
- Scenario 5: ping –Q 0x55 2.2.2.1
- Scenario 6: ping –Q 0x144 2.2.2.1

Execute successively the following commands from a bash terminal of MESCAL21, check then what is received in eth0 of MESCAL31:

- Scenario 7: ping –Q 0xc8 3.3.3.5
- Scenario 8: ping –Q 0xd0 3.3.3.5
- Scenario 9: ping –Q 0xd8 3.3.3.5
- Scenario 10: ping 2.2.2.1
- Scenario 11: ping –Q 0x55 3.3.3.5

• Scenario 12: ping –Q 0x144 3.3.3.5

Execute successively the following commands from a bash terminal of MESCAL51, and check what is received in eth0 of MESCAL31:

- Scenario 7: ping –Q 0x28 3.3.3.5
- Scenario 8: ping –Q 0x30 3.3.3.5
- Scenario 9: ping –Q 0x38 3.3.3.5
- Scenario 10: ping 3.3.3.5
- Scenario 11: ping –Q 0x55 3.3.3.5
- Scenario 12: ping –Q 0x144 3.3.3.5

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xc8.
- Scenario 2: ping requests are received with a TOS value of 0xd0.
- Scenario 3: ping requests are received with a TOS value of 0xc8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0xe8.
- Scenario 8: ping requests are received with a TOS value of 0xf0.
- Scenario 9: ping requests are received with a TOS value of 0xf8.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.
- Scenario 13: ping requests are received with a TOS value of 0xe8.
- Scenario 14: ping requests are received with a TOS value of 0xf0.
 - Scenario 15: ping requests are received with a TOS value of 0xf8.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: Unreachable destination.
- Scenario 18: Unreachable destination.

Execution date : 15/09/04

- *Result* : Obtained results are as follows:
 - Scenario 1: ping requests are received with a TOS value of 0xc8.
 - Scenario 2: ping requests are received with a TOS value of 0xd0.
 - Scenario 3: ping requests are received with a TOS value of 0xc8.
 - Scenario 4: ping requests are received with a TOS value of 0x00.
 - Scenario 5: Unreachable destination.
 - Scenario 6: Unreachable destination.
 - Scenario 7: ping requests are received with a TOS value of 0xe8.
 - Scenario 8: ping requests are received with a TOS value of 0xf0.
 - Scenario 9: ping requests are received with a TOS value of 0xf8.
 - Scenario 10: ping requests are received with a TOS value of 0x00.
 - Scenario 11: Unreachable destination.
 - Scenario 12: Unreachable destination.
 - Scenario 13: ping requests are received with a TOS value of 0xe8.
 - Scenario 14: ping requests are received with a TOS value of 0xf0.
 - Scenario 15: ping requests are received with a TOS value of 0xf8.
 - Scenario 16: ping requests are received with a TOS value of 0x00.
 - Scenario 17: Unreachable destination.
 - Scenario 18: Unreachable destination.

Failure level Remarks	: None : None
Test Reference	: TB_P1_FUNCT/DSSW/16
Test Purpose	: Verify DSCP swapping at ingress of MESCAL43.
Procedure	: Ensure that BGP is activated between AS4 and AS6. Execute gsdel in MESCAL61 and gsa in MESCAL43.
	 Execute successively the following commands from a bash terminal of MESCAL61, and check what is received in eth3 of MESCAL41: Scenario 1: ping -Q 0x48 41.41.41.5 Scenario 2: ping -Q 0x50 41.41.41.5 Scenario 3: ping -Q 0x58 41.41.41.5 Scenario 4: ping 41.41.41.5 Scenario 5: ping -Q 0x55 41.41.41.5 Scenario 6: ping -Q 0x144 41.41.41.5 Execute successively the following commands from a bash terminal of MESCAL61, and check what is received in eth3 of MESCAL42: Scenario 7: ping -Q 0x48 42.42.42.1
	• Scenario 8: $ping - Q 0x50 42.42.42.1$
	 Scenario 9: ping -Q 0x58 42.42.42.1 Scenario 10: ping 42 42 42 1
	• Scenario 11: ping –Q 0x55 42.42.42.1
	• Scenario 12: ping –Q 0x144 42.42.42.1
Expected result	t: Within the traffic analyzer, following results must be obtained:
	 Scenario 1: ping requests are received with a TOS value of 0x88. Scenario 2: ping requests are received with a TOS value of 0x90. Scenario 3: ping requests are received with a TOS value of 0x98. Scenario 4: ping requests are received with a TOS value of 0x00. Scenario 5: Unreachable destination. Scenario 6: Unreachable destination. Scenario 7: ping requests are received with a TOS value of 0x88 Scenario 8: ping requests are received with a TOS value of 0x98 Scenario 9: ping requests are received with a TOS value of 0x98 Scenario 10: ping requests are received with a TOS value of 0x00. Scenario 11: Unreachable destination.
Execution date Result	: 15/09/04 : Obtained results are as follows:
	 Scenario 1: ping requests are received with a TOS value of 0x88. Scenario 2: ping requests are received with a TOS value of 0x90. Scenario 3: ping requests are received with a TOS value of 0x98. Scenario 4: ping requests are received with a TOS value of 0x00. Scenario 5: Unreachable destination. Scenario 6: Unreachable destination. Scenario 7: ping requests are received with a TOS value of 0x88 Scenario 8: ping requests are received with a TOS value of 0x98 Scenario 9: ping requests are received with a TOS value of 0x98

Tost Defense	. тр р	1 EUNCT/DSSW/17
Remarks	: None	
Failure level	: None	
	٠	Scenario 12: Unreachable destination.
	•	Scenario 11: Unreachable destination.
	•	Scenario 10: ping requests are received with a TOS value of 0x00.

Test Reference	2 : TB_PI_FUNCT/DSSW/17
Test Purpose	: Verify DSCP swapping at egress of MESCAL51.
Procedure	: Ensure that BGP is activated between AS5 and AS2 and between AS4 and AS5. Execute gsdel in MESCAL21 and MESCAL42. And execute gsa in MESCAL51.
	 Execute successively the following commands from a bash terminal of MESCAL21, and check what is received in eth5 of MESCAL42: Scenario 1: ping -Q 0xe8 42.42.42.5 Scenario 2: ping -Q 0xf0 42.42.42.5 Scenario 3: ping -Q 0xf8 42.42.42.5 Scenario 4: ping 42.42.42.5 Scenario 5: ping -Q 0x55 42.42.42.5 Scenario 6: ping -Q 0x144 42.42.42.5
	Execute successively the following commands from a bash terminal of MESCAL42, and check what is received in eth5 of MESCAL21:
	• Scenario 7: ping –Q 0x28 2.2.2.5
	• Scenario 8: ping –Q 0x30 2.2.2.5
	• Scenario 9: ping $-Q$ 0x38 2.2.2.5 • Scenario 10: ping 2.2.2.5
	• Scenario 10: ping $-0.0x552225$
	 Scenario 12: ping –Q 0x144 2.2.2.5
Expected resul	<i>t</i> : Within the traffic analyzer, following results must be obtained:
	 Scenario 1: ping requests are received with a TOS value of 0x28. Scenario 2: ping requests are received with a TOS value of 0x30. Scenario 3: ping requests are received with a TOS value of 0x38. Scenario 4: ping requests are received with a TOS value of 0x00. Scenario 5: Unreachable destination. Scenario 6: Unreachable destination. Scenario 7: ping requests are received with a TOS value of 0x68. Scenario 8: ping requests are received with a TOS value of 0x68. Scenario 9: ping requests are received with a TOS value of 0x68. Scenario 10: ping requests are received with a TOS value of 0x60. Scenario 10: ping requests are received with a TOS value of 0x00. Scenario 11: Unreachable destination. Scenario 12: Unreachable destination.
Execution date Result	2 : 15/09/043 : Obtained results are as follows:
	 Scenario 1: ping requests are received with a TOS value of 0x28. Scenario 2: ping requests are received with a TOS value of 0x30. Scenario 3: ping requests are received with a TOS value of 0x38. Scenario 4: ping requests are received with a TOS value of 0x00. Scenario 5: Unreachable destination. Scenario 6: Unreachable destination.

- Scenario 7: Cancelled
- Scenario 8: Cancelled
- Scenario 9: Cancelled
- Scenario 10: Cancelled
- Scenario 11: Cancelled
- Scenario 12: Cancelled

Failure level: NoneRemarks: None

Test Reference : **TB_P1_FUNCT/DSSW/18**

Test Purpose : Verify DSCP swapping at egress of MESCAL61.

Procedure : Ensure that BGP is activated between AS6 and AS7, AS6 and AS8 and between AS4 and AS6. Execute **qsdel** in MESCAL41, MESCAL43, MESCAL71 and MESCAL81. And execute **qsa** in MESCAL61.

Execute successively the following commands from a bash terminal of MESCAL41, and check what is received in eth2 of MESCAL81:

- Scenario 1: ping –Q 0x48 6.6.6.6.6
- Scenario 2: ping –Q 0x50 6.6.6.6
- Scenario 3: ping –Q 0x58 6.6.6.6.6
- Scenario 4: ping 6.6.6.6.6
- Scenario 5: ping –Q 0x55 6.6.6.6.6
- Scenario 6: ping –Q 0x144 6.6.6.6.6

Execute successively the following commands from a bash terminal of MESCAL43, and check what is received in eth2 of MESCAL81:

- Scenario 7: ping –Q 0x48 6.6.6.6.6
- Scenario 8: ping –Q 0x50 6.6.6.6.6
- Scenario 9: ping –Q 0x58 6.6.6.6.6
- Scenario 10: ping 6.6.6.6.6
- Scenario 11: ping –Q 0x55 6.6.6.6.6
- Scenario 12: ping –Q 0x144 6.6.6.6.6

Execute successively the following commands from a bash terminal of MESCAL71, and check what is received in eth2 of MESCAL81:

- Scenario 13: ping –Q 0xa8 6.6.6.6
- Scenario 14: ping –Q 0xa0 6.6.6.6
- Scenario 15: ping –Q 0xa8 6.6.6.6
- Scenario 16: ping 6.6.6.6
- Scenario 17: ping –Q 0x55 6.6.6.6
- Scenario 18: ping –Q 0x144 6.6.6.6

Execute successively the following commands from a bash terminal of MESCAL81, and check what is received in eth1 of MESCAL71:

- Scenario 19: ping –Q 0xe8 6.6.6.2
- Scenario 20: ping –Q 0xf0 6.6.6.2
- Scenario 21: ping –Q 0xf8 6.6.6.2
- Scenario 22: ping 6.6.6.2
- Scenario 23: ping –Q 0x55 6.6.6.2
- Scenario 24: ping –Q 0x144 6.6.6.2

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0xe8.
- Scenario 8: ping requests are received with a TOS value of 0xf0.
- Scenario 9: ping requests are received with a TOS value of 0xf8.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.
- Scenario 13: ping requests are received with a TOS value of 0xe8.
- Scenario 14: ping requests are received with a TOS value of 0xf0.
- Scenario 15: ping requests are received with a TOS value of 0xf8.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: Unreachable destination.
- Scenario 18: Unreachable destination.
- Scenario 19: ping requests are received with a TOS value of 0xa8.
- Scenario 20: ping requests are received with a TOS value of 0xb0.
- Scenario 21: ping requests are received with a TOS value of 0xb8.
- Scenario 22: ping requests are received with a TOS value of 0x00.
- Scenario 23: Unreachable destination.
- Scenario 24: Unreachable destination.

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.
- Scenario 3: ping requests are received with a TOS value of 0xf8.
- Scenario 4: ping requests are received with a TOS value of 0x00.
- Scenario 5: Unreachable destination.
- Scenario 6: Unreachable destination.
- Scenario 7: ping requests are received with a TOS value of 0xe8.
- Scenario 8: ping requests are received with a TOS value of 0xf0.
- Scenario 9: ping requests are received with a TOS value of 0xf8.
- Scenario 10: ping requests are received with a TOS value of 0x00.
- Scenario 11: Unreachable destination.
- Scenario 12: Unreachable destination.
- Scenario 13: ping requests are received with a TOS value of 0xe8.
- Scenario 14: ping requests are received with a TOS value of 0xf0.
- Scenario 15: ping requests are received with a TOS value of 0xf8.
- Scenario 16: ping requests are received with a TOS value of 0x00.
- Scenario 17: Unreachable destination.
- Scenario 18: Unreachable destination.
- Scenario 19: ping requests are received with a TOS value of 0xa8.
- Scenario 20: ping requests are received with a TOS value of 0xb0.
- Scenario 21: ping requests are received with a TOS value of 0xb8.
- Scenario 22: ping requests are received with a TOS value of 0x00.
- Scenario 23: Unreachable destination.
- Scenario 24: Unreachable destination.

Failure level Remarks	: None : None		
Test Reference	: TB_P1_FUNCT/DSSW/19		
Test Purpose	: Verify DSCP swapping at egress of MESCAL71.		
Procedure	: Ensure that BGP is activated between AS6 and AS7. Execute qsdel in MESCAL61. And execute qsa in MESCAL71.		
	 Execute successively the following commands from a bash terminal of MESCAL61: Scenario 1: ping -Q 0xa8 6.6.6.2 Scenario 2: ping -Q 0xb0 6.6.6.2 Scenario 3: ping -Q 0xb8 6.6.6.2 Scenario 4: ping 6.6.6.2 Scenario 5: ping -Q 0x55 6.6.6.2 Scenario 6: ping -Q 0x144 6.6.6.2 		
Expected resul	t : Within the traffic analyzer, following results must be obtained:		
	 Scenario 1: ping requests are received with a TOS value of 0xa8. Scenario 2: ping requests are received with a TOS value of 0xb0. Scenario 3: ping requests are received with a TOS value of 0xb8. Scenario 4: ping requests are received with a TOS value of 0x00. Scenario 5: Unreachable destination. Scenario 6: Unreachable destination. 		
Execution date Result	: 15/09/04 : Obtained results are as follows:		
Eailung	 Scenario 1: ping requests are received with a TOS value of 0xa8. Scenario 2: ping requests are received with a TOS value of 0xb0. Scenario 3: ping requests are received with a TOS value of 0xb8. Scenario 4: ping requests are received with a TOS value of 0x00. Scenario 5: Unreachable destination. Scenario 6: Unreachable destination. 		
Failure level Remarks	: None : None		
Test Reference	: TB_P1_FUNCT/DSSW/20		
Test Purpose	: Verify DSCP swapping at egress of MESCAL81.		
Procedure	: execute qsde1 in MESCAL61. And execute qsa in MESCAL81.		
	 Execute successively the following commands from a bash terminal of MESCAL61: Scenario 1: ping -Q 0xe8 6.6.6.6 Scenario 2: ping -Q 0xf0 6.6.6.6 		

- Scenario 3: ping –Q 0xf8 6.6.6.6
- Scenario 4: ping 6.6.6.6
- Scenario 5: ping –Q 0x55 6.6.6.6
- Scenario 6: ping –Q 0x144 6.6.6.6

Expected result : Within the traffic analyzer, following results must be obtained:

- Scenario 1: ping requests are received with a TOS value of 0xe8.
- Scenario 2: ping requests are received with a TOS value of 0xf0.

	 Scenario 3: ping requests are received with a TOS value of 0xf8. Scenario 4: ping requests are received with a TOS value of 0x00. Scenario 5: Unreachable destination. Scenario 6: Unreachable destination.
Execution date Result	: 15/09/04 : Obtained results are as follows:
Failure level	 Scenario 1: ping requests are received with a TOS value of 0xe8. Scenario 2: ping requests are received with a TOS value of 0xf0. Scenario 3: ping requests are received with a TOS value of 0xf8. Scenario 4: ping requests are received with a TOS value of 0x00. Scenario 5: Unreachable destination. Scenario 6: Unreachable destination.
Kemarks	. None
Test Reference : TB_P1_FUNCT/DSSW/21 Test Purpose : Verify the QoS configuration of the whole testbed. Procedure :	
1 roceuure	 Log to MESCAL11, then execute the pfgping script. The scenarios are linked together; the starting configuration of scenario N+1 is the one for Scenario N. Scenario 1: Execute qsa script in all router, or execute initqos from MESCAL_ADM. Scenario 2: execute qsdel in MESCAL31 and MESCAL21. Scenario 3: execute qsi-ethl in MESCAL31 and qsi-eth6 in MESCAL21. Scenario 4: execute qsa in MESCAL31 and MESCAL21.
Expected result	: the following results must be obtained as output of pfgping script.
	 Scenario 1: All destinations are reachable in all meta-QoS-class planes. Scenario 2: All destinations are reachable in best effort plane. But only MESCAL31 and MESCAL21 interfaces are reachable in the rest of meta-QoS-class planes. Scenario 3: All destinations are reachable in best effort plane. But only MESCAL31 and MESCAL21 interfaces are reachable in the rest of meta-QoS-class planes. Scenario 4: All destinations are reached in all meta-QoS-class planes.
Execution date Result	: 15/09/04 : Obtained results are as follows:
Failure level	 Scenario 1: All destinations are reachable in all meta-QoS-class planes. Scenario 2: All destinations are reachable in best effort plane. But only MESCAL31 and MESCAL21 interfaces are reachable in the rest of meta-QoS-class planes. Scenario 3: All destinations are reachable in best effort plane. But only MESCAL31 and MESCAL21 interfaces are reachable in the rest of meta-QoS-class planes. Scenario 4: All destinations are reached in all meta-QoS-class planes. None
Remarks	: None

10.1.3TB_P1_FUNCT/SHAP

Tests that are to be carried in this section assume that the following bandwidth configuration is deployed in testbed. Before beginning executing these tests, verify that TC scripts are conform to this configuration.⁴

AS-AS	MC	BW (Mbit/s)	BW (Mbit/s)
AS1-AS2	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 5	Maximum is 5
		Min is 2	Min is 2
AS1-AS3	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 5	Maximum is 5
		Min is 2	Min is 2
AS2-AS4	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 7	Maximum is 7
		Min is 4	Min is 4
AS2-AS5	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 5	Maximum is 5
	MOL	Min is 2	Min is 2
AS3-AS4	MCI	1	1
	MC2	1	1
	MCA	⊥ Morrimum ig 7	⊥ Morrimum iq 7
	MC4	Maximum is / Min is 4	Maximum is / Min is 4
<u> 294-295</u>	MC1	1	1
ADT ADJ	MC2	1	1
	MC3	1	1
	MC4	⊥ Maximum is 7	⊥ Maximum is 7
		Min is 4	Min is 4
AS4-AS6	MC1	3	3
	MC2	3	3
	MC3	3	3
	MC4	Max is 10	Max is 10
		Min is 1	Min is 1
AS6-AS7	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 5	Maximum is 5
		Min is 2	Min is 2
AS6-AS8	MC1	1	1
	MC2	1	1
	MC3	1	1
	MC4	Maximum is 5	Maximum is 5
		Min is 2	Min is 2

Table 36 - Bandwidth Threshold per meta-QoS-class

⁴ The link between AS2 and AS5 is no more alive. Thus, test related to the link between these two routers will be concelled.

Test Reference	: TB_P1_F	FUNCT/SHAP/1		
Test Purpose	: Verify sha	: Verify shaping configuration in MESCAL11.		
Procedure	: Log to MI	ESCAL11, Configure Smartbit to generate traffic as follows:		
	•	Scenario 1: Stream with tos= $0x00$ and throughput = 5Mbit/s towards $21.0.0.2$		
	•	Scenario 2: Stream with tos= $0x28$ and throughput = 5Mbit/s towards 21 0 0 2		
	•	Scenario 3: Stream with tos= $0x30$ and throughput = 5Mbit/s towards 21002		
	-	Scenario 4: Stream with tos= $0x38$ and throughput = 5Mbit/s towards		
	-	Scenario 5: Stream with tos= $0x00$ and throughput = 5Mbit/s towards		
	-	31.0.0.2 Scenario 6: Stream with tos=0x28 and throughput = 5Mbit/s towards		
		31.0.0.2		
	•	Scenario 7: Stream with tos= $0x30$ and throughput = 5Mbit/s towards $31.0.0.2$		
	•	Scenario 8: Stream with tos= $0x38$ and throughput = 5Mbit/s towards $31.0.0.2$		
	Execute qs	sa script from a bash prompt.		
	-	Scenario 9: Stream with tos= $0x00$ and throughput = 0.5 Mbit/s towards $21.0.0.2$		
	-	Scenario 10: Stream with tos= $0x28$ and throughput = 0.5 Mbit/s towards 21 0 0 2		
	-	Scenario 11: Stream with tos= $0x30$ and throughput = 0.5 Mbit/s towards		
	•	Scenario 12: Stream with tos= $0x38$ and throughput = 0.5 Mbit/s towards		
	•	Scenario 13: Stream with tos= $0x00$ and throughput = 5Mbit/s towards		
	•	Scenario 14: Stream with tos=0x28 and throughput = 5Mbit/s towards		
	-	21.0.0.2 Scenario 15: Stream with tos=0x30 and throughput = 5Mbit/s towards		
		21.0.0.2 Scenario 16: Stream with tos=0x38 and throughput = 5Mbit/s towards		
		21.0.0.2 Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards		
		21.0.0.2 Scenario 18: Stream with tos= $0x28$ and throughput = 1Mbit/s towards		
	_	Scenario 18. Sucam with $\cos - 0x28$ and throughput – Twole's towards $21.0.0.2$		
	•	Scenario 19: Stream with tos= $0x30$ and throughput = 1Mbit/s towards $21.0.0.2$		
	•	Scenario 20: Stream with tos= $0x38$ and throughput = 1Mbit/s towards $21.0.0.2$		
	•	Scenario 21: Stream with tos= $0x00$ and throughput = 7Mbit/s towards 21.0.0.2		
	•	Scenario 22: Stream with tos= $0x28$ and throughput = 2Mbit/s towards 21.0.0.2		
	•	Scenario 23: Stream with tos= $0x30$ and throughput = 2Mbit/s towards 21.0.0.2		
	-	Scenario 24: Stream with tos= $0x38$ and throughput = 2Mbit/s towards		

21.0.0.2

- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 26: Stream with tos=0x28 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 27: Stream with tos=0x30 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 28: Stream with tos=0x38 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 30: Stream with tos=0x28 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 31: Stream with tos=0x30 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 32: Stream with tos=0x38 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 34: Stream with tos=0x28 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 35: Stream with tos=0x30 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 36: Stream with tos=0x38 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 38: Stream with tos=0x28 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 39: Stream with tos=0x30 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 40: Stream with tos=0x38 and throughput = 2Mbit/s towards 31.0.0.2

Expected result : The following results must be obtained:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.

- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s
- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 5Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 12: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 17: No drop.
 Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.

Failure level Remarks	: None : None	Scenario 30: Accepted traffic has a rate of 1Mbit/s Scenario 31: Accepted traffic has a rate of 1Mbit/s Scenario 32: Accepted traffic has a rate of 1Mbit/s Scenario 33: No drop. Scenario 34: No drop. Scenario 35: No drop. Scenario 36: No drop. Scenario 37: Accepted traffic has a rate of 5Mbit/s Scenario 38: Accepted traffic has a rate of 1Mbit/s Scenario 39: Accepted traffic has a rate of 1Mbit/s Scenario 40: Accepted traffic has a rate of 1Mbit/s			
Test Reference	: TB_P1_F	FUNCT/SHAP/2			
Test Purpose	: Verify shaping configuration in MESCAL71.				
Procedure	: Log to M	ESCAL71, Configure Smartbit to generate traffic as follows:			
	•	Scenario 1: Stream with tos= $0x00$ and throughput = 5Mbit/s towards $61.0.0.2$			
	•	Scenario 2: Stream with tos= $0xe8$ and throughput = 5Mbit/s towards $61.0.0.2$			
	•	Scenario 3: Stream with tos= $0xf0$ and throughput = 5Mbit/s towards $61.0.0.2$			
	•	Scenario 4: Stream with tos= $0xf8$ and throughput = 5Mbit/s towards $61.0.0.2$			
	Execute qsa script from a bash prompt.				
	-	Scenario 5: Stream with tos= $0x00$ and throughput = 0.5 Mbit/s towards $61.0.0.2$			
	-	Scenario 6: Stream with tos= $0xe8$ and throughput = $0.5Mbit/s$ towards $61.0.0.2$			
	-	Scenario 7: Stream with tos= $0xf0$ and throughput = $0.5Mbit/s$ towards $61.0.0.2$			
	-	Scenario 8: Stream with tos= $0xf8$ and throughput = $0.5Mbit/s$ towards $61.0.0.2$			
	-	Scenario 9: Stream with tos= $0x00$ and throughput = 5Mbit/s towards $61.0.0.2$			
	-	Scenario 10: Stream with tos= $0xe8$ and throughput = 5Mbit/s towards $61.0.0.2$			
	-	Scenario 11: Stream with tos= $0xf0$ and throughput = 5Mbit/s towards $61.0.0.2$			
	•	Scenario 12: Stream with tos= $0xf8$ and throughput = 5Mbit/s towards $61.0.0.2$			
	•	Scenario 13: Stream with tos= $0x00$ and throughput = 1Mbit/s towards $61.0.0.2$			
	•	Scenario 14: Stream with tos=0xe8 and throughput = 1Mbit/s towards 61.0.0.2			

- Scenario 15: Stream with tos=0xf0 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 16: Stream with tos=0xf8 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 18: Stream with tos=0xe8 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 19: Stream with tos=0xf0 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 20: Stream with tos=0xf8 and throughput = 2Mbit/s towards 61.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 1Mbit/s
- Scenario 11: Accepted traffic has a rate of 1Mbit/s
- Scenario 12: Accepted traffic has a rate of 1Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 5Mbit/s.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s.
- Scenario 19: Accepted traffic has a rate of 1Mbit/s.
- Scenario 20: Accepted traffic has a rate of 1Mbit/s.

Execution date : 15/09/04

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.

- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 1Mbit/s
- Scenario 11: Accepted traffic has a rate of 1Mbit/s
- Scenario 12: Accepted traffic has a rate of 1Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 5Mbit/s.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s.
- Scenario 19: Accepted traffic has a rate of 1Mbit/s.
- Scenario 20: Accepted traffic has a rate of 1Mbit/s.

Failure level: NoneRemarks: None

Test Reference : TB P1 FUNCT/SHAP/3 Test Purpose : Verify shaping configuration in MESCAL81. Procedure : Log to MESCAL81, Configure Smartbit to generate traffic as follows: Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2 Scenario 2: Stream with tos=0x28 and throughput = 5Mbit/s towards 61.0.0.2 Scenario 3: Stream with tos=0x30 and throughput = 5Mbit/s towards 61.0.0.2 Scenario 4: Stream with tos=0x38 and throughput = 5Mbit/s towards 61.0.0.2 Execute **qsa** script from a bash prompt. Scenario 5: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 61.0.0.2 Scenario 6: Stream with tos=0x28 and throughput = 0.5Mbit/s towards 61.0.0.2 Scenario 7: Stream with tos=0x30 and throughput = 0.5Mbit/s towards 61.0.0.2 Scenario 8: Stream with tos=0x38 and throughput = 0.5Mbit/s towards 61.0.0.2

- Scenario 9: Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 10: Stream with tos=0x28 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 11: Stream with tos=0x30 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 12: Stream with tos=0x38 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 14: Stream with tos=0x28 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 15: Stream with tos=0x30 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 16: Stream with tos=0x38 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 18: Stream with tos=0x28 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 19: Stream with tos=0x30 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 20: Stream with tos=0x38 and throughput = 2Mbit/s towards 61.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 1Mbit/s
- Scenario 11: Accepted traffic has a rate of 1Mbit/s
- Scenario 12: Accepted traffic has a rate of 1Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.

- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 5Mbit/s.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s.
- Scenario 19: Accepted traffic has a rate of 1Mbit/s.
- Scenario 20: Accepted traffic has a rate of 1Mbit/s.

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 1Mbit/s
- Scenario 11: Accepted traffic has a rate of 1Mbit/s
- Scenario 12: Accepted traffic has a rate of 1Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 5Mbit/s.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s.
- Scenario 19: Accepted traffic has a rate of 1Mbit/s.
- Scenario 20: Accepted traffic has a rate of 1Mbit/s.

Failure level	: None
Remarks	: None

Test Reference	: TB_P1_F	UNCT/SHAP/4			
Test Purpose	: Verify shaping configuration in MESCAL51.				
Procedure	: Log to MESCAL51, Configure Smartbit to generate traffic as follows:				
	•	Scenario 1: Stream with tos= $0x00$ and throughput = 5Mbit/s towards 21.0.0.2			
	•	Scenario 2: Stream with tos= $0xa8$ and throughput = 5Mbit/s towards $21.0.0.2$			
	•	Scenario 3: Stream with tos= $0xb0$ and throughput = 5Mbit/s towards $21.0.0.2$			

- Scenario 4: Stream with tos=0xb8 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 6: Stream with tos=0xa8 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 7: Stream with tos=0xb0 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 8: Stream with tos=0xb8 and throughput = 7Mbit/s towards 42.0.0.2

Execute **qsa** script from a bash prompt.

- Scenario 9: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 10: Stream with tos=0xa8 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 11: Stream with tos=0xb0 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 12: Stream with tos=0xb8 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 14: Stream with tos=0xa8 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 15: Stream with tos=0xb0 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 16: Stream with tos=0xb8 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 18: Stream with tos=0xa8 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 19: Stream with tos=0xb0 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 20: Stream with tos=0xb8 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 22: Stream with tos=0xa8 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 23: Stream with tos=0xb0 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 24: Stream with tos=0xb8 and throughput = 2Mbit/s towards 21.0.0.2

- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 26: Stream with tos=0xa8 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 27: Stream with tos=0xb0 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 28: Stream with tos=0xb8 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 30: Stream with tos=0xa8 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 31: Stream with tos=0xb0 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 32: Stream with tos=0xb8 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 34: Stream with tos=0xa8 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 35: Stream with tos=0xb0 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 36: Stream with tos=0xb8 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 8Mbit/s towards 42.0.0.2
- Scenario 38: Stream with tos=0xa8 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 39: Stream with tos=0xb0 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 40: Stream with tos=0xb8 and throughput = 2Mbit/s towards 42.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.

- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s
- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 7Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

- Scenario 1: Cancelled.
- Scenario 2: Cancelled.
- Scenario 3: Cancelled.
- Scenario 4: Cancelled.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: Cancelled.
- Scenario 10: Cancelled.
- Scenario 11: Cancelled.
- Scenario 12: Cancelled.
- Scenario 13: Cancelled.
- Scenario 13: Cancelled
- Scenario 14: Cancelled
 Scenario 15: Cancelled
- Scenario 15: Cancelled
 Scenario 16: Cancelled
- Scenario 17: Cancelled.
- Scenario 17: Cancelled.
 Scenario 18: Cancelled.
- Scenario 18: Cancelled.
- Scenario 19: Cancelled.
- Scenario 20: Cancelled.
- Scenario 21: Cancelled
- Scenario 22: Cancelled
- Scenario 23: Cancelled

- Scenario 24: Cancelled
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s
- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 7Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

Failure level: NoneRemarks: None

Test Reference : **TB_P1_FUNCT/SHAP/5**

Test Purpose : Verify shaping configuration in MESCAL43.

- *Procedure* : Log to MESCAL43, Configure Smartbit to generate traffic as follows:
 - Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
 - Scenario 2: Stream with tos=0x88 and throughput = 5Mbit/s towards 61.0.0.2
 - Scenario 3: Stream with tos=0x90 and throughput = 5Mbit/s towards 61.0.0.2
 - Scenario 4: Stream with tos=0x98 and throughput = 5Mbit/s towards 61.0.0.2

Execute **qsa** script from a bash prompt.

- Scenario 5: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 6: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 7: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 8: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 9: Stream with tos=0x00 and throughput = 9Mbit/s towards 61.0.0.2
- Scenario 10: Stream with tos=0x88 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 11: Stream with tos=0x90 and throughput = 5Mbit/s towards 61.0.0.2

- Scenario 12: Stream with tos=0x98 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 14: Stream with tos=0x88 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 15: Stream with tos=0x90 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 16: Stream with tos=0x98 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 15Mbit/s towards 61.0.0.2
- Scenario 18: Stream with tos=0x88 and throughput = 4Mbit/s towards 61.0.0.2
- Scenario 19: Stream with tos=0x90 and throughput = 4Mbit/s towards 61.0.0.2
- Scenario 20: Stream with tos=0x98 and throughput = 4Mbit/s towards 61.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 3Mbit/s
- Scenario 11: Accepted traffic has a rate of 3Mbit/s
- Scenario 12: Accepted traffic has a rate of 3Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 10Mbit/s
- Scenario 18: Accepted traffic has a rate of 3Mbit/s
- Scenario 19: Accepted traffic has a rate of 3Mbit/s
- Scenario 20: Accepted traffic has a rate of 3Mbit/s

Result	: Obtained results	are as follows:
--------	--------------------	-----------------

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 3Mbit/s
- Scenario 11: Accepted traffic has a rate of 3Mbit/s
- Scenario 12: Accepted traffic has a rate of 3Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 10Mbit/s
- Scenario 18: Accepted traffic has a rate of 3Mbit/s
- Scenario 19: Accepted traffic has a rate of 3Mbit/s
- Scenario 20: Accepted traffic has a rate of 3Mbit/s

Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/SHAP/6

Test Purpose : Verify shaping configuration in MESCAL41.

Procedure : Log to MESCAL41, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 2: Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 3: Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 4: Stream with tos=0x98 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 61.0.0.2

- Scenario 6: Stream with tos=0x88 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 7: Stream with tos=0x90 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 8: Stream with tos=0x98 and throughput = 7Mbit/s towards 61.0.0.2

Execute **qsa** script from a bash prompt.

- Scenario 9: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 10: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 11: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 12: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 14: Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 15: Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 16: Stream with tos=0x98 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 18: Stream with tos=0x88 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 19: Stream with tos=0x90 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 20: Stream with tos=0x98 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 9Mbit/s towards 31.0.0.2
- Scenario 22: Stream with tos=0x88 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 23: Stream with tos=0x90 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 24: Stream with tos=0x98 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 26: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 61.0.0.2

- Scenario 27: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 28: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 30: Stream with tos=0x88 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 31: Stream with tos=0x90 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 32: Stream with tos=0x98 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 34: Stream with tos=0x88 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 35: Stream with tos=0x90 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 36: Stream with tos=0x98 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 10Mbit/s towards 61.0.0.2
- Scenario 38: Stream with tos=0x88 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 39: Stream with tos=0x90 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 40: Stream with tos=0x98 and throughput = 2Mbit/s towards 61.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.

- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 7Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s.
- Scenario 24: Accepted traffic has a rate of 1Mbit/s.
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 3Mbit/s
- Scenario 31: Accepted traffic has a rate of 3Mbit/s
- Scenario 32: Accepted traffic has a rate of 3Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 7Mbit/s
- Scenario 38: Accepted traffic has a rate of 3Mbit/s
- Scenario 39: Accepted traffic has a rate of 3Mbit/s
- Scenario 40: Accepted traffic has a rate of 3Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.

- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 7Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s.
- Scenario 24: Accepted traffic has a rate of 1Mbit/s.
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 3Mbit/s
- Scenario 31: Accepted traffic has a rate of 3Mbit/s
- Scenario 32: Accepted traffic has a rate of 3Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 7Mbit/s
- Scenario 38: Accepted traffic has a rate of 3Mbit/s
- Scenario 39: Accepted traffic has a rate of 3Mbit/s
- Scenario 40: Accepted traffic has a rate of 3Mbit/s

Failure level: NoneRemarks: None

Test Reference : **TB_P1_FUNCT/SHAP/7**

Test Purpose : Verify shaping configuration in MESCAL42.

Procedure : Log to MESCAL42, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 2: Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 3: Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 4: Stream with tos=0x98 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 6: Stream with tos=0x88 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 7: Stream with tos=0x90 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 8: Stream with tos=0x98 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 9: Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 10: Stream with tos=0x88 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 11: Stream with tos=0x90 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 12: Stream with tos=0x98 and throughput = 7Mbit/s towards 51.0.0.2

Execute **qsa** script from a bash prompt.

- Scenario 13: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 14: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 15: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 16: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 18: Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 19: Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 20: Stream with tos=0x98 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 1Mbit/s towards 31.0.0.2

- Scenario 22: Stream with tos=0x88 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 23: Stream with tos=0x90 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 24: Stream with tos=0x98 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 9Mbit/s towards 31.0.0.2
- Scenario 26: Stream with tos=0x88 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 27: Stream with tos=0x90 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 28: Stream with tos=0x98 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 30: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 29: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 31: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 32: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 34: Stream with tos=0x88 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 35: Stream with tos=0x90 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 36: Stream with tos=0x98 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 38: Stream with tos=0x88 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 39: Stream with tos=0x90 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 40: Stream with tos=0x98 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 41: Stream with tos=0x00 and throughput = 9Mbit/s towards 21.0.0.2
- Scenario 42: Stream with tos=0x88 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 43: Stream with tos=0x90 and throughput = 2Mbit/s towards 21.0.0.2

- Scenario 44: Stream with tos=0x98 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 45: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 46: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 47: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 48: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 49: Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 50: Stream with tos=0x88 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 51: Stream with tos=0x90 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 52: Stream with tos=0x98 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 53: Stream with tos=0x00 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 54: Stream with tos=0x88 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 55: Stream with tos=0x90 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 56: Stream with tos=0x98 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 57: Stream with tos=0x00 and throughput = 9Mbit/s towards 51.0.0.2
- Scenario 58: Stream with tos=0x88 and throughput = 2Mbit/s towards 51.0.0.2
- Scenario 59: Stream with tos=0x90 and throughput = 2Mbit/s towards 51.0.0.2
- Scenario 60: Stream with tos=0x98 and throughput = 2Mbit/s towards 51.0.0.2

Expected result : Following results must be obtained:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.

- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 7Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop.
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s

- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop.
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.
- Scenario 55: No drop.
- Scenario 56: No drop.
- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 7Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop.
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop.
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.
- Scenario 55: No drop.
- Scenario 56: No drop.

- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s

Failure level: NoneRemarks: None

Test Reference : TB P1 FUNCT/SHAP/8 Test Purpose : Verify shaping configuration in MESCAL21. Procedure : Log to MESCAL21, Configure Smartbit to generate traffic as follows: Scenario 1: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2 Scenario 2: Stream with tos=0x48 and throughput = 7Mbit/s towards 11.0.0.2 Scenario 3: Stream with tos=0x50 and throughput = 7Mbit/s towards 11.0.0.2 Scenario 4: Stream with tos=0x58 and throughput = 7Mbit/s towards 11.0.0.2 Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2 Scenario 6: Stream with tos=0x48 and throughput = 7Mbit/s towards 42.0.0.2 Scenario 7: Stream with tos=0x50 and throughput = 7Mbit/s towards 42.0.0.2 Scenario 8: Stream with tos=0x58 and throughput = 7Mbit/s towards 42.0.0.2 Scenario 9: Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2 Scenario 10: Stream with tos=0x48 and throughput = 7Mbit/s towards 51.0.0.2 Scenario 11: Stream with tos=0x50 and throughput = 7Mbit/s towards 51.0.0.2 Scenario 12: Stream with tos=0x58 and throughput = 7Mbit/s towards 51.0.0.2 Execute **qsa** script from a bash prompt. Scenario 13: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 11.0.0.2 Scenario 14: Stream with tos=0x48 and throughput = 0.5Mbit/s towards 11.0.0.2

- Scenario 15: Stream with tos=0x50 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 16: Stream with tos=0x58 and throughput = 0.5Mbit/s towards 11.0.0.2

- Scenario 17: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 18: Stream with tos=0x48 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 19: Stream with tos=0x50 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 20: Stream with tos=0x58 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 22: Stream with tos=0x48 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 23: Stream with tos=0x50 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 24: Stream with tos=0x58 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 26: Stream with tos=0x48 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 27: Stream with tos=0x50 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 28: Stream with tos=0x58 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 30: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 29: Stream with tos=0x48 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 31: Stream with tos=0x50 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 32: Stream with tos=0x58 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 34: Stream with tos=0x48 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 35: Stream with tos=0x50 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 36: Stream with tos=0x58 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 38: Stream with tos=0x48 and throughput = 1Mbit/s towards 42.0.0.2

- Scenario 39: Stream with tos=0x50 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 40: Stream with tos=0x58 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 41: Stream with tos=0x00 and throughput = 10Mbit/s towards 42.0.0.2
- Scenario 42: Stream with tos=0x48 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 43: Stream with tos=0x50 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 44: Stream with tos=0x58 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 45: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 46: Stream with tos=0x48 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 47: Stream with tos=0x50 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 48: Stream with tos=0x58 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 49: Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 50: Stream with tos=0x48 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 51: Stream with tos=0x50 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 52: Stream with tos=0x58 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 53: Stream with tos=0x00 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 54: Stream with tos=0x48 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 55: Stream with tos=0x50 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 56: Stream with tos=0x58 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 57: Stream with tos=0x00 and throughput = 10Mbit/s towards 51.0.0.2
- Scenario 58: Stream with tos=0x48 and throughput = 2Mbit/s towards 51.0.0.2
- Scenario 59: Stream with tos=0x50 and throughput = 2Mbit/s towards 51.0.0.2
- Scenario 60: Stream with tos=0x58 and throughput = 2Mbit/s towards 51.0.0.2

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 5Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s

- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.
- Scenario 55: No drop.
- Scenario 56: No drop.
- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s

Result : Obtained results are as follows:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Cancelled.

- Scenario 11: Cancelled.
- Scenario 12: Cancelled.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 5Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: Cancelled.
- Scenario 46: Cancelled.
- Scenario 47: Cancelled.

- Scenario 48: Cancelled.
- Scenario 49: Cancelled
- Scenario 50: Cancelled
- Scenario 51: Cancelled
- Scenario 52: Cancelled
- Scenario 53: Cancelled.
- Scenario 54: Cancelled.
- Scenario 55: Cancelled.
- Scenario 56: Cancelled.
- Scenario 57: Cancelled
- Scenario 58: Cancelled
- Scenario 59: Cancelled
- Scenario 60: Cancelled

Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/SHAP/9 Test Purpose : Verify shaping configuration in MESCAL31. : Log to MESCAL31, Configure Smartbit to generate traffic as follows: Procedure Scenario 1: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2 Scenario 2: Stream with tos=0x68 and throughput = 7Mbit/s towards 11.0.0.2 Scenario 3: Stream with tos=0x70 and throughput = 7Mbit/s towards 11.0.0.2 Scenario 4: Stream with tos=0x78 and throughput = 7Mbit/s towards 11.0.0.2 Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2 Scenario 6: Stream with tos=0x68 and throughput = 7Mbit/s towards 42.0.0.2 Scenario 7: Stream with tos=0x70 and throughput = 7Mbit/s towards 42.0.0.2 Scenario 8: Stream with tos=0x78 and throughput = 7Mbit/s towards 42.0.0.2 Scenario 9: Stream with tos=0x00 and throughput = 7Mbit/s towards 41.0.0.2 Scenario 10: Stream with tos=0x68 and throughput = 7Mbit/s towards 41.0.0.2 Scenario 11: Stream with tos=0x70 and throughput = 7Mbit/s towards 41.0.0.2

Scenario 12: Stream with tos=0x78 and throughput = 7Mbit/s towards 41.0.0.2

Execute **qsa** script from a bash prompt.

- Scenario 13: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 14: Stream with tos=0x68 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 15: Stream with tos=0x70 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 16: Stream with tos=0x78 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 5Mbit/s towards 11.0.0.2
- Scenario 18: Stream with tos=0x68 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 19: Stream with tos=0x70 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 20: Stream with tos=0x78 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 22: Stream with tos=0x68 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 23: Stream with tos=0x70 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 24: Stream with tos=0x78 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 26: Stream with tos=0x68 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 27: Stream with tos=0x70 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 28: Stream with tos=0x78 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 30: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 29: Stream with tos=0x68 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 31: Stream with tos=0x70 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 32: Stream with tos=0x78 and throughput = 0.5Mbit/s towards 42.0.0.2

- Scenario 33: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 34: Stream with tos=0x68 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 35: Stream with tos=0x70 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 36: Stream with tos=0x78 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 38: Stream with tos=0x68 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 39: Stream with tos=0x70 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 40: Stream with tos=0x78 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 41: Stream with tos=0x00 and throughput = 10Mbit/s towards 42.0.0.2
- Scenario 42: Stream with tos=0x68 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 43: Stream with tos=0x70 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 44: Stream with tos=0x78 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 45: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 46: Stream with tos=0x68 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 47: Stream with tos=0x70 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 48: Stream with tos=0x78 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 49: Stream with tos=0x00 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 50: Stream with tos=0x68 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 51: Stream with tos=0x70 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 52: Stream with tos=0x78 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 53: Stream with tos=0x00 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 54: Stream with tos=0x68 and throughput = 1Mbit/s towards 41.0.0.2

- Scenario 55: Stream with tos=0x70 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 56: Stream with tos=0x78 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 57: Stream with tos=0x00 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 58: Stream with tos=0x68 and throughput = 2Mbit/s towards 41.0.0.2
- Scenario 59: Stream with tos=0x70 and throughput = 2Mbit/s towards 41.0.0.2
- Scenario 60: Stream with tos=0x78 and throughput = 2Mbit/s towards 41.0.0.2

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 5Mbit/s

- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.
- Scenario 55: No drop.
- Scenario 56: No drop.
- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s

Result : Obtained results are as follows:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 5Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.

- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.
- Scenario 55: No drop.
- Scenario 56: No drop.
- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s

Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/SHAP/10

Test Purpose : Verify shaping configuration in MESCAL61.

Procedure : Log to MESCAL61, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 2: Stream with tos=0xc8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 3: Stream with tos=0xd0and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 4: Stream with tos=0xd8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 5: Stream with tos=0x00 and throughput = 5Mbit/s towards 81.0.0.2

- Scenario 6: Stream with tos=0xc8 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 7: Stream with tos=0xd0and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 8: Stream with tos=0xd8 and throughput = 5Mbit/s towards 81.0.0.2

Execute **qsa** script from a bash prompt.

- Scenario 9: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 71.0.0.2
- Scenario 10: Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 71.0.0.2
- Scenario 11: Stream with tos=0xd0and throughput = 0.5Mbit/s towards 71.0.0.2
- Scenario 12: Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 71.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 14: Stream with tos=0xc8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 15: Stream with tos=0xd0and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 16: Stream with tos=0xd8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards 71.0.0.2
- Scenario 18: Stream with tos=0xc8 and throughput = 1Mbit/s towards 71.0.0.2
- Scenario 19: Stream with tos=0xd0and throughput = 1Mbit/s towards 71.0.0.2
- Scenario 20: Stream with tos=0xd8 and throughput = 1Mbit/s towards 71.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 7Mbit/s towards 71.0.0.2
- Scenario 22: Stream with tos=0xc8 and throughput = 2Mbit/s towards 71.0.0.2
- Scenario 23: Stream with tos=0xd0and throughput = 2Mbit/s towards 71.0.0.2
- Scenario 24: Stream with tos=0xd8 and throughput = 2Mbit/s towards 71.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 81.0.0.2
- Scenario 26: Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 81.0.0.2

- Scenario 27: Stream with tos=0xd0and throughput = 0.5Mbit/s towards 81.0.0.2
- Scenario 28: Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 81.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 30: Stream with tos=0xc8 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 31: Stream with tos=0xd0and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 32: Stream with tos=0xd8 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 81.0.0.2
- Scenario 34: Stream with tos=0xc8 and throughput = 1Mbit/s towards 81.0.0.2
- Scenario 35: Stream with tos=0xd0and throughput = 1Mbit/s towards 81.0.0.2
- Scenario 36: Stream with tos=0xd8 and throughput = 1Mbit/s towards 81.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 38: Stream with tos=0xc8 and throughput = 2Mbit/s towards 81.0.0.2
- Scenario 39: Stream with tos=0xd0and throughput = 2Mbit/s towards 81.0.0.2
- Scenario 40: Stream with tos=0xd8 and throughput = 2Mbit/s towards 81.0.0.2

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.

- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s
- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 2Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

Result : Obtained results are as follows:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.

- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s
- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 2Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/SHAP/11

Test Purpose : Verify shaping configuration in MESCAL61.

Procedure : Log to MESCAL61, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 2: Stream with tos=0xc8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 3: Stream with tos=0xd0and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 4: Stream with tos=0xd8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 5: Stream with tos=0x00 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 6: Stream with tos=0xc8 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 7: Stream with tos=0xd0and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 8: Stream with tos=0xd8 and throughput = 5Mbit/s towards 43.0.0.2

Execute **qsa** script from a bash prompt.

- Scenario 9: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 10: Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 11: Stream with tos=0xd0and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 12: Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 10Mbit/s towards 41.0.0.2
- Scenario 14: Stream with tos=0xc8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 15: Stream with tos=0xd0and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 16: Stream with tos=0xd8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 18: Stream with tos=0xc8 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 19: Stream with tos=0xd0and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 20: Stream with tos=0xd8 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 12Mbit/s towards 41.0.0.2

- Scenario 22: Stream with tos=0xc8 and throughput = 4Mbit/s towards 41.0.0.2
- Scenario 23: Stream with tos=0xd0and throughput = 4Mbit/s towards 41.0.0.2
- Scenario 24: Stream with tos=0xd8 and throughput = 4Mbit/s towards 41.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 43.0.0.2
- Scenario 26: Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 43.0.0.2
- Scenario 27: Stream with tos=0xd0and throughput = 0.5Mbit/s towards 43.0.0.2
- Scenario 28: Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 43.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 30: Stream with tos=0xc8 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 31: Stream with tos=0xd0and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 32: Stream with tos=0xd8 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 43.0.0.2
- Scenario 34: Stream with tos=0xc8 and throughput = 1Mbit/s towards 43.0.0.2
- Scenario 35: Stream with tos=0xd0and throughput = 1Mbit/s towards 43.0.0.2
- Scenario 36: Stream with tos=0xd8 and throughput = 1Mbit/s towards 43.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 15Mbit/s towards 43.0.0.2
- Scenario 38: Stream with tos=0xc8 and throughput = 6Mbit/s towards 43.0.0.2
- Scenario 39: Stream with tos=0xd0and throughput = 6Mbit/s towards 43.0.0.2
- Scenario 40: Stream with tos=0xd8 and throughput = 6Mbit/s towards 43.0.0.2

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.

- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 3Mbit/s
- Scenario 15: Accepted traffic has a rate of 3Mbit/s
- Scenario 16: Accepted traffic has a rate of 3Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 10Mbit/s
- Scenario 22: Accepted traffic has a rate of 3Mbit/s
- Scenario 23: Accepted traffic has a rate of 3Mbit/s
- Scenario 24: Accepted traffic has a rate of 3Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop
- Scenario 30: Accepted traffic has a rate of 3Mbit/s
- Scenario 31: Accepted traffic has a rate of 3Mbit/s
- Scenario 32: Accepted traffic has a rate of 3Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 10Mbit/s
- Scenario 38: Accepted traffic has a rate of 3Mbit/s
- Scenario 39: Accepted traffic has a rate of 3Mbit/s
- Scenario 40: Accepted traffic has a rate of 3Mbit/s

Result	: Obtained results	are as follows:
--------	--------------------	-----------------

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 3Mbit/s
- Scenario 15: Accepted traffic has a rate of 3Mbit/s
- Scenario 16: Accepted traffic has a rate of 3Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 10Mbit/s
- Scenario 22: Accepted traffic has a rate of 3Mbit/s
- Scenario 23: Accepted traffic has a rate of 3Mbit/s
- Scenario 24: Accepted traffic has a rate of 3Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop
- Scenario 30: Accepted traffic has a rate of 3Mbit/s
- Scenario 31: Accepted traffic has a rate of 3Mbit/s
- Scenario 32: Accepted traffic has a rate of 3Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.

- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 10Mbit/s
- Scenario 38: Accepted traffic has a rate of 3Mbit/s
- Scenario 39: Accepted traffic has a rate of 3Mbit/s
- Scenario 40: Accepted traffic has a rate of 3Mbit/s

Failure level : None

Remarks : None

10.1.4TB_P1_FUNCT/POLI

Test Reference	: TB_P1_FUNCT/POLI/1		
Test Purpose	: Verify policing configuration in MESCAL11.		
Procedure	: Log to MESCAL11, Configure Smartbit to generate traffic as follows:		
	 Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 21.0.0.2 		
	 Scenario 2: Stream with tos=0x28 and throughput = 5Mbit/s towards 21.0.0.2 		
	 Scenario 3: Stream with tos=0x30 and throughput = 5Mbit/s towards 21.0.0.2 		
	 Scenario 4: Stream with tos=0x38 and throughput = 5Mbit/s towards 21.0.0.2 		
	 Scenario 5: Stream with tos=0x00 and throughput = 5Mbit/s towards 31.0.0.2 		
	 Scenario 6: Stream with tos=0x28 and throughput = 5Mbit/s towards 31.0.0.2 		
	 Scenario 7: Stream with tos=0x30 and throughput = 5Mbit/s towards 31.0.0.2 		
	 Scenario 8: Stream with tos=0x38 and throughput = 5Mbit/s towards 31.0.0.2 		
	Execute qsa script in immediate neighbours and qsdel in local router from a bash prompt.		
	Scenario 9: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 21.0.0.2		
	 Scenario 10: Stream with tos=0x28 and throughput = 0.5Mbit/s towards 21.0.0.2 		
	 Scenario 11: Stream with tos=0x30 and throughput = 0.5Mbit/s towards 21.0.0.2 		
	 Scenario 12: Stream with tos=0x38 and throughput = 0.5Mbit/s towards 21.0.0.2 		
	 Scenario 13: Stream with tos=0x00 and throughput = 5Mbit/s towards 21.0.0.2 		
	Scenario 14: Stream with tos=0x28 and throughput = 5Mbit/s towards 21.0.0.2		

- Scenario 15: Stream with tos=0x30 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 16: Stream with tos=0x38 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 18: Stream with tos=0x28 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 19: Stream with tos=0x30 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 20: Stream with tos=0x38 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 22: Stream with tos=0x28 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 23: Stream with tos=0x30 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 24: Stream with tos=0x38 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 26: Stream with tos=0x28 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 27: Stream with tos=0x30 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 28: Stream with tos=0x38 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 30: Stream with tos=0x28 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 31: Stream with tos=0x30 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 32: Stream with tos=0x38 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 34: Stream with tos=0x28 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 35: Stream with tos=0x30 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 36: Stream with tos=0x38 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 38: Stream with tos=0x28 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 39: Stream with tos=0x30 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 40: Stream with tos=0x38 and throughput = 2Mbit/s towards 31.0.0.2

Expected result : The following results must be obtained:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s
- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 5Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

Execution date : 15/09/04

	Result	: Obtained results are
--	--------	------------------------

- ained results are as follows:Scenario 1: No drop.
 - Scenario 1: No drop.
 Scenario 2: No drop.
 - Scenario 2: No drop.Scenario 3: No drop.
 - Scenario 3: No drop.
 Scenario 4: No drop.
 - Scenario 4: No drop.
 Scenario 5: No drop.
 - Scenario 5: No drop.
 Scenario 6: No drop.
 - Scenario 7: No drop.
 - Scenario 7: No drop.
 Scenario 8: No drop.

 - Scenario 9: No drop.
 - Scenario 10: No drop.

- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s
- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 5Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

Failure level: None.Remarks: None.

Test Reference : TB_P1_FUNCT/POLI/2

Test Purpose : Verify policing configuration in MESCAL71.

Procedure : Log to MESCAL71, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 2: Stream with tos=0xe8 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 3: Stream with tos=0xf0 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 4: Stream with tos=0xf8 and throughput = 5Mbit/s towards 61.0.0.2

Execute **qsa** script in immediate neighbours and **qsdel** in local router from a bash prompt.

• Scenario 5: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 61.0.0.2

- Scenario 6: Stream with tos=0xe8 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 7: Stream with tos=0xf0 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 8: Stream with tos=0xf8 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 9: Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 10: Stream with tos=0xe8 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 11: Stream with tos=0xf0 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 12: Stream with tos=0xf8 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 14: Stream with tos=0xe8 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 15: Stream with tos=0xf0 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 16: Stream with tos=0xf8 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 18: Stream with tos=0xe8 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 19: Stream with tos=0xf0 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 20: Stream with tos=0xf8 and throughput = 2Mbit/s towards 61.0.0.2

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 1Mbit/s

- Scenario 11: Accepted traffic has a rate of 1Mbit/s
- Scenario 12: Accepted traffic has a rate of 1Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 5Mbit/s.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s.
- Scenario 19: Accepted traffic has a rate of 1Mbit/s.
- Scenario 20: Accepted traffic has a rate of 1Mbit/s.

Result : Obtained results are as follows:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 1Mbit/s
- Scenario 11: Accepted traffic has a rate of 1Mbit/s
- Scenario 12: Accepted traffic has a rate of 1Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 5Mbit/s.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s.
- Scenario 19: Accepted traffic has a rate of 1Mbit/s.
- Scenario 20: Accepted traffic has a rate of 1Mbit/s.

Failure level: None.Remarks: None.

Test Reference : TB_P1_FUNCT/POLI/3

Test Purpose : Verify policing configuration in MESCAL81.

Procedure : Log to MESCAL81, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 2: Stream with tos=0x28 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 3: Stream with tos=0x30 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 4: Stream with tos=0x38 and throughput = 5Mbit/s towards 61.0.0.2

Execute **qsa** script in immediate neighbours and **qsdel** in local router from a bash prompt.

- Scenario 5: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 6: Stream with tos=0x28 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 7: Stream with tos=0x30 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 8: Stream with tos=0x38 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 9: Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 10: Stream with tos=0x28 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 11: Stream with tos=0x30 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 12: Stream with tos=0x38 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 14: Stream with tos=0x28 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 15: Stream with tos=0x30 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 16: Stream with tos=0x38 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 18: Stream with tos=0x28 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 19: Stream with tos=0x30 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 20: Stream with tos=0x38 and throughput = 2Mbit/s towards 61.0.0.2
- Check with Smartbit statistics if there is traffic drops.

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 1Mbit/s
- Scenario 11: Accepted traffic has a rate of 1Mbit/s
- Scenario 12: Accepted traffic has a rate of 1Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 5Mbit/s.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s.
- Scenario 19: Accepted traffic has a rate of 1Mbit/s.
- Scenario 20: Accepted traffic has a rate of 1Mbit/s.

Result : Obtained results are:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 1Mbit/s
- Scenario 11: Accepted traffic has a rate of 1Mbit/s
- Scenario 12: Accepted traffic has a rate of 1Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.

- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 5Mbit/s.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s.
- Scenario 19: Accepted traffic has a rate of 1Mbit/s.
- Scenario 20: Accepted traffic has a rate of 1Mbit/s.

Failure level	: None.
Remarks	: None.

-			
Test Reference	: TB_P1_FUNCT/POLI/4		
Test Purpose	: Verify policing configuration in MESCAL51.		
Procedure	: Log to MESCAL51, Configure Smartbit to generate traffic as follows:		
	 Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 21.0.0.2 		
	 Scenario 2: Stream with tos=0xa8 and throughput = 5Mbit/s towards 21.0.0.2 		
	 Scenario 3: Stream with tos=0xb0 and throughput = 5Mbit/s towards 21.0.0.2 		
	 Scenario 4: Stream with tos=0xb8 and throughput = 5Mbit/s towards 21.0.0.2 		
	 Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2 		
	 Scenario 6: Stream with tos=0xa8 and throughput = 7Mbit/s towards 42.0.0.2 		
	 Scenario 7: Stream with tos=0xb0 and throughput = 7Mbit/s towards 42.0.0.2 		
	 Scenario 8: Stream with tos=0xb8 and throughput = 7Mbit/s towards 42.0.0.2 		
	Execute qsa script in immediate neighbours and qsdel in local router from a bash prompt.		
	 Scenario 9: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 21.0.0.2 		
	 Scenario 10: Stream with tos=0xa8 and throughput = 0.5Mbit/s towards 21.0.0.2 		
	 Scenario 11: Stream with tos=0xb0 and throughput = 0.5Mbit/s towards 21.0.0.2 		
	 Scenario 12: Stream with tos=0xb8 and throughput = 0.5Mbit/s towards 21.0.0.2 		
	 Scenario 13: Stream with tos=0x00 and throughput = 5Mbit/s towards 21.0.0.2 		
	 Scenario 14: Stream with tos=0xa8 and throughput = 5Mbit/s towards 21.0.0.2 		

- Scenario 15: Stream with tos=0xb0 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 16: Stream with tos=0xb8 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 18: Stream with tos=0xa8 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 19: Stream with tos=0xb0 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 20: Stream with tos=0xb8 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 22: Stream with tos=0xa8 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 23: Stream with tos=0xb0 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 24: Stream with tos=0xb8 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 26: Stream with tos=0xa8 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 27: Stream with tos=0xb0 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 28: Stream with tos=0xb8 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 30: Stream with tos=0xa8 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 31: Stream with tos=0xb0 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 32: Stream with tos=0xb8 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 34: Stream with tos=0xa8 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 35: Stream with tos=0xb0 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 36: Stream with tos=0xb8 and throughput = 1Mbit/s towards 42.0.0.2

- Scenario 37: Stream with tos=0x00 and throughput = 8Mbit/s towards 42.0.0.2
- Scenario 38: Stream with tos=0xa8 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 39: Stream with tos=0xb0 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 40: Stream with tos=0xb8 and throughput = 2Mbit/s towards 42.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s
- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 7Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

Execution date : 15/09/04

- : Obtained results are:
 - Scenario 1: Cancelled.
 - Scenario 2: Cancelled.
 - Scenario 3: Cancelled.
 - Scenario 4: Cancelled.
 - Scenario 5: No drop.
 - Scenario 6: No drop.
 - Scenario 7: No drop.
 - Scenario 8: No drop.
 - Scenario 9: Cancelled.
 - Scenario 10: Cancelled.
 - Scenario 11: Cancelled.
 - Scenario 12: Cancelled.
 - Scenario 13: Cancelled.
 - Scenario 14: Cancelled
 - Scenario 15: Cancelled
 - Scenario 16: Cancelled
 - Scenario 17: Cancelled.
 - Scenario 18: Cancelled.
 - Scenario 19: Cancelled.
 - Scenario 20: Cancelled.
 - Scenario 21: Cancelled
 - Scenario 22: Cancelled
 - Scenario 23: Cancelled
 - Scenario 24: Cancelled
 - Scenario 25: No drop.
 - Scenario 26: No drop.
 - Scenario 27: No drop.
 - Scenario 28: No drop.
 - Scenario 29: No drop.
 - Scenario 30: Accepted traffic has a rate of 1Mbit/s
 - Scenario 31: Accepted traffic has a rate of 1Mbit/s
 - Scenario 32: Accepted traffic has a rate of 1Mbit/s
 - Scenario 33: No drop.
 - Scenario 34: No drop.
 - Scenario 35: No drop.
 - Scenario 36: No drop.
 - Scenario 37: Accepted traffic has a rate of 7Mbit/s
 - Scenario 38: Accepted traffic has a rate of 1Mbit/s
 - Scenario 39: Accepted traffic has a rate of 1Mbit/s
 - Scenario 40: Accepted traffic has a rate of 1Mbit/s

Failure level: None.Remarks: None.

Test Reference : TB_P1_FUNCT/POLI/5

Test Purpose : Verify policing configuration in MESCAL43.

Procedure : Log to MESCAL43, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 2: Stream with tos=0x88 and throughput = 5Mbit/s towards 61.0.0.2

- Scenario 3: Stream with tos=0x90 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 4: Stream with tos=0x98 and throughput = 5Mbit/s towards 61.0.0.2

Execute **qsa** script in immediate neighbours and **qsdel** in local router from a bash prompt.

- Scenario 5: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 6: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 7: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 8: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 9: Stream with tos=0x00 and throughput = 9Mbit/s towards 61.0.0.2
- Scenario 10: Stream with tos=0x88 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 11: Stream with tos=0x90 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 12: Stream with tos=0x98 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 14: Stream with tos=0x88 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 15: Stream with tos=0x90 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 16: Stream with tos=0x98 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 15Mbit/s towards 61.0.0.2
- Scenario 18: Stream with tos=0x88 and throughput = 4Mbit/s towards 61.0.0.2
- Scenario 19: Stream with tos=0x90 and throughput = 4Mbit/s towards 61.0.0.2
- Scenario 20: Stream with tos=0x98 and throughput = 4Mbit/s towards 61.0.0.2

Check with Smartbit statistics if there is traffic drops.

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.

- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 3Mbit/s
- Scenario 11: Accepted traffic has a rate of 3Mbit/s
- Scenario 12: Accepted traffic has a rate of 3Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 10Mbit/s
- Scenario 18: Accepted traffic has a rate of 3Mbit/s
- Scenario 19: Accepted traffic has a rate of 3Mbit/s
- Scenario 20: Accepted traffic has a rate of 3Mbit/s

Result : Obtained results are:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Accepted traffic has a rate of 3Mbit/s
- Scenario 11: Accepted traffic has a rate of 3Mbit/s
- Scenario 12: Accepted traffic has a rate of 3Mbit/s
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: Accepted traffic has a rate of 10Mbit/s
- Scenario 18: Accepted traffic has a rate of 3Mbit/s
- Scenario 19: Accepted traffic has a rate of 3Mbit/s
- Scenario 20: Accepted traffic has a rate of 3Mbit/s

Failure level: None.Remarks: None.

Test Reference : TB P1 FUNCT/POLI/6 Test Purpose : Verify policing configuration in MESCAL41. : Log to MESCAL41, Configure Smartbit to generate traffic as follows: Procedure Scenario 1: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 2: Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 3: Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 4: Stream with tos=0x98 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 61.0.0.2 Scenario 6: Stream with tos=0x88 and throughput = 7Mbit/s towards 61.0.0.2 Scenario 7: Stream with tos=0x90 and throughput = 7Mbit/s towards 61.0.0.2 Scenario 8: Stream with tos=0x98 and throughput = 7Mbit/s towards 61.0.0.2 Execute **qsa** script in immediate neighbours and **qsdel** in local router from a bash prompt. Scenario 9: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 31.0.0.2 Scenario 10: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 31.0.0.2 Scenario 11: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 31.0.0.2 Scenario 12: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 31.0.0.2 Scenario 13: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 14: Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 15: Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 16: Stream with tos=0x98 and throughput = 7Mbit/s towards

Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards 31.0.0.2

Copyright © MESCAL Consortium, July 2005

31.0.0.2

- Scenario 18: Stream with tos=0x88 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 19: Stream with tos=0x90 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 20: Stream with tos=0x98 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 9Mbit/s towards 31.0.0.2
- Scenario 22: Stream with tos=0x88 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 23: Stream with tos=0x90 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 24: Stream with tos=0x98 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 26: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 27: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 28: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 30: Stream with tos=0x88 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 31: Stream with tos=0x90 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 32: Stream with tos=0x98 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 34: Stream with tos=0x88 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 35: Stream with tos=0x90 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 36: Stream with tos=0x98 and throughput = 1Mbit/s towards 61.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 10Mbit/s towards 61.0.0.2
- Scenario 38: Stream with tos=0x88 and throughput = 2Mbit/s towards 61.0.0.2
- Scenario 39: Stream with tos=0x90 and throughput = 2Mbit/s towards 61.0.0.2

Scenario 40: Stream with tos=0x98 and throughput = 2Mbit/s towards 61.0.0.2

Check with Smartbit statistics if there is traffic drops.

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 7Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s.
- Scenario 24: Accepted traffic has a rate of 1Mbit/s.
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 3Mbit/s
- Scenario 31: Accepted traffic has a rate of 3Mbit/s
- Scenario 32: Accepted traffic has a rate of 3Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.

- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 7Mbit/s
- Scenario 38: Accepted traffic has a rate of 3Mbit/s
- Scenario 39: Accepted traffic has a rate of 3Mbit/s
- Scenario 40: Accepted traffic has a rate of 3Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 7Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s.
- Scenario 24: Accepted traffic has a rate of 1Mbit/s.
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.

Scenario 30: Accepted traffic has a rate of 3Mbit/s Scenario 31: Accepted traffic has a rate of 3Mbit/s Scenario 32: Accepted traffic has a rate of 3Mbit/s Scenario 33: No drop. Scenario 34: No drop. Scenario 35: No drop. Scenario 36: No drop. Scenario 37: Accepted traffic has a rate of 7Mbit/s Scenario 38: Accepted traffic has a rate of 3Mbit/s Scenario 39: Accepted traffic has a rate of 3Mbit/s Scenario 40: Accepted traffic has a rate of 3Mbit/s Failure level : None. Remarks : None. Test Reference : TB P1 FUNCT/POLI/7 Test Purpose : Verify policing configuration in MESCAL42. Procedure : Log to MESCAL42, Configure Smartbit to generate traffic as follows: Scenario 1: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 2: Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 3: Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 4: Stream with tos=0x98 and throughput = 7Mbit/s towards 31.0.0.2 Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 21.0.0.2 Scenario 6: Stream with tos=0x88 and throughput = 7Mbit/s towards 21.0.0.2 Scenario 7: Stream with tos=0x90 and throughput = 7Mbit/s towards 21.0.0.2 Scenario 8: Stream with tos=0x98 and throughput = 7Mbit/s towards 21.0.0.2 Scenario 9: Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2 Scenario 10: Stream with tos=0x88 and throughput = 7Mbit/s towards 51.0.0.2 Scenario 11: Stream with tos=0x90 and throughput = 7Mbit/s towards 51.0.0.2 Scenario 12: Stream with tos=0x98 and throughput = 7Mbit/s towards

51.0.0.2

Execute **qsa** script in immediate neighbours and **qsdel** in local router from a bash prompt.

- Scenario 13: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 14: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 15: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 16: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 18: Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 19: Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 20: Stream with tos=0x98 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 22: Stream with tos=0x88 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 23: Stream with tos=0x90 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 24: Stream with tos=0x98 and throughput = 1Mbit/s towards 31.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 9Mbit/s towards 31.0.0.2
- Scenario 26: Stream with tos=0x88 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 27: Stream with tos=0x90 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 28: Stream with tos=0x98 and throughput = 2Mbit/s towards 31.0.0.2
- Scenario 30: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 29: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 31: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 32: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 7Mbit/s towards 21.0.0.2

- Scenario 34: Stream with tos=0x88 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 35: Stream with tos=0x90 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 36: Stream with tos=0x98 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 38: Stream with tos=0x88 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 39: Stream with tos=0x90 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 40: Stream with tos=0x98 and throughput = 1Mbit/s towards 21.0.0.2
- Scenario 41: Stream with tos=0x00 and throughput = 9Mbit/s towards 21.0.0.2
- Scenario 42: Stream with tos=0x88 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 43: Stream with tos=0x90 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 44: Stream with tos=0x98 and throughput = 2Mbit/s towards 21.0.0.2
- Scenario 45: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 46: Stream with tos=0x88 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 47: Stream with tos=0x90 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 48: Stream with tos=0x98 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 49: Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 50: Stream with tos=0x88 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 51: Stream with tos=0x90 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 52: Stream with tos=0x98 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 53: Stream with tos=0x00 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 54: Stream with tos=0x88 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 55: Stream with tos=0x90 and throughput = 1Mbit/s towards 51.0.0.2

- Scenario 56: Stream with tos=0x98 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 57: Stream with tos=0x00 and throughput = 9Mbit/s towards 51.0.0.2
- Scenario 58: Stream with tos=0x88 and throughput = 2Mbit/s towards 51.0.0.2
- Scenario 59: Stream with tos=0x90 and throughput = 2Mbit/s towards 51.0.0.2
- Scenario 60: Stream with tos=0x98 and throughput = 2Mbit/s towards 51.0.0.2

Expected result : Following results must be obtained:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 7Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s

- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop.
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop.
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.
- Scenario 55: No drop.
- Scenario 56: No drop.
- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.

- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 7Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop.
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.

- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop.
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.
- Scenario 55: No drop.
- Scenario 56: No drop.
- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s
- Failure level: None.Remarks: None.

Test Reference : TB_P1_FUNCT/POLI/8

Test Purpose : Verify policing configuration in MESCAL21.

- *Procedure* : Log to MESCAL21, Configure Smartbit to generate traffic as follows:
 - Scenario 1: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
 - Scenario 2: Stream with tos=0x48 and throughput = 7Mbit/s towards 11.0.0.2
 - Scenario 3: Stream with tos=0x50 and throughput = 7Mbit/s towards 11.0.0.2
 - Scenario 4: Stream with tos=0x58 and throughput = 7Mbit/s towards 11.0.0.2
 - Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
 - Scenario 6: Stream with tos=0x48 and throughput = 7Mbit/s towards 42.0.0.2

- Scenario 7: Stream with tos=0x50 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 8: Stream with tos=0x58 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 9: Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 10: Stream with tos=0x48 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 11: Stream with tos=0x50 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 12: Stream with tos=0x58 and throughput = 7Mbit/s towards 51.0.0.2

Execute **qsa** script in immediate neighbours and **qsdel** in local router from a bash prompt.

- Scenario 13: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 14: Stream with tos=0x48 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 15: Stream with tos=0x50 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 16: Stream with tos=0x58 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 18: Stream with tos=0x48 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 19: Stream with tos=0x50 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 20: Stream with tos=0x58 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 22: Stream with tos=0x48 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 23: Stream with tos=0x50 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 24: Stream with tos=0x58 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 26: Stream with tos=0x48 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 27: Stream with tos=0x50 and throughput = 2Mbit/s towards 11.0.0.2

- Scenario 28: Stream with tos=0x58 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 30: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 29: Stream with tos=0x48 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 31: Stream with tos=0x50 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 32: Stream with tos=0x58 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 34: Stream with tos=0x48 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 35: Stream with tos=0x50 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 36: Stream with tos=0x58 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 38: Stream with tos=0x48 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 39: Stream with tos=0x50 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 40: Stream with tos=0x58 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 41: Stream with tos=0x00 and throughput = 10Mbit/s towards 42.0.0.2
- Scenario 42: Stream with tos=0x48 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 43: Stream with tos=0x50 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 44: Stream with tos=0x58 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 45: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 46: Stream with tos=0x48 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 47: Stream with tos=0x50 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 48: Stream with tos=0x58 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 49: Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2

- Scenario 50: Stream with tos=0x48 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 51: Stream with tos=0x50 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 52: Stream with tos=0x58 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 53: Stream with tos=0x00 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 54: Stream with tos=0x48 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 55: Stream with tos=0x50 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 56: Stream with tos=0x58 and throughput = 1Mbit/s towards 51.0.0.2
- Scenario 57: Stream with tos=0x00 and throughput = 10Mbit/s towards 51.0.0.2
- Scenario 58: Stream with tos=0x48 and throughput = 2Mbit/s towards 51.0.0.2
- Scenario 59: Stream with tos=0x50 and throughput = 2Mbit/s towards 51.0.0.2
- Scenario 60: Stream with tos=0x58 and throughput = 2Mbit/s towards 51.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.

- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 5Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.

- Scenario 55: No drop.
- Scenario 56: No drop.
- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: Cancelled.
- Scenario 11: Cancelled.
- Scenario 12: Cancelled.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 5Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.

- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: Cancelled.
- Scenario 46: Cancelled.
- Scenario 47: Cancelled.
- Scenario 48: Cancelled.
- Scenario 49: Cancelled
- Scenario 50: Cancelled
- Scenario 51: Cancelled
- Scenario 52: Cancelled
- Scenario 53: Cancelled.
- Scenario 54: Cancelled.
- Scenario 55: Cancelled.
- Scenario 56: Cancelled.
- Scenario 57: Cancelled
- Scenario 58: Cancelled
- Scenario 59: Cancelled
- Scenario 60: Cancelled

Failure level	: None.
Remarks	: None.

Test Reference	: TB_P1_FUNCT/POLI/9
Test Purpose	: Verify policing configuration in MESCAL31.

Procedure : Log to MESCAL31, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 2: Stream with tos=0x68 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 3: Stream with tos=0x70 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 4: Stream with tos=0x78 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 5: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 6: Stream with tos=0x68 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 7: Stream with tos=0x70 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 8: Stream with tos=0x78 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 9: Stream with tos=0x00 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 10: Stream with tos=0x68 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 11: Stream with tos=0x70 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 12: Stream with tos=0x78 and throughput = 7Mbit/s towards 41.0.0.2

Execute **qsa** script in immediate neighbours and **qsdel** in local router from a bash prompt.

- Scenario 13: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 14: Stream with tos=0x68 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 15: Stream with tos=0x70 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 16: Stream with tos=0x78 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 5Mbit/s towards 11.0.0.2
- Scenario 18: Stream with tos=0x68 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 19: Stream with tos=0x70 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 20: Stream with tos=0x78 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 1Mbit/s towards 11.0.0.2

- Scenario 22: Stream with tos=0x68 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 23: Stream with tos=0x70 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 24: Stream with tos=0x78 and throughput = 1Mbit/s towards 11.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 26: Stream with tos=0x68 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 27: Stream with tos=0x70 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 28: Stream with tos=0x78 and throughput = 2Mbit/s towards 11.0.0.2
- Scenario 30: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 29: Stream with tos=0x68 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 31: Stream with tos=0x70 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 32: Stream with tos=0x78 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 34: Stream with tos=0x68 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 35: Stream with tos=0x70 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 36: Stream with tos=0x78 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 38: Stream with tos=0x68 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 39: Stream with tos=0x70 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 40: Stream with tos=0x78 and throughput = 1Mbit/s towards 42.0.0.2
- Scenario 41: Stream with tos=0x00 and throughput = 10Mbit/s towards 42.0.0.2
- Scenario 42: Stream with tos=0x68 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 43: Stream with tos=0x70 and throughput = 2Mbit/s towards 42.0.0.2

- Scenario 44: Stream with tos=0x78 and throughput = 2Mbit/s towards 42.0.0.2
- Scenario 45: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 46: Stream with tos=0x68 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 47: Stream with tos=0x70 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 48: Stream with tos=0x78 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 49: Stream with tos=0x00 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 50: Stream with tos=0x68 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 51: Stream with tos=0x70 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 52: Stream with tos=0x78 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 53: Stream with tos=0x00 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 54: Stream with tos=0x68 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 55: Stream with tos=0x70 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 56: Stream with tos=0x78 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 57: Stream with tos=0x00 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 58: Stream with tos=0x68 and throughput = 2Mbit/s towards 41.0.0.2
- Scenario 59: Stream with tos=0x70 and throughput = 2Mbit/s towards 41.0.0.2
- Scenario 60: Stream with tos=0x78 and throughput = 2Mbit/s towards 41.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.

- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s
- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 5Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s

- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.
- Scenario 55: No drop.
- Scenario 56: No drop.
- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: No drop.
- Scenario 15: No drop.
- Scenario 16: No drop.
- Scenario 17: No drop.
- Scenario 18: Accepted traffic has a rate of 1Mbit/s
- Scenario 19: Accepted traffic has a rate of 1Mbit/s

- Scenario 20: Accepted traffic has a rate of 1Mbit/s
- Scenario 21: No drop.
- Scenario 22: No drop.
- Scenario 23: No drop.
- Scenario 24: No drop.
- Scenario 25: Accepted traffic has a rate of 5Mbit/s
- Scenario 26: Accepted traffic has a rate of 1Mbit/s
- Scenario 27: Accepted traffic has a rate of 1Mbit/s
- Scenario 28: Accepted traffic has a rate of 1Mbit/s
- Scenario 29: No drop.
- Scenario 30: No drop.
- Scenario 31: No drop.
- Scenario 32: No drop.
- Scenario 33: No drop
- Scenario 34: Accepted traffic has a rate of 1Mbit/s
- Scenario 35: Accepted traffic has a rate of 1Mbit/s
- Scenario 36: Accepted traffic has a rate of 1Mbit/s
- Scenario 37: No drop.
- Scenario 38: No drop.
- Scenario 39: No drop.
- Scenario 40: No drop.
- Scenario 41: Accepted traffic has a rate of 7Mbit/s
- Scenario 42: Accepted traffic has a rate of 1Mbit/s
- Scenario 43: Accepted traffic has a rate of 1Mbit/s
- Scenario 44: Accepted traffic has a rate of 1Mbit/s
- Scenario 45: No drop.
- Scenario 46: No drop.
- Scenario 47: No drop.
- Scenario 48: No drop.
- Scenario 49: No drop
- Scenario 50: Accepted traffic has a rate of 1Mbit/s
- Scenario 51: Accepted traffic has a rate of 1Mbit/s
- Scenario 52: Accepted traffic has a rate of 1Mbit/s
- Scenario 53: No drop.
- Scenario 54: No drop.
- Scenario 55: No drop.
- Scenario 56: No drop.

- Scenario 57: Accepted traffic has a rate of 7Mbit/s
- Scenario 58: Accepted traffic has a rate of 1Mbit/s
- Scenario 59: Accepted traffic has a rate of 1Mbit/s
- Scenario 60: Accepted traffic has a rate of 1Mbit/s

Failure level: None.Remarks: None.

Test Reference : TB_P1_FUNCT/POLI/10

Test Purpose : Verify policing configuration in MESCAL61.

Procedure : Log to MESCAL61, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 2: Stream with tos=0xc8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 3: Stream with tos=0xd0and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 4: Stream with tos=0xd8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 5: Stream with tos=0x00 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 6: Stream with tos=0xc8 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 7: Stream with tos=0xd0and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 8: Stream with tos=0xd8 and throughput = 5Mbit/s towards 81.0.0.2

Execute **qsa** script in immediate neighbours and **qsdel** in local router from a bash prompt.

- Scenario 9: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 71.0.0.2
- Scenario 10: Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 71.0.0.2
- Scenario 11: Stream with tos=0xd0and throughput = 0.5Mbit/s towards 71.0.0.2
- Scenario 12: Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 71.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 14: Stream with tos=0xc8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 15: Stream with tos=0xd0and throughput = 5Mbit/s towards 71.0.0.2

- Scenario 16: Stream with tos=0xd8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards 71.0.0.2
- Scenario 18: Stream with tos=0xc8 and throughput = 1Mbit/s towards 71.0.0.2
- Scenario 19: Stream with tos=0xd0and throughput = 1Mbit/s towards 71.0.0.2
- Scenario 20: Stream with tos=0xd8 and throughput = 1Mbit/s towards 71.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 7Mbit/s towards 71.0.0.2
- Scenario 22: Stream with tos=0xc8 and throughput = 2Mbit/s towards 71.0.0.2
- Scenario 23: Stream with tos=0xd0and throughput = 2Mbit/s towards 71.0.0.2
- Scenario 24: Stream with tos=0xd8 and throughput = 2Mbit/s towards 71.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 81.0.0.2
- Scenario 26: Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 81.0.0.2
- Scenario 27: Stream with tos=0xd0and throughput = 0.5Mbit/s towards 81.0.0.2
- Scenario 28: Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 81.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 30: Stream with tos=0xc8 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 31: Stream with tos=0xd0and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 32: Stream with tos=0xd8 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 81.0.0.2
- Scenario 34: Stream with tos=0xc8 and throughput = 1Mbit/s towards 81.0.0.2
- Scenario 35: Stream with tos=0xd0and throughput = 1Mbit/s towards 81.0.0.2
- Scenario 36: Stream with tos=0xd8 and throughput = 1Mbit/s towards 81.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 5Mbit/s towards 81.0.0.2

- Scenario 38: Stream with tos=0xc8 and throughput = 2Mbit/s towards 81.0.0.2
- Scenario 39: Stream with tos=0xd0and throughput = 2Mbit/s towards 81.0.0.2
- Scenario 40: Stream with tos=0xd8 and throughput = 2Mbit/s towards 81.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s

- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 2Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 1Mbit/s
- Scenario 15: Accepted traffic has a rate of 1Mbit/s
- Scenario 16: Accepted traffic has a rate of 1Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 5Mbit/s
- Scenario 22: Accepted traffic has a rate of 1Mbit/s
- Scenario 23: Accepted traffic has a rate of 1Mbit/s
- Scenario 24: Accepted traffic has a rate of 1Mbit/s
- Scenario 25: No drop.

- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop.
- Scenario 30: Accepted traffic has a rate of 1Mbit/s
- Scenario 31: Accepted traffic has a rate of 1Mbit/s
- Scenario 32: Accepted traffic has a rate of 1Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 2Mbit/s
- Scenario 38: Accepted traffic has a rate of 1Mbit/s
- Scenario 39: Accepted traffic has a rate of 1Mbit/s
- Scenario 40: Accepted traffic has a rate of 1Mbit/s

Failure level: None.Remarks: None.

Tost	Roforonco	TR	P1	FUNCT/POLI/11
resi	Reference	ID	ГІ	FUNCI/FULI/II

Test Purpose : Verify policing configuration in MESCAL61.

Procedure : Log to MESCAL61, Configure Smartbit to generate traffic as follows:

- Scenario 1: Stream with tos=0x00 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 2: Stream with tos=0xc8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 3: Stream with tos=0xd0and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 4: Stream with tos=0xd8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 5: Stream with tos=0x00 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 6: Stream with tos=0xc8 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 7: Stream with tos=0xd0and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 8: Stream with tos=0xd8 and throughput = 5Mbit/s towards 43.0.0.2

Execute **qsa** script in immediate neighbours and **qsdel** in local router from a bash prompt.

Scenario 9: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 41.0.0.2

- Scenario 10: Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 11: Stream with tos=0xd0and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 12: Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 13: Stream with tos=0x00 and throughput = 10Mbit/s towards 41.0.0.2
- Scenario 14: Stream with tos=0xc8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 15: Stream with tos=0xd0and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 16: Stream with tos=0xd8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 17: Stream with tos=0x00 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 18: Stream with tos=0xc8 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 19: Stream with tos=0xd0and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 20: Stream with tos=0xd8 and throughput = 1Mbit/s towards 41.0.0.2
- Scenario 21: Stream with tos=0x00 and throughput = 12Mbit/s towards 41.0.0.2
- Scenario 22: Stream with tos=0xc8 and throughput = 4Mbit/s towards 41.0.0.2
- Scenario 23: Stream with tos=0xd0and throughput = 4Mbit/s towards 41.0.0.2
- Scenario 24: Stream with tos=0xd8 and throughput = 4Mbit/s towards 41.0.0.2
- Scenario 25: Stream with tos=0x00 and throughput = 0.5Mbit/s towards 43.0.0.2
- Scenario 26: Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 43.0.0.2
- Scenario 27: Stream with tos=0xd0and throughput = 0.5Mbit/s towards 43.0.0.2
- Scenario 28: Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 43.0.0.2
- Scenario 29: Stream with tos=0x00 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 30: Stream with tos=0xc8 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 31: Stream with tos=0xd0and throughput = 5Mbit/s towards 43.0.0.2

- Scenario 32: Stream with tos=0xd8 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 33: Stream with tos=0x00 and throughput = 1Mbit/s towards 43.0.0.2
- Scenario 34: Stream with tos=0xc8 and throughput = 1Mbit/s towards 43.0.0.2
- Scenario 35: Stream with tos=0xd0and throughput = 1Mbit/s towards 43.0.0.2
- Scenario 36: Stream with tos=0xd8 and throughput = 1Mbit/s towards 43.0.0.2
- Scenario 37: Stream with tos=0x00 and throughput = 15Mbit/s towards 43.0.0.2
- Scenario 38: Stream with tos=0xc8 and throughput = 6Mbit/s towards 43.0.0.2
- Scenario 39: Stream with tos=0xd0and throughput = 6Mbit/s towards 43.0.0.2
- Scenario 40: Stream with tos=0xd8 and throughput = 6Mbit/s towards 43.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5: No drop.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8: No drop.
- Scenario 9: No drop.
- Scenario 10: No drop.
- Scenario 11: No drop.
- Scenario 12: No drop.
- Scenario 13: No drop.
- Scenario 14: Accepted traffic has a rate of 3Mbit/s
- Scenario 15: Accepted traffic has a rate of 3Mbit/s
- Scenario 16: Accepted traffic has a rate of 3Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.

- Scenario 21: Accepted traffic has a rate of 10Mbit/s
- Scenario 22: Accepted traffic has a rate of 3Mbit/s
- Scenario 23: Accepted traffic has a rate of 3Mbit/s
- Scenario 24: Accepted traffic has a rate of 3Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop
- Scenario 30: Accepted traffic has a rate of 3Mbit/s
- Scenario 31: Accepted traffic has a rate of 3Mbit/s
- Scenario 32: Accepted traffic has a rate of 3Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 10Mbit/s
- Scenario 38: Accepted traffic has a rate of 3Mbit/s
- Scenario 39: Accepted traffic has a rate of 3Mbit/s
- Scenario 40: Accepted traffic has a rate of 3Mbit/s

- Result
- : Obtained results are: • Scenario 1: No drop.
 - Soonario 1. 110 arop.
 - Scenario 2: No drop.
 - Scenario 3: No drop.
 - Scenario 4: No drop.
 - Scenario 5: No drop.
 - Scenario 6: No drop.
 - Scenario 7: No drop.
 - Scenario 8: No drop.
 - Scenario 9: No drop.
 - Scenario 10: No drop.
 - Scenario 11: No drop.
 - Scenario 12: No drop.
 - Scenario 13: No drop.
 - Scenario 14: Accepted traffic has a rate of 3Mbit/s
 - Scenario 15: Accepted traffic has a rate of 3Mbit/s

- Scenario 16: Accepted traffic has a rate of 3Mbit/s
- Scenario 17: No drop.
- Scenario 18: No drop.
- Scenario 19: No drop.
- Scenario 20: No drop.
- Scenario 21: Accepted traffic has a rate of 10Mbit/s
- Scenario 22: Accepted traffic has a rate of 3Mbit/s
- Scenario 23: Accepted traffic has a rate of 3Mbit/s
- Scenario 24: Accepted traffic has a rate of 3Mbit/s
- Scenario 25: No drop.
- Scenario 26: No drop.
- Scenario 27: No drop.
- Scenario 28: No drop.
- Scenario 29: No drop
- Scenario 30: Accepted traffic has a rate of 3Mbit/s
- Scenario 31: Accepted traffic has a rate of 3Mbit/s
- Scenario 32: Accepted traffic has a rate of 3Mbit/s
- Scenario 33: No drop.
- Scenario 34: No drop.
- Scenario 35: No drop.
- Scenario 36: No drop.
- Scenario 37: Accepted traffic has a rate of 10Mbit/s
- Scenario 38: Accepted traffic has a rate of 3Mbit/s
- Scenario 39: Accepted traffic has a rate of 3Mbit/s
- Scenario 40: Accepted traffic has a rate of 3Mbit/s

Failure level: None.Remarks: None.

10.1.5TB_P1_FUNCT/BWMA

Test Reference	: TB_P1_FUNCT/BWMA/1			
Test Purpose	: Verify bandwidth management configuration in MESCAL11.			
Procedure	: Log to MESCAL11, Configure Smartbit to generate traffic as follows:			
 Scenario 1: 				
	• Stream with tos= $0x00$ and throughput = $8Mbit/s$ towards 21.0.0.2			

- Stream with tos=0x28 and throughput = 8Mbit/s towards 21.0.0.2
- Stream with tos=0x30 and throughput = 8Mbit/s towards 21.0.0.2

- \circ Stream with tos=0x38 and throughput = 8Mbit/s towards 21.0.0.2
- Scenario 2:
 - Stream with tos=0x00 and throughput = 8Mbit/s towards 31.0.0.2
 - Stream with tos=0x28 and throughput = 8Mbit/s towards 31.0.0.2
 - Stream with tos=0x30 and throughput = 8Mbit/s towards 31.0.0.2
 - Stream with tos=0x38 and throughput = 8Mbit/s towards 31.0.0.2

Execute qsa script from a bash prompt.

- Scenario 3:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 21.0.0.2
 - \circ Stream with tos=0x28 and throughput = 0.5Mbit/s towards 21.0.0.2
 - \circ Stream with tos=0x30 and throughput = 0.5Mbit/s towards 21.0.0.2
 - \circ Stream with tos=0x38 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 4:
 - Stream with tos=0x00 and throughput = 5Mbit/s towards 21.0.0.2
 - Stream with tos=0x28 and throughput = 5Mbit/s towards 21.0.0.2
 - Stream with tos=0x30 and throughput = 5Mbit/s towards 21.0.0.2
 - Stream with tos=0x38 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 5:
 - \circ Stream with tos=0x00 and throughput = 5Mbit/s towards 21.0.0.2
 - Stream with tos=0x30 and throughput = 5Mbit/s towards 21.0.0.2
 - Stream with tos=0x38 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 6:
 - \circ Stream with tos=0x00 and throughput = 5Mbit/s towards 21.0.0.2
 - \circ Stream with tos=0x28 and throughput = 5Mbit/s towards 21.0.0.2
- Scenario 7:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x28 and throughput = 0.5Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x30 and throughput = 0.5Mbit/s towards 31.0.0.2
 - Stream with tos=0x38 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 8:
 - \circ Stream with tos=0x00 and throughput = 5Mbit/s towards 31.0.0.2

- \circ Stream with tos=0x28 and throughput = 5Mbit/s towards 31.0.0.2
- Stream with tos=0x30 and throughput = 5Mbit/s towards 31.0.0.2
- Stream with tos=0x38 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 9:
 - Stream with tos=0x00 and throughput = 5Mbit/s towards 31.0.0.2
 - o Stream with tos=0x28 and throughput = 5Mbit/s towards 31.0.0.2
 - o Stream with tos=0x30 and throughput = 5Mbit/s towards 31.0.0.2
- Scenario 10:
 - Stream with tos=0x00 and throughput = 5Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x28 and throughput = 5Mbit/s towards 31.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - o Stream with tos=0x28 is received with a throughput of 1Mbit/s
 - Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 5:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 6:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x28 is received with a throughput of 1Mbit/s
- Scenario 7: No drop.
- Scenario 8:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - Stream with tos=0x28 is received with a throughput of 1Mbit/s
 - Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s

Copyright © MESCAL Consortium, July 2005

- Scenario 9:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 10:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x28 is received with a throughput of 1Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - Stream with tos=0x28 is received with a throughput of 1Mbit/s
 - Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 5:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - o Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 6:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x28 is received with a throughput of 1Mbit/s
- Scenario 7: No drop.
- Scenario 8:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - Stream with tos=0x28 is received with a throughput of 1Mbit/s
 - Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 9:

- o Received traffic has a throughput of 5Mbit/s
- Stream with tos=0x00 is received with a throughput of 3Mbit/s
- Stream with tos=0x30 is received with a throughput of 1Mbit/s
- Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 10:
 - o Received traffic has a throughput of 5Mbit/s
 - Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x28 is received with a throughput of 1Mbit/s

Failure level	: None.
Remarks	: None.

Test Reference : TB_P1_FUNCT/BWMA/2

Test Purpose : Verify bandwidth management configuration in MESCAL71.

Procedure : Log to MESCAL71, Configure Smartbit to generate traffic as follows:

- Scenario 1:
 - \circ Stream with tos=0x00 and throughput = 8Mbit/s towards 61.0.0.2
 - Stream with tos=0xe8 and throughput = 8Mbit/s towards 61.0.0.2
 - Stream with tos=0xf0 and throughput = 8Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0xf8 and throughput = 8Mbit/s towards 61.0.0.2

Execute qsa script from a bash prompt.

- Scenario 2:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0xe8 and throughput = 0.5Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0xf0 and throughput = 0.5Mbit/s towards 61.0.0.2
 - Stream with tos=0xf8 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 3:
 - Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
 - Stream with tos=0xe8 and throughput = 5Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0xf0 and throughput = 5Mbit/s towards 61.0.0.2
 - Stream with tos=0xf8 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 4:
 - \circ Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
 - Stream with tos=0xf0 and throughput = 5Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0xf8 and throughput = 5Mbit/s towards 61.0.0.2
- Scenario 5:
 - \circ Stream with tos=0x00 and throughput = 5Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0xe8 and throughput = 1Mbit/s towards 61.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - Stream with tos=0xe8 is received with a throughput of 1Mbit/s
 - Stream with tos=0xf0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xf8 is received with a throughput of 1Mbit/s
- Scenario 4:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - Stream with tos=0xf0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xf8 is received with a throughput of 1Mbit/s
- Scenario 5:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0xe8 is received with a throughput of 1Mbit/s

Execution date : 15/09/04

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - Stream with tos=0xe8 is received with a throughput of 1Mbit/s
 - Stream with tos=0xf0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xf8 is received with a throughput of 1Mbit/s
- Scenario 4:
 - Received traffic has a throughput of 5Mbit/s
 - Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - Stream with tos=0xf0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xf8 is received with a throughput of 1Mbit/s

- Scenario 5:
 - o Received traffic has a throughput of 5Mbit/s
 - Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0xe8 is received with a throughput of 1Mbit/s

Failure level Remarks	: None. : None.			
Test Reference	: TB_P1_FUNCT/BWMA/3			
Test Purpose	: Verify bandwidth management configuration in MESCAL81.			
Procedure	: Log to MESCAL81, Configure Smartbit to generate traffic as follows:			
	 Scenario 1: 			
	• Stream with tos= $0x00$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x28$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x30$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x38$ and throughput = 5Mbit/s towards 61.0.0.2			
	Execute qsa script from a bash prompt.			
	 Scenario 2: 			
	• Stream with tos= $0x00$ and throughput = 0.5 Mbit/s towards $61.0.0.2$			
	• Stream with tos= $0x28$ and throughput = 0.5 Mbit/s towards $61.0.0.2$			
	• Stream with tos= $0x30$ and throughput = 0.5 Mbit/s towards $61.0.0.2$			
	• Stream with tos=0x38 and throughput = 0.5Mbit/s towards 61.0.0.2			
	• Scenario 3:			
	• Stream with tos= $0x00$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x28$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x30$ and throughput = $5Mbit/s$ towards $61.0.0.2$			
	• Stream with tos= $0x38$ and throughput = $5Mbit/s$ towards $61.0.0.2$			
	 Scenario 4: 			
	• Stream with tos= $0x00$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x30$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x38$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Scenario 5:			
	• Stream with tos= $0x00$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x28$ and throughput = 5Mbit/s towards 61.0.0.2			
	Check with Smartbit statistics if there is traffic drops.			

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - o Stream with tos=0x28 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 4:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 5:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x28 is received with a throughput of 1Mbit/s

Execution date : 15/09/04

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - Stream with tos=0x28 is received with a throughput of 1Mbit/s
 - Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 4:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - Stream with tos=0x30 is received with a throughput of 1Mbit/s
 - Stream with tos=0x38 is received with a throughput of 1Mbit/s
- Scenario 5:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x28 is received with a throughput of 1Mbit/s

Failure level Remarks	: None. : None.				
Test Reference	: TB_P1_I	FUNCT/	BWMA/4		
Test Purpose	: Verify ba	: Verify bandwidth management configuration in MESCAL51.			
Procedure	: Log to M	ESCAL:	51, Configure Smartbit to generate traffic as follows:		
	-	Scenar	io 1:		
		0	Stream with tos= $0x00$ and throughput = $8Mbit/s$ towards 21.0.0.2		
		0	Stream with tos= $0xa8$ and throughput = $8Mbit/s$ towards $21.0.0.2$		
		0	Stream with tos= $0xb0$ and throughput = $8Mbit/s$ towards $21.0.0.2$		
		0	Stream with tos=0xb8 and throughput = 8Mbit/s towards 21.0.0.2		
	•	Scenar	io 2:		
		0	Stream with tos= $0x00$ and throughput = $8Mbit/s$ towards $42.0.0.2$		
		0	Stream with tos=0xa8 and throughput = 8Mbit/s towards 42.0.0.2		
		0	Stream with tos=0xb0 and throughput = 8Mbit/s towards 42.0.0.2		
		0	Stream with tos=0xb8 and throughput = 8Mbit/s towards 42.0.0.2		
	Execute q	Execute qsa script from a bash prompt.			
	•	Scenar	io 3:		
		0	Stream with tos= $0x00$ and throughput = 0.5 Mbit/s towards $21.0.0.2$		
		0	Stream with tos= $0xa8$ and throughput = $0.5Mbit/s$ towards $21.0.0.2$		
		0	Stream with tos= $0xb0$ and throughput = $0.5Mbit/s$ towards $21.0.0.2$		
		0	Stream with tos= $0xb8$ and throughput = $0.5Mbit/s$ towards $21.0.0.2$		
	-	Scenar	io 4:		
		0	Stream with tos= $0x00$ and throughput = $5Mbit/s$ towards $21.0.0.2$		
		0	Stream with tos= $0xa8$ and throughput = $5Mbit/s$ towards $21.0.0.2$		
		0	Stream with tos= $0xb0$ and throughput = $5Mbit/s$ towards $21.0.0.2$		
		0	Stream with tos= $0xb8$ and throughput = $5Mbit/s$ towards $21.0.0.2$		
	•	Scenar	io 5:		
		0	Stream with tos= $0x00$ and throughput = 5 Mbit/s towards 21.0.0.2		
		0	Stream with tos= $0xb0$ and throughput = $5Mbit/s$ towards $21.0.0.2$		
		0	Stream with tos= $0xb8$ and throughput = $5Mbit/s$ towards $21.0.0.2$		
	-	Scenar	io 6:		
		0	Stream with tos= $0x00$ and throughput = 5 Mbit/s towards 21.0.0.2		
		0	Stream with tos=0xa8and throughput = 5Mbit/s towards 21.0.0.2		

• Scenario 7:

- \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 42.0.0.2
- \circ Stream with tos=0xa8 and throughput = 0.5Mbit/s towards 42.0.0.2
- Stream with tos=0xb0 and throughput = 0.5Mbit/s towards 42.0.0.2
- \circ Stream with tos=0xb8 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 8:
 - Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
 - Stream with tos=0xa8 and throughput = 7Mbit/s towards 42.0.0.2
 - Stream with tos=0xb0 and throughput = 7Mbit/s towards 42.0.0.2
 - Stream with tos=0xb8 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 9:
 - Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
 - Stream with tos=0xb0 and throughput = 7Mbit/s towards 42.0.0.2
 - Stream with tos=0xb8 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 10:
 - Stream with tos=0x00 and throughput = 7Mbit/s towards 42.0.0.2
 - Stream with tos=0xa8 and throughput = 7Mbit/s towards 42.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - o Stream with tos=0xa8 is received with a throughput of 1Mbit/s
 - o Stream with tos=0xb0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xb8 is received with a throughput of 1Mbit/s
- Scenario 5:
 - o Received traffic has a throughput of 5Mbit/s
 - Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - Stream with tos=0xb0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xb8 is received with a throughput of 1Mbit/s
- Scenario 6:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s

- o Stream with tos=0xa8 is received with a throughput of 1Mbit/s
- Scenario 7: No drop.
- Scenario 8:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0xa8 is received with a throughput of 1Mbit/s
 - Stream with tos=0xb0 is received with a throughput of 1Mbit/s
 - o Stream with tos=0xb8 is received with a throughput of 1Mbit/s

Scenario 9:

- o Received traffic has a throughput of 7Mbit/s
- o Stream with tos=0x00 is received with a throughput of 5Mbit/s
- o Stream with tos=0xb0 is received with a throughput of 1Mbit/s
- Stream with tos=0xb8 is received with a throughput of 1Mbit/s
- Scenario 10:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - o Stream with tos=0xa8 is received with a throughput of 1Mbit/s

Execution date : 15/09/04

- Scenario 1: Cancelled.
- Scenario 2: No drop.
- Scenario 3: Cancelled.
- Scenario 4: Cancelled
- Scenario 5: Cancelled
- Scenario 6: Cancelled
- Scenario 7: No drop.
- Scenario 8:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0xa8 is received with a throughput of 1Mbit/s
 - Stream with tos=0xb0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xb8 is received with a throughput of 1Mbit/s
- Scenario 9:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - o Stream with tos=0xb0 is received with a throughput of 1Mbit/s
 - o Stream with tos=0xb8 is received with a throughput of 1Mbit/s

- Scenario 10:
 - o Received traffic has a throughput of 7Mbit/s
 - Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - o Stream with tos=0xa8 is received with a throughput of 1Mbit/s

Failure level Remarks	: None. : None.			
Test Reference	: TB_P1_FUNCT/BWMA/5			
Test Purpose	: Verify bandwidth management configuration in MESCAL43.			
<i>Procedure</i> : Log to MESCAL43, Configure Smartbit to generate traffic as follows:				
	 Scenario 1: 			
	• Stream with tos= $0x00$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x88$ and throughput = $5Mbit/s$ towards $61.0.0.2$			
	• Stream with tos= $0x90$ and throughput = $5Mbit/s$ towards $61.0.0.2$			
	• Stream with tos= $0x98$ and throughput = $5Mbit/s$ towards $61.0.0.2$			
	Execute qsa script from a bash prompt.			
	 Scenario 2: 			
	• Stream with tos= $0x00$ and throughput = 0.5 Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x88$ and throughput = 0.5 Mbit/s towards 61.0.0.2			
	• Stream with tos=0x90 and throughput = 0.5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x98$ and throughput = 0.5 Mbit/s towards 61.0.0.2			
	 Scenario 3: 			
	• Stream with tos=0x00 and throughput = 10Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x88$ and throughput = $5Mbit/s$ towards $61.0.0.2$			
	• Stream with tos= $0x90$ and throughput = 5Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x98$ and throughput = $5Mbit/s$ towards $61.0.0.2$			
	• Scenario 4:			
	• Stream with tos=0x00 and throughput = 10Mbit/s towards 61.0.0.2			
	• Stream with tos= $0x90$ and throughput = $5Mbit/s$ towards $61.0.0.2$			
	• Stream with tos= $0x98$ and throughput = $5Mbit/s$ towards $61.0.0.2$			
	 Scenario 5: 			
	• Stream with tos=0x00 and throughput = 10Mbit/s towards 61.0.0.2			

• Stream with tos=0x88 and throughput = 5Mbit/s towards 61.0.0.2 Check with Smartbit statistics if there is traffic drops.

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3:
 - o Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x88 is received with a throughput of 3Mbit/s
 - Stream with tos=0x90 is received with a throughput of 3Mbit/s
 - Stream with tos=0x98 is received with a throughput of 3Mbit/s
- Scenario 4:
 - o Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x90 is received with a throughput of 3Mbit/s
 - o Stream with tos=0x98 is received with a throughput of 3Mbit/s
- Scenario 5:
 - Received traffic has a throughput of 10Mbit/s
 - Stream with tos=0x00 is received with a throughput of 7Mbit/s
 - Stream with tos=0x88 is received with a throughput of 3Mbit/s

Execution date : 15/09/04

- *Result* : Obtained results are:
 - Scenario 1: No drop.
 - Scenario 2: No drop.
 - Scenario 3:
 - Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 1Mbit/s
 - Stream with tos=0x88 is received with a throughput of 3Mbit/s
 - Stream with tos=0x90 is received with a throughput of 3Mbit/s
 - Stream with tos=0x98 is received with a throughput of 3Mbit/s
 - Scenario 4:
 - o Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x90 is received with a throughput of 3Mbit/s
 - Stream with tos=0x98 is received with a throughput of 3Mbit/s
 - Scenario 5:
 - Received traffic has a throughput of 10Mbit/s

- Stream with tos=0x00 is received with a throughput of 7Mbit/s
- Stream with tos=0x88 is received with a throughput of 3Mbit/s

Failure level Remarks	: None. : None.
Test Reference	: TB_P1_FUNCT/BWMA/6
Test Purpose	: Verify bandwidth management configuration in MESCAL41.
Procedure	: Log to MESCAL41, Configure Smartbit to generate traffic as follows:
	 Scenario 1:
	• Stream with tos= $0x00$ and throughput = 7Mbit/s towards 31.0.0.2
	• Stream with tos= $0x88$ and throughput = 7Mbit/s towards 31.0.0.2
	• Stream with tos= $0x90$ and throughput = 7Mbit/s towards $31.0.0.2$
	• Stream with tos= $0x98$ and throughput = 7Mbit/s towards $31.0.0.2$
	 Scenario 2:
	• Stream with tos= $0x00$ and throughput = 7Mbit/s towards 61.0.0.2
	• Stream with tos= $0x88$ and throughput = 7Mbit/s towards 61.0.0.2
	• Stream with tos= $0x90$ and throughput = 7Mbit/s towards 61.0.0.2
	• Stream with tos= $0x98$ and throughput = 7Mbit/s towards 61.0.0.2
	Execute qsa script from a bash prompt.
	 Scenario 3:
	• Stream with tos= $0x00$ and throughput = 0.5 Mbit/s towards $31.0.0.2$
	• Stream with tos=0x88 and throughput = 0.5Mbit/s towards 31.0.0.2
	• Stream with tos= $0x90$ and throughput = 0.5 Mbit/s towards $31.0.0.2$
	• Stream with tos= $0x98$ and throughput = 0.5 Mbit/s towards $31.0.0.2$
	 Scenario 4:
	• Stream with tos= $0x00$ and throughput = 10 Mbit/s towards $31.0.0.2$
	• Stream with tos= $0x88$ and throughput = 7Mbit/s towards 31.0.0.2
	• Stream with tos= $0x90$ and throughput = 7Mbit/s towards $31.0.0.2$
	• Stream with tos= $0x98$ and throughput = 7Mbit/s towards $31.0.0.2$
	 Scenario 5:
	• Stream with tos= $0x00$ and throughput = 10 Mbit/s towards $31.0.0.2$
	• Stream with tos= $0x90$ and throughput = 7Mbit/s towards $31.0.0.2$
	• Stream with tos= $0x98$ and throughput = $7Mbit/s$ towards $31.0.0.2$

- Scenario 6:
 - \circ Stream with tos=0x00 and throughput = 10Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 7:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 61.0.0.2
 - Stream with tos=0x88 and throughput = 0.5Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0x90 and throughput = 0.5Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0x98 and throughput = 0.5Mbit/s towards 61.0.0.2
- Scenario 8:
 - \circ Stream with tos=0x00 and throughput = 12Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0x88 and throughput = 7Mbit/s towards 61.0.0.2
 - Stream with tos=0x90 and throughput = 7Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0x98 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 9:
 - Stream with tos=0x00 and throughput = 12Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0x90 and throughput = 7Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0x98 and throughput = 7Mbit/s towards 61.0.0.2
- Scenario 10:
 - Stream with tos=0x00 and throughput = 12Mbit/s towards 61.0.0.2
 - \circ Stream with tos=0x88 and throughput = 7Mbit/s towards 61.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x88 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 4:
 - Received traffic has a throughput of 7Mbit/s

- Stream with tos=0x00 is received with a throughput of 5Mbit/s
- Stream with tos=0x90 is received with a throughput of 1Mbit/s
- Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 5:
 - Received traffic has a throughput of 7Mbit/s
 - Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8:
 - Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 1Mbit/s
 - Stream with tos=0x88 is received with a throughput of 3Mbit/s
 - Stream with tos=0x90 is received with a throughput of 3Mbit/s
 - Stream with tos=0x98 is received with a throughput of 3Mbit/s
- Scenario 9:
 - o Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x90 is received with a throughput of 3Mbit/s
 - o Stream with tos=0x98 is received with a throughput of 3Mbit/s
- Scenario 10:
 - Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 7Mbit/s
 - o Stream with tos=0x88 is received with a throughput of 3Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s
 - Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 4:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 5Mbit/s

- Stream with tos=0x90 is received with a throughput of 1Mbit/s
- Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 5:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s.
- Scenario 6: No drop.
- Scenario 7: No drop.
- Scenario 8:
 - o Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x88 is received with a throughput of 3Mbit/s
 - o Stream with tos=0x90 is received with a throughput of 3Mbit/s
 - Stream with tos=0x98 is received with a throughput of 3Mbit/s
- Scenario 9:
 - Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x90 is received with a throughput of 3Mbit/s
 - o Stream with tos=0x98 is received with a throughput of 3Mbit/s
- Scenario 10:
 - o Received traffic has a throughput of 10Mbit/s
 - Stream with tos=0x00 is received with a throughput of 7Mbit/s
 - o Stream with tos=0x88 is received with a throughput of 3Mbit/s

Failure level: None.Remarks: None.

Test Reference : TB_P1_FUNCT/BWMA/7

Test Purpose : Verify bandwidth management configuration in MESCAL42.

Procedure : Log to MESCAL42, Configure Smartbit to generate traffic as follows:

- Scenario 1:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 31.0.0.2
 - Stream with tos=0x88 and throughput = 9Mbit/s towards 31.0.0.2
 - Stream with tos=0x90 and throughput = 9Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x98 and throughput = 9Mbit/s towards 31.0.0.2
- Scenario 2:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 21.0.0.2
 - \circ Stream with tos=0x88 and throughput = 9Mbit/s towards 21.0.0.2

- \circ Stream with tos=0x90 and throughput = 9Mbit/s towards 21.0.0.2
- Stream with tos=0x98 and throughput = 9Mbit/s towards 21.0.0.2
- Scenario 3:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 51.0.0.2
 - Stream with tos=0x88 and throughput = 9Mbit/s towards 51.0.0.2
 - Stream with tos=0x90 and throughput = 9Mbit/s towards 51.0.0.2
 - Stream with tos=0x98 and throughput = 9Mbit/s towards 51.0.0.2

Execute qsa script from a bash prompt.

- Scenario 4:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x88 and throughput = 0.5Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x90 and throughput = 0.5Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x98 and throughput = 0.5Mbit/s towards 31.0.0.2
- Scenario 5:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 31.0.0.2
 - Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2
 - Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x98 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 6:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x90 and throughput = 7Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x98 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 7:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 31.0.0.2
 - \circ Stream with tos=0x88 and throughput = 7Mbit/s towards 31.0.0.2
- Scenario 8:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 21.0.0.2
 - Stream with tos=0x88 and throughput = 0.5Mbit/s towards 21.0.0.2
 - \circ Stream with tos=0x90 and throughput = 0.5Mbit/s towards 21.0.0.2
 - \circ Stream with tos=0x98 and throughput = 0.5Mbit/s towards 21.0.0.2
- Scenario 9:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 21.0.0.2

- \circ Stream with tos=0x88 and throughput = 7Mbit/s towards 21.0.0.2
- Stream with tos=0x90 and throughput = 7Mbit/s towards 21.0.0.2
- Stream with tos=0x98 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 10:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 21.0.0.2
 - Stream with tos=0x90 and throughput = 7Mbit/s towards 21.0.0.2
 - \circ Stream with tos=0x98 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 11:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 21.0.0.2
 - \circ Stream with tos=0x88 and throughput = 7Mbit/s towards 21.0.0.2
- Scenario 12:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x88 and throughput = 0.5Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x90 and throughput = 0.5Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x98 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 13:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 51.0.0.2
 - Stream with tos=0x88 and throughput = 7Mbit/s towards 51.0.0.2
 - Stream with tos=0x90 and throughput = 7Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x98 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 14:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 51.0.0.2
 - Stream with tos=0x90 and throughput = 7Mbit/s towards 51.0.0.2
 - Stream with tos=0x98 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 15:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 51.0.0.2
 - Stream with tos=0x88 and throughput = 7Mbit/s towards 51.0.0.2

Expected result : Following results must be obtained:

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5:
 - Received traffic has a throughput of 7Mbit/s

- Stream with tos=0x00 is received with a throughput of 4Mbit/s
- Stream with tos=0x88 is received with a throughput of 1Mbit/s
- Stream with tos=0x90 is received with a throughput of 1Mbit/s
- Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 6:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 7:
 - o Received traffic has a throughput of 7Mbit/s
 - Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s
- Scenario 8: No drop.
- Scenario 9:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s
 - Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 10:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - o Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 11:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s
- Scenario 12: No drop.
- Scenario 13:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s
 - Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 14:

- o Received traffic has a throughput of 7Mbit/s
- Stream with tos=0x00 is received with a throughput of 5Mbit/s
- Stream with tos=0x90 is received with a throughput of 1Mbit/s
- Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 15:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - o Stream with tos=0x88 is received with a throughput of 1Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x88 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 6:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - o Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 7:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s
- Scenario 8: No drop.
- Scenario 9:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s
 - Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 10:

- o Received traffic has a throughput of 7Mbit/s
- Stream with tos=0x00 is received with a throughput of 5Mbit/s
- Stream with tos=0x90 is received with a throughput of 1Mbit/s
- Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 11:
 - Received traffic has a throughput of 7Mbit/s
 - Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s
- Scenario 12: No drop.
- Scenario 13:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x88 is received with a throughput of 1Mbit/s
 - Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 14:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - Stream with tos=0x90 is received with a throughput of 1Mbit/s
 - Stream with tos=0x98 is received with a throughput of 1Mbit/s
- Scenario 15:
 - Received traffic has a throughput of 7Mbit/s
 - Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - o Stream with tos=0x88 is received with a throughput of 1Mbit/s

Failure level: None.Remarks: None.

Test Reference	: TB_P1_FUNCT/BWMA/8	
Test Purpose	: Verify bandwidth management configuration in MESCAL21.	
Procedure	: Log to MESCAL21, Configure Smartbit to generate traffic as follows:	
	 Scenario 1: 	
	• Stream with tos= $0x00$ and throughput = 9 Mbit/s towards 11.0.0.2	
	• Stream with tos= $0x48$ and throughput = 9 Mbit/s towards 11.0.0.2	
	• Stream with tos= $0x50$ and throughput = 9 Mbit/s towards 11.0.0.2	
	• Stream with tos= $0x58$ and throughput = 9 Mbit/s towards 11.0.0.2	
	 Scenario 2: 	

• Stream with tos=0x00 and throughput = 9Mbit/s towards 42.0.0.2

- \circ Stream with tos=0x48 and throughput = 9Mbit/s towards 42.0.0.2
- Stream with tos=0x50 and throughput = 9Mbit/s towards 42.0.0.2
- Stream with tos=0x58 and throughput = 9Mbit/s towards 42.0.0.2
- Scenario 3:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 51.0.0.2
 - Stream with tos=0x48 and throughput = 9Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x50 and throughput = 9Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x58 and throughput = 9Mbit/s towards 51.0.0.2

Execute qsa script from a bash prompt.

- Scenario 4:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x48 and throughput = 0.5Mbit/s towards 11.0.0.2
 - Stream with tos=0x50 and throughput = 0.5Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x58 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 5:
 - \circ Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x48 and throughput = 7Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x50 and throughput = 7Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x58 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 6:
 - \circ Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x50 and throughput = 7Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x58 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 7:
 - \circ Stream with tos=0x00 and throughput = 7Mbit/s towards 11.0.0.2
 - Stream with tos=0x48 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 8:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 42.0.0.2
 - \circ Stream with tos=0x48 and throughput = 0.5Mbit/s towards 42.0.0.2
 - \circ Stream with tos=0x50 and throughput = 0.5Mbit/s towards 42.0.0.2
 - \circ Stream with tos=0x58 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 9:

- \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 42.0.0.2
- Stream with tos=0x48 and throughput = 7Mbit/s towards 42.0.0.2
- Stream with tos=0x50 and throughput = 7Mbit/s towards 42.0.0.2
- Stream with tos=0x58 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 10:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 42.0.0.2
 - Stream with tos=0x50 and throughput = 7Mbit/s towards 42.0.0.2
 - \circ Stream with tos=0x58 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 11:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 42.0.0.2
 - Stream with tos=0x48 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 12:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x48 and throughput = 0.5Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x50 and throughput = 0.5Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x58 and throughput = 0.5Mbit/s towards 51.0.0.2
- Scenario 13:
 - Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2
 - Stream with tos=0x48 and throughput = 7Mbit/s towards 51.0.0.2
 - Stream with tos=0x50 and throughput = 7Mbit/s towards 51.0.0.2
 - \circ Stream with tos=0x58 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 14:
 - \circ Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2
 - Stream with tos=0x50 and throughput = 7Mbit/s towards 51.0.0.2
 - Stream with tos=0x58 and throughput = 7Mbit/s towards 51.0.0.2
- Scenario 15:
 - Stream with tos=0x00 and throughput = 7Mbit/s towards 51.0.0.2
 - Stream with tos=0x48 and throughput = 7Mbit/s towards 51.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5:

- o Received traffic has a throughput of 5Mbit/s
- Stream with tos=0x00 is received with a throughput of 2Mbit/s
- o Stream with tos=0x48 is received with a throughput of 1Mbit/s
- Stream with tos=0x50 is received with a throughput of 1Mbit/s
- o Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 6:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 7:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x48 is received with a throughput of 1Mbit/s
- Scenario 8: No drop.
- Scenario 9:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 10:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 11:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s
- Scenario 12: No drop.
- Scenario 13:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s

- Scenario 14:
 - o Received traffic has a throughput of 5Mbit/s
 - Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 15:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: Cancelled.
- Scenario 4: No drop.
- Scenario 5:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 6:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 7:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0x48 is received with a throughput of 1Mbit/s
- Scenario 8: No drop.
- Scenario 9:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - \circ Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x58 is received with a throughput of 1Mbit/s

- Scenario 10:
 - o Received traffic has a throughput of 7Mbit/s
 - Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
 - Scenario 11:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - Scenario 12: Cancelled.
 - Scenario 13: Cancelled
 - Scenario 14: Cancelled
 - Scenario 15: Cancelled

Failure level : None.

Remarks : None.

Test Reference : TB_P1_FUNCT/BWMA/9

Test Purpose : Verify bandwidth management configuration in MESCAL31.

Procedure : Log to MESCAL31, Configure Smartbit to generate traffic as follows:

- Scenario 1:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x68 and throughput = 9Mbit/s towards 11.0.0.2
 - Stream with tos=0x70 and throughput = 9Mbit/s towards 11.0.0.2
 - Stream with tos=0x78 and throughput = 9Mbit/s towards 11.0.0.2
- Scenario 2:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 42.0.0.2
 - Stream with tos=0x68 and throughput = 9Mbit/s towards 42.0.0.2
 - Stream with tos=0x70 and throughput = 9Mbit/s towards 42.0.0.2
 - Stream with tos=0x78 and throughput = 9Mbit/s towards 42.0.0.2
- Scenario 3:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 41.0.0.2
 - Stream with tos=0x68 and throughput = 9Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0x70 and throughput = 9Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0x78 and throughput = 9Mbit/s towards 41.0.0.2

Execute qsa script from a bash prompt.

- Scenario 4:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 11.0.0.2

- \circ Stream with tos=0x68 and throughput = 0.5Mbit/s towards 11.0.0.2
- \circ Stream with tos=0x70 and throughput = 0.5Mbit/s towards 11.0.0.2
- \circ Stream with tos=0x78 and throughput = 0.5Mbit/s towards 11.0.0.2
- Scenario 5:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x68 and throughput = 7Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x70 and throughput = 7Mbit/s towards 11.0.0.2
 - Stream with tos=0x78 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 6:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 11.0.0.2
 - Stream with tos=0x70 and throughput = 7Mbit/s towards 11.0.0.2
 - Stream with tos=0x78 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 7:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 11.0.0.2
 - \circ Stream with tos=0x68 and throughput = 7Mbit/s towards 11.0.0.2
- Scenario 8:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 42.0.0.2
 - \circ Stream with tos=0x68 and throughput = 0.5Mbit/s towards 42.0.0.2
 - Stream with tos=0x70 and throughput = 0.5Mbit/s towards 42.0.0.2
 - Stream with tos=0x78 and throughput = 0.5Mbit/s towards 42.0.0.2
- Scenario 9:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 42.0.0.2
 - \circ Stream with tos=0x68 and throughput = 7Mbit/s towards 42.0.0.2
 - Stream with tos=0x70 and throughput = 7Mbit/s towards 42.0.0.2
 - Stream with tos=0x78 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 10:
 - \circ Stream with tos=0x00 and throughput = 9Mbit/s towards 42.0.0.2
 - Stream with tos=0x70 and throughput = 7Mbit/s towards 42.0.0.2
 - Stream with tos=0x78 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 11:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 42.0.0.2
 - Stream with tos=0x68 and throughput = 7Mbit/s towards 42.0.0.2
- Scenario 12:

- \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 41.0.0.2
- \circ Stream with tos=0x68 and throughput = 0.5Mbit/s towards 41.0.0.2
- Stream with tos=0x70 and throughput = 0.5Mbit/s towards 41.0.0.2
- \circ Stream with tos=0x78 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 13:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 41.0.0.2
 - Stream with tos=0x68 and throughput = 7Mbit/s towards 41.0.0.2
 - Stream with tos=0x70 and throughput = 7Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0x78 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 14:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0x70 and throughput = 7Mbit/s towards 41.0.0.2
 - Stream with tos=0x78 and throughput = 7Mbit/s towards 41.0.0.2
- Scenario 15:
 - Stream with tos=0x00 and throughput = 9Mbit/s towards 41.0.0.2
 - Stream with tos=0x68 and throughput = 7Mbit/s towards 41.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 6:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 7:
 - Received traffic has a throughput of 5Mbit/s

- o Stream with tos=0x00 is received with a throughput of 4Mbit/s
- Stream with tos=0x48 is received with a throughput of 1Mbit/s
- Scenario 8: No drop.
- Scenario 9:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 10:
 - o Received traffic has a throughput of 7Mbit/s
 - Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 11:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - Stream with tos=0x48 is received with a throughput of 1Mbit/s
- Scenario 12: No drop.
- Scenario 13:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 14:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 15:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s

Result : Obtained results are:

• Scenario 1: No drop.

- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4: No drop.
- Scenario 5:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 6:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 7:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s
- Scenario 8: No drop.
- Scenario 9:
 - Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0x48 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - o Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 10:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 5Mbit/s
 - o Stream with tos=0x50 is received with a throughput of 1Mbit/s
 - Stream with tos=0x58 is received with a throughput of 1Mbit/s
- Scenario 11:
 - o Received traffic has a throughput of 7Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 6Mbit/s
 - Stream with tos=0x48 is received with a throughput of 1Mbit/s
- Scenario 12: No drop.
- Scenario 13:
 - Received traffic has a throughput of 7Mbit/s

		0	Stream with tos=0x00 is received with a throughput of 4Mbit/s
		0	Stream with tos=0x48 is received with a throughput of 1Mbit/s
		0	Stream with tos=0x50 is received with a throughput of 1Mbit/s
		0	Stream with tos=0x58 is received with a throughput of 1Mbit/s
	•	Scenar	io 14:
		0	Received traffic has a throughput of 7Mbit/s
		0	Stream with tos=0x00 is received with a throughput of 5Mbit/s
		0	Stream with tos=0x50 is received with a throughput of 1Mbit/s
		0	Stream with tos=0x58 is received with a throughput of 1Mbit/s
	•	Scenar	io 15:
		0	Received traffic has a throughput of 7Mbit/s
		0	Stream with tos=0x00 is received with a throughput of 6Mbit/s
		0	Stream with tos=0x48 is received with a throughput of 1Mbit/s
Failure level Remarks	: None : None		
Test Reference	: TB_P1_I	FUNCT/	BWMA/10
Test Purpose	: Verify ba	ndwidth	management configuration in MESCAL61.
Procedure	: Log to MESCAL61, Configure Smartbit to generate traffic as follows:		
	-	Scenar	io 1:
		0	Stream with tos= $0x00$ and throughput = 7Mbit/s towards 71.0.0.2
		0	Stream with tos=0xc8 and throughput = 7Mbit/s towards 71.0.0.2
		0	Stream with tos=0xd0and throughput = 7Mbit/s towards 71.0.0.2
		0	Stream with tos=0xd8 and throughput = 7Mbit/s towards 71.0.0.2
	•	Scenar	io 2:
		0	Stream with tos= $0x00$ and throughput = 7Mbit/s towards 81.0.0.2
		0	Stream with tos=0xc8 and throughput = 7Mbit/s towards 81.0.0.2
		0	Stream with tos=0xd0and throughput = 7Mbit/s towards 81.0.0.2
		0	Stream with tos=0xd8 and throughput = 7Mbit/s towards 81.0.0.2
	Execute q	sa script	from a bash prompt.
	•	Scenar	io 3:
		0	Stream with tos= $0x00$ and throughput = 0.5 Mbit/s towards $71.0.0.2$
		0	Stream with tos= $0xc8$ and throughput = $0.5Mbit/s$ towards $71.0.0.2$
		0	Stream with tos= $0xd0and$ throughput = $0.5Mbit/s$ towards $71.0.0.2$
		0	Stream with tos= $0xd8$ and throughput = $0.5Mbit/s$ towards $71.0.0.2$

- Scenario 4:
 - \circ Stream with tos=0x00 and throughput = 7Mbit/s towards 71.0.0.2
 - Stream with tos=0xc8 and throughput = 5Mbit/s towards 71.0.0.2
 - Stream with tos= $0 \times d0$ and throughput = 5 M bit/s towards 71.0.0.2
 - Stream with tos=0xd8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 5:
 - \circ Stream with tos=0x00 and throughput = 7Mbit/s towards 71.0.0.2
 - Stream with tos= $0 \times d0$ and throughput = 5 M bit/s towards 71.0.0.2
 - \circ Stream with tos=0xd8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 6:
 - \circ Stream with tos=0x00 and throughput = 7Mbit/s towards 71.0.0.2
 - Stream with tos=0xc8 and throughput = 5Mbit/s towards 71.0.0.2
- Scenario 7:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 81.0.0.2
 - \circ Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 81.0.0.2
 - \circ Stream with tos=0xd0and throughput = 0.5Mbit/s towards 81.0.0.2
 - \circ Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 81.0.0.2
- Scenario 8:
 - Stream with tos=0x00 and throughput = 7Mbit/s towards 81.0.0.2
 - Stream with tos=0xc8 and throughput = 5Mbit/s towards 81.0.0.2
 - Stream with tos=0xd0and throughput = 5Mbit/s towards 81.0.0.2
 - \circ Stream with tos=0xd8 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 9:
 - \circ Stream with tos=0x00 and throughput = 7Mbit/s towards 81.0.0.2
 - Stream with tos= $0 \times d0$ and throughput = 5 M bit/s towards 81.0.0.2
 - Stream with tos=0xd8 and throughput = 5Mbit/s towards 81.0.0.2
- Scenario 10:
 - \circ Stream with tos=0x00 and throughput = 7Mbit/s towards 81.0.0.2
 - Stream with tos=0xc8 and throughput = 5Mbit/s towards 81.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4:

- o Received traffic has a throughput of 5Mbit/s
- Stream with tos=0x00 is received with a throughput of 2Mbit/s
- o Stream with tos=0xc8 is received with a throughput of 1Mbit/s
- Stream with tos=0xd0 is received with a throughput of 1Mbit/s
- o Stream with tos=0xd8 is received with a throughput of 1Mbit/s
- Scenario 5:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - o Stream with tos=0xd0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xd8 is received with a throughput of 1Mbit/s
- Scenario 6:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0xc8 is received with a throughput of 1Mbit/s
- Scenario 7: No drop.
- Scenario 8:
 - o Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - Stream with tos=0xc8 is received with a throughput of 1Mbit/s
 - Stream with tos=0xd0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xd8 is received with a throughput of 1Mbit/s
- Scenario 9:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - o Stream with tos=0xd0 is received with a throughput of 1Mbit/s
 - Stream with tos=0xd8 is received with a throughput of 1Mbit/s
- Scenario 10:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0xc8 is received with a throughput of 1Mbit/s.

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4:
 - Received traffic has a throughput of 5Mbit/s

- Stream with tos=0x00 is received with a throughput of 2Mbit/s
- Stream with tos=0xc8 is received with a throughput of 1Mbit/s
- Stream with tos=0xd0 is received with a throughput of 1Mbit/s
- Stream with tos=0xd8 is received with a throughput of 1Mbit/s
- Scenario 5:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - o Stream with tos=0xd0 is received with a throughput of 1Mbit/s
 - o Stream with tos=0xd8 is received with a throughput of 1Mbit/s
- Scenario 6:
 - o Received traffic has a throughput of 5Mbit/s
 - Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0xc8 is received with a throughput of 1Mbit/s
- Scenario 7: No drop.
- Scenario 8:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 2Mbit/s
 - o Stream with tos=0xc8 is received with a throughput of 1Mbit/s
 - Stream with tos=0xd0 is received with a throughput of 1Mbit/s
 - o Stream with tos=0xd8 is received with a throughput of 1Mbit/s
- Scenario 9:
 - Received traffic has a throughput of 5Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 3Mbit/s
 - o Stream with tos=0xd0 is received with a throughput of 1Mbit/s
 - o Stream with tos=0xd8 is received with a throughput of 1Mbit/s
- Scenario 10:
 - Received traffic has a throughput of 5Mbit/s
 - Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0xc8 is received with a throughput of 1Mbit/s
- Failure level: NoneRemarks: None

Test Reference : TB_P1_FUNCT/BWMA/11

Test Purpose : Verify bandwidth management configuration in MESCAL61.

Procedure : Log to MESCAL61, Configure Smartbit to generate traffic as follows:

- Scenario 1:
 - \circ Stream with tos=0x00 and throughput = 12Mbit/s towards 41.0.0.2

- \circ Stream with tos=0xc8 and throughput = 5Mbit/s towards 41.0.0.2
- \circ Stream with tos=0xd0and throughput = 5Mbit/s towards 41.0.0.2
- \circ Stream with tos=0xd8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 2:
 - \circ Stream with tos=0x00 and throughput = 12Mbit/s towards 43.0.0.2
 - Stream with tos=0xc8 and throughput = 5Mbit/s towards 43.0.0.2
 - \circ Stream with tos=0xd0and throughput = 5Mbit/s towards 43.0.0.2
 - \circ Stream with tos=0xd8 and throughput = 5Mbit/s towards 43.0.0.2

Execute qsa script from a bash prompt.

- Scenario 3:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0xd0and throughput = 0.5Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 41.0.0.2
- Scenario 4:
 - Stream with tos=0x00 and throughput = 12Mbit/s towards 41.0.0.2
 - Stream with tos=0xc8 and throughput = 5Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0xd0and throughput = 5Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0xd8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 5:
 - Stream with tos=0x00 and throughput = 12Mbit/s towards 41.0.0.2
 - Stream with tos=0xd0 and throughput = 5Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0xd8 and throughput = 5Mbit/s towards 41.0.0.2
- Scenario 6:
 - \circ Stream with tos=0x00 and throughput = 12Mbit/s towards 41.0.0.2
 - \circ Stream with tos=0xc8 and throughput = 8Mbit/s towards 41.0.0.2
- Scenario 7:
 - \circ Stream with tos=0x00 and throughput = 0.5Mbit/s towards 43.0.0.2
 - \circ Stream with tos=0xc8 and throughput = 0.5Mbit/s towards 43.0.0.2
 - o Stream with tos=0xd0and throughput = 0.5Mbit/s towards 43.0.0.2

- \circ Stream with tos=0xd8 and throughput = 0.5Mbit/s towards 43.0.0.2
- Scenario 8:
 - o Stream with tos=0x00 and throughput = 12Mbit/s towards 43.0.0.2
 - Stream with tos=0xc8 and throughput = 5Mbit/s towards 43.0.0.2
 - Stream with tos=0xd0and throughput = 5Mbit/s towards 43.0.0.2
 - \circ Stream with tos=0xd8 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 9:
 - o Stream with tos=0x00 and throughput = 12Mbit/s towards 43.0.0.2
 - \circ Stream with tos=0xd0and throughput = 5Mbit/s towards 43.0.0.2
 - Stream with tos=0xd8 and throughput = 5Mbit/s towards 43.0.0.2
- Scenario 10:
 - \circ Stream with tos=0x00 and throughput = 12Mbit/s towards 43.0.0.2
 - Stream with tos=0xc8 and throughput = 5Mbit/s towards 43.0.0.2

Expected result :

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4:
 - Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 1Mbit/s
 - Stream with tos=0xc8 is received with a throughput of 3Mbit/s
 - Stream with tos=0xd0 is received with a throughput of 3Mbit/s
 - Stream with tos=0xd8 is received with a throughput of 3Mbit/s
- Scenario 5:
 - Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - Stream with tos=0xd0 is received with a throughput of 3Mbit/s
 - Stream with tos=0xd8 is received with a throughput of 3Mbit/s
- Scenario 6:
 - Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 7Mbit/s
 - o Stream with tos=0xc8 is received with a throughput of 3Mbit/s
- Scenario 7: No drop.
- Scenario 8:

- o Received traffic has a throughput of 10Mbit/s
- Stream with tos=0x00 is received with a throughput of 1Mbit/s
- o Stream with tos=0xc8 is received with a throughput of 3Mbit/s
- Stream with tos=0xd0 is received with a throughput of 3Mbit/s
- o Stream with tos=0xd8 is received with a throughput of 3Mbit/s
- Scenario 9:
 - o Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0xd0 is received with a throughput of 3Mbit/s
 - Stream with tos=0xd8 is received with a throughput of 3Mbit/s
- Scenario 10:
 - Received traffic has a throughput of 10Mbit/s
 - Stream with tos=0x00 is received with a throughput of 7Mbit/s
 - o Stream with tos=0xc8 is received with a throughput of 3Mbit/s

- Scenario 1: No drop.
- Scenario 2: No drop.
- Scenario 3: No drop.
- Scenario 4:
 - o Received traffic has a throughput of 10Mbit/s
 - Stream with tos=0x00 is received with a throughput of 1Mbit/s
 - o Stream with tos=0xc8 is received with a throughput of 3Mbit/s
 - o Stream with tos=0xd0 is received with a throughput of 3Mbit/s
 - o Stream with tos=0xd8 is received with a throughput of 3Mbit/s
- Scenario 5:
 - Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0xd0 is received with a throughput of 3Mbit/s
 - Stream with tos=0xd8 is received with a throughput of 3Mbit/s
- Scenario 6:
 - o Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 7Mbit/s
 - Stream with tos=0xc8 is received with a throughput of 3Mbit/s
- Scenario 7: No drop.
- Scenario 8:
 - Received traffic has a throughput of 10Mbit/s

- Stream with tos=0x00 is received with a throughput of 1Mbit/s
- Stream with tos=0xc8 is received with a throughput of 3Mbit/s
- Stream with tos=0xd0 is received with a throughput of 3Mbit/s
- Stream with tos=0xd8 is received with a throughput of 3Mbit/s
- Scenario 9:
 - Received traffic has a throughput of 10Mbit/s
 - o Stream with tos=0x00 is received with a throughput of 4Mbit/s
 - o Stream with tos=0xd0 is received with a throughput of 3Mbit/s
 - o Stream with tos=0xd8 is received with a throughput of 3Mbit/s
- Scenario 10:
 - Received traffic has a throughput of 10Mbit/s
 - Stream with tos=0x00 is received with a throughput of 7Mbit/s
 - Stream with tos=0xc8 is received with a throughput of 3Mbit/s

Failure level: None.Remarks: None.

10.2 Phase 2

The "null" value means no value.

10.2.1 TB_P2_FUNCT/CMES

Test Reference	: TB_P2_FUNCT/CMES/1
Test Purpose	: Verify the capability length.
Procedure	: Configure MESCAL11 to send the following QoS service capability to MESCAL21
	Group 1
	• Group 2

Expected result : The capability length must be set to 2.

Execution date : 22/09/04 *Result* : The capability length is set to 2.

⊞ Frame 10 (109 bytes on wire, 109 bytes captured) 🗄 Transmission Control Protocol, Src Port: 53966 (53966), Dst Port: bgp (179), Seq: 2217784803, Ack: 2247019363, Len: 43 Border Gateway Protocol 🗆 OPEN Message Marker: 16 bytes Length: 43 bytes Type: OPEN Message (1) Version: 4 My AS: 1 Hold time: 15 BGP identifier: 11.11.11.1 Optional parameters length: 14 bytes Optional parameters 🗆 Capabilities Advertisement (8 bytes) Parameter type: Capabilities (2) Parameter length: 6 bytes 🗆 Multiprotocol extensions capability (6 bytes) Capability code: Multiprotocol extensions (1) Capability length: 4 bytes 🗆 Capability value Address family identifier: IPv4 (1) Reserved: 1 byte Subsequent address family identifier: Unicast (1) 🗆 Capabilities Advertisement (6 bytes) Parameter type: Capabilities (2) Parameter length: 4 bytes Unknown capability (4 bytes) Capability code: Private use (129) Capability length: 2 bytes Capability value: Unknown

<	///
0000 00 50 04 48 c3 63 00 d0 b7 21 50 bd 08 00 4	5 00 .P.H.c!PE.
0010 00 5f e1 1e 40 00 01 06 94 6e 01 01 01 05 0	1 01@n
0020 01 06 d2 ce 00 b3 84 30 b5 e3 85 ee cb 63 8	0 180c
0030 16 d0 95 9e 00 00 01 01 08 0a 03 fa 54 fe 0	3 18
0040 d4 44 ff	f ff .D
0050 ff ff 00 2b 01 04 00 01 00 <u>0f 0b 0b 0b</u> 01 0	e 02 + <u></u>
0060 06 01 04 00 01 00 01 02 04 <mark>31 02 00 ff</mark>	····· ·

Failure level	: None
Remarks	: None

Test Reference	: TB_P2_FUNCT/CMES/2
Test Purpose	: Verify the QoS service capability field length.
Procedure	: Configure MESCAL11 to send the following QoS service capability to MESCAL21
	Group 1
	• Group 2

Expected result : The length of the QoS service capability must be 2 bytes.

Execution date	: 22/09/04
Result	: The length of the QoS service capability is 2 bytes.
Failure level	: None
Remarks	: None
Result Failure level Remarks	: The length of the QoS service capability is 2 bytes : None : None

Test Reference	: TB_P2_FUNCT/CMES/3
Test Purpose	: Verify that Group 1 QoS service capability is supported.
Procedure	: Configure MESCAL11 to send a QoS service capability G1 to MESCAL21.
Expected result	: QoS service capability field is two bytes. The first byte must be set to 0xFF.
Result	: The first byte is set to 0xFF.

Failure level Remarks	: None : None	
Test Reference Test Purpose Procedure	: TB_P2_FUNCT/CMES/4 : Verify that Group 2 QoS service c : Configure MESCAL11 to send	apability is supported. a QoS service capability G2 to MESCAL21.
Expected result	: QoS service capability field is two	bytes. The second byte is set to 0xFF.
Execution date Result	: 22/09/04 : The second byte is set to 0xFF	
 ➡ Frame 10 (109 by ➡ Ethernet II, Srot ➡ Internet Protocod ➡ Transmission Con ➡ Border Gateway f ➡ OPEN Message Marker: 1 Length: 4 Type: OPE Version: My AS: 1 Hold time ■ GP ident Optional ➡ Capab: Pa ■ Mu ■ Capab: Pa Un 	<pre>ites on wire, 109 bytes captured) :: 00:d0:b7:21:50:bd, Dst: 00:50:04:48:c3:63 il, Src Addr: 1.1.1.5 (1.1.1.5), Dst Addr: 1 trol Protocol, Src Port: 53966 (53966), Dst rotocol 6 bytes 3 bytes N Message (1) 4 : 15 ifier: 11.11.11.1 parameters length: 14 bytes parameters lities Advertisement (8 bytes) rameter length: 6 bytes ltiprotocol extensions capability (6 bytes) Capability code: Multiprotocol extensions Capability value Address family identifier: IPv4 (1) Reserved: 1 byte Subsequent address family identifier: U lities Advertisement (6 bytes) rameter type: Capabilities (2) rameter length: 4 bytes Capability value Address family identifier: IPv4 (1) Reserved: 1 byte Subsequent address family identifier: U lities Advertisement (6 bytes) rameter length: 4 bytes Capability code: Private use (129) Capability code: Private use (129) Capability value: Unknown Cap</pre>	.1.1.6 (1.1.1.6) Port: bgp (179), Seq: 2217784803, Ack: 2247019363, Len: 43 (1) nicast (1)
-1		11111

N		
0000 00 50 04 48 c3 63 00 d0	b7 21 50 bd 08 00 45 00	.P.H.c!PE.
0010 00 5f e1 1e 40 00 01 06	94 6e 01 01 01 05 01 01	@n
0020 01 06 d2 ce 00 b3 84 30	b5 e3 85 ee cb 63 80 18	0c
0030 16 d0 95 9e 00 00 01 01	08 0a 03 fa 54 fe 03 18	T
0040 d4 44 ff ff ff ff ff ff	ff ff ff ff ff ff ff ff ff	.D
0050 ff ff 00 2b 01 04 00 01	00 Of Ob <u>Ob Ob</u> 01 Oe 02	···· ⁺ ····· ··· <u>··</u> ···
0060 06 01 04 00 01 00 01 02	04 81 02 <u>00 ff</u>	••••••••• •••• <mark>••</mark>
<i>Failure level</i> : None		
<i>Remarks</i> : None		

Test Reference	: TB_P2_FUNCT/CMES/5
Test Purpose	: Validate the conformance of QoS information length.
Procedure	: Configure MESCAL11 to send BGP UPDATE message with QoS information codes to MESCAL21.

The following QoS information is to be configured. The value of the QoS Information length is to be checked with a traffic analyzer:
- Scenario 1: no QoS information
- Scenario 2: reserved rate 52
- Scenario 3: loss rate 60
- Scenario 4: reserved rate 52 and loss rate 60
- Scenario 5: minimum one way delay 50 and reserved rate 52
- Scenario 6: minimum one way delay 50 and loss rate 60
- Scenario 7: minimum one way delay 50, loss rate 60 and reserved rate 52
- Scenario 8: minimum one way delay 50, maximum one way delay 150 and loss rate 60
- Scenario 9: minimum one way delay 50, average one way delay 100 and loss rate 60
- Scenario 10: minimum one way delay 50, average one way delay 100 and reserved rate 52
- Scenario 11: minimum one way delay 50, average one way delay 100, maximum one way delay 150 and loss rate 60
- Scenario 12: minimum one way delay 50, average one way delay 100, loss rate 60 and reserved rate 52
- Scenario 13: minimum one way delay 50, maximum one way delay 150, average one way delay 100, loss rate 60 and reserved rate 52
- Scenario 14: minimum one way delay 50, maximum one way delay 150, average one way delay 100, loss rate 60 and jitter 5

Expected result : QoS information length is one byte field. The value checked within a traffic analyzer must be set as follows:

- Scenario 1: 0
- Scenario 2: 1
- Scenario 3: 1
- Scenario 4: 2
- Scenario 5: 2
- Scenario 6: 2
- Scenario 7: 3
- Scenario 8: 3
- Scenario 9: 3
- Scenario 10: 3
- Scenario 11: 4
- Scenario 12: 4
- Scenario 13: 5
- Scenario 14: 5

Execution date : 22/09/04

Result

: The value checked within a traffic analyzer are set as follows:

- Scenario 1: 0
- Scenario 2: 1
- Scenario 3: 1
- Scenario 4: 2
- Scenario 5: 2
- Scenario 6: 2
- Scenario 7: 3
- Scenario 8: 3
- Scenario 9: 3
- Scenario 10: 3
- Scenario 11: 4
- Scenario 12: 4
- Scenario 13: 5

	•	Scenario	14:5
Failure level	: None		
Remarks	: None		

Test Reference Test Purpose Procedure	 : TB_P2_FUNCT/CMES/6 : Verify that "Packet Rate QoS Code" and its associated Sub-codes are supported. : Configure MESCAL11 to send BGP UPDATE message with following QoS information to MESCAL21: Scenario 1: loss rate 30 Scenario 2: reserved rate 25 Scenario 3: available rate 25 Scenario 4: loss rate 45 and reserved rate 454 Scenario 5: loss rate 55 and available rate 100 Scenario 6: reserved rate 52 and available rate 100 Scenario 7: loss rate 60, available rate 100 and reserved rate 52
	A traffic analyzer must be launched in the egress of MESCAL11 in order to examine the values of QoS information Codes and QoS information Sub-Codes.
Expected result	 In all Scenario listed above, QoS information field is coded in 4 bits and QoS information Sub Code is also coded in 4 bits. The values of these fields must be as follows depending on the scenarios listed above: Scenario 1: QoS information Code = 1, QoS information Sub Code = 3 Scenario 2: QoS information Code = 1, QoS information Sub Code = 1 Scenario 3: QoS information Code = 1, QoS information Sub Code = 2 Scenario 4: QoS information Code = 1, QoS information Sub Code = 3 QoS information Code = 1, QoS information Sub Code = 3 Scenario 4: QoS information Code = 1, QoS information Sub Code = 1
	 QoS information Code = 1, QoS information Sub Code = 3 QoS information Code = 1, QoS information Sub Code = 2 Scenario 6: QoS information Code = 1, QoS information Sub Code = 1 QoS information Code = 1, QoS information Sub Code = 2 Scenario 7: QoS information Code = 1, QoS information Sub Code = 1 QoS information Code = 1, QoS information Sub Code = 2 QoS information Code = 1, QoS information Sub Code = 2 QoS information Code = 1, QoS information Sub Code = 2 QoS information Code = 1, QoS information Sub Code = 3
Execution date Result	: 22/09/04 : Obtained results are as follows:
	 Scenario 1: QoS information Code = 1, QoS information Sub Code = 3 Scenario 2: QoS information Code = 1, QoS information Sub Code = 1 Scenario 3: QoS information Code = 1, QoS information Sub Code = 2 Scenario 4: QoS information Code = 1, QoS information Sub Code = 3 QoS information Code = 1, QoS information Sub Code = 1 Scenario 5: QoS information Code = 1, QoS information Sub Code = 3

- \circ QoS information Code = 1, QoS information Sub Code = 2
- Scenario 6:
 - QoS information Code = 1, QoS information Sub Code = 1
 - \circ QoS information Code = 1, QoS information Sub Code = 2

• Scenario 7:

⊞ Frame 24 (139 bytes on wire, 139 bytes captured)

- QoS information Code = 1, QoS information Sub Code = 1
- \circ QoS information Code = 1, QoS information Sub Code = 2
- \circ QoS information Code = 1, QoS information Sub Code = 3

⊞ Ethernet II, Src: 00:d0:b7:21:50:bd, Dst: 00:50:04:48:c3:63 ⊞ Internet Protocol, Src Addr: 1.1.1.5 (1.1.1.5), Dst Addr: 1.1.1.6 (1.1.1.6) 🗄 Transmission Control Protocol, Src Port: 54348 (54348), Dst Port: bgp (179), Seq: 4123177748, Ack: 4118484193, Len: 73 Border Gateway Protocol UPDATE Message Marker: 16 bytes Length: 73 bytes Type: UPDATE Message (2) Unfeasible routes length: 0 bytes Total path attribute length: 50 bytes Path attributes ⊞ ORIGIN: IGP (4 bytes) ⊞ AS_PATH: 1 (7 bytes) ENEXT_HOP: 1.1.1.5 (7 bytes) ⊞ COMMUNITIES: 1:1 (7 bytes) 🗆 Unknown (25 bytes) ➡ Flags: 0xc0 (Optional, Transitive, Complete) Type code: Unknown (17) Length: 22 bytes Unknown (22 bytes) • 0000 00 50 04 48 c3 63 00 d0 0010 00 7d be d9 40 00 01 06 b7 21 50 bd 08 00 45 00 b6 95 01 01 01 05 01 01 .P.H.c.. .!P...E. .}..@... ... b3 14 f5 7b 14 e1 80 18 0020 01 06 d4 4c 00 b3 f5 c2 ...L....{.... 0030 43 e0 db 75 00 00 01 01 08 0a 03 fd 0c d9 03 1b C...u.... 0040 8b b8 ff 0050 ff ff 00 49 02 00 00 00 32 40 01 01 00 40 02 04 ...I.... 20...0.. 0060 02 01 00 01 40 03 04 01 01 01 05 c0 08 04 00 01 00 01 c0 11 16 03 12 00 00 00 01 01 01 01 01 05 0070 64 13 00 <u>3c 11 00 34 1a</u> 0a 06 00 0080 *Failure level* : None Remarks : None Test Reference : TB P2 FUNCT/CMES/7 : Validate that "One Way Delay QoS Code" and its associated Sub-codes are Test Purpose supported. : Configure MESCAL11 to send BGP UPDATE message with following QoS Procedure information to MESCAL21. Scenario 1: minimum one-way delay 50 Scenario 2: maximum one-way delay 150 Scenario 3: average one-way delay 100 Scenario 4: minimum one-way delay 50 and maximum one-way delay 150 Scenario 5: minimum one-way delay 50 and average one-way delay 100 Scenario 6: maximum one-way delay 150 and average one-way delay 100 Scenario 7: minimum one-way delay 50, maximum one-way delay 150 and average one-wav delav 100 A traffic analyzer must be launched in the egress of MESCAL11 in order to examine the values that are set in QoS information Code and QoS information Sub-Code. Expected result: In all Scenario listed above, QoS information field is coded in 4 bits and QoS information Sub Code is also coded in 4 bits. The values of these fields must be as follows depending on the scenarios listed above: Scenario 1: QoS information Code = 2, QoS information Sub Code = 4Copyright © MESCAL Consortium, July 2005

- Scenario 2: QoS information Code = 2, QoS information Sub Code = 5
- Scenario 3: QoS information Code = 2, QoS information Sub Code = 6
 Scenario 4:
- Scenario 4:
 - QoS information Code = 2, QoS information Sub Code = 4
 - QoS information Code = 2, QoS information Sub Code = 5 Scenario 5:
 - QoS information Code = 2, QoS information Sub Code = 4
 - \circ QoS information Code = 2, QoS information Sub Code = 6
- Scenario 6:
 - \circ QoS information Code = 2, QoS information Sub Code = 5
 - \circ QoS information Code = 2, QoS information Sub Code = 6
- Scenario 7:
 - QoS information Code = 2, QoS information Sub Code = 4
 - QoS information Code = 2, QoS information Sub Code = 5
 - \circ QoS information Code = 2, QoS information Sub Code = 6

Execution date: 22/09/04

Result : Obtained results are as follows:

- Scenario 1: QoS information Code = 2, QoS information Sub Code = 4
- Scenario 2: QoS information Code = 2, QoS information Sub Code = 5
- Scenario 3: QoS information Code = 2, QoS information Sub Code = 6
- Scenario 4:
 - QoS information Code = 2, QoS information Sub Code = 4
 - QoS information Code = 2, QoS information Sub Code = 5
- Scenario 5:
 - QoS information Code = 2, QoS information Sub Code = 4
 - \circ QoS information Code = 2, QoS information Sub Code = 6
- Scenario 6:
 - QoS information Code = 2, QoS information Sub Code = 5
 - \circ QoS information Code = 2, QoS information Sub Code = 6
- Scenario 7:
 - QoS information Code = 2, QoS information Sub Code = 4
 - \circ QoS information Code = 2, QoS information Sub Code = 5
 - \circ QoS information Code = 2, QoS information Sub Code = 6

Failure level : None

Remarks : None

: TB_P2_FUNCT/CMES/8 : Validate that "Inter-Packet Delay Variation QoS Code" and its associated Sub-codes are supported.
: Configure MESCAL11 to send BGP UPDATE message with following QoS information to MESCAL21: jitter 5
 : QoS information field is coded in 4 bits and QoS information Sub Code is also coded in 4 bits. The value of these two fields must be: QoS information Code = 3, QoS information Sub Code = 0
: 22/09/04 : The value of these two fields are: QoS information Code = 3, QoS information Sub Code = 0
: None : None

Test Reference : TB_P2_FUNCT/CMES/9

Test Purpose: Validate the QoS information value.Procedure: Configure MESCAL11 to send BGP UPDATE message with following QoS
information to MESCAL21.

- Scenario 1: minimum one-way delay 50, maximum one-way delay 150 and average one-way delay 100
- Scenario 2: reserved rate 100, available rate 100
- Scenario 3: reserved rate 1
- Scenario 4: jitter 5
- Scenario 5: minimum one-way delay -50, maximum one-way delay -150 and average one-way delay -100
- Scenario 6: reserved rate -100, available rate -100

A traffic analyzer must be launched in the egress of MESCAL11 in order to examine the values of QoS information Value field of QoS_NLRI attribute.

Expected result : QoS Information value must be coded in 2 bytes. The value of this field must be as follows (depending on the scenarios listed above)

- Scenario 1:
 - \circ QoS Information Value = 50
 - \circ QoS Information Value = 150
 - \circ QoS Information Value = 100
- Scenario 2:
 - QoS Information Value = 100
 - \circ QoS Information Value = 100
 - Scenario 3: QoS Information Value = 1
- Scenario 4: jitter 5
- Scenario 5: Error messages must be returned.
- Scenario 6: Error messages must be returned.

Execution date : 22/09/04

- *Result* : Obtained results are as follows:
 - Scenario 1:
 - \circ QoS Information Value = 50
 - \circ QoS Information Value = 150
 - QoS Information Value = 100

0070

0080 01 05 0a 0b 00

00 01 c0 11 10 01 30 00 05 1a 00 00 01 01 01 01

⊞ Frame 30 (139 bytes on wire, 139 bytes captured) ⊞ Ethernet II, Src: 00:d0:b7:21:50:bd, Dst: 00:50:04:48:c3:63 ⊞ Internet Protocol, Src Addr: 1.1.1.5 (1.1.1.5), Dst Addr: 1.1.1.6 (1.1.1.6) 🗄 Transmission Control Protocol, Src Port: 54465 (54465), Ist Port: bgp (179), Seq: 421938161, Ack: 411281136, Len: 73 Border Gateway Protocol 🖯 UPDATE Message Marker: 16 bytes Length: 73 bytes Type: UPDATE Message (2) Unfeasible routes length: 0 bytes Total path attribute length: 50 bytes Path attributes ⊞ ORIGIN: IGP (4 bytes) ⊞ AS_PATH: 1 (7 bytes) ⊞ NEXT_HOP: 1,1,1,5 (7 bytes) 🗆 Unknown (25 bytes) ➡ Flags: 0xc0 (Optional, Transitive, Complete) Type code: Unknown (17) Length: 22 bytes Unknown (22 bytes) ∢| 0000 00 50 04 48 c3 63 00 d0 b7 21 50 bd 08 00 45 00 .P.H.c.. .!P...E. 00 7d a6 b0 40 00 01 06 ce be 01 01 01 05 01 01 0010 .}..@.... 01 06 d4 c1 00 b3 19 26 43 f1 18 83 a6 f0 80 18& C...... 0020 0030 43 e0 9a c2 00 00 01 01 08 0a 03 fd e5 97 03 1c С..... 0040 64 74 ff dt..... 0050 ff ff 00 49 02 00 00 00 32 40 01 01 00 40 02 04 20...0.. 02 01 00 01 40 03 04 01 01 01 05 c0 08 04 00 01 00 01 c0 11 16 03 26 00 64 25 00 96 24 00 32 1aB 0060 .&. d%. 64 25 00 0a 0b 00 0070 00 96 24 00 32 1a .\$.2. 00 00 01 01 01 01 01 05 0080 Scenario 2: QoS Information Value = 1000 QoS Information Value = 100 0 Scenario 3: QoS Information Value = 1 Scenario 4: jitter 5 ⊞ Frame 26 (133 bytes on wire, 133 bytes captured) ⊞ Ethernet II, Src: 00:d0:b7:21:50:bd, Dst: 00:50:04:48:c3:63 ⊞ Internet Protocol, Src Addr: 1.1.1.5 (1.1.1.5), Dst Addr: 1.1.1.6 (1.1.1.6) 🗄 Transmission Control Protocol, Src Port: 54531 (54531), Dst Port: bgp (179), Seq: 723472046, Ack: 718170805, Len: 67 Border Gateway Protocol 🗆 UPDATE Message Marker: 16 bytes Length: 67 bytes Type: UPDATE Message (2) Unfeasible routes length: 0 bytes Total path attribute length: 44 bytes 🗆 Path attributes ⊞ AS_PATH: 1 (7 bytes) COMMUNITIES: 1:1 (7 bytes) 🗆 Unknown (19 bytes) ➡ Flags: 0xc0 (Optional, Transitive, Complete) Type code: Unknown (17) Length: 16 bytes Unknown (16 bytes) ∢ | 00 50 04 48 c3 63 00 d0 b7 21 50 bd 08 00 45 00 00 77 16 0f 40 00 01 06 5f 66 01 01 01 05 01 01 0000 .P.H.c.. .!P...E. .w..@... _f.....+. N.*.j... 0010 01 06 d5 03 00 b3 2b 1f 0020 4e ae 2a ce 6a b5 80 18 0030 43 e0 79 47 00 00 01 01 08 0a 03 fe 58 b5 03 1c C.yG....X... 0040 d7 92 ff 0050 ff ff 00 43 02 00 00 00 2c 40 01 01 00 40 02 04 ...C.... ,@...@.. 0060 02 01 00 01 40 03 04 01 01 01 05 c0 08 04 00 01

.0.

- Scenario 5: "Unknown command" message is returned. Scenario 6: "Unknown command" message is returned. •

Failure level	: None
Remarks	: None

Test Reference Test Purpose Procedure	: TB_P2_FUNCT/CMES/10 : Validate the QoS information codes, sub-codes and values. : Configure MESCAL11 to send BGP UPDATE message with following QoS information to MESCAL21:
	 available rate 130, average one-way delay 100, jitter 5, loss rate 60, maximum one-way delay 150, minimum one-way delay 50 and reserved rate 95
	A traffic analyzer must be launched in the egress of MESCAL11 in order to examine the values of QoS information code, sub-codes and value field of QoS_NLRI attribute.
Expected result	t: The value of the QoS information length field must be 0x07. There must be 7 QoS Informations each composed of a QoS information code (1 byte), sub-code (1 byte) and value (2 byte) fields. The following unordered QoS Informations must be announced:
	 QoS Information code = 1, QoS Information code = 2, QoS Information value = 0x0082 QoS Information code = 2, QoS Information code = 6, QoS Information value = 0x0064 QoS Information code = 3, QoS Information code = 0, QoS Information value = 0x0005 QoS Information code = 1, QoS Information code = 3, QoS Information value = 0x003c QoS Information code = 2, QoS Information code = 5, QoS Information value = 0x0096 QoS Information code = 2, QoS Information code = 4, QoS Information value = 0x0032 QoS Information code = 1, QoS Information code = 1, QoS Information value = 0x0032
Execution date Result	: 22/09/04 : The following unordered QoS Informations are announced:
	 QoS Information code = 1, QoS Information code = 2, QoS Information value = 0x0082 QoS Information code = 2, QoS Information code = 6, QoS Information value = 0x0064 QoS Information code = 3, QoS Information code = 0, QoS Information value = 0x0005 QoS Information code = 1, QoS Information code = 3, QoS Information value = 0x003c QoS Information code = 2, QoS Information code = 5, QoS Information value = 0x0096 QoS Information code = 2, QoS Information code = 4, QoS Information value = 0x0032

 QoS Information code = 1, QoS Information code = 1, QoS Information value = 0x005f

⊞ Frame 20 (151 bytes on wire, 151 bytes captured)	
⊞ Ethernet II, Src: 00:d0:b7:21:50:bd, Dst: 00:50:04:48:c3:63	
🗄 Internet Protocol, Src Addr: 1.1.1.5 (1.1.1.5), Dst Addr: 1.1.1.6 (1.1.1.6)	
🗄 Transmission Control Protocol, Src Port: 51956 (51956), Dst Port: bgp (179), Seq: 2781780667, Ack: 27777903	18, Len: 85
E Border Gateway Protocol	
UPDATE Message	
Marker: 16 bytes	
Length: 85 bytes	
Type: UPDATE Message (2)	
Unfeasible routes length: 0 bytes	
Total path attribute length: 62 bytes	
□ Path attributes	
DRIGIN: IGP (4 bytes)	
□ Flags: 0x40 (Well-known, Transitive, Complete)	
iype code: UKIGIN (1)	
Length; I Dyte	
HS_FHIR: I (/ bytes) HEISLand (wid) (HeISLand Transitive Complete)	
Ture codet (CEDITERION), Hansitive, Complete)	
lepeth (butes	
The state 1	
□ Flaos: 0x40 (Well-known, Transitive, Complete)	
Tupe code: NEXT HOP (3)	
Length: 4 bytes	
Next hop: 1.1.1.5	
COMMUNITIES: 1:1 (7 bytes)	
⊞ Flags: 0xc0 (Optional, Transitive, Complete)	
Type code: COMMUNITIES (8)	
Length: 4 bytes	
🗄 Communities: 1:1	
🗆 Unknown (37 bytes)	
⊞ Flags: 0xc0 (Optional, Transitive, Complete)	
Type code: Unknown (17)	
Length: 34 bytes	
Unknown (34 bytes)	
0050 02 01 00 01 40 03 04 01 01 05 00 08 04 00 01 8	
0080 00 3c 25 00 96 24 00 32 11 00 5f 00 00 00 01 01 .<%\$.2	
0090 01 01 05 0a 0b 00	
Frilms land None	
<i>Failure level</i> : None	

Failure level: NoneRemarks: None

Test Reference	: TB_P2_FUNCT/CMES/11
Test Purpose	: Verify that QoS class identifier can be set to a value that is between 0 and 63.
Procedure	: Configure MESCAL11 to send BGP UPDATE messages to MESCAL21 in the
	following meta-QoS-class plans:
	Plan 1: 0
	 Plan 2: -1
	 Plan 3: 25
	 Plan 4: 63

- Plan 5: 70
- Plan 6: 55
- Plan 7: 33

Traffic must be sniffed in the egress of MESCAL11 or in ingress of MESCAL21 to verify the value of QoS class identifier in BGP UPDATE messages.

Expected result : QoS class identifier is coded in one byte and must be set to the value that is indicated below or an error message must be returned to the administrator:

- Plan 1: 0
- Plan 2: An error message must be returned to the administrator
- Plan 3: 25
- Plan 4: 63
- Plan 5: An error message must be returned to the administrator
- Plan 6: 55
- Plan 7: 33

Execution date: 22/09/04 : Obtained results are as follows: Result Plan 1: The OoS class identifier is set to 0. Plan 2: "Unknown command" message is returned. Plan 3: The QoS class identifier is set to 25. Plan 4: The QoS class identifier is set to 63. Plan 5: "Unknown command" message is returned. Plan 6: The OoS class identifier is set to 55. Plan 7: The QoS class identifier is set to 33. . Failure level : None Remarks : None *Test Reference* : **TB_P2_FUNCT/CMES/12** Test Purpose : Validate the QoS Origin field. Procedure : Clear the network prefixes announced by MESCAL11. And then make the following operations: Configure a route static towards 193.251.128.0/19 via eth1 Add the following command lines in MESCAL11 BGP configuration "network 212.167.0.0/21" and "network 62.42.0.0/16" the following commands "redistribute Add connected" and "redistribute static" Under BGP router configuration, add the following line Neighbour 1.1.1.6 route-map SetOrigin out Define this prefix-list: ip prefix-list test11 permit 62.42.0.0/16 ip prefix-list test11 deny any Define this route-map: route-map SetOrigin permit 10 match ip address prefix-list test11 set origin egp route-map SetOrigin permit 20 Note that a BGP session must be activated between MESCAL11 and MESCAL12.

Expected result : Execute this CLI command in MESCAL11: "**sh ip bgp**". The value of the origin value must be positioned as follows:

For **212.167.0.0/21** entry origin must be IGP For **62.42.0.0/16** entry origin value must be IGP For **1.1.1.0/30** entry origin value must be incomplete For **193.251.128.0/19** origin value must be incomplete

Execute this CLI command in MESCAL21: "**sh ip bgp**". The value of the origin value must be positioned as follows:

	For 212.167.0.0/21 entry origin must be IGP For 62.42.0.0/16 entry origin value must be EGP For 193.251.128.0/19 origin value must be incomplete	
Execution date Result	<pre>: 22/09/04 : The command "sh ip bgp" in MESCAL11 shows that: 212.167.0.0/21 entry origin is IGP 62.42.0.0/16 entry origin value is IGP 1.1.1.0/30 entry origin value is incomplete 193.251.128.0/19 origin value is incomplete The command "sh ip bgp" in MESCAL11 shows that: 212.167.0.0/21 entry origin is IGP 62.42.0.0/16 entry origin value is EGP 193.251.128.0/19 origin value is incomplete</pre>	
Failure level Remarks	: None : None	
Test Reference Test Purpose	: TB_P2_FU : Check the this field con Number	UNCT/CMES/13 validity of Address Family Identifier (AFI). For information, the value of uld be as follows: Description
	0	Reserved
	1	IP (IP version 4)
	2	IP6 (IP version 6)
	3	NSAP
	4	HDLC (8-bit multidrop)
	5	BBN 1822
	6	802
	7	E.163
	8	E.164 (SMDS, Frame Relay, ATM)
	9	F.69 (Telex)
	10	X.121 (X.25, Frame Relay)
	11	IPX
	12	Appletalk
	13	Decnet IV
	14	Banyan Vines
	65535	Reserved
Procedure	: Configuration of MESCAL11 is the same as for the previous test. Launch a traffic analyzer in MESCAL11. Execute the following command "clear ip bgp *".	

Expected result : Within the traffic analyzer, Check the value of AFI field of QoS_NLRI attribute. The value of AFI must be: 1.

Execution date : 22/09/04

Result : The value of AFI in the QoS_NLRI attribute is 1.

Failure level Remarks	: None : None		
Test Reference Test Purpose	 : TB_P2_FUNCT/CMES/14 : Check the validity of Subsequent Address Family Identifier (SAFI). For information, the value of this field could be as follows: 1: Network Layer Reachability Information used for unicast forwarding 		
	 2: Network Layer Reachability Information used for multicast forwarding 		
	 3: Network Layer Reachability Information used for both unicast and multicast forwarding 		
Procedure	: Configuration of MESCAL11 is the same as for the previous test. Launch a traffic analyzer in MESCAL11. Execute the following command "clear ip bgp *".		
Expected result	: Within the traffic analyzer, Check the value of SAFI field of QoS_NLRI attribute. The value of this field must be: 1. Multicast is out of scope.		
Execution date Result Failure level Remarks	: 22/09/04 : The value of SAFI in the QoS_NLRI attribute is 1. : None : None		
Test Reference Test Purpose Procedure	 : TB_P2_FUNCT/CMES/15 : Check the validity of Network Address of Next Hop. : Configuration of MESCAL11 is the same as for the previous test. Log to MESCAL21 and execute the following command line "sh ip bgp". 		
	Configure MESCAL31 to send the following network "216.191.64.0/20". Log to MESCAL41 and execute the following command line "sh ip bgp". Also log to MESCAL43 and execute the following command line "sh ip bgp". The BGP session between MESCAL31 and MESCAL42 must be down.		
Expected result	: When executing the command above in MESCAL21, in Next Hop column, the following IP address must be present: For 212.167.0.0/21 entry 1.1.1.5 For 62.42.0.0/16 entry 1.1.1.5 For 1.1.1.0/30 entry 1.1.1.5		
	When executing the command above in MESCAL41, in Next Hop column, the following IP address must be present: For 216.191.64.0/20 entry 3.3.3.1		
	When executing the command above in MESCAL43, in Next Hop column, the following IP address must be present: For 216.191.64.0/20 entry 3.3.3.1		
Execution date Result	: 22/09/04 : When executing the command above in MESCAL21, in Next Hop column, the following IP address are present: For 212.167.0.0/21 entry 1.1.1.5		

	For 62.42.0.0/16 entry 1.1.1.5 For 1.1.1.0/30 entry 1.1.1.5
	When executing the command above in MESCAL41, in Next Hop column, the following IP address are present: For 216.191.64.0/20 entry 3.3.3.1
	When executing the command above in MESCAL43, in Next Hop column, the following IP address is present: For 216.191.64.0/20 entry 3.3.3.1
Failure level Remarks	: None : None
Test Reference Test Purpose Procedure	: TB_P2_FUNCT/CMES/16 : Verify the conformance of NLRI field. : Log to MESCAL11 and define three local-QoS-classes for MESCAL11.
 193.251 212.167 62.42.0 193.251 	Scenario 1: Add the following commands: .128.0/19 local-qos-class 0 .0.0/21 local-qos-class 0 .0/16 local-qos-class 0 .240.0/20 local-qos-class 0
 193.251 212.167 62.42.0 193.251 	Scenario 2: Add the following commands: .128.0/19 local-qos-class 1 .0.0/21 local-qos-class 1 .0/16 local-qos-class 1 .240.0/20 local-qos-class 1
 193.251 212.167 62.42.0 193.251 	Scenario 3: Add the following commands: .128.0/19 local-qos-class 0 1 2 .0.0/21 local-qos-class 0 1 2 .0/16 local-qos-class 0 1 2 .240.0/20 local-qos-class 0 1 2
 193.251 212.167 62.42.0 193.251 	Scenario 4: Add the following commands: .128.0/19 local-qos-class 0 .0.0/21 local-qos-class 1 2 .0/16 local-qos-class 2 .240.0/20 local-qos-class 0 1
	A traffic analyser must be launched in MESCAL11 and capture the traffic that is sent in interface used to connect to MESCAL21.
Expected resul	<i>t</i> : With the traffic analyzer, we must visualize the following results according to the Scenario we are studing:

• Scenario 1: Only one QoS_NLRI message is sent to MESCAL21. The NLRI field contains the following prefixes: 193.251.0/19, 212.167.0.0/21, 62.42.0.0/16 and 193.251.240.0/20.

- Scenario 2: Only one QoS_NLRI message is sent to MESCAL21. The NLRI field contains the following prefixes: 193.251.0/19, 212.167.0.0/21, 62.42.0.0/16 and 193.251.240.0/20.
- Scenario 3: Three QoS_NLRI messages are sent to MESCAL21. The NLRI field of each QoS_NLRI message contains the following prefixes: 193.251.0/19, 212.167.0.0/21, 62.42.0.0/16 and 193.251.240.0/20.
- Scenario 4: Three QoS_NLRI messages are sent to MESCAL21.
 - The NLRI field of the first message contains 193.251.0/19 and 193.251.240.0/20.
 - The NLRI field of the second message contains 212.167.0.0/21 and 62.42.0.0/16.
 - The NLRI field of the third message contains 212.167.0.0/21 and 193.251.240.0/20.

Execution date : 22/09/04

Result : Obtained results are as follows:

- Scenario 1: Only one QoS_NLRI message is sent to MESCAL21. The NLRI field contains the following prefixes: 193.251.0/19, 212.167.0.0/21, 62.42.0.0/16 and 193.251.240.0/20.
- Scenario 2: Only one QoS_NLRI message is sent to MESCAL21. The NLRI field contains the following prefixes: 193.251.0/19, 212.167.0.0/21, 62.42.0.0/16 and 193.251.240.0/20.

 Border Gateway Protocol 🖯 UPDATE Message Marker: 16 bytes Length: 76 bytes Type: UPDATE Message (2) Unfeasible routes length: 0 bytes Total path attribute length: 53 bytes □ Path attributes ⊞ ORIGIN: IGP (4 bytes) ⊞ AS_PATH: 1 (7 bytes) ⊞ COMMUNITIES: 1:1 (7 bytes) 🗆 Unknown (28 bytes) ➡ Flags: 0xc0 (Optional, Transitive, Complete) Type code: Unknown (17) Length: 25 bytes Unknown (25 bytes)

•		
0000	00 50 04 48 c3 63 00 d0 b7 21 50 bd 08 00 45 00	.P.H.c!PE.
0010	00 80 82 c3 40 00 01 06 f2 a8 01 01 01 05 01 01	
0020	01 06 00 b3 d9 6a e4 0a f7 6d e5 21 5b 5c 80 18	jm.![\
0030	16 a0 4e cc 00 00 01 01 08 0a 04 15 76 0f 03 33	N
0040	f4 cd ff	
0050	ff ff 00 4c 02 00 00 00 35 40 01 01 00 40 02 04	L 500
0060	02 01 00 01 40 03 04 01 01 01 05 c0 08 04 00 01	@ <u></u>
0070	00 01 c0 11 19 00 1a 00 00 01 01 01 01 01 05 10	· · · · · · · · · · · · · · · · · · ·
0080	3e 2a 15 d4 a7 00 14 c1 fb f0 13 c1 fb 80	>*

- Scenario 3: Three QoS_NLRI messages are sent to MESCAL21. The NLRI field of each QoS_NLRI message contains the following prefixes: 193.251.0/19, 212.167.0.0/21, 62.42.0.0/16 and 193.251.240.0/20.
- Scenario 4: Three QoS_NLRI messages are sent to MESCAL21.
 - The NLRI field of the first message contains 193.251.0/19 and 193.251.240.0/20.

- The NLRI field of the second message contains 212.167.0.0/21 and 62.42.0.0/16.
- The NLRI field of the third message contains 212.167.0.0/21 and 193.251.240.0/20.

Failure level: NoneRemarks: None

10.2.2 TB_P2_FUNCT/DSCP

Test Reference Test Purpose	: TB_P2_FUNCT/DSCP/1 : Validate that egress DSCP swapping operation is correctly achieved when receiving BGP UPDATE messages.
Procedure	 : Configure MESCAL11'q-BGP process to be instantiated in the local QoS classes bellow: Lqc0.dscp = 0 Lqc1.dscp = 10 Lqc2.dscp = 12 Lqc3.dscp = 14 Lqc4.dscp = 26 Lqc5.dscp = 13 Lqc6.dscp = 63
	In addition, the following mapping between local QoS classes and meta-QoS-classes is also configured in MESCAL11: • 0 is mapped to 0 • 10 is mapped to 26 • 12 is mapped to 28 • 14 is mapped to 30
	MESCAL11 Network prefixes are to be announced to MESCAL21 in the local QoS classes as specified bellow: Scenario 1: 193.251.128.0/19 is to be announced in lqc0 Scenario 2: 193.251.128.0/19 is to be announced in lqc0, lqc1, lqc2 and lqc3 Scenario 3: 193.251.128.0/19 is to be announced in lqc0, lqc4, lqc5 and lqc6 Scenario 4: 193.251.128.0/19 is to be announced in lqc0, lqc1, lqc2 and lqc6 A traffic analyzer must be used in the egress of MESCAL11 in order to verify the value carried in q-BGP UPDATE messages.
Expected result	 in egress of MESCAL11, QoS class identifier of QoS_NLRI attribute must be set to the value that is listed below: Scenario 1: QoS_NLRI message must have a QoS class identifier set to 0 for the prefix 193.251.128.0/19 Scenario 2: Four q-BGP UPDATE messages for the prefix 193.251.128.0/19 must be sent to MESCAL21 with different QoS_NLRI attributes. QoS class identifier of the first q-BGP UPDATE message is set to 0. QoS class identifier of the second q-BGP UPDATE message is set to 26. QoS class identifier of the third q-BGP UPDATE message is set to 28 and QoS class identifier of the fourth q-BGP UPDATE message is set to 30.

- Scenario 3: Only one q-BGP UPDATE message for the prefix 193.251.128.0/19 must be sent to MESCAL21. QoS class identifier value in QoS NLRI attribute is 0.
- Scenario 4: Three q-BGP UPDATE messages for the prefix 193.251.128.0/19 must be sent to MESCAL21 with different QoS_NLRI attributes. QoS class identifier of the first q-BGP UPDATE message is set to 0. QoS class identifier of the second q-BGP UPDATE message is set to 26. QoS class identifier of the third q-BGP UPDATE message is set to 28.

Execution date : 22/09/04

<i>Result</i> : Obtained results are as follow
--

- Scenario 1: QoS_NLRI message has a QoS class identifier set to 0 for the prefix 193.251.128.0/19
- Scenario 2: Four q-BGP UPDATE messages for the prefix 193.251.128.0/19 are sent to MESCAL21 with different QoS_NLRI attributes. QoS class identifier of the first q-BGP UPDATE message is set to 0. QoS class identifier of the second q-BGP UPDATE message is set to 26. QoS class identifier of the third q-BGP UPDATE message is set to 28 and QoS class identifier of the fourth q-BGP UPDATE message is set to 30.
- Scenario 3: Only one q-BGP UPDATE message for the prefix 193.251.128.0/19 is sent to MESCAL21. QoS class identifier value in QoS_NLRI attribute is 0.
- Scenario 4: Three q-BGP UPDATE messages for the prefix 193.251.128.0/19 are sent to MESCAL21 with different QoS_NLRI attributes. QoS class identifier of the first q-BGP UPDATE message is set to 0. QoS class identifier of the second q-BGP UPDATE message is set to 26. QoS class identifier of the third q-BGP UPDATE message is set to 28.

Failure level: NoneRemarks: None

Test Reference : **TB_P2_FUNCT/DSCP/2** *Test Purpose* : Validate that ingress DSCP swapping operation is correctly achieved when receiving BGP UPDATE messages.

- *Procedure* : Configure MESCAL21'q-BGP process to be instantiated in the local QoS classes bellow:
 - Lqc0.dscp = 0
 - Lqc1.dscp = 18
 - Lqc2.dscp = 20
 - Lqc3.dscp = 22
 - Lqc4.dscp = 26
 - Lqc5.dscp = 13
 - Lqc6.dscp = 63

Configure MESCAL11'q-BGP process to be instantiated in the local QoS classes bellow:

- Lqc0.dscp = 0
- Lqc1.dscp = 26
- Lqc2.dscp = 28
- Lqc3.dscp = 30
- Lqc4.dscp = 44
- Lqc5.dscp = 63

Lqc6.dscp = 55

The following mapping between local QoS classes and meta-QoS-classes is also configured in MESCAL21:

- 0 is mapped to 0
- 18 is mapped to 26
- 20 is mapped to 28
- 22 is mapped to 30

Network prefixes are to be announced to MESCAL21 in the local QoS classes as specified bellow:

- Scenario 1: **193.251.128.0/19** is to be announced in lqc0
- Scenario 2: 193.251.128.0/19 is to be announced in lqc0, lqc1, lqc2 and lqc3
- Scenario 3: 193.251.128.0/19 is to be announced in lqc4, lqc5 and lqc6
- Scenario 4: 193.251.128.0/19 is to be announced in lqc1, lqc2, lqc3 and lqc6

A traffic analyzer must be used in the egress of MESCAL11 in order to verify the value carried in q-BGP UPDATE messages.

In MESCAL21, the following 'sh ip route' or 'sh ip bgp' commands must be executed in order to verify the existence of the network prefix announced by MESCAL11 in the appropriate local QoS class.

Expected result: when executing the 'sh ip route' command in MESCAL21, the 193.251.128.0/19 entry must be in the appropriate local QoS class.

- Scenario 1: The prefix 193.251.128.0/19 exists only in the local QoS class 0
- Scenario 2: The prefix 193.251.128.0/19 exists in the following local-QoS-class planes:
 - \circ Lqc0=0
 - \circ Lqc1=18
 - \circ Lqc2=20
 - \circ Lqc3=22
- Scenario 3: The prefix 193.251.128.0/19 does not exist in any local QoS class
- Scenario 4: The prefix 193.251.128.0/19 exists in the following local-QoS-class planes:
 - \circ Lqc1=18
 - \circ Lqc2=20
 - \circ Lqc3=22

Execution date : 22/09/04

- *Result* : Obtained results are as follows:
 - Scenario 1: The prefix 193.251.128.0/19 exists only in the local QoS class 0
 - Scenario 2: The prefix **193.251.128.0/19** exists only in the following local-QoS-class planes:
 - \circ Lqc0=0
 - o Lqc1=18
 - \circ Lqc2=20
 - \circ Lqc3=22
 - Scenario 3: The prefix 193.251.128.0/19 does not exist in any local QoS class

Scenario 4: The prefix 193.251.128.0/19 exists only in the following local-QoS-class planes:

```
\circ Lqc1=18
                              \circ Lqc2=20
                              \circ Lqc3=22
         MESCAL21# sh ip bgp
         No BGP network in the local-gos-class 0 exists
         BGP table version is 0, local router ID is 21.21.21.1
         Status codes: s suppressed, d damped, h history, p stale, * valid, > best, i - internal
         Origin codes: i - IGP, e - EGP, ? - incomplete

        local-qos-class 1 (loose solution options) :
Network
        Metric LocPrf Weight Path

        *> 193.251.128.0/19
        18 1.1.1.5
        0
        1

                                                                                 0 1 i
         Total number of prefixes in the local-gos-class 1 : 1
         local-qos-class 2 (loose solution options) :
                           QC-id Next Hop
            Network
                                                            Metric LocPrf Weight Path
         *> 193.251.128.0/19
                                    20 1.1.1.5
                                                                                0 1 i
         Total number of prefixes in the local-qos-class 2 : 1
         local-qos-class 3 (loose solution options) :
                             QC-id Next Hop Metric LocPrf Weight Path
           Network
         *> 193.251.128.0/19
                                     22 1.1.1.5
                                                                          0 1 i
         Total number of prefixes in the local-qos-class 3 : 1
         No BGP network in the local-qos-class 4 exists
         No BGP network in the local-qos-class 5 exists
         No BGP network in the local-gos-class 6 exists
Failure level : None
Remarks
                : None
```

10.2.3 TB_P2_FUNCT/QCMP

Conformance status is optional for all attributes in the following tests.

Test Reference Test Purpose	: TB_P2_FUNCT/QCMP/1
	Check that the <u>reserved-rate</u> QoS parameter is correctly computed by the receiving ASBR
Procedure	 Establish a pSLS between AS1 and AS2 for exchanging MC1 traffic. Successively carry out the following elementary tests with the following parameters values: A - Set l-QC11.reserved-rate = 500 and l-QC21.reserved-rate = 250. B - Set l-QC11.reserved-rate = 250 and l-QC21.reserved-rate = 500. C - Set l-QC11.reserved-rate = 0 and l-QC21.reserved-rate = 250. D - Set l-QC11.reserved-rate = 500 and l-QC21.reserved-rate = 0. E - Set l-QC11.reserved-rate = 65535 and l-QC21.reserved-rate = 250. F - Set l-QC11.reserved-rate = 500 and l-QC21.reserved-rate = 250. F - Set l-QC11.reserved-rate = 65535 and l-QC21.reserved-rate = 65535. G - Set l-QC11.reserved-rate = 65535 and l-QC21.reserved-rate = 65535. H - Set l-QC11.reserved-rate = 0 and l-QC21.reserved-rate = 65535. I - Set l-QC11.reserved-rate = 0 and l-QC21.reserved-rate = 65535. I - Set l-QC11.reserved-rate = 0 and l-QC21.reserved-rate = 65535. I - Set l-QC11.reserved-rate = 0 and l-QC21.reserved-rate = 65535. I - Set l-QC11.reserved-rate = 0 and l-QC21.reserved-rate = 65535. I - Set l-QC11.reserved-rate = null and l-QC21.reserved-rate = 65535. I - Set l-QC11.reserved-rate = null and l-QC21.reserved-rate = 100. J - Set l-QC11.reserved-rate = 100 and l-QC21.reserved-rate = 100. J - Set l-QC11.reserved-rate = 100 and l-QC21.reserved-rate = 100. J - Set l-QC11.reserved-rate = 100 and l-QC21.reserved-rate = 100. J - Set l-QC11.reserved-rate = 100 and l-QC21.reserved-rate = 100. J - Set l-QC11.reserved-rate = 100 and l-QC21.reserved-rate = 100. J - Set l-QC11.reserved-rate = 100 and l-QC21.reserved-rate = 100. J - Set l-QC11.reserved-rate = 100 and l-QC21.reserved-rate = 100. K - Set l-QC11.reserved-rate = 100 and l-QC21.reserved-rate = 100. K - Set l-QC11.reserved-rate = 100 and l-QC21.reserved-rate = 100.

network 194.52.168.0/21. Note on AS1 the value of the computed reserved-rate QoS parameter for this network.

Expected result :

- A reserved-rate = 250 B - reserved-rate = 250 C - reserved-rate = 0 D - reserved-rate = 0 E - reserved-rate = 250 F - reserved-rate = 500 G - reserved-rate = 65535 H - reserved-rate = 0 I - reserved-rate = null J - reserved-rate = null K - reserved-rate = null
- Execution date: 22/09/04Result: Obtained results are as follows:A reserved-rate = 250B reserved-rate = 250C reserved-rate = 0D reserved-rate = 0E reserved-rate = 250F reserved-rate = 500G reserved-rate = 500G reserved-rate = 65535H reserved-rate = 0I reserved-rate = nullJ reserved-rate = nullK reserved-rate = null

Failure level: NoneRemarks: None

Test Reference	· TB P2 FUNCT/OCMP/2
Test Purpose	:
1	Check that invalid reserved-rate values are rejected by the command-line interface.
Procedure	· · · · · · · · · · · · · · · · · · ·
	Specify the following invalid values on AS1:
	A - Set $1-QC11$.reserved-rate = 65736
	B - Set l-QC11.reserved-rate = 9999999999999999999
	C - Set $1-QC11$.reserved-rate = -1
	D - Set $1-QC11$.reserved-rate = -65736
	E - Set l-QC11.reserved-rate = string
	F - Set l-QC11.reserved-rate = 1string
Expected result	
*	All the above values must be rejected by the command-line interface.
Execution date	: 22/09/04
Result	: All the above values are rejected by the command-line interface.
Failure level	: None
Remarks	: None

Test Reference : TB_P2_FUNCT/QCMP/3

Test Purpose	:
I I I I I I I I I I I I I I I I I I I	Check that the available-rate QoS parameter is correctly computed by the receiving
	ASBR.
Procedure	:
	Establish a pSLS between AS1 and AS2 for exchanging MC1 traffic. Successively
	carry out the following elementary tests with the following parameters values:
	A - Set I-QC11 available-rate = 500 and I-QC21 available-rate = 250 .
	B - Set I-QC11 available rate = 0 and I-QC21 available rate = 250
	C - Set I-QC11 available rate = $500 \text{ and } 1 \text{ QC21}$ available rate = 0
	D = Set I-QC11 available-rate = 65535 and I-QC21 available-rate = 250
	E = Set I-QC11 available-rate = 500 and I-QC21 available-rate = 65535
	G - Set I-OC11 available-rate = 65535 and I-OC21 available-rate = 65535.
	H - Set I-OC11.available-rate = 0 and I-OC21.available-rate = 65535 .
	I - Set I-QC11.available-rate = null and I-QC21.available-rate = 500.
	J - Set $1-QC11$.available-rate = 500 and $1-QC21$.available-rate = null.
	K - Set $1-QC11$.available-rate = null and $1-QC21$.available-rate = null.
	All other QoS parameters are un-valued. Configure q-BGP so that AS2 announces
	network 194.52.168.0/21. Note on AS1 the value of the computed available-rate QoS
	parameter for this network.
Expected result	
	A - available-rate = 250
	D - available rate = 0
	D - available-rate = 0
	E - available-rate = 250
	F - available-rate = 500
	G - available-rate = 65535
	H - available-rate = 0
	I - available-rate = null
	J - available-rate = null
	K - available-rate = null
Execution data	· 22/00/04
Result	· Obtained results are as follows:
Restiti	A - available-rate = 250
	B - available-rate = 250
	C - available-rate = 0
	D - available-rate = 0
	E - available-rate = 250
	F - available-rate = 500
	G - available-rate = 65535
	H - available-rate = 0
	I - available-rate = null
	J - available-rate = null
Failure level	: None
Remarks	: None

Test Reference : TB_P2_FUNCT/QCMP/4

Test Purpose

Check that invalid **<u>available-rate</u>** values are rejected by the command-line interface.

Procedure	: Specify the following invalid values on AS1: A - Set l-QC11.available-rate = 65736 B - Set l-QC11.available-rate = 99999999999999999999999999999999999
Expected result	All the above values must be rejected by the command-line interface.
Execution date Result Failure level Remarks	: 22/09/04 : All the above values are rejected by the command-line interface. : None : None
Test Reference Test Purpose	: TB_P2_FUNCT/QCMP/5
Procedure	Check that the <u>min-owd</u> (minimum one-way-delay) QoS parameter is correctly computed by the receiving ASBR.
	Establish a pSLS between AS1 and AS2 for exchanging MC1 traffic. Successively carry out the following elementary tests with the following parameters values: A - Set I-QC11.min-owd = 500 and I-QC21.min-owd = 250. B - Set I-QC11.min-owd = 0 and I-QC21.min-owd = 500. C - Set I-QC11.min-owd = 500 and I-QC21.min-owd = 0. E - Set I-QC11.min-owd = 65535 and I-QC21.min-owd = 250. F - Set I-QC11.min-owd = 500 and I-QC21.min-owd = 65535. G - Set I-QC11.min-owd = 65535 and I-QC21.min-owd = 65535. H - Set I-QC11.min-owd = 0 and I-QC21.min-owd = 65535. I - Set I-QC11.min-owd = 0 and I-QC21.min-owd = 65535. J - Set I-QC11.min-owd = null and I-QC21.min-owd = 500. J - Set I-QC11.min-owd = 500 and I-QC21.min-owd = mull. K - Set I-QC11.min-owd = s00 and I-QC21.min-owd = null. All other QoS parameters are un-valued. Configure q-BGP so that AS2 announces network 194.52.168.0/21. Note on AS1 the value of the computed min-owd QoS parameter for this network.
Expected result	A - min-owd = 750 B - min-owd = 750 C - min-owd = 250 D - min-owd = 500 E - min-owd = 65535 F - min-owd = 65535 G - min-owd = 65535 H - min-owd = 65535 I - min-owd = null J - min-owd = null K - min-owd = null
Execution date Result	: 22/09/04 : Obtained results are as follows: A - min-owd = 750 B - min-owd = 750

	C - min-owd = 250
	D - min-owd = 500
	E - min-owd = 65535
	F - min-owd = 65535
	G - min-owd = 65535
	H - min-owd = 65535
	I - min-owd = null
	J - min-owd = null
	K - min-owd = null
Failure level	: None
Remarks	: None
Tost Defense	TR D2 EUNCT/OCMD/C

Test Reference	: TB_P2_FUNCT/QCMP/6
Test Purpose	:
	Check that invalid <u>min-owd</u> values are rejected but the command-line interface.
Procedure	:
	Specify the following invalid values on AS1:
	A - Set l-QC11.min-owd = 65736
	B - Set l-QC11.min-owd = 9999999999999999999
	C - Set 1 -QC11.min-owd = -1
	D - Set l-QC11.min-owd = -65736
	E - Set 1-QC11.min-owd = string
	F - Set l-QC11.min-owd = 1 string
Expected result	<i>t</i> :
	All the above values must be rejected by the command-line interface.
Execution date	: 22/09/04
Result	: All the above values are rejected by the command-line interface.
Failure level	: None
Remarks	: None
Test Reference	: TB_P2_FUNCT/QCMP/7
Test Purpose	:
	Check that the max-owd (maximum one-way-delay) QoS parameter is correctly
	computed by the receiving ASBR.
Procedure	
	Same procedure as TB_P2_FUNCT/QCMP/5. Replace min-owd by <u>max-owd</u> .
Expected result	<i>t</i> :
	A - max-owd = 750
	B - max - owd = 750
	C - max-owd = 250
	D - max-owd = 500
	E - max-owd = 65535
	F - max-owd = 65535
	$G - \max - owd = 65535$
	H - max-owd = 65535
	I - max-owd = null
	J - max-owd = null
	K - max-owd = null
Execution date	: 22/09/04
Result	: Obtained results are as follows:
	A - max-owd = 750
	B - max - owd = 750
	Copyright © MESCAL Consortium, July 2005

	C - max-owd = 250 D - max-owd = 500 E - max-owd = 65535 F - max-owd = 65535 G - max-owd = 65535 H - max-owd = 65535 I - max-owd = null J - max-owd = null K - max-owd = null
Failure level Remarks	: None : None
Test Reference Test Purpose	: TB_P2_FUNCT/QCMP/8
Procedure	Check that invalid $\underline{max-owd}$ values are rejected by the command-line interface.
Expected result	Same procedure as TB_P2_FUNCT/QCMP/6. Replace min-owd by <u>max-owd</u> .
Expected result	All the above values must be rejected by the command-line interface.
Execution date Result Failure level Remarks	: 22/09/04 : All the above values are rejected by the command-line interface. : None : None
Test Reference Test Purpose	: TB_P2_FUNCT/QCMP/9
1	Check that the <u>average-owd</u> (average one-way-delay) QoS parameter is correctly computed by the receiving ASBR.
Procedure	: Same procedure as TB P2 FUNCT/QCMP/5. Replace min-owd by average-owd .
Expected result	t:
-	A - average-owd = 750
	B - average - $owd = 750$
	C - average $-owd = 250$ D average $-owd = 500$
	E - average - owd = 65535
	F - average -owd = 65535
	G - average -owd = 65535
	H - average - owd = 65535
	I - average - owd = null
	K - average - owd = null
Execution date	: 22/09/04
Result	: Obtained results are as follows: A = average aver
	R - average - owd = 750
	C - average - owd = 250
	D - average - $owd = 500$
	E - average - owd = 65535
	F - average - owd = 65535
	G - avoiago - Owu - 05555

	H - average -owd = 65535
	I - average - $owd = null$
	J - average - owd = null
	K - average -owd – null
Failure level	·None
Remarks	· None
Tast Rafaranca	TR P2 FUNCT/OCMP/10
Test Purnose	. ID_I2_F UNCI/QCMI/IU
10571 00 pose	Check that invalid average-owd values are rejected by the command-line interface.
Procedure	
	Same procedure as TB_P2_FUNCT/QCMP/6. Replace min-owd by average-owd.
Expected result	t:
	All the above values must be rejected by the command-line interface.
Execution date	· 22/09/04
Result	· All the above values are rejected by the command-line interface
Failure level	: None
Remarks	: None
Test Reference	TB P2 FUNCT/OCMP/11
Test Purpose	
Ĩ	Check that the loss-rate QoS parameter is correctly computed by the receiving ASBR.
Procedure	:
	Establish a pSLS between AS1 and AS2 for exchanging MC1 traffic. Successively
	carry out the following elementary tests with the following parameters values:
	A - Set I-QC11.loss-rate = 500 and I-QC21.loss-rate = 250 . B - Set I-QC11 loss rate = 250 and I-QC21 loss rate = 500
	$C = Set I - QC11 \log_{10} = 2.50 \text{ and } I - QC21 \log_{10} = 2.50$
	D - Set I-OC11 loss-rate = 500 and I-OC21 loss-rate = 0 .
	E - Set I-QC11.loss-rate = 65535 and I-QC21.loss-rate = 250.
	F - Set $1-QC11$.loss-rate = 500 and $1-QC21$.loss-rate = 65535.
	G - Set l-QC11.loss-rate = 65535 and l-QC21.loss-rate = 65535 .
	H - Set l-QC11.loss-rate = 0 and l-QC21.loss-rate = 65535 .
	I - Set I-QC11.loss-rate = null and I-QC21.loss-rate = 500 .
	J - Set I-QC11.loss-rate = 500 and I-QC21.loss-rate = null. $K = Set I = QC11 \log_2 rate = null and I = QC21 \log_2 rate = null.$
	All other OoS parameters are un-valued. Configure a-BGP so that AS2 appoinces
	network 194.52.168.0/21. Note on AS1 the value of the computed loss-rate OoS
	parameter for this network.
	A loss-rate = 1 means 0.001%
Expected result	t:
	A - loss-rate = 748 or 749 depending on the round-off
	B - loss-rate = 748 or 749 depending on the round-off
	C = 105S-rate = 500
	E - loss-rate = 65535
	F - loss-rate = 65535
	G - loss-rate = 65535
	H - loss-rate = 65535
	I - loss-rate = null
	J - loss-rate = null
	K - loss-rate = null

Execution date Result	: $22/09/04$: Obtained results are as follows: A - loss-rate = 748 B - loss-rate = 748 C - loss-rate = 250 D - loss-rate = 500 E - loss-rate = 65535 F - loss-rate = 65535 G - loss-rate = 65535 H - loss-rate = 65535 I - loss-rate = null J - loss-rate = null K - loss-rate = null
Failure level Remarks	: None : None
Test Reference Test Purpose	: TB_P2_FUNCT/QCMP/12 :
Procedure	Check that invalid <u>loss-rate</u> values are rejected by the command-line interface.
	Specify the following invalid values on AS1:
	A - Set I-QC11.loss-rate to 65736
	B - Set I-QC11.loss-rate to -1
	D - Set I-QC11 loss-rate to -65736
	E - Set I-OC11.loss-rate to string
	F - Set I-QC11.loss-rate to 1string
Expected result	
	All the above values must be rejected by the command-line interface.
Execution date	: 22/09/04
Result	: All the above values are rejected by the command-line interface.
Failure level	: None
Remarks	: None
Test Reference	: TB_P2_FUNCT/QCMP/13
Procedure	Check that the <u>jitter</u> QoS parameter is correctly computed by the receiving ASBR.
Troccume	 Establish a pSLS between AS1 and AS2 for exchanging MC1 traffic. Successively carry out the following elementary tests with the following parameters values: A - Set 1-QC11.jitter = 500 and 1-QC21.jitter = 250. B - Set 1-QC11.jitter = 0 and 1-QC21.jitter = 500. C - Set 1-QC11.jitter = 0 and 1-QC21.jitter = 250. D - Set 1-QC11.jitter = 500 and 1-QC21.jitter = 0. E - Set 1-QC11.jitter = 65535 and 1-QC21.jitter = 250. F - Set 1-QC11.jitter = 500 and 1-QC21.jitter = 65535. G - Set 1-QC11.jitter = 65535 and 1-QC21.jitter = 65535. H - Set 1-QC11.jitter = 0 and 1-QC21.jitter = 65535. I - Set 1-QC11.jitter = 0 and 1-QC21.jitter = 65535. I - Set 1-QC11.jitter = null and 1-QC21.jitter = 500. J - Set 1-QC11.jitter = null and 1-QC21.jitter = mull. K - Set 1-QC11.jitter = null and 1-QC21.jitter = null.

All other QoS parameters are un-valued. Configure q-BGP so that AS2 announces network 194.52.168.0/21. Note on AS1 the value of the computed jitter QoS parameter for this network.

Expected result :

- A jitter = 750
- B jitter = 750
- C jitter = 250
- D jitter = 500
- E jitter = 65535
- F jitter = 65535
- G jitter = 65535
- H jitter = 65535
- I jitter = null
- J jitter = null
 K jitter = null

Execution date : 22/09/04

Result : Obtained results are as follows:

- A jitter = 750
- B jitter = 750
- C jitter = 250
- D jitter = 500
- E jitter = 65535
- F jitter = 65535
- G jitter = 65535
- H jitter = 65535
- I jitter = null
- J jitter = null
- K jitter = null

Failure level: NoneRemarks: None

Test Reference	: TB_P2_FUNCT/QCMP/14
Test Purpose	: Check that invalid <u>jitter</u> values are rejected by the command-line interface.
Procedure	:
	Specify the following invalid values on AS1:
	• A - Set $1-QC11$.jitter = 65736
	• B - Set l-QC11.jitter = 9999999999999999999
	• C - Set l-QC11.jitter = -1
	• D - Set $1-QC11$.jitter = -65736
	• E - Set l-QC11.jitter = string
	• F - Set l-QC11.jitter = 1string
Expected result	
	All the above values must be rejected by the command-line interface.
Execution date	: 22/09/04
Result	: All the above values are rejected by the command-line interface.
Failure level	: None
Remarks	: None

Test Reference Test Purpose	: TB_P2_FUNCT/QCMP/15 :
Procedure	Check that the receiving ASBR is able to compute multiple QoS parameters contained in an announcement.
Frocedure	Establish a pSLS between AS1 and AS2 for exchanging MC1 traffic.
	Set the l-QC11 parameter values as follow: Set l-OC11.reserved-rate = 500.
	Set $1-QC11$.available-rate = 200.
	Set $1-QC11$.loss-rate = 1000.
	Set $1-QC11$.min-owd = 10.
	Set $1-QC11.max-owd = 25$.
	Set $1-QC11$.average-owd = 15.
	Set I-QC11.jitter = 5.
	Set the l-QC21 parameter values as follow:
	Set $1-QC21$.reserved-rate = 300.
	Set I-QC21 available-rate = 250 .
	Set I-QC21.loss-rate = 2500 .
	Set 1 \circ C21 max ovd = 60
	Set 1-QC21 average-owd = 55
	Set 1-QC21 iitter = 15
	501 (0-1,100 10)
	Configure q-BGP so that AS2 announces network 194.52.168.0/21. Note on AS1 the value of the computed QoS parameters for this network.
Expected result	:
	reserved-rate = 300 .
	available-rate = 200 .
	loss-rate = 34/5.
	$\lim_{n \to \infty} -40.$
	average-owd = 70
	itter = 20.
Execution date	: 22/09/04
Result	: Obtained results are as follows: reserved rate $= 200$
	available-rate = 200
	loss-rate = 3475
	\min -owd = 40.
	max-owd = 85.
	average-owd = 70.
	jitter = 20.
MESCAL11# sh ip bgp lo BGP table version is 0 Status codes: s suppres Origin codes: i - IGP.	cal-qos-class 11 , local router ID is 11.11.11.1 ssed, d damped, h history, p stale, * valid, > best, i - internal e - FGP. 2 - incomplete
local-qos-class 11 (loo Network	ose solution options) : QC-id Next Hop Metric LocPrf Weight reserved-rate available-rate loss-rate min-owd max-owd avg-owd jitter Path
*> 194.52.168.0/21	63 1.1.1.6 0 300 200 3475 40 85 70 20 2 i
Total number of prefixe	es in the local-qos-class 11 : 1
Failure level	: None
Remarks	: None

Test Reference Test Purpose	: TB_P2_FUNCT/QCMP/16 :
	Check that the receiving ASBR is able to compute multiple QoS parameters for a same prefix announced within different meta-QoS-planes.
Procedure	
	Establish a pSLS between AS1 and AS2 for exchanging MC1, MC2 and MC3 traffic.
	Configure I-QC12 and I-QC13 with the same QoS parameter values as I-QC11.
	Configure I-QC22 and I-QC23 with the same QoS parameter values as I-QC21.
	Configure q-BGP so that AS2 announces network 194.52.168.0/21 on meta-QoS- planes MC1, MC2 and MC3.
	Note on AS1 the value of the computed QoS parameters for this network and per meta-QoS-plane.
Expected result	
-	Announcements received for this prefix within the 3 meta-QoS-class planes must have the same values as expected in TB_P2_FUNCT/QCMP/15.
Execution date	: 22/09/04
Result	: Announcements received for this prefix within the 3 meta-QoS-class planes have the same values as expected in TB P2 FUNCT/OCMP/15.
Failure level	: None
Remarks	: None

10.2.4TB_P2_FUNCT/RSEL

Sub-group preamble

Unless specified, tests hereafter described involve AS1, AS2, AS3 and AS4. Peering need to be set-up between: AS1 & AS2, AS1 & AS3, AS3 & AS4, AS2 & AS4 as described in the testbed configuration section. Only Mescal-42 ASBR from AS4 is involved in this series of tests.

pSLSs established between ASs concern MC1 and Best-effort only.

AS1 announces only the network prefix 193.251.128.0/19 within MC1. Others ASs propagate the route but do not announce any of their own networks.

OoS	guarantees	of I-OC im	nlementing	MC1 i	n each A	S are	defined	as follo	w.
QUS	guarantees	ULL-QC III	prementing	MC11		is are	ucinicu	as ion	<i>J</i> w.

Attributes	Priority	I-QC11	l-QC21	I-QC31	l-QC41
Reserved-rate	1	600	800	400	2000
Available-rate	2	500	400	600	2000
Loss-rate	3	1000	500	1500	200
Min-owd	4	10	15	25	5
Max-owd	5	50	25	10	15
Average-owd	6	30	20	15	10
Jitter	7	20	4	5	5

Table 37: Local QoS Class Characteristics

The priority level and the conformance (mandatory/optional) status of each QoS attribute is specified by each test.

The precision is set to 0%, for all attributes of all l-QCs, unless it is explicitly specified.

Group 2 QoS Service Capability is also configured unless explicit related configuration is recommended by a given test.

Test Reference Test Purpose	: TB_P2_FUNCT/RSEL/1
	Check that several ASs involved in the loose service option are able to exchange route updates containing correctly computed QoS information.
Procedure	:
	Start as defined by this sub-group preamble. Conformance status is mandatory for all attributes.
	Check the number of route received by AS4 for the network prefix 193.251.128.0/19 .
	Check that a route has been selected.
	Check that QoS parameters have the expected values.
Expected result	
	Route431 via AS4,AS3,AS1:
	reserved-rate = 400 .
	avaliable-late = 300.
	min-owd = 40
	max-owd = 75
	average-owd = 55
	jitter = 30.
	Route421 via AS4,AS2,AS1:
	reserved-rate = 600 .
	available-rate = 400 .
	loss-rate = 1692 .
	\min -owd = 30.
	max-owd = 90.
	average-owd = 60 .
	Jiller – 29.

AS4 must select route421

```
Execution date : 22/09/04Result: Obtained results are as follows:Route431 via AS4,AS3,AS1:<br/>reserved-rate = 400.<br/>available-rate = 500.<br/>loss-rate = 2680.<br/>min-owd = 40.<br/>max-owd = 75.<br/>average-owd = 55.<br/>jitter = 30.
```

Route421 via AS4,AS2,AS1:

reserved-rate = 600. available-rate = 400. loss-rate = 1692. min-owd = 30. max-owd = 90. average-owd = 60. jitter = 29.

Route421 is selected.

MESCAL42# sh ip bgp local-qos-class 41 BGP table version is 0, local router ID is 42.42.42.1 Status codes: s suppressed, d damped, h history, p stale, * valid, > best, i - internal Origin codes: i - IGP, e - EGP, ? - incomplete local-qos-class 41 (loose solution options) : QC-id Next Hop 34 2.2.2.1 Metric LocPrf Weight reserved-rate available-rate loss-rate min-owd max-owd avg-owd jitter Path 0 600 400 1692 30 90 60 29 2 Network *> 193.251.128.0/19 90 75 60 29 21i 55 30 31i 34 3.3.3.5 0 400 500 2680 40 Total number of prefixes in the local-qos-class 41 : 1

Failure level	: None
Remarks	: None

Test Reference Test Purpose	: TB_P2_FUNCT/RSEL/2 : Check, in simple Scenarios, that the route selection process takes into account the					
Procedure	priority level of each QoS attribute. : As defined by this sub-group preamble. Conformance status is mandatory for all attributes.					
	 Using the table described in the preamble as a reference for each sub-test, exchange the priority value of: Scenario A1 - Reserved-rate and the available-rate QoS parameter Scenario B1 - Reserved-rate and the loss-rate QoS parameter Scenario C1 - Reserved-rate and the min-owd QoS parameter Scenario D1 - Reserved-rate and the max-owd QoS parameter Scenario E1 - Reserved-rate and the average-owd QoS parameter Scenario F1 - Reserved-rate and the jitter QoS parameter One performed, exchange the QoS attribute values of 1-QC21 and 1-QC31. Perform the 6 tests again (A2-F2) 					

Expected result :

- Scenario A1 Route431 is selected
- Scenario B1 Route421 is selected
- Scenario C1 Route421 is selected
- Scenario D1 Route431 is selected
- Scenario E1 Route431 is selected
- Scenario F1 Route421 is selected
- Scenario A2 Route421 is selected
- Scenario B2 Route431 is selected
- Scenario C2 Route431 is selected
- Scenario D2 Route421 is selected
- Scenario E2 Route421 is selected
- Scenario F2 Route431 is selected

Execution date : 22/09/04

Result

: Obtained results are as follows:

- Scenario A1 Route431 is selected
- Scenario B1 Route421 is selected
- Scenario C1 Route421 is selected
- Scenario D1 Route431 is selected
- Scenario E1 Route431 is selected
- Scenario F1 Route421 is selected
- Scenario A2 Route421 is selected
- Scenario B2 Route431 is selected
- Scenario C2 Route431 is selected
- Scenario D2 Route421 is selected
- Scenario E2 Route421 is selected
- Scenario F2 Route431 is selected

Failure level	: None
Remarks	: None

Test Reference	: TB_P2_FUNCT/RSEL/3
Test Purpose	:
	Check that the route selection process takes into account the QoS attributes which have a lower priority when the previous attributes (with higher priority) are equivalent.
Procedure	: Start as defined by this sub-group preamble. Conformance status is mandatory for all attributes.

Give l-QC31 the same values as l-QC21. Then, perform the following sub-tests:

- Scenario A Set l-QC31.available-rate=450
- Scenario B Set l-QC31.available-rate=350
- Scenario C Set l-QC31.available-rate=400 and set l-QC31.loss-rate=400
- Scenario D Set l-QC31.available-rate=400 and set l-QC31.loss-rate=600
- Scenario E Set I-QC31.loss-rate=500 and set I-QC31.min-owd=20
- Scenario F Set I-QC31.loss-rate=500 and set I-QC31.min-owd=10
- Scenario G Set I-QC31.min-owd=15 and set I-QC31.max-owd=30
- Scenario H Set I-QC31.min-owd=15 and set I-QC31.max-owd=20
- Scenario I Set l-QC31.max-owd=25 and set l-QC31.average-owd=25
- Scenario J Set l-QC31.max-owd=25 and set l-QC31.average-owd=15
- Scenario K Set 1-QC31.average-owd=20 and set 1-QC31.jitter=5
- Scenario L Set l-QC31.average-owd=20 and set l-QC31.jitter=3

Expected result :

For each sub-test, check that AS4 received 2 routes with the same QoS attribute values except one of them. One of them must have selected.

- Scenario A Available-rate is different Route via AS3 must be selected
- Scenario B Available-rate is different Route via AS2 must be selected
- Scenario C Loss-rate is different Route via AS3 must be selected
- Scenario D Loss-rate is different Route via AS2 must be selected
- Scenario E Min-owd is different Route via AS2 must be selected
- Scenario F Min-owd is different Route via AS3 must be selected
- Scenario G Max-owd is different Route via AS2 must be selected
- Scenario H Max-owd is different Route via AS3 must be selected
- Scenario I Average-owd is different Route via AS2 must be selected
- Scenario J Average-owd is different Route via AS3 must be selected
- Scenario K Jitter is different Route via AS2 must be selected
- Scenario L Jitter is different Route via AS3 must be selected

Execution date : 22/09/04

Result

: Obtained results are as follows:

- Scenario A Available-rate is different Route via AS3 is selected
- Scenario B Available-rate is different Route via AS2 is selected
- Scenario C Loss-rate is different Route via AS3 is selected
- Scenario D Loss-rate is different Route via AS2 is selected
- Scenario E Min-owd is different Route via AS2 is selected
- Scenario F Min-owd is different Route via AS3 is selected
- Scenario G Max-owd is different Route via AS2 is selected
- Scenario H Max-owd is different Route via AS3 is selected
- Scenario I Average-owd is different Route via AS2 is selected
- Scenario J Average-owd is different Route via AS3 is selected
- Scenario K Jitter is different Route via AS2 is selected

MESCAL42# sh ip bgp local-qos-class 41 BGP table version is 0, local router ID is 42.42.42.1 Status codes: s suppressed, d damped, h history, p stale, * valid, > best, i - internal Origin codes: i - IGP, e - EGP, ? - incomplete

10	local-gos-class 41 (loose solution options) :												
	Network	QC-id Next Hop	Metric LocPrf	Weight	reserved-rate	available-rate	loss-rate	min-owd	max-owd	avg-owd	jitter	Path	
*	193.251.128.0/19	34 3.3.3.5		0	600	400	1692	30	90	60	30	31	i
*>		34 2.2.2.1		0	600	400	1692	30	90	60	29	21	i

Total number of prefixes in the local-qos-class 41 : 1

• Scenario L – Jitter is different - Route via AS3 is selected

D3.2: Final Experimental Results

MESCAL42# sh ip bgp local-qos-class 41 BGP table version is 0, local router ID is 42.42.42.1 Status codes: s suppressed, d damped, h history, p stale, * valid, > best, i - internal Origin codes: i - IGP, e - EGP, ? - incomplete local-gos-class 41 (loose solution options) : Network QC-id Next Hop Metric LocPrf Weight reserved-rate available-rate loss-rate min-owd max-owd avg-owd jitter Path *> 193.251.128.0/19 34 3.3.3.5 600 400 1692 30 90 60 28 31i 21i 0 34 2.2.2.1 0 600 400 1692 30 90 60 29 Total number of prefixes in the local-gos-class 41 : 1 Failure level : None Remarks : None Test Reference : TB_P2_FUNCT/RSEL/4 Test Purpose Check that the precision command line parameter is correctly handled for the reserved-rate QoS attribute. Procedure Start as defined by this sub-group preamble. Conformance status is mandatory for all attributes. Scenario 1: Set the precision for 1-QC41.reserved-rate=50%. The related QoS attribute of the 2 routes overlap. Scenario 2: Set the precision for I-QC41.reserved-rate=10%. The related QoS attribute of the 2 routes DO NOT overlap. Expected result : Scenario 1: Route via AS3 must be selected (the decision is enforced by the available-rate) Scenario 2: Route via AS2 must be selected (the decision is enforced by the reserved-rate) Execution date: 22/09/04 Result : Obtained results are as follows: Scenario 1: Route via AS3 is selected MESCAL42# sh ip bgp local-qos-class 41 BGP table version is 0, local router ID is 42.42.42.1 Status codes: s suppressed, d damped, h history, p stale, * valid, > best, i - internal Origin codes: i - IGP, e - EGP, ? - incomplete local-qos-class 41 (loose solution options) : QC-id Next Hop Network Metric LocPrf Weight reserved-rate available-rate loss-rate min-owd max-owd avg-owd jitter Path *> 193.251.128.0/19 2680 34 3.3.3.5 0 400 500 40 75 55 30 31i 2 1 i 34 2.2.2.1 0 600 400 1692 30 90 60 29 Total number of prefixes in the local-qos-class 41 : 1 Scenario 2: Route via AS2 is selected MESCAL42# sh ip bgp local-qos-class 41 BGP table version is 0, local router ID is 42.42.42.1 Status codes: s suppressed, d damped, h history, p stale, * valid, > best, i - internal Origin codes: i - IGP, e - EGP, ? - incomplete local-qos-class 41 (loose solution options) : Network QC-id Next Hop Metric LocPrf Weight reserved-rate available-rate loss-rate min-owd max-owd avg-owd jitter Path 34 3.3.3.5 . 193.251.128.0/19 30 3 1 i 400 500 2680 40 75 0 55 *> 34 2.2.2.1 0 600 400 1692 30 90 60 29 2 1 i Total number of prefixes in the local-qos-class 41 : 1 Failure level : None Remarks : None

Test Reference : TB_P2_FUNCT/RSEL/5

- *Test Purpose* : Check that the precision command line parameter is correctly handled for the available-rate QoS attribute.
- *Procedure* : Start as defined by this sub-group preamble.

Conformance status is mandatory for all attributes.

Set the precision for l-QC41.reserved-rate=50%. The related QoS attribute of the 2 routes overlap.

- Scenario 1: Set the precision for l-QC41.available-rate=50%. The related QoS attribute of the 2 routes overlap.
- Scenario 2: Set the precision for l-QC41.available-rate=10%. The related QoS attribute of the 2 routes DO NOT overlap.

Expected result : The following results must be obtained:

- Scenario 1: Route via AS2 must be selected (the decision is enforced by the loss-rate)
- Scenario 2: Route via AS3 must be selected (the decision is enforced by the available-rate)

Execution date : 22/09/04

: Obtained results are as follows:

- Scenario 1: Route via AS2 is selected
- Scenario 2: Route via AS3 is selected

Failure level : None

Result

Remarks : None

Test Reference : TB_P2_FUNCT/RSEL/6

- *Test Purpose* : Check that the precision command line parameter is correctly handled for the loss-rate QoS attribute.
- *Procedure* : Start as defined by this sub-group preamble. Conformance status is mandatory for all attributes.

Set the precision for l-QC41.reserved-rate=50%. The related QoS attribute of the 2 routes overlap.

Set the precision for l-QC41.available-rate=50%. The related QoS attribute of the 2 routes overlap.

- Scenario 1: Set the precision for l-QC41.loss-rate=50%. The related QoS attribute of the 2 routes overlap.
- Scenario 2: Set the precision for l-QC41.loss-rate=10%. The related QoS attribute of the 2 routes DO NOT overlap.

Expected result : The following results must be obtained:

- Scenario 1: Route via AS2 must be selected (the decision is enforced by the minowd)
- Scenario 2: Route via AS2 must be selected (the decision is enforced by the loss-rate)

Execution date Result	 22/09/04 Obtained results are as follows: Scenario 1: Route via AS2 is selected 							
	 Scenario 2: Route via AS2 is selected 							
Failure level Remarks	: None : None							
Test Reference	: TB_P2_FUNCT/RSEL/7							
Test Purpose	: Check that the precision command line parameter is correctly handled for the min- owd QoS attribute.							
Procedure	: Start as defined by this sub-group preamble. Conformance status is mandatory for all attributes.							
	Set the precision for l-QC41.reserved-rate=50%. The related QoS attribute of the 2 routes overlap.							
	Set the precision for l-QC41.available-rate=50%. The related QoS attribute of the 2 routes overlap.							
	Set the precision for l-QC41.loss-rate=50%. The related QoS attribute of the 2 routes overlap.							
	 Scenario 1: Set the precision for l-QC41.min-owd=20%. The related QoS attribute of the 2 routes overlap. 							
	 Scenario 2: Set the precision for l-QC41.min-owd=5%. The related QoS attribute of the 2 routes DO NOT overlap. 							
Expected result	: The following results must be obtained:							
	 Scenario 1: Route via AS3 must be selected (the decision is enforced by the max- owd) 							
	 Scenario 2: Route via AS2 must be selected (the decision is enforced by the min- owd) 							
Execution date Result	: 22/09/04 : Obtained results are as follows: • Scenario 1: Route via AS3 is selected							
	 Scenario 2: Route via AS2 is selected 							
Failure level Remarks	: None : None							
Test Reference	: TB_P2_FUNCT/RSEL/8							
Test Purpose	: Check that the precision command line parameter is correctly handled for the max-owd QoS attribute.							

Procedure : Start as defined by this sub-group preamble. Conformance status is mandatory for all attributes.

Set the precision for l-QC41.reserved-rate=50%. The related QoS attribute of the 2 routes overlap.

Set the precision for l-QC41.available-rate=50%. The related QoS attribute of the 2 routes overlap.

Set the precision for l-QC41.loss-rate=50%. The related QoS attribute of the 2 routes overlap.

Set the precision for l-QC41.min-owd=20%. The related QoS attribute of the 2 routes overlap.

- Scenario 1: Set the precision for 1-QC41.max-owd=50%. The related QoS attribute of the 2 routes overlap.
- Scenario 2: Set the precision for l-QC41.max-owd=5%. The related QoS attribute of the 2 routes DO NOT overlap.

Expected result : The following results must be obtained:

- Scenario 1: Route via AS3 must be selected (the decision is enforced by the average-owd)
- Scenario 2: Route via AS3 must be selected (the decision is enforced by the maxowd)

Execution date Result	 : 22/09/04 : Obtained results are as follows: Scenario 1: Route via AS3 is selected Scenario 2: Route via AS3 is selected
Failure level Remarks	: None : None
Test Reference	: TB_P2_FUNCT/RSEL/9
Test Purpose	: Check that the precision command line parameter is correctly handled for the average-owd QoS attribute.
Procedure	: Start as defined by this sub-group preamble. Conformance status is mandatory for all attributes.
	Set the precision for l-QC41.reserved-rate=50%. The related QoS attribute of the 2 routes overlap.
	Set the precision for l-QC41.available-rate=50%. The related QoS attribute of the 2 routes overlap.
	Set the precision for l-QC41.loss-rate=50%. The related QoS attribute of the 2 routes overlap.
	Set the precision for l-QC41.min-owd=20%. The related QoS attribute of the 2 routes

overlap.

Set the precision for l-QC41.max-owd=50%. The related QoS attribute of the 2 routes overlap.

- Scenario 1: Set the precision for 1-QC41.average-owd=50%. The related QoS attribute of the 2 routes overlap.
- Scenario 2: Set the precision for l-QC41.average-owd=2%. The related QoS attribute of the 2 routes DO NOT overlap.

Expected result : The following results must be obtained:

- Scenario 1: Route via AS2 must be selected (the decision is enforced by the jitter)
- Scenario 2: Route via AS3 must be selected (the decision is enforced by the average-owd)

Execution date : 22/09/04

Result

: Obtained results are as follows:

Scenario 1: Route via AS2 is selected

MESCAL42# sh local-qos-cl local qos-class 41 qos-class-id: 34 available-rate 2000 pr average-owd 10 priorit jitter 5 priority 7 ma loss-rate 200 priority max-owd 15 priority 5 min-owd 5 priority 4 m reserved-rate 2000 pri solution-option: loose	ass 41 iority 2 mandatory precis y 6 mandatory precision 5 ndatory 3 mandatory precision 50 mandatory precision 20 ority 1 mandatory precisi	ion 50 0 on 50									
MESCAL42# sh ip bgp local BGP table version is 0, l Status codes: s suppresse Origin codes: i - IGP, e	-qos-class 41 ocal router ID is 42.42.4 d, d damped, h history, p - EGP, ? - incomplete	2.1 stale, * valid, >	best, i - int	ernal							
local-qos-class 41 (loose	solution options) :										
Network	QC-id Next Hop	Metric LocPrf Wei	ght reserved-	rate available	-rate loss	s-rate min	-owd max-	owd avg-	owd jit	ter Pat	th
*> 193.251.128.0/19	34 2.2.2.1		0	600	400	1692	30	90	60	29	21i
Total number of prefixes	in the local-qos-class 41	: 1									
-	Scenario 2: R	oute via AS3	is select	ed							
MESCAL42# sh local-qos-cl local qos-class 41 qos-class-id: 34 available-rate 2000 pr average-owd 10 priorit jitter 5 priority 7 ma loss-rate 200 priority max-owd 15 priority 5 min-owd 5 priority 4 m reserved-rate 2000 pri solution-option: loose	ass 41 iority 2 mandatory precis y 6 mandatory precision 2 ndatory 3 mandatory precision 50 mandatory precision 50 andatory precision 20 ority 1 mandatory precisi	ion 50 on 50									
MESCAL42# sh ip bgp local BGP table version is 0, 1 Status codes: s suppresse Origin codes: i - IGP, e	-qos-class 41 ocal router ID is 42.42.4 d, d damped, h history, p - EGP, ? - incomplete	2.1 stale, * valid, >	best, i - int	ernal							
<pre>local-qos-class 41 (loose Network * 193.251.128.0/19 *></pre>	solution options) : QC-id Next Hop 34 2.2.2.1 34 3.3.3.5	Metric LocPrf Wei	ght reserved- 0 0	rate available 600 400	-rate loss 400 500	-rate min 1692 2680	-owd max- 30 40	owd avg- 90 75	owd jit 60 55	ter Pat 29 30	th 21i 31i
Total number of prefixes	in the local-qos-class 41	: 1									
Failure level :	None										

Remarks : None
Test Reference : TB_P2_FUNCT/RSEL/10

- *Test Purpose* : Check that the precision command line parameter is correctly handled for the jitter QoS attribute.
- *Procedure* : Start as defined by this sub-group preamble. Conformance status is mandatory for all attributes.

Set the precision for l-QC41.reserved-rate=50%. The related QoS attribute of the 2 routes overlap.

Set the precision for l-QC41.available-rate=50%. The related QoS attribute of the 2 routes overlap.

Set the precision for l-QC41.loss-rate=50%. The related QoS attribute of the 2 routes overlap.

Set the precision for l-QC41.min-owd=20%. The related QoS attribute of the 2 routes overlap.

Set the precision for l-QC41.max-owd=50%. The related QoS attribute of the 2 routes overlap.

Set the precision for l-QC41.average-owd=50%. The related QoS attribute of the 2 routes overlap.

- Scenario 1: Set the precision for l-QC41.jitter=10%. The related QoS attribute of the 2 routes overlap.
- Scenario 2: Set the precision for l-QC41.jitter=1%. The related QoS attribute of the 2 routes DO NOT overlap.

Expected result : The following results must be obtained:

- Scenario 1: The same Route in best effort and in l-QC41 plan must be selected (depends on BGP route selection process)
- Scenario 2: Route via AS2 must be selected (the decision is enforced by the jitter)

Execution date Result	 : 22/09/04 : Obtained results are as follows: Scenario 1: Route via AS2 is selected in best effort and in l-QC41 plan. Scenario 2: Route via AS2 is selected
Failure level Remarks	: None : The default bestpath selection configured compare router-id for identical q-eBGP paths. The router-id of MESCAL21 is set to 21.21.21.1 and the router-id of MESCAL31 is set to 31.31.31.1. So route via AS2 is selected in scenario 1.
Test Reference Test Purpose	: TB_P2_FUNCT/RSEL/11 : Validate the behaviour of q-BGP when mandatory parameters aren't received.
Procedure	: q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of reserved rate to mandatory:

- Scenario 1: Configure AS1, AS2 and AS3 to send reserved-rate as described in the introduction of this test group.
- Scenario 2: Configure only AS1 and AS3 to send reserved-rate as described in the introduction of this test group.
- Scenario 3: Configure only AS1 and AS2 to send reserved-rate as described in the introduction of this test group.
- Scenario 4: Configure only AS2 and AS3 to send reserved-rate as described in the introduction of this test group.

Expected result : The following results must be obtained:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. Selected route is R421.
- Scenario 2: Only one Route is received from AS3. This route is selected.
- Scenario 3: Only one Route is received from AS2. This route is selected.
- Scenario 4: No route has been received for the specified prefix.

Execution date : 22/09/04

Result : Obtained results are as follows:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. Selected route is R421.
- Scenario 2: Only one Route is received from AS3. This route is selected.
- Scenario 3: Only one Route is received from AS2. This route is selected.
- Scenario 4: No route has been received for the specified prefix

Failure level: NoneRemarks: None

Test Reference : **TB_P2_FUNCT/RSEL/12**

Test Purpose : Validate the behaviour of q-BGP when mandatory parameters aren't received.

Procedure

q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of available rate to mandatory:

- Scenario 1: Configure AS1, AS2 and AS3 to send available-rate as described in the introduction of this test group.
- Scenario 2: Configure only AS1 and AS3 to send available -rate as described in the introduction of this test group.
- Scenario 3: Configure only AS1 and AS2 to send available -rate as described in the introduction of this test group.
- Scenario 4: Configure only AS2 and AS3 to send available -rate as described in the introduction of this test group.

Expected result : The following results must be obtained:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
- Scenario 2: Only one Route is received from AS3. This route is selected.
- Scenario 3: Only one Route is received from AS2. This route is selected.

• Scenario 4: No route has been received for the specified prefix.

Execution date : 22/09/04

Result : Obtained results are as follows:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
- Scenario 2: Only one Route is received from AS3. This route is selected.
- Scenario 3: Only one Route is received from AS2. This route is selected.
- Scenario 4: No route has been received for the specified prefix.

Failure level : None *Remarks* : None

Test Reference : TB P2 FUNCT/RSEL/13 *Test Purpose* : Validate the behaviour of q-BGP when mandatory parameters aren't received. Procedure q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of loss rate to mandatory: Scenario 1: Configure AS1, AS2 and AS3 to send loss-rate as described in the introduction of this test group. Scenario 2: Configure only AS1 and AS3 to send loss-rate as described in the introduction of this test group. Scenario 3: Configure only AS1 and AS2 to send loss-rate as described in the introduction of this test group. Scenario 4: Configure only AS2 and AS3 to send loss-rate as described in the introduction of this test group. *Expected result* : The following results must be obtained: Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected Scenario 2: Only one Route is received from AS3. This route is selected. Scenario 3: Only one Route is received from AS2. This route is selected. Scenario 4: No route has been received for the specified prefix. Execution date · 22/09/04 Result : Obtained results are as follows: Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected. Scenario 2: Only one Route is received from AS3. This route is selected. Scenario 3: Only one Route is received from AS2. This route is selected. Scenario 4: No route has been received for the specified prefix. Failure level : None Remarks : None Test Reference : TB P2 FUNCT/RSEL/14

Test Purpose : Validate the behaviour of q-BGP when mandatory parameters aren't received.

Procedure

q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of min-owd to mandatory:

- Scenario 1: Configure AS1, AS2 and AS3 to send min-owd as described in the introduction of this test group.
- Scenario 2: Configure only AS1 and AS3 to send min-owd as described in the introduction of this test group.
- Scenario 3: Configure only AS1 and AS2 to send min-owd as described in the introduction of this test group.
- Scenario 4: Configure only AS2 and AS3 to send min-owd as described in the introduction of this test group.

Expected result : The following results must be obtained:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
- Scenario 2: Only one Route is received from AS3. This route is selected.
- Scenario 3: Only one Route is received from AS2. This route is selected.
- Scenario 4: No route has been received for the specified prefix.

Execution date : 22/09/04

Result : Obtained results are as follows:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
- Scenario 2: Only one Route is received from AS3. This route is selected.
- Scenario 3: Only one Route is received from AS2. This route is selected.
- Scenario 4: No route has been received for the specified prefix.

Failure level : None *Remarks* : None

Test Reference : **TB_P2_FUNCT/RSEL/15** *Test Purpose* : Validate the behaviour of a-BGP when mandatory parameter

Test Purpose : Validate the behaviour of q-BGP when mandatory parameters aren't received.

Procedure

q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of max-owd to mandatory:

- Scenario 1: Configure AS1, AS2 and AS3 to send max-owd as described in the introduction of this test group.
- Scenario 2: Configure only AS1 and AS3 to send max-owd as described in the introduction of this test group.
- Scenario 3: Configure only AS1 and AS2 to send max-owd as described in the introduction of this test group.
- Scenario 4: Configure only AS2 and AS3 to send max-owd as described in the introduction of this test group.

Expected result : The following results must be obtained:

	•	Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
	•	Scenario 2: Only one Route is received from AS3. This route is selected.
	•	Scenario 3: Only one Route is received from AS2. This route is selected.
	•	Scenario 4: No route has been received for the specified prefix.
Execution date Result	: 2 : C	2/09/04 Obtained results are as follows:
	•	Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
	•	Scenario 2: Only one Route is received from AS3. This route is selected.
	•	Scenario 3: Only one Route is received from AS2. This route is selected.
	•	Scenario 4: No route has been received for the specified prefix.
Failure level Remarks	: N : N	Ione
Test Reference Test Purpose	: T : V	B_P2_FUNCT/RSEL/16 Validate the behaviour of q-BGP when mandatory parameters aren't received.
Procedure	: q-I Se	BGP configuration will be update as specified in the description of each scenario: t the conformance status of average-owd to mandatory:
	•	Scenario 1: Configure AS1, AS2 and AS3 to send average-owd as described in the introduction of this test group.
	•	Scenario 2: Configure only AS1 and AS3 to send average-owd as described in the introduction of this test group.
	•	Scenario 3: Configure only AS1 and AS2 to send average-owd as described in the introduction of this test group.
	•	Scenario 4: Configure only AS2 and AS3 to send average-owd as described in the introduction of this test group.
Expected result	t : T	he following results must be obtained:
	•	Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
	•	Scenario 2: Only one Route is received from AS3. This route is selected.
	•	Scenario 3: Only one Route is received from AS2. This route is selected.
	•	Scenario 4: No route has been received for the specified prefix.
Execution date Result	: 2 : C	2/09/04 Obtained results are as follows:
	•	Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
	•	Scenario 2: Only one Route is received from AS3. This route is selected.
	•	Scenario 3: Only one Route is received from AS2. This route is selected.

• Scenario 4: No route has been received for the specified prefix.

Failure level Remarks	: None : None
Test Reference Test Purpose	: TB_P2_FUNCT/RSEL/17 : Validate the behaviour of q-BGP when mandatory parameters aren't received.
Procedure	: q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of jitter to mandatory:
	• Scenario 1: Configure AS1, AS2 and AS3 to send jitter as described in the introduction of this test group.
	• Scenario 2: Configure only AS1 and AS3 to send jitter as described in the introduction of this test group.
	• Scenario 3: Configure only AS1 and AS2 to send jitter as described in the introduction of this test group.
	• Scenario 4: Configure only AS2 and AS3 to send jitter as described in the introduction of this test group.
Expected result	t: The following results must be obtained:
	• Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
	 Scenario 2: Only one Route is received from AS3. This route is selected.
	 Scenario 3: Only one Route is received from AS2. This route is selected.
	 Scenario 4: No route has been received for the specified prefix.
Execution date Result	: 22/09/04 : Obtained results are as follows:
	• Scenario 1: Two routes are received by AS4, from AS3 and AS2. Route Route421 is selected.
	• Scenario 2: Only one Route is received from AS3. This route is selected.
	• Scenario 3: Only one Route is received from AS2. This route is selected.
	 Scenario 4: No route has been received for the specified prefix.
Failure level Remarks	: None : None
Test Reference Test Purpose	: TB_P2_FUNCT/RSEL/18 : Validate the behaviour of q-BGP when optional parameters aren't received.
Procedure	: q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of reserved rate to optional:
	• Scenario 1: Configure AS1, AS2 and AS3 to send reserved-rate as described in the introduction of this test group.
	• Scenario 2: Configure only AS1 and AS3 to send reserved-rate as described in the introduction of this test group.

- Scenario 3: Configure only AS1 and AS2 to send reserved-rate as described in the introduction of this test group.
- Scenario 4: Configure only AS2 and AS3 to send reserved-rate as described in the introduction of this test group.

Expected result : The following results must be obtained:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 2: Two routes are received by AS4, from AS3 and AS2. R431 is selected.
- Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 4: Two routes are received by AS4, from AS3 and AS2. R431 is selected.

Execution date : 22/09/04

Result : Obtained results are as follows:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 2: Two routes are received by AS4, from AS3 and AS2. R431 is selected.

MESCAL42# sh ip bgp local-qos-class 41 BGP table version is 0, local router ID is 42.42.42.1 Status codes: s suppressed, d damped, h history, p stale, * valid, > best, i - internal Origin codes: i - IGP, e - EGP, ? - incomplete local-qos-class 41 (loose solution options) : Metric LocPrf Weight reserved-rate available-rate loss-rate min-owd max-owd avg-owd jitter Path Network OC-id Next Hop * 193.251.128.0/19 34 2.2.2.1 400 1692 30 21i 31i 90 60 29 34 3.3.3.5 400 75 55 30 0 500 2680 40 Total number of prefixes in the local-gos-class 41 : 1 Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected. MESCAL42# sh ip bgp local-qos-class 41 BGP table version is 0, local router ID is 42.42.42.1 Status codes: s suppressed, d damped, h history, p stale, * valid, > best, i - internal Origin codes: i - IGP, e - EGP, ? - incomplete local-gos-class 41 (loose solution options) : Metric LocPrf Weight reserved-rate available-rate loss-rate min-owd max-owd avg-owd jitter Path Network QC-id Next Hop *> 193.251.128.0/19 34 2.2.2.1 2 1 i 0 600 4001692 30 90 60 29 55 30 31i 34 3.3.3.5 0 500 2680 40 75 Total number of prefixes in the local-qos-class 41 : 1 Scenario 4: Two routes are received by AS4, from AS3 and AS2. R431 is selected. MESCAL42# sh ip bgp local-qos-class 41 BGF table version is 0, local router ID is 42.42.42.1 Status codes: s suppressed, d damped, h history, p stale, * valid, > best, i - internal Origin codes: i - IGP, e - EGP, ? - incomplete local-gos-class 41 (loose solution options)

* 193.251.128.0/19 34 2.2.2.1 0 - 400 1692 30 90 6 *> 34 3.3.3.5 0 - 500 2680 40 75 5				
*> 34 3.3.3.5 0 - 500 2680 40 75 5) 29 21i	1 i	1 i	i
	30 31i	1 i	1 i	i

Total number of prefixes in the local-qos-class 41 : 1

Failure level: NoneRemarks: None

Test Reference : TB_P2_FUNCT/RSEL/19

Test Purpose : Validate the behaviour of q-BGP when optional parameters aren't received.

Procedure

q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of available rate to optional:

- Scenario 1: Configure AS1, AS2 and AS3 to send available-rate as described in the introduction of this test group.
- Scenario 2: Configure only AS1 and AS3 to send available -rate as described in the introduction of this test group.
- Scenario 3: Configure only AS1 and AS2 to send available -rate as described in the introduction of this test group.
- Scenario 4: Configure only AS2 and AS3 to send available -rate as described in the introduction of this test group.

Expected result : The following results must be obtained:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 2: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 4: Two routes are received by AS4, from AS3 and AS2. R421 is selected.

Execution date : 22/09/04

Result : Obtained results are as follows:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 2: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 4: Two routes are received by AS4, from AS3 and AS2. R421 is selected.

Failure level: NoneRemarks: None

Test Reference : **TB_P2_FUNCT/RSEL/20**

Test Purpose : Validate the behaviour of q-BGP when optional parameters aren't received.

Procedure

q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of loss rate to optional:

- Scenario 1: Configure AS1, AS2 and AS3 to send loss-rate as described in the introduction of this test group.
- Scenario 2: Configure only AS1 and AS3 to send loss-rate as described in the introduction of this test group.

- Scenario 3: Configure only AS1 and AS2 to send loss-rate as described in the introduction of this test group.
- Scenario 4: Configure only AS2 and AS3 to send loss-rate as described in the introduction of this test group.

Expected result : The following results must be obtained:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 2: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 4: Two routes are received by AS4, from AS3 and AS2. R421 is selected.

Execution date : 22/09/04

Result : Obtained results are as follows:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 2: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 4: Two routes are received by AS4, from AS3 and AS2. R421 is selected.

Failure level: NoneRemarks: None

Test Reference : TB_P2_FUNCT/RSEL/21

Test Purpose : Validate the behaviour of q-BGP when optional parameters aren't received.

Procedure

q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of min-owd to optional:

- Scenario 1: Configure AS1, AS2 and AS3 to send min-owd as described in the introduction of this test group.
- Scenario 2: Configure only AS1 and AS3 to send min-owd as described in the introduction of this test group.
- Scenario 3: Configure only AS1 and AS2 to send min-owd as described in the introduction of this test group.
- Scenario 4: Configure only AS2 and AS3 to send min-owd as described in the introduction of this test group.

Expected result : The following results must be obtained:

• Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.

- Scenario 2: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 4: Two routes are received by AS4, from AS3 and AS2. R421 is selected.

Execution date : 22/09/04

Result : Obtained results are as follows:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 2: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 4: Two routes are received by AS4, from AS3 and AS2. R421 is selected.

Failure level: NoneRemarks: None

Test Reference : TB P2 FUNCT/RSEL/22

Test Purpose : Validate the behaviour of q-BGP when optional parameters aren't received.

Procedure

q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of max-owd to optional:

- Scenario 1: Configure AS1, AS2 and AS3 to send max-owd as described in the introduction of this test group.
- Scenario 2: Configure only AS1 and AS3 to send max-owd as described in the introduction of this test group.
- Scenario 3: Configure only AS1 and AS2 to send max-owd as described in the introduction of this test group.
- Scenario 4: Configure only AS2 and AS3 to send max-owd as described in the introduction of this test group.

Expected result : The following results must be obtained:

- Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 2: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
- Scenario 4: Two routes are received by AS4, from AS3 and AS2. R421 is selected.

Execution date : 22/09/04

Result : Obtained results are as follows:

	•	Scenario selected.	1:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
	•	Scenario selected.	2:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
	•	Scenario selected.	3:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
	•	Scenario selected.	4:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
Failure level Remarks	: N : N	one													
Test Reference Test Purpose	: T : V	B_P2_FU alidate the	NC e be	C T/RS haviou	EL/23 ur of q-]	BGP	when opt	iona	ıl para	meters	aren'	t rece	eived.		
Procedure	: q-E Set	BGP configure the confo	gura orma	ation v ance s	vill be u tatus of	ipdat aver	te as speci age-owd	ified to op	in the	descr	iption	of ea	ach sce	enario:	
	•	Scenario introduct	1: (tion	Config of thi	gure AS s test gr	1, A oup.	S2 and A	S3 to	o send	avera	ge-ow	d as	descril	bed in t	the
	•	Scenario introduct	2: (Config of thi	gure on s test gr	ly As oup.	S1 and AS	53 to	o send	averaş	ge-ow	d as	descrit	bed in t	the
	•	Scenario introduct	3: 0	Config of thi	gure on s test gr	ly As oup.	S1 and AS	52 to	send	averaş	ge-ow	d as	descrit	bed in t	the
	•	Scenario introduct	4: (Config of thi	gure on s test gr	ly As oup.	S2 and AS	53 to	o send	averaş	ge-ow	d as (descrit	bed in t	the
Expected result	: T	he followi	ngı	results	s must b	e ob	tained:								
	•	Scenario selected.	1:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
	•	Scenario selected.	2:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
	•	Scenario selected.	3:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
	•	Scenario selected.	4:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
Execution date Result	: 22 : O	2/09/04 btained re	sult	s are a	as follov	ws:									
	•	Scenario selected.	1:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
	•	Scenario selected.	2:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
	•	Scenario selected.	3:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is
	•	Scenario selected.	4:	Two	routes	are	received	by	AS4,	from	AS3	and	AS2.	R421	is

Failure level Remarks	: None : None
Test Reference Test Purpose	: TB_P2_FUNCT/RSEL/24 : Validate the behaviour of q-BGP when optional parameters aren't received.
Procedure	: q-BGP configuration will be update as specified in the description of each scenario: Set the conformance status of jitter to optional:
	• Scenario 1: Configure AS1, AS2 and AS3 to send jitter as described in the introduction of this test group.
	• Scenario 2: Configure only AS1 and AS3 to send jitter as described in the introduction of this test group.
	• Scenario 3: Configure only AS1 and AS2 to send jitter as described in the introduction of this test group.
	• Scenario 4: Configure only AS2 and AS3 to send jitter as described in the introduction of this test group.
Expected result	t: The following results must be obtained:
-	 Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
	• Scenario 2: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
	• Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
	• Scenario 4: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
Execution date Result	: 22/09/04 : Obtained results are as follows:
	• Scenario 1: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
	• Scenario 2: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
	• Scenario 3: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
	• Scenario 4: Two routes are received by AS4, from AS3 and AS2. R421 is selected.
Failure level Remarks	: None : None
10.2.5TB_F	P2_FUNCT/INT

Test Reference : **TB_P2_FUNCT/INT/1** *Test Purpose* : Validate the behaviour of a BGP speaker when receiving unsupported capability.

Procedure

q-BGPD process is activated in MESCAL11. BGPD is activated in MESCAL21.

- Scenario 1: Configure MESCAL11 as a neighbor of MESCAL21. MESCAL21 is to be configured in MESCAL11 as a neighbor. Configure MESCAL11 to support QoS Service Capability Group 1. Check the messages exchanged between MESCAL21 and MESCAL31 with a traffic analyzer.
- Scenario 2: Configure MESCAL11 as a neighbor of MESCAL21. MESCAL21 is to be configured in MESCAL11 as a neighbor. Configure MESCAL11 to support QoS Service Capability Group 2. Check the messages exchanged between MESCAL21 and MESCAL31 with a traffic analyzer.
- Scenario 3: Configure MESCAL11 as a neighbor of MESCAL21. MESCAL21 is to be configured in MESCAL11 as a neighbor. Configure MESCAL11 to support QoS Service Capability Group 1 and Group 2. Check the messages exchanged between MESCAL21 and MESCAL31 with a traffic analyzer.

Expected result : The following results must be obtained:

- Scenario 1: MESCAL21 has to send a notification message with the Error Sub Code set to Unsupported Capability. MESCAL11 should re-attempt to open a BGP session with MESCAL21 but without sending to the peer the Capabilities Optional Parameter.
- Scenario 2: MESCAL21 has to send a notification message with the Error Sub Code set to Unsupported Capability. MESCAL11 should re-attempt to open a BGP session with MESCAL21 but without sending to the peer the Capabilities Optional Parameter.
- Scenario 3: MESCAL21 has to send a notification message with the Error Sub Code set to Unsupported Capability. MESCAL11 should re-attempt to open a BGP session with MESCAL21 but without sending to the peer the Capabilities Optional Parameter.

Execution date : 22/09/04

Result : Obtained results are as follows:

- Scenario 1: The BGP session is established and MESCAL21 has not sent any notification message.
- Scenario 2: The BGP session is established and MESCAL21 has not sent any notification message.
- Scenario 3: The BGP session is established and MESCAL21 has not sent any notification message.

Failure level : None

Remarks : ZeboS BGP process has been implemented so that unknown capabilities are by default ignored. In order to validate the test, MESCAL11 must be configured as a neighbor of MESCAL21 with the "strict capability match" command. This command permits to close the BGP connection if capability value does not completely match to remote peer. If this command is used and if an unknown capability. If the capability code is greater or equal to 128, the capability is a "for private use" capability and is only ignored. If the capability code is lower than 128, a notification message with the Error Sub Code set to Unsupported Capability is sent. In this case the results of the tests match the expected ones.

Test Reference : TB_P2_FUNCT/INT/2

Test Purpose : Validate the behaviour of a q-BGP speaker when receiving notification set to unsupported capabilities from BGP speaker.

Procedure

q-BGPD process is activated in MESCAL11. BGPD is activated in MESCAL21.

Configure local-QoS-class in MESCAL11 as specified in the introduction of the previous test group. Also AS1 networks that are listed in testbed configuration are also to be announced by MESCAL11 in l-QC1 and best effort.

AS2 Networks that are listed in testbed configuration are also to be announced by MESCAL21.

Launch a traffic analyzer in eth1 of MESCAL11. Check the messages that are exchanges between MESCAL11 and MESCAL12.

Expected result : MESCAL11 must not send QoS_NLRI messages to MESCAL21. MESCAL21 must send NLRI information to MESCAL11. MESCAL11 must send NLRI information to MESCAL21.

Execution date	: 22/09/04
Result	:
	MESCAL11 does not send any QoS_NLRI messages to MESCAL21.
	MESCAL21 sends NLRI information to MESCAL11.
	MESCAL11 sends NLRI information to MESCAL21.
Failure level	: None
Remarks	: None

Test Reference : TB_P2_FUNCT/INT/3

Test Purpose : Validate the q-BGP router installs routes received from BGP speaker in best effort plane.

Procedure :

q-BGPD process is activated in MESCAL11. BGPD is activated in MESCAL21.

Configure local-QoS-class in MESCAL11 as specified in the introduction of the previous test group. Also AS1 networks that are listed in testbed configuration are also to be announced by MESCAL11 in l-QC1 and best effort.

AS2 Networks that are listed in testbed configuration are also to be announced by MESCAL21.

Log to MESCAL11 and execute this command line: "**sh ip bgp**". Check if MESCAL21 networks are listed.

Expected result : All MESCAL21's networks must be present in the MESCAL11 best effort RIB.

Execution date	: 22/09/04
Result	: All MESCAL21's networks are present in the MESCAL11 best effort RIB.
Failure level	: None
Remarks	: None

Test Reference : TB_P2_FUNCT/INT/4

Test Purpose : Validate the BGP router installs routes received from q-BGP speaker.

Procedure

q-BGPD process is activated in MESCAL11. BGPD is activated in MESCAL21.

Configure local-QoS-class in MESCAL11 as specified in the introduction of the previous test group. Also AS1 networks that are listed in testbed configuration are also to be announced by MESCAL11 in l-QC1 and best effort.

AS2 Networks that are listed in testbed configuration are also to be announced by MESCAL21.

Log to MESCAL21 and execute this command line: "**sh ip bgp**". Check if MESCAL11 networks are listed.

Expected result : All MESCAL11's networks must be present in the MESCAL21 BGP RIB.

Execution date	: 22/09/04
Result	: All MESCAL11's networks are present in the MESCAL21 BGP RIB.
Failure level	: None
Remarks	: None

10.3Phase 3

The "null" value means no value.

10.3.1TB_P3_FUNCT/CMES

10.3.1.1 Reminder

Each PCP message consists of the PCP header followed by a number of arguments depending on the nature of the operation.

0	1	2	3
Version	Op Code	Message	Length

10.3.1.1.1 Open message



10.3.1.1.2 Accept message



10.3.1.1.3 Close message





10.3.1.1.4 Path Error message



10.3.1.1.5 Cancel message



10.3.1.1.6 Acknowledge message



10.3.1.1.7 Request message



10.3.1.1.8 Response message





10.3.1.2 pSLS agreement

Configure the following pSLS between AS1 and AS2 (AS2 offers this pSLS to AS1):

- List of Meta-QoS-classes and bandwidth:
 - MQ1: TOS value 0x68 bandwidth 1Mbit
 - MQ2: TOS value 0x71 bandwidth 1Mbit
 - MQ3: TOS value 0x78 bandwidth 1Mbit
 - MQ4: TOS value 0x00 bandwidth 2Mbit
- Total bandwidth: 5Mbit
- MESCAL11'PCSID: 11.11.11.11
- MESCAL21'PCSID: 21.21.21.21

Test Reference : **TB_P3_FUNCT/CMES/1**

Test Purpose	: Check the format of OPEN, CLOSE and ACCEPT messages.
Procedure	: Launch a traffic analyzer that captures MESCAL11' eth1 traffic.
	Establish a PCP session between MESCAL11'PCS and MESCAL21'PCS.
	When session is accepted by MESCAL21'PCS, close the session.
	Verify that OPEN, CLOSE and ACCEPT messages are conform to [D1.2] specifications. For more details see introduction of this test group.

Expected result :

MESCAL21'PCS should answer with CLOSE or ACCEPT message to OPEN request received from MESCAL11'PCS.

OPEN, CLOSE and ACCEPT messages should have a format as indicated in introduction of this test group.

Op Code contained in common header is:

- 1: OPEN
- 2: ACCEPT
- 3: CLOSE

Execution date : 21/04/05

Result : The format of the exchanged messages is aligning with the specifications. Opcode of recived messaged are as follows:

Cat Costas Disatau	Tests			- 0
East Capture Dispitay	Toos	Bustown	r lue	<u></u>
+ Time Source	Destination	Protocol	C IND	
2 0,000088 mescal111	mencal212	TOP	40022 > bgp (ACK) Seg=397828488 Ack=3933938325 Win=16705 Len=0	
3 1.494417 14.0.0.2	71.0.0.2	UDP	Source port: 32006 Destination port: 5050	
4 1.564027 mescal111	mescal212	RCb	KEEPILIVE Message	
5 1,564211 mescal212	nescal111	TEP	bgp > 40022 [ROK] Seq1993936525 Rok:1997828507 Wint17220 Lent0 00411 > 1040 [Stml] Seq1993936525 Rok:1997828507 Wint17220 Lent0	
7 2.068182 14.0.0.2	24.0.0.2	TCP	1040 > 60411 [SYN, 40K] Sev=3760868461 Ack=3781884946 Min=5782 Len=0	
8 2,088299 24,0,0,2	14,0,0,2	TCP	60411 > 1040 [ACK] Seq=3701004946 Ack=3760060462 Win=5040 Len=0	
9 2,088996 24,0.0,2	14.0.0.2	TCP	60411 > 1040 [PSH, ACK] Seq=5781884945 Ack=5760888452 Win=5840 Len=8	
10 2,089078 16,0,0,2	26,0,0,2	TOP	1040 > 50411 [HUK] Seq#5760888462 HUK-5751884554 Min45752 Len40 1040 > 50411 [HUK] Seq#3700988462 HUK-5791884954 Min45792 Len40	
2 2,090264 24,0,0,2	14.0.0.2	TCP	60411 > 1040 [ACK] Seq=3781884954 Ack=3760868470 Win=5840 Len=0	
13 2,092330 14,0,0,2	26,0,0,2	TCP	34409 > 1040 [SYN] Seqt3771127280 Rck:r0 Winn5040 Lenr0	
4 2,052446 24,0.0.2	14.0.0.2	TCP	1040 > 34489 [SYN, RCK] Seq=3790544462 Ack=3771127281 Win=5792 Len=0	
15 2.092907 14.0.0.2	24.0.0.2	TCP	34489 > 1040 [ACK] Sen=3771127281 Ack=3790544463 Win=5840 Len=0	
16 2,093965 14,0,0,2	26,0,0,2	TCP	34483 > 1040 [PSR, HCK] Segr5771127281 Ack:5790544465 Wint5840 Lent0 1040 > 24489 [0/2] Segr577127280644652 Col=2771127299 MissE292 Lent0	
18 2 276766 24 0 0 7	14.0.0.2	TOP	1040 > 34499 (FKH (1) Several 9544485 Hits of 1273117799 Hits 7542 Laters	
3 2,304046 14,0.0.2	24.0.0.2	TUP	34403 > 1040 [ROC] Sec=3771127209 Ack=3790544471 Win=5040 Len=0	
0 2,311953 14,0,0,2	71.0.0.2	UDP	Source port: 32806 Destination port: 5050	
1 5,009195 mescal212	mescal111	ICP	KEEPHLIVE Message	
Z 5.009293 mescal111	mescal212	TCP	40022 > bgp [RCK] Seq=997828507 Rok=993936344 Win=16705 Len=0	
2 5.003233 mercal111 3 6.574380 mercal111 4 6 576322 mercal111	mescal212 mescal212	TCP BCP	40022 > bgp [RCK] Seq=997828507 Ack=993936344 Win=16705 Len=0 KETPALIVE Neccode Lan A Advance (Arc) E-comparamental Ack-advanceDe (Nove12700 Long)	
2 5.003233 mescal111 3 6.574380 mescal111 4 6.575023 mescal212 5 10.009262 mescal212	mescal212 mescal212 mescal111 mescal111	TCP BCP TCP BCP	4002 2 bpp (4K) 544707020507 6K×35336544 Win=16705 Lene0 UETPHLIVE Hessage bp3 > 40022 (4K) 544705306344 Ack-897020526 Win=12520 Leve0 UETPELIVE Hermane	
22 5.07235 mercal111 23 5.574800 mercal111 24 5.575903 mercal122 35 10.009525 mercal212 36 10.009526 mercal212 36 10.009526 mercal212 36 10.009526 mercal212 36 10.009526 mercal212 36 10.009526 mercal21 36 10.009526 mercal21 36 10.009526 mercal21 36 10.001526 mercal21 36 10.001526 mercal21 37 10.001526 mercal21 36 10.001526 mercal21 37 10.001526 merca	mescal212 mescal212 mescal111 mescal111 mescal111 setsil11 setsil11 setsil11 setsil11 setsil11 setsil11 setsil111 setsil1111 setsil11111 setsil111111 setsil1111111 setsil111111111111111111111111111111111111	ICP BCP TCP BCP TCP 1:50:56d t Addr: 14.0.0 1), Det Port: ;	4002 / bpc (FAIG) Sept2010070 Fix-313335244 Win-12170 Levno ETTM: UT Finance 2010 Fix-31335244 Ani-4335245254 Win-1220 Levno Martin UT Fix-1000 Fix-31352544 Ani-433525555 Win-1220 Levno Martin UT Fix-1000 Fix-31352544 Ani-433525555 Win-1220 Levno Martin UT Fix-31355554 Ani-4335255555 Win-1220 Levno Martin UT Fix-31355554 Ani-4335255555 Win-1220 Levno Ani-220 Levno	
22 3.07233 meraill 25 5.07480 menaill 25 5.07480 menaill 25 10.00252 menaill 25 10.00252 menaill 25 10.00252 menaill merail 10.00550 menaill meraill	mercal212 mercal212 mercal212 mercal212 mercal222 bdret separation 24.0.0.2 (24.0.0.2), hr 5 % Parts 60411 (6041 40) 8 8080554 90080462	100 160 100 100 100 100 1150:bd 1:50:bd 1:50:bd 1:50:bd 1:50:bd 1:50:bd 1:50:bd	4002 / bp [KG] Sep2702007 As13335244 Win-12076 Levn TETN LTF Resp. bg . Stor [HO] Sep27021524 Ac1-30320526 Win-1220 Levn 4002 / bp [KG] Sep27021526 Win-103038535 Win-12076 Levn 4002 / bp [KG] Sep2702152 Win-1000 [km] Sep2702152 Win-12076 Win-1	
22 3.00235 metalli 22 3.00255 metalli 5 0.00255 m	metcal 212; metcal 212; metcal 212; metcal 212; metcal 212; metcal 212; metcal 212; metcal 212; doi:10.1002/02; doi:0.1002/02; doi:0.1002/02; <td>107 167 167 179 179 179 179 179 179 179 179 179 17</td> <td>4002 / bp [KG] Sep2702007 As:3533544 Win:12076 Lev0 TETRI UT Ansam Mg / A002 [A0] Sep3335264 AcA:35325524 Win:1225 Lev0 4002 / bp [KG] Sep37025555 Ha:353535833 Win:15076 Lev0 4002 / bp [KG] Sep37025555 Ha:353558833 Win:15076 Lev0 12(14.0.72) 1040 (1040), Sep3 3721588446, Ha:3 3750958862, Lev; 0</td> <td></td>	107 167 167 179 179 179 179 179 179 179 179 179 17	4002 / bp [KG] Sep2702007 As:3533544 Win:12076 Lev0 TETRI UT Ansam Mg / A002 [A0] Sep3335264 AcA:35325524 Win:1225 Lev0 4002 / bp [KG] Sep37025555 Ha:353535833 Win:15076 Lev0 4002 / bp [KG] Sep37025555 Ha:353558833 Win:15076 Lev0 12(14.0.72) 1040 (1040), Sep3 3721588446, Ha:3 3750958862, Lev; 0	

• 2: ACCEPT

and

❤ <capture> - Ethernal</capture>			- 6 8
Eile Edit Capture Display I	[ools		Help
No., Time Source 1 0.000000 mescal212	Destination Protoc recoal111 BCP	ICOI INO IEEPIILIVE Nessage	-
2 0,000083 mescal111 3 1,494417 14,0,0,2	mescal212 TCP 71.0.0.2 UBP	40022 > bgp [AOX] Seu4897828488 Ack=893338325 Win=16705 Len=0 Source port: 32006 Destination port: 5050	
4 1.564027 mescal111 5 1.564211 mescal212	mescal212 BGP mescal111 TCP	NEOPALIVE Message bgp > 40022 (ACX) Seq=9938386325 Ack=9978289507 Win=17220 Len=0	
6 2.088094 24.0.0.2 7 2.088182 14.0.0.2	14.0.0.2 TCP 24.0.0.2 TCP	60411 > 1040 (SYN) Seq=3781884945 Hok=0 Win=5840 Len=0 1040 > 60411 (SYN, HOK) Seq=3760868461 Hok=3781884946 Win=5782 Len=0	
9 2.000203 24,0,0,2 9 2.000395 24,0,0,2 10 2.089079 14 0.0 2	14.0.0.2 TCP 14.0.0.2 TCP 24.0.0.2 TCP	99-81.8 / 2009 [2:64,3 5847370100408 HEX7370000462 V11775000 (ANTO 60411 > 1040 [2:94, EXX] 5847370100496 Ext.3705000462 V1175040 Lenr0 1040 > 50411 [0:01] 55447370505025 (Aut.270505025 (Aut.27050025)0	
11 2.090164 14,0,0,2 12 2.090264 24,0,0,2	24,0,0,2 TCP 14,0,0,2 TCP	1040 > 50411 [F9H, KOK] Seve370088462 (kH-578184454 Win-5782 Len-8 60411 > 1040 (ACK) Seq:3701884954 Ack:376088470 Win:5840 Len:0	
13 2,092330 14,0,0,2 14 2,092446 24,0,0,2	24,0,0,2 TCP 14,0,0,2 TCP	34409 > 1040 [STN] Seq=3771127200 Ask=0 Win=5040 Len=0 1040 > 34489 [STN, ACX] Seq=3790544462 Ack=3771127201 Win=5792 Len=0	
15 2.092507 14.0.0.2 16 2.093665 14.0.0.2	24.0.0.2 TCP 24.0.0.2 TCP	34489 > 1040 [ACK] Seq=3771127281 Ack=3790544453 Win+5840 Len=0 34489 > 1040 [PSH, ACK] Seq=3771127281 Ack=3790544463 Win+5840 Len=8	
17 2.093769 24.0.0.2 18 2.276766 24.0.0.2	14.0.0.2 TEP 14.0.0.2 TEP	1040 > 34489 [ACX] Seg=3790544463 Ack=3771127289 Win=5792 Len=0 1040 > 34489 [PSH, ACX] Seg=3790544463 Ack=3771127289 Win=5792 Len=0	
19 2.304046 14.0.0.2 20 2.311953 14.0.0.2	24.0.0.2 TCP 71.0.0.2 UDP	8489 > 1040 (ACK) Seq=87/1127289 Ack=879644471 Win=8840 Lem=0 Source port: 32806 Destination port: 5050	
21 5,009195 mescal212 22 5,009293 mescal111	sescal212 TCP	KEEP94_IVE_Tensage 40022 > bgp_(ACX) Seq=997029507 Ack=993906344 Win=16705 Len=0	
23 6.574800 mescal111 24 6.575023 mescal212 25 10 000000 mescal212	seccal111 TCP	NELTYNL IVE Heessage byg > 40022 (RCK) Seq=993336344 Rck=997828525 Win=17220 Len=0 HETTRM The Heessan	
25 10,005252 mescal212 26 10,009349 mescal111	mescal212 TCP	ACC2796, IVE: Resider 40002 > bop [ADX] Seqr997829526 Ack::993936363 Winn16705 Lenn0	
millowerk II, Seci Weshidthing Branchistor Rotern Proceed. B Transmission Rotern Proceed. Services meder: Throwson Bertrated more 154411 (Brit Services meder: Throwson Britage Longit: Styles Britage Longit: Styles Britage Longit: Styles Dackmark: Object Development Deckmark: Object Development Big Longit: Cit Bytes Tate (Ebstes) Tate (Ebstes)	(551, taeor 180077156	0.4.2 (24.0.4.2) 86411 (66411), See: 3700086442, Ask: 37818864964. Len: 0	
★1 0020 00 2c 58 42 40 00 40 05 bc 0020 00 02 04 15 50 00 c0 05 bc 0039 15 a 15 00 00 01 00 10 0 0040 83 24 13 00 00 01 00 50 00	76 0e 00 00 02 18 00	Manana Inggana Manana	•
Filter		V Reset Apply Data (data), 8 bytes	
	mescal O N	MESCAI @mescal@MESCAI @rccapture> - Ethen Path Computation @(Network Configur. 4)	u Apr 21, 14:40
■ 3·C	'I OSE		
5.0	LODE		
 captures - Ethereal File Edit Canture Display T 	ools		- C X
No. Time Source	Destination Protoco	col * info	
15 4.444131 14.0.0.2 20 4.451933 14.0.0.2	71,0.0.2 UDP 71,0.0.2 UDP	Source port: 32006 Destination port: 5050 Source port: 32006 Destination port: 5050	
24 4.495360 14.0.0.2 2 0.000071 mescal111 2 0.000071 mescal111	71,0,0,2 UDP mescal212 TOP	Source port: 32006 Destination port: 5050 #0022 > bigs (MAX) Seeu-0397530730 McA-0393338567 Win-15705 Law-0	
5 0.852568 14.0.0.2 4 0.832750 24.0.0.2 5 1.015419 24.0.0.2	24.0.0.2 10P 14.0.0.2 T0P 14.0.0.2 T0P	1040 / 60450 [FSH, HCL] beel-55/285224 HoH-555823405 Win/57/2 Len/4 60450 / 1040 [A0X] Seq435923555 Adu/337258228 Win/5840 Len/0 60450 / 1040 [FSH, HCL] Seq4359825705 Adu/337258228 Win/5840 Len/d	
6 1.019490 14.0.0.2 7 1.019646 24.0.0.2	24.0.0.2 TOP 14.0.0.2 TOP	1040 > 60450 [ACK] Seqi3372983228 Acki3988238/57 Wint5792 Lenit0 1040 > 34528 [P5H, ACK] Seqi3365347134 Acki3881188507 Wint5792 Lenit4	3
8 1.032548 14.0.0.2 9 1.032742 24.0.0.2	24.0.0.2 TOP 14.0.0.2 TOP	34520 > 1040 [PSH, 4CX] Seq#3901100507 Ackx3986547130 Wine5640 Levn4 1040 > 34528 [ACX] Seq#3906347138 Ackx3981108511 Wine5792 Levn0	
11 2,097000 mescal212 12 4,441120 14,0,0,2	sencal111 TOP 24,0.0.2 TOP	bgs > 40022 (ACK) Sec-903038567 Ack-907030749 Winc17220 Level 1040 > (04450 (MB), ACX) Sec-9022010520 Ack-90023037 Winc5202 Level 40465 - ack-forch 4, ack-9022010520 Ack-90023037 Winc5202 Level	
15 4,44556 14,0,0,2 16 4,44556 14,0,0,2	14.0.0.2 10 ^p 14.0.0.2 10 ^p 24.0.0.2 10 ^p	0450 / 2440 [Fith, BCC] Seq 2000223007 Ref: 202206206 Mint Select Center 0450 / 2440 [Fith, BCC] Seq 2000233067 Ref: 202206 Mint Select Lenno 1404 / Badfo [Fith, BCC] Seq 20202328 Ref: 200023240 Mint Select Lenno	
17 4,445741 24,0,0,2 18 4,449913 14,0,0,2	14.0.0.2 TOP 24.0.0.2 TOP	00450 > 1040 [ACX] Seq-3306233058 Ask-3372303237 Win=5840 Lenv0 34520 > 1040 [PSH, ACX] Seq-3301105511 Ack-3306347130 Win=5840 Lenv8	- 1
19 4.450099 24.0.0.2 21 4.451909 24.0.0.2	14.0.0.2 TOP 14.0.0.2 TOP	1040 > 34528 [ADK] 5eq=3306347138 Adk=3301100519 Win=5792 Len=0 1040 > 34528 [FIN, ADK] 5eq=3308547138 Adk=3301100519 Win=5792 Len=0	
22 4,453694 14,0,0,2 23 4,453886 24,0,0,2	24.0.0.2 TOP 14.0.0.2 TOP	34528 > 1040 [F1N, R0K] Seq=3981188619 Ack=3986347139 Win=5840 Lem=0 1040 > 34628 [R0K] Seq=3986347139 Ack=398118850 Win=5928 Lem=0	
26 5.009595 mescal111 28 7.097858 mescal212 70 10 009525 mescal212	mescal212 10P mescal111 TCP	40022 > bgp: [AC] Seq-957333743 Hok-95933866 Vin-157050 Lenv0 bgp > 40022 [ACK] Seq-93333566 Ack-937330768 Vin-157050 Lenv0	- 1
30 10,009525 mescal111 32 12,098446 mescal212	MESCAI212 10P	40022 > bgg (HKX, See-959'59/H5 HK-9595380/F6 HK-15/75 LmH0 bge > 40022 (HKX) See-953338505 HK-957830/767 Min+17220 Len+0	-
<pre>0 Free 12 Cf bytes on size. 74 (0:Element L), circl dividual/circl 0:Element L), circl dividual/circl 0:Element L), circl dividual/circl 0:Elements.circl dividual/circl 0:Elements.circl dividual/circl 0:Elements.circl dividual/circl 0:Elements.circl dividual/circl 0:Elements.circl dividual/circl 0:Elements.circl dividual/circl 0:Elements.circl dividual/circl 0:Elements.circl dividual/circl 0:Elements.circ</pre>	Mete General System General 2014, Bet 05/02/064 2014, Bet 05/02/06 Ben Pert 1808 (1948), Bet Port 5 50 50 1025 203007	0.2 (24,0,0,2) 19409 (59459), Seq: 3372593228, Hoi: 3398233957, Len: 8	<u></u>
■ ■ ■ ■ ■ ■ ■ ■ ■ ■ ■ ■ ■ ■	12 0+ 00 00 00 18 00 - (-, 4, 4, 4) 6+ 50 18 40 19 0 0 0 0 0 17 63 0+ 44 	Mescal @ MESCAL @+capture - Ethen _Path Computation _Phenock Configur, @ Thu A	pr 21, 14:43
<i>vel</i> : None. : None			
ence : TB_P3	B FUNCT/C	CMES/2	

Procedure

:

Launch a traffic analyzer that captures MESCAL11' eth1 traffic.

Establish a PCP session between MESCAL11'PCS and MESCAL21'PCS.

- Scenario 1: When PCP session is accepted by MESCAL21'PCS, configure MESCAL11'PCS to send a request for an LSP terminating in 2.2.2.1 in MC1 and a bandwidth of 1Mbit. No QoS parameters to be included in the request.
- Scenario 2: Configure MESCAL11'PCS to send a request for an LSP terminating in 2.2.2.1 in MC2 and a bandwidth of 2Mbit. No QoS parameters to be included in the request.

Verify that REQUEST, RESPONSE, PATH-ERROR and ACKNOWLEDGE messages are conform to [D1.2] specifications. For more details see introduction of this test group.

Expected result :

- Scenario 1:
 - MESCAL21'PCS should answer with CLOSE or ACCEPT message to OPEN request received from MESCAL11'PCS.
 - o MESCAL21'PCS should send a RESPONSE-PATH to MESCAL11
 - REQUEST, RESPONSE and ACKNOWLEDGE messages should have a format as indicated in introduction of this test group. Op Code contained in common header is:
 - 4: REQUEST
 - 5: RESPONSE
 - 8: ACKNOWLEDGE
- Scenario 2:
 - o MESCAL12'PCS should send a PATH-ERROR to MESCAL11.

Execution date : 21/04/05

Result : Obtained results are as follows:

- Scenario 1:
 - MESCAL21'PCS sent an ACCEPT message to OPEN request received from MESCAL11'PCS.
 - o MESCAL21'PCS sent a RESPONSE-PATH to MESCAL11
 - REQUEST, RESPONSE and ACKNOWLEDGE messages have a format as indicated in introduction of this test group. Op Code contained in common header is:
 - 4: REQUEST

Contract Print of Contract Contract				
Elle Edit Capture Display	Tools			Help
No. Time Source	Destination	Protocol *	info	
3 0.421344 14.0.0.2 2 0.000100 mescal111	71.0.0.2 weecal212	TOP	Source port: 52806 Destination port: 5080 40022 > bgp [ACK] Seq=987852934 Ack=993940771 Vin=16706 Lem=0	
5 2,006881 seecal212 5 2,129745 14 0,0.2	nescal111	TOP	bgp > 40022 [40X] Seq=993940771 Ack=997832953 V[re=17220 Ler=0 24772 > 1000 [F914] W11 Seq=555747981 Get#69708240 V[re=0840 Ler=18	
7 2.130704 24.0.0.2	14.0.0.2	TOP	1040 > 34722 [ACK] Seq=68/284340 Ack-666747131 Vin=5792 Len=0	
9 2.311089 14.0.0.2	24.0.0.2	TOP	34722 > 1040 [ACK] Seq=666747131 Ack+680284382 Win+5840 Len=0	
10 3,145202 14,0,0,2 11 3,145399 24,0,0,2	24.0.0.2	TOP	34722 > 1040 [PSH, HCK] Seq=886747131 Rck=880284382 Vin=5840 Len=18 1040 > 34722 [ACK] Seq=680284382 Rck=666747149 Vin=5792 Len=0	
13 5,010087 mescal111 15 7.012721 mescal212	mescal212	TOP	40/02 > bgp [ADX] Seq=997832953 Ack=993940790 Mix=16705 Lev=0 bas > 40/02 [ADX] Seq=997832953 Ack=993940790 Mix=16705 Lev=0	
1 0,000000 mescal212	mescal111	BOP	KEEPNLIVE Hessage	
4 2,006407 mescal111 12 5,009990 mescal212	mescal111	BOP	KEEPKLIVE Hessaye KEEPKLIVE Hessaye	
14 7.012599 mescal111	mescal212	BOP	KEEPVLID/E Message	
IB Frame 6 (114 bytes on wire, 1	14 bytes castured)			Le
00 Ethernet 11, Src1 001d01b7121	150:bd. But: 00:50:04:	48103163	2 (24 0 0 2)	
Transmission Control Protocol	, Src Port: 34722 (347)	22), Det Fort; 1	0440/04/02/02/ 0400 (1040), Seq: 666747083, Ack: 680284340, Len: 48	
Source port: 34722 (34722 Destination port: 1040 (1)	040)			
Sequence number; 66674708	3			
Reknowledgement number: 605	747133 80284340			
Header Tength: 32 bytes El El aver 0x0018 (PSH, aCK)				
Window size: 5840				
El Options: (12 bytes))			
NOP NOP				
Time stamp: toval 1979	75464, tsecr 193193698			
Contraction of the second s				
1				•
4			out per	12
0050 13 13 00 01 00 00 02 01	$00 \ 01 \ 00 \ 06 \ 11 \ 03 \ 64 \ 00 \ 01 \ 01 \ 01 \ 01 \ 06 \ 15 \ 15 \ 15 \ 15 \ 15 \ 15 \ 15 \ 1$	22		
0070 00 00	01 00 04 18 11 00 01 00			
Educ			Beauti Arestal Data (data) 48 butes	
	2		- Canad (Abbit) care (care):	
4 🗖 🥪 🖓 🦓 📢	2	mescal@ME!	SCAL11:/h Imescal@MESCAL11:/h 🕲 <capture> - Ethereal [Setwork Configuration]</capture>	D D D Thu Apr 21, 14:54
- 5.1		ICE		
	кнурги			
0.1	KESI OI	NSE		
5.1		NSE		
 ccaptures : Ethereal Ethereal 		NSE		- 6
 <capture> Ethereal</capture> File Edt Capture Display Hand Exercise 		NSE	- Las	Eef
<captures -="" ethereal<br="">Elle Edit Capture Display No. Time Source 3.0.421344 14.0.0.2</captures>	Tools		100	⊨ 6 Het
ccapture> Ethernal Elle Edt Capture Display No. Time Source 3 9,42344 14,9,9,2 2 9,00000 iercell111 5 9,00000 iercell111	Tools	Protocol UDP TCP	##6 Source port: 2006 Bestivation port: 5050 Wood 2 Abg (CA) See407020534 46040580271 WinuED05 Lend	- 0 Het
Capture> • Ethereal Elle EdtaptureDisplay No. TimeSource 3 0,42344 14,0,0,2 2 0,0000 mescal111 5 2,005561 mescal121 6 2,12575 14,0,0,2	Tools Destination 71.0.0.2 secoal212	Pressool VDP TCP TCP TCP	enfs Source spirit, 300% Bustionation prets 500% Source spirit, 300% Fundionation prets 500% Source spirit, 300% Support Source spirit	- đ Hel
<td>Tools Destination 71,0,0,2 metcal111 20,0,0,2 14,0,0,2 14,0,0,2</td> <td>Protocol 7</td> <td>Sector Sector Sector<</td> <td>- 18 Hel</td>	Tools Destination 71,0,0,2 metcal111 20,0,0,2 14,0,0,2 14,0,0,2	Protocol 7	Sector Sector<	- 18 Hel
ccaptures Etheroal File Edt Capture Diptoy No. Time Source Diptoy 3 0,423444,0,0,2 0,00000 mercalisti Source 2 0,00000 mercalisti Source Diptoy 2 2,10004 34,0,0,2 2,10004 34,0,0,2 Source 2 2,10004 34,0,0,2 2,10004 34,0,0,2 Source 2 2,10004 34,0,0,2 2,10004 34,0,0,2 Source	Iools Destination 71,0,0,2 mecal122 mecal122 mecal123 14,0,0,2 14,0,0,0 14,0,0,0 14,0,0,0 14,0,0,0 14,0,0,0 14,0,0,0 14,0,0,0 14,0,0,0 14,	NDE Pressool UP TCP TCP TCP TCP TCP TCP	Info Source part: 2006 Bestivation parts 5950 Source parts (CO) Seed00002534 Hol00000271 Winat0005 Lendo Source parts 2006 Bestivation parts 5950 Source parts 2006 Bestivation parts 2007 Bestivation parts 2	je Bu
*Capture> Ethereal Effe Edf Capture Display No Time Source 20,00200 <	Iools Destination 71,0,0,2 mescal122 mescal113 28,0,0,2 14,0,0,2 28,0,0,2 28,0,0,2 28,0,0,2 28,0,0,2 28,0,0,2	NDE Pretscol UP TCP TCP TCP TCP TCP TCP	Print Space operts SDW6 Bestionation prets SDW6 Space operationation Support Assistance operation SDW6 Space operationation Support Assistance operation SDW6 Space operation Support Support SDW6 SDW6 <t< td=""><td>(a) (b) 194</td></t<>	(a) (b) 194
ccaptures Ethernal Ele Edit Capture Display Tone Capture Display Tone Source Control Con	Iools Destination 713-0,02 mecal212	Prefaced UP TOP TOP TOP TOP TOP TOP TOP	bric Source perti: 2009 Bestinstips pert: 2009 Source pert: 2009 Bestinstips pertinstip pertinstippertinstipertipertinstip pertinstippertipertinstipertipertinstip p	- e Het
ccaptures - Etheral Ene Edt Capture 8 0.62344 34.0.0.2 2 0.00200 messaillit 5 0.62344 34.0.0.2 5 0.22396 14.0.0.2 7 2.13309 34.0.0.2 9 2.13409 34.0.0.2 9 2.13409 14.0.0.2 15 3.0007 messaillit 15 3.0007 messaillit 16 3.02771 messaillit 16 9.02771 messaillit	Destination 71,0,0,02 mescal222 mescal223 mescal224 mescal225 mescal225 mescal225 mescal225 mescal225	Pretocel ¹	Pdb Source opril: 2006 Bestiontion port: 5000	- 6 194
scaphares Tehnel Edit Captore Deploy No Time Source 2 0.00030 messilili 5 2.00030 messilili 5 2.00030 messilili 5 2.00030 messilili 5 2.00030 messilili 5 2.00140 Mo.0.2 5 2.00160 messilili 5 2.00000 messilili 5 7.00000 messilili 5 7.00000 messilili 5 7.00000 messililii 2 5.00000 messililiii 10 5.00000 messililiiiiiii 10 5.00000 messililiiiiiiii 10 5.00000 messililiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiii	Iools Destination 71:0.0.2 meccal222 meccal121 24:0.0.2 14:0.0.2 meccal121 24:0.0.2 14:0.0.2 meccal111 meccal111 meccal122 meccal111 meccal111 meccal111 meccal111 meccal111 meccal111 meccal111	Pretocol ¹ Preto	enfo Space operts SDME Bestionation perts SDME Space operationation Bestionationation perts SDME Bestionationation perts Space operationation Bestionationationationationationationationa	104 104
scapheres Effect Captere Diption No. Time Source Source 3 0.62124 3.4.9.6.2 Source 5 2.0081 Source Source 7 3.0072 Source Source 7 2.0081 Source Source 8 2.0081 Source Source 9 2.0081 Source Source 9 2.0081 Source Source 9 3.0007 Source Source 13 5.0007 Source Source 13 5.0007 Source Source 13 5.0007 Source Source 14 2.00607 Source Source 15 2.00607 Source Source 14 2.00607 Source Source 15 2.00607 Source Source 16 2.00607 Source Source 17	Ipoin Destination 74.0.0.2 mescaliti 24.0.0.2 24.0.1 24.0.2 24.0.2 24.0.2 24.0.2 24.0.2 24.0.2 24.0.2 24.0.2 24.0.2 24.0.2 24.0.2 24.0.2 24.0.2 <	Pretocol UP UP TCP TCP TCP TCP TCP TCP TCP TCP TCP TC	Info Source pert; 2006 Bestivation pert; 5006 Source pert; 2006 Source pert; 2006 Source pert; 2006 Bestivation pert; 5006 Source pert; 2006 Source pert; 2006 Source pert; 2006 Source pert; 2006 Source pert; 2006 Source pert; 2007 Source pert; 2006 Source pert; 2007 Source pert; 2006 Source pert; 2007 Source pert; 2007 Source pert; 2007 Source pert; 2008 Source pert; 2007 Source pert; 2008 Source pert; 2008 Source pert; 2	- 6 <u>1</u> 94
<captures -="" etheral<br="">Ene Edit _Capture _Display No. Time</captures>	Ipoin Destination 70.00.2 mecal22 mecal22 10.00.2 mecal22 10.00.2 20.00.2 20.00.2 20.00.2 20.00.2 40.00.2 mecal2121 mecal2121 mecal2121 mecal2121 mecal2121 mecal2121 mecal2121	NSE Protocoli UP TO TO TO TO TO TO TO TO TO TO	Pdf: Source opril: 2006 Bestionation port: 5060 Source opril: 2007 Bestionation port: 5060 Source opril: 2007 Bestionation port: 5060 Source opril: 2007 Bestionation port: 2007 <td>- 6 194</td>	- 6 194
ccaptures Chethered Edit Cuthere English 8 Time Source 9 -0.00100 Ano,0,-2 5 2.000100 Ano,0,-2 5 2.000100 Ano,0,-2 5 2.000100 Ano,0,-2 5 2.00010 Ano,0,-2 7 2.00000 Monol.2 7 2.00000 Monol.2 7 2.00000 Monol.2 1 3.40000 Monol.2 1 5.40000 Monol.2 1 2.00000 Monol.2 1 2.00000 Monol.2 1 2.00000 Monol.2 1 4.00000 Monol.2 1 4.00000 Monol.2	Ionit Ionit 71,0,0,2 71,0,0,2 metallili 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 36,	NDE Protocol UP TO TO TO TO TO TO TO TO TO TO	Info Source perts: 2009 Instantion perts: 2009	(= 6)
scapheres Elevent Edit Edit Captere 10 Editadit Standard 10 Editadit Standard 11 Stafeste Standard 12 Stafeste Stafeste 13 Stafeste Stafeste 14 Zoffeste Stafeste 14 Zoffeste Stafeste	Loots Destination 70.0.02 mecal221 mecal221 mecal221 14.0.02 24.0.02 14.0.02 24.0.02 14.0.02 24.0.02 mecal221 mecal222 mecal222 mecal222	NDE	bit Source perts, 2006 Bestivation perts 5950 Source perts, 1003 Sequences Source perts, 1003 Bestivation perts 5950 Source perts, 1003 Bestivation perts, 10050 Source perts, 1003 Bestivation perts, 10050 Source perts, 1004 Bestivation perts, 10050 Source perts, 1005 Bestivation perts, 10050 Bestivation perts, 1005 Bestivation perts, 10050 Bestivation perts, 10050 Bestivation perts, 10050 Bestivation perts, 10050 Bestivation perts, 10050 Bestivation perts, 10050 Bestivation perts, 10050 Bestivation perts, 100500	- 6 <u>1</u> 94
scaptures Fibre all Edit Captore English No. Time Source 2 0.00310 messialli 5 0.00310 messialli 6 2.00310 messialli 7 0.00310 messialli 8 2.00310 messialli 8 2.00310 messialli 8 2.00300 messialli 8 2.00300 messialli 8 2.00407 messialli 8 2.00407 messialli 15 0.0071 messialli 16 2.00407 messialli 16 2.00407 messialli 16 2.00407 messialli 16 2.00407 messialli 14 2.00407 messialli 14 7.01599 messialli	Ipoin Destination 71,0,0,2 mecaliti 21,0,0,2 mecaliti 24,0,0,2 14,0,0,2 24,0,0,2 14,0,0,2 14,0,0,2 14,0,0,2 14,0,0,2 14,0,0,2 14,0,0,2 14,0,0,2 14,0,0,2 mecaliti mecaliti mecaliti mecaliti mecaliti mecaliti mecaliti mecaliti	NDE	Pdf: Source opril: 2006 Bestionation port: 5060 Source opril: 2006 Bestionation port: 5060 Source opril: 2006 Security Status Source opril: 2007	- 6 19
Captures Etheral Edit Capture Enplay 3 642144 34,0,62 3 642144 34,0,62 3 642144 34,0,62 3 642144 34,0,62 5 2,00814 84,0,62 7 2,20084 84,0,62 7 2,20084 84,0,62 9 2,31409 34,0,62 10 3,44092 14,0,62 11 3,46092 14,0,62 15 7,42079 8886422 2 5,00000 8886121 2 5,00000 8886121 2 5,00000 8886121 2 5,00000 8886121 2 5,00000 8886121 2 5,00000 8886121 2 5,00000 8886121 2 5,00000 8886121 2 5,00000 8886121 2 5,00000 8886121 2	Tools [Destination 77,0,0,2 secont211 24,0,0,2 14,0,0,2 34,0,0,2 44,0,0,2 44,0,0,2 44,0,0,2 secont212 secont212 secont212 secont212	NSE Pretocol 10 10 10 10 10 10 10 10 10 10	Info Source parts 2006 Bestionstance perts 2007. Source parts 2007 Bestionstance perts 2007. Source perts 2007. Bestionstance perts 2007. Source perts 2007. Bestionstance perts 2007. Source perts 2007. Bestionstance perts 2007. Bestionstance perts 2007. Bestionstance perts 2007. Bestionstance perts 2007. Bestionstance perts 2007. Bestionstance perts 2007. Bestionstance perts 2007. BESTION Perts 2007. Bestionstance perts 2007. BESTION Perts	- 6 14
scapharzo - Effetendi Ella Edit Captere Display No. Time Source 3 0.62244 34.0.62 Source 2 0.02244 34.0.62 Source 3 0.02244 34.0.62 Source 2 0.0205 Source Source 3 0.02244 34.0.62 Source 3 0.0217 Source Source 3 3.0007 Source Source 3 3.0007 Source Source 3 3.0007 Source Source 3 5.0007 Source Source 3 3.0007 Source Source 3 4.0004 Source Source 3 3.0007 Source Source 3 4.0004 Source Source 3 4.0004 Source Source 3 4.0004 Source Source	Internation 70.0.02 mecal221 14.0.02 20.002 20.002 20.002 20.002 20.002 20.002 20.002 20.002 20.002 20.002 20.002 mecal222 mecal222	NDE Protocoli 10 10 10 10 10 10 10 10 10 10	Info Source port; \$20% Bestivation port; 50% Source port; \$20%	- 6 54
Captores Tehend. Em Edit Cuptore Deploy 80: Time Source 2 0,000100 mecalilit 2 0,000100 mecalilit 5 2,000810 mecalilit 5 2,00081 mecalilit 1 5,00090 mecalilit 1 4,00000 mecalilit 1 4,000000 mecalilit 1 4,00000 mecalilit 1 4,000000 mecalilit 1 4,000000 mecalilit 1 4,000000 mecalilit 1 4,000000000000000000000000000000000000	Ipoin Destination 7,0,0,0,2 mecolilli 24,0,0,2	NDE Pretocol 107 107 107 107 107 107 107 107	Info Sever pert: 2001 Pertinetion pert: 5001 Sever pert: 2001 Sever pert: 2002 Sever pert: 2002 Sever pert: 2002 Sever pert: 2002 Sever pert: 2002 Sever pert: 2004 Sever pert: 2002 </td <td>- 6 H</td>	- 6 H
Captores Etheral Edit Captore Depty Captore Depty Captore Depty Captore Captore Captore Captore Captore	Iools Iools 70,0,0,2 secont2111 24,0,0,2 secont2111 24,0,0,2 secont2111 24,0,0,2 secont2121 34,0,0,2 secont2121 secont2121 secont2121 secont2121 secont2121 secont2121 secont2121 secont2121 secont2122 secont2122 secont2122 secont2122 secont2122 secont2122 secont2122	Pretocel: (09) 107 107 107 107 107 107 107 107	Info Sovre prist_2006 Bestivision port; 5000 Source port; 5000 Bestivision port; 5000 Source port; 5000 Bestivision port; 5000 Source port; 5000 Bestivision port; 5000 BESTIVIC France Bestivision port; 5000 BESTIVIC France <td>- 6 <u>H</u></td>	- 6 <u>H</u>
Captores - Etheral Edit Capter Depuisor Time Source	Image:	Protocol: Protocol: 107 107 107 107 107 107 107 107	Info Source port; 2006 Bestivation port; 5006 Source port; 2006 Bestivation port; 5007 Source port; 2006 Bestivation port; 5006 Source port; 2006 Bestivation port; 5007 Source port; 2007 Bestivation port; 5007 Source p	- 6 54
Copperson Tethend Entropy Copperson Coppe	Ipoin Destination 73,0,0,2 mesolilili 30,0,0,2 14,0,0,7 24,0,0,2 34,0,0,2 34,0,0,2 mesolilili	NDE	Into Sovers (prist: Solf): Septimized Astronomy 1: 900; (pp: 4002) ECO: Septimized Astronomy 2: 900; (pp: 4002) ECO: Septimized Astronomy 2: 900; 3272: 3100; (pp: Septimized Astronomy 2: 900; (pp: Septimized Astronomy 2	(= 6 () () () () () () () () () () () () ()
Captores Etheral Edit Captore Department Captore	Iools Iools 71,0,0,2 secont2121 ascont2121 secont2121 24,0,0,2 secont2121 24,0,0,2 secont2121 24,0,0,2 secont2121 ascont2121 secont2121 ascont2121 secont2121 ascont2121 secont2121 ascont2121 secont2122 ascont2121 secont2122 ascont2121 secont2122 ascont2122 secont2122 bitoticities, Int: orospharp secont2122	Pretocol (PP 10P 10P 10P 10P 10P 10P 10P 1	Info Sover print: 2006 Bentingstopping Status Bentingstopping	- 6
captores Tehend Effe Edf Captore Depty Se Time Source 2 0,00000 wenciliti 2 0,0000 wenciliti 2 0,00000 wenciliti 2 0,0000000000000000000000000000000	Image:	NDE	Info Sover op 11, 2006 Bentinnin porti 5050 Sover op 11, 2007 Bentinnin porti 5050 Sover op 11, 2007 Bentinnin porti 5050 Sover op 11, 2008 Bentinnin porti 10000 Sover op 11, 2008 Bentinnin porti 100000 Sover op 11, 2008 Bentinnin porti 1000000 Sover op 11, 2008 Bentinnin porti 1000000 Sover op 11, 2008 Bentinnin porti 10000000000 Sover op 11, 2008 Bentinnin porti 1000000000000000000000000000000000000	<u>- 6</u>
Captores Etheral Edit Cuptore Depty Control No.01	Iools Iools 17.0.0.2 mecolisis 17.0.0.2 mecolisis 18.0.0.2 Mecolisis 19.0.0.2 Mecolisis 10.0.0.2 Mecolisis 10.0.0.2 Mecolisis 10.0.0.0.2 Mecolisis 10.0.0.0.2 Mecolisis 10.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.	NDE Pretocell UP TO TO TO TO TO TO TO TO TO TO	brd Sverig 1941 Staff Institution provid 900 Sverig 1941 Sverig 1941 Sverig 1941 Sverig 1941 Sverig 1941 Sverig 1941 Sverig 1941 Sverig 1941 Sverig 1941 Sverig 1941 Sverig 1941 SVerig 1941 Sverig 1941 Sverig 1941 Sverig 1941 SVE2 1946 Sverig 1941 Sverig 1941 Sverig 1941 SVE2 1947 Sverig 1941 Sverig 1941 Sverig 1941 SVE2 1947 Sverig 1941 Sverig 1941 Sverig 1941 SVE2 1947 Sverig 1941 Sverig 1941 Sverig 1941 SVE2 1947 </td <td>(= 6) (4)</td>	(= 6) (4)
Captores Etheral Edit Captores Etheral Edit Captore Display Time Source Sourc	Loois Destination 71,0,0,2 mecal221 mecal221 mecal221 10,0,0,2 mecal221 11,0,0,2 mecal221 12,0,0,2 mecal221 14,0,0,2 mecal221 14,0,0,2 mecal2121 mecal2121 mecal2121 mecal2122 mecal2121 mecal2122 mecal2122 mecal2122 mecal2122 100 bytes contract mecal212 101,5e per; 1000;007 34,0,0,2 102,5e per; 1000;007 34,0,0,2 103,6e per; 1000;007 34,0,0,2 104,0e per contract mecol212 105,6e per; 1000;007 34,0,0,2 106,6e per; 1000;007 34,0,0,2 107,27 34,0,0,2 108,002;04,0,02,7;1 34,0,02 109,0000;04,02,7;1 34,0,02 100,0000;04,02,7;1 34,0,02 100,0000;04,02,7;1 34,0,02 100,0000;04,0000;04,0000;04,0000;04,0000;04,0000;04,0000;04,0000;04,0000;04,0000;04,0000;04,0000;04,0000;04,000;04,0000;04,0000;04,0000;04,0000;04,000;04,0000;04,0000;04,000;04,000;04,000;04,000	NDE	Info Sovers print; 2006 Bertinnitin port; 5006 Starting; Park (GC) Sept0302234; Aux3030407 Bund1070; Laudi SV22; Jole (Park, AC) Sept03120; Starting; SV222; Laudi Bund1070; Laudi SV22; Jole (Park, AC) Sept03120; Starting; SV221; Bund1070; Laudi Bund1070; Laudi SV22; Jole (Park, AC) Sept03120; Starting; SV221; Bund1070; Laudi Bund1070; Laudi SV22; Jole (Park, AC) Sept03120; Starting; SV221; Bund1070; Laudi Bund1070; Laudi SV22; Jole (Park, AC) Sept03120; Starting; SV221; Bund1070; Laudi Bund1070; Laudi SV22; Jole (Park, AC) Sept03120; Starting; SV221; Bund1070; Laudi Bund1070; Laudi SV22; Jole (Park, SV2; Bund1070; Bund1070; Laudi Bund1070; Laudi SV22; Jole (SV3; SV4; Bund1070; Bund1070; Laudi Bund1070; Laudi SV22; Jole (SV3; SV4; Bund1070; Bund1070; Laudi Bund1070; Laudi SV22; Jole (SV3; SV4; Bund1070; Bund1070; Laudi Bund1070; Laudi SV22; Jole (SV3; Bund100; Bund100; Bund100; Laudi	(= 6 (5)
Image: Standbarren	Ipoin Destination 71,0,0,2 mecalitit 24,0,0,2 14,0,0,2 124,0,0,2 14,0,0,2 104,0,0,2 14,0,0,1 mecalitit mecali	NSE Pretocel UP TCP TCP TCP TCP TCP TCP TCP TC	φtc Control Sector Control Sector Control Sector Sector Sector	<u>- 6</u>
Captores Etheral Edit Cuptore (page) Tesse Edit Cupt	Iools [Joetsnation] 71,0,0,2 association 71,0,0,2 association 24,0,0,2 14,0,0,2 association 24,0,0,2 14,0,0,2 association	NDE (Pretocol) (P) (P) (P) (P) (P) (P) (P) (P	Info Sover print: SSVB: Destination peril: 590; (spc: 4002; CGO) Sept3580(7); 40:439812577; 40:4012576; Lev0 3472; 100; [Pin; ACC] Sept3580(7); 40:439812570; Muni220; Lev0 3472; 100; [Pin; ACC] Sept3580(7); 40:43025520; Lev0 3572; 100; [Pin; ACC] Sept3580(7); 40:430515272; 20:01220; Lev0 4574; 116; Hense 1274; 117; Hense 1274; 116; Hense 1274; 117; 117; 117; 117; 117; 117; 117; 1	- 6 <u>1</u> 44
Captores - Etheral Edit Captores - Etheral Edit Captore Societ Soc	Losis Destination 71,0,0,2 mecal221 mecal221 mecal221 34,0,0,2 secol221 34,0,0,2 secol221 34,0,0,2 secol222 34,0,0,2 secol122 mecal122 mecal122 mecal122 mecal122 36,0,0,2 secol122 mecal122 mecal122 36,0,0,2 secol111 mecal122 mecal122 36,0,2,7,1 secol112 mecal122 secol113 mecal122 secol113 Mecal122 secol113 Mecal122 secol113 Mecal122 secol113 Mecal122 secol113 Mecal122 secol114 Mecal123 secol115 Mecal124 secol114 Mecal125 secol115 Mecal126 secol114 Mecal127 secol114 Mecal128 secol114 Mecal129 secol114 Mecal114 secol14	NSE Preduced UP 107 107 107 107 107 107 107 107	Info Sverey parts, 25% Restingtion parts (5%) SVEREy p	- 6 <u>1</u> 94
Captores Etheral Edit Captore Deploy Res Edit Re	Ioris Destination 71,0,0,2 mecolitili 20,0,0,2 14,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 100 bytes captored) 000 bytes captored) 000 bytes (aptored) 0 0 0 0 0 0 0 0 0 0 0	NSE Pretocell UP TO TO TO TO TO TO TO TO TO TO	Intel Sverige (sci) September (sci) Sep	- 6
Captorso: Etheral Edit Captorso: Etheral Edit Captorso: Etheral Sociation 314/4/21 Sociation	Internation Total [Destination] 71,0,0,2 association 24,0,0,2 14,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 association 24,0,0,2 association association <	NSE Pretacel UP TC TC TC TC TC TC TC TC TC TC	Info Sover print: 25% Bertinnian port: 5% BERNET Bertinnian port: 5%	- 6 <u>1</u> 9
Captores Tehend Edit Captore Depty Ref Captore Depty	Losis Destination 72.0.0.2 mecalititi 30.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0.2 34.0.0 34.0.0.2 34.0.0 34.0.0.2 34.0.0 34.0.0.2 34.0.0 34.0.0.2 34.0.0 34.0.0.2 34.0.0 34.0.0.2 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0.0 34.0	NSE Preduced UP TO TO TO TO TO TO TO TO TO TO	Vet. Svett Svet	- 6 59
Captores Ethel Edit Cuptore Depuis Total Anno. Control Anno. C	Iools Ioestnation 71,0,0,2 association 71,0,0,2 association 24,0,0,2 14,0,0,2 association 24,0,0,2 34,0,0,2 association biologic biologic association association association association association association biologic association association association	NDE	Intel Svere print 3200 Bestingtion peril 900 Svere peril 900 <td>- 6 <u>1</u>44</td>	- 6 <u>1</u> 44
Captions - Etheral Edit Edit Captore Depty 36: Time Source 2 0,00010 encoded 2 0	Internation Internation 71,0,0,2 Telestination metcal212 metcal212 metcal212 Telestination Telestination	NSE Preduced UP 107 107 107 107 107 107 107 107	Info Svere print: 20% Entimiting peri: 50% Svere print: 20% Entimiting period Svere period Svere print: 20% Entimiting period Svere period <	1- 6 194
Captores Tehend End Edit Captore Deploy Res Tele Edit Captore Deploy Science D	Ioris Destination 71,0,0,2 mecolitit 20,0,0,2 14,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 24,0,0,2 mecolliti mecolititi mecolitititi </td <td>NSE Pretocell UP TO TO TO TO TO TO TO TO TO TO</td> <td>Vit Svare (VAI) September 2000 (VA</td> <td>18 - 1</td>	NSE Pretocell UP TO TO TO TO TO TO TO TO TO TO	Vit Svare (VAI) September 2000 (VA	18 - 1
Captors: Etheral Edit Captors: Etheral Edit Captor Social	Internation 71,0,0,2 association 71,0,0,2 association 11,0,0,2 association 12,0,0,2 association 14,0,0,2 association 24,0,0,2 14,0,0,2 association	NDE	ν. γ.	
Captores Tehend Effe Edit Captore Deploy Se Time Source 2 0,00010 Mencial11 2 0,0000 Mencial11 4 0,002 1 1,0000 Mencial11 2 0,0000 Mencial11 4 0,002 1 1,0000 Mencial11 1 0,0000 Mencial11 1 0,000 Mencial11 1 0,000 Mencial11 1 0,0000 Mencial11 1 0,000 Mencial11	Losis Destination 72.0.0.2 ************************************	NSE Preduced UP 107 107 107 107 107 107 107 107	Vita State State State	94 <u>1</u> 94 <u>1</u>
Captores Ethela Energy Ele Edit Captore Deploy Roman Ethela Edit Scotter Ethela Edit Scot	Internation Total T	NDE	Info Section 12001 Section 120011 Section 12001	
Copperson - Etheral Edit Capter Department Teme Starten Societte 34,4,62 Societte 34,62 Societte 34	Internation Total [Destination] 71,0,0,2 accollege accollege 100 bytes captured accollege	NSE	Intel Street Stree Stree Stree	
Captores Theoder Control of the	Ipetsnation 72.0.0.2 30000 72.0.0.2 association 72.0.0.2 association 72.0.0.2 10.0.0.0.0 10.0.0.0.0 10.0.0.0.0 10.0.0.0.0 10.0.0.0.0 10.0.0.0.0 10.0.0.0.0 10.0.0.0.0 10.0.0.0.0 10.0.0.0.0.0 10.0.0.0.0.0 10.0.0.0.0.0.0 10.0.0.0.0.0.0.0.0.0 10.0.0.0.0.0.0.0.0.0.0.0.0.0 10.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.	NDE	Image: Source (SAC) Specific Source (SAC)	
Computers - Etherad Computers - Etherad Computer - Ethe	Ioris [Jostnation] 71,0,0,2 association 71,0,0,2 biological 71,0,0,2 biological 71,0,0,2 biological 71,0,0,2 biological 71,0,0,2 biological 71,0,0,2 biologiological 71,0,0,2	Pretocol UP TO TO TO TO TO TO TO TO TO TO	Info Svere print: SVM: Bretisticion perit: 50% SVER: Bretisticion perit: 50% <	2 (4) (4) (4) (4) (4) (4) (4) (4)

• 8: ACKNOWLEDGE

Time		Tools			В
	Source	Destination	Protocol *	Info	
3 0,42134	4 14.0.0.2	71,0,0,2	UDP	Source port; 32006 Destination port; 5050	
5 2.00558	0 mescalill 1 mescal212	mescal212 mescal111	TOP	40022 > 5ge (403) 544/351/35234 468/35380/77 Wints/705 Lento han > 40227 [AD1] 540/3531/35234 478/35380/77 Wints/705 Lento	
6 2,12974	5 14.0.0.2	24,0,0,2	TOP	34722 > 1040 [PSH, ACK] Secold6747003 Ack-600204340 kin-5040 Lenn48	
7 2,13070	4 24,0,0,2	24.0.0.2	TOP	1040 > 34722 [ACK] Seq=680284340 Ackx666747131 Winn5792 Lenx0	
8 2,27889	5 24.0.0.2	14.0.0.2	TOP	1040 > 34722 (PSH, RCK) September24340 Rebree547133 Winn5752 Lemis2 M222 > 1046 D2V7 September2313 Rebree50000000000000000000000000000000000	
10 3,14520	2 14,0.0,2	24,0.0,2	109	34722 > 1040 (PSH, 40Y) Sequ565747131 Ack-680264382 Win45640 Len-18	
11 3,14539	8 24,0.0.2	14.0.0.2	TOP	1040 > 34722 [ACK] Seq=680284382 Ack=686747149 Win=5792 Len=0	
15 7.012222	/ mescallil	mescal212	102	4002 > 5gc (HK) 54073732255 Hold 300730 Windo Control Lenio has > 40022 (HK) 540133140750 (Hold 9018292 (Hold 301720) (Hold 90	
1 0,000000	0 mescal212	seccal111	302	NDNLIVE Ressage	
4 2,00540	7 wescalill	mescal212	102	VEPHLIVE Herrage	
12 5,00999	0 Rescal212	eescal111	HCP	KEEPKLIVE Recision	
Frame 10 (8 Discret II Internet Pri formationio Source p Destinat	4 bytes on wire. E , Src: 00c90cb7;27 otocol. Src: Hddr: nontrol Protocol ort: 34722 (14722 (14722) in port: 2140 (1) runber: 68674735	4 bytes captured) 150:bd, Sht: 00;50:04; 140,0.2 (44,0.2), 8 440,0.2 (44,0.2), 9 47143 17143	40:c3:63 st Rddr: 24.0.0. 22). Bat Port: 10	2 (26.0.2) 00 (1000), Set 0074711, 60: 00704302, Let 10	
Next sev Reknowle Header 1	dysenit number: 6 ength: 32 bytes				
Next sev Actrowle Header 1 10 Flaget 0	dyment number: B ength: 32 bytes by0018 (PSH, RCK)				
Next sev Next sev Nex	dyment number: B ength: 32 bytes by0018 (PSH, ACK) 1128: 5040 b; 0xb30c (correct)			
Next sey Reinoule Header 1 Biflage: 0 Window s Dheckson Bi Options	dysment number: 6 ength: 32 bytes h0013 (PSH, ACK) 1201 5040 s: 0xb30c (correct : (12 bytes)				
Next see Next see Next see Next see Next see Nor NOP NOP	dyement number: 6 ength: 12 bytes h0018 (PSH, ADK) 128: 5040 s: 0xb30c (correct : (12 bytes)				
Next sey Next sey Next sey Next sey Next sey Next sey Nor Nor Nor Time	dgement number: 6 ength: 12 bytes h0018 (FSH, BCK) iIze: 5840 : 0.bb3b (correct : (12 bytes) stamp: taval 1979) 75566, tamor 183194282			
Next sei Richneile Header 1 Bi Flagst 0 Vinder s Decksum NOP NOP Time	dgement number: 6 ength: 12 bytes h0018 (PSN, 40X) 1128: 5040 (0.0530k (correct : (12 bytes) etamp: tavel 1579 top)) 75566. toecr 188194282			
Next sev Recincule Header 1 Bi Flagst 0 Vindov s Decksum Bi Optionst NOP NOP Tise	dyneet number: 5 ength: 32 bytes holds (PSN, 40() 120: 5040 c: 0xb30c (correct : (12 bytes) stamp: tavel 1579 tes)) 75566, tamar 183194282			
Next sev Next sev Next sev Next sev Next sev Nor Nor Nor Nor Time	dyneent number: 5 ength: 32 bytes holds (PSN, ACK) ize; 5040 c; 0xb30c (correct : (12 bytes) etamp: taval 1979 ter))) 75566, tamar 183194382			
Next see Next see Next see Next see Vindow s Decksum Decksum NOP Time	dgement number: 5 ength: 32 bytes words (PSM, SGC) 120: 5040 : OubBit (correct (12 bytes) steep: tavel 1579) 75566. taeor 188194282			
Next see Reitroule Reitroule Befangt 0 Vindor 1 Decisus 10 Octoors NOP NOP Time	dyment nuder: 5 ergth; 12 bytes w003 (FSR, 60%) unit 5540 (12 bytes) etamp: toval 1979 erg)) 75566. tamor 180194202			
Next see Reitreele Reitreele Bisder 1 Bister Declaus Bister NOP Time Reitreele	dyment nucley: 5 ergsh: 12 bytes 10038 (PSR, 60C) 1101: 5040 ; OABBe (correct : (12 bytes) starp: taval 1579 erb9) 75566, tamor 182194282			
Next see Reit see Reitroule Header 1 BiFlaget 0 Under 1 Declaum NOP Time NOP Time	dynamit nudor: B ergit: 12 bytes (0016 (PSR, BOC) 1201 (SMO (1201 (SMO (120 bytes)) c) 0.010((120 bytes)) c) 0.010((120 bytes)) c) 0.010((120 bytes)) c) 0.010((120 bytes)) c) 0.010((120 bytes))) 75666. taeor 188194202			
Next see Reit see Header 1 BF Flagst 0 Vindou s Decision NOP Tise Toe NOP Tise 0 00 02 87 0 16 do bio 52 as	dynamet nutber 1 meght 12 bytes wolds (PSH, sOC) 1281 5040 ; OubBk (correct (12 bytes) stamp; tavel 1979 cctb) stamp; tavel 1979 cctb) stamp; tavel 1979 cctb)) 17556, taor 1833402 c) P. 30 & 50 de 30 1 20 de 30 c de 30 a c	s	2	1
Next see Richt see Richtmoule Header 1 Biflangt 0 Vindou 1 Dheckson Dheckson Dheckson Dheckson Dheckson Dheckson NDP Time Time The the the the the NDP Time	dyment nutber: D bytes regist: 20 bytes wolds (PSR, SOV) izst: 5040 ; 0xbBk (correct ; (12 bytes) stamp: tavel 1979 rec) stamp: tavel 1979 rec) stamp: tavel 1979 rec)) 17556, taor 183194022 of Ph 20 Re 50 de 50 1 20 Ga 20 co de 60 5 a d	·		[]
Mext see Reinneile Header 1 Bi Filaget 0 Window 1 Declaus NOP NOP NOP NOP NOP NOP NOP NOP	dynamet number 10 ength: 13 bytes wolls (PSR, BOL) 128: 5040 ; 0.65Bk (correct (12 bytes) stage; taval 1579 cr)) ; a2 04 10 27 bd ; b2 00 00 1.01 ; b2 00 10 00 com) 175566, teacr 183394002 c0 Pb 20 Bc 50 de 60 1 18 G 60 cc da 60 ca 40 19 G 20 de 50 de 50 de		P ▼ [Peed] Appig Ditts (trats), 18 bytes	1

- Scenario 2:
 - MESCAL21'PCS sent an PATH-ERROR to MESCAL11 as shown in the figure below:

Time	obsue Deshing		- 24	
Time	Source	Destination	Protocol *	info
2 0 000178	14.0.0.2 metcal212	/1.0.0.2	TOP	Source port: 52000 Destination port: 5000 here: 2 double Internet Source 2022 2021 Environment Provide
3 2,544703	\$ 14.0.0.2	24,0,0,2	TOP	34722 > 1040 [PSH, ACK] Seq=666747227 Ack=680284446 Min=5640 Len=48
4 2.544912	24.0.0.2	14,0,0,2	TCP	1040 > 34722 [RCK] Seq=680234446 Ack=666747275 Win=5732 Len=0
5 2,556478	24.0.0.2	14,0,0,2	TOP	1040 > \$4722 [PSH, HCK] Sep=80784446 Hok=868747275 Win=9792 Len=18
9 2.754503	14.0.0.2	24.0.0.2 nesca1212	TOP	34/22 / 1040 (HLR) 36(566)47/27 HCR500/24464 40(16:040 Leno) 40/22 / box (GCK) 56(6)504/245 Gck-900324000 Leno)
1 5.010114	mescal212	nescal111	TOP	bap > 40022 [ACX] Seg-593542082 Ack+997834264 Uun=17220 Lenn0
\$ 7,755655	9 mescaliii	nescal212	TOP	40022 > bgp [ACK] Seq:997834264 Ack:993942101 Win:16705 Len:0
0.000000	mescaliii	mescal212	BGP	KEEPVLIVE Hessage
\$ 2,754404	sescal212	mescal111	BUP	RELPALIY, Persage
2 7.755563	aescal212	mescaliziz	RCP	NETFRATE INCOMPT
awe 5 (84 hernet II	bytes on wire, B Src: 00:50:04:4 stocol, Src Addr:	4 bytes captured) 0:c3:63, Dst: 00:d0:b7: 24.0.0.2 (24.0.0.2), D 1, Src Port: 1040 (1040	21:50;bd hat Addr: 14.0.0.), Dat Port: 347	2 (14.0.0.2) 22 (34.722), Segi 680/28446, Ack: 665747275, Len: 18
ternet Pro ministeion Source p Bestinat Sequence Next seq Acknowle Header 1 Header 1	<pre>Lontrol Protoco ort: 1040 (1040) ion port: 34722 (number: 68028444 uence number: 680 dgement number: 68 dgement number: 68 dgement number: 68 uots (PSR, ACK) ize: 5732 : 0x7c87 (correct (12 bytes)</pre>	(34722) 16 1204464 2004464 2006747275		
ternet Pro ansmitter Source p Destinat Sequence Acknowle Header 1 3 Flags; 0 Window s Options: NOP NOP Time	<pre>t Lontrol Protoco ort: 1040 (1040) ion port: 34722 (number: 68028444 uence number: 680 dgement number: 6 ength: 32 bytes wolls (PSH, ACK) ize: 5732 : 0x7687 (correct (12 bytes) stamp: tava] 1833</pre>	(34722) 16 1205464 986747275 1) 220578, tasor 150009774		
ternet Pro anishistic Source p Bestinat Sequence Next seq Acknowle Header 1 B Flags: 0 Window s Drecksus B Options: NOP Time to 458 bpt	<pre>Lontrol Protoco ort: 1040 (1040) ion port: 34/220 (umaber: 680/2844 uence number: 680 dgement number: 6 ength: 32 bytes x0018 (PSH, ACK) ize: 5732 ; 0x7087 (correct (12 byter) stamp: taval 1933 tes)</pre>	34722) HE 1284454 HE 1284775 HE 1) 228570, tasor 130009774		
ternet Pro ministerio Source p Bestinat Sequence Next seq Acknowle Header 1 (Plags: 0 Window s Checksum (Dptions: NOP NOP Time a (18 bpt	Lonirol Mrotoco ort; 1040 (1040) ion port; 34720 (mumber: 6000844 wence number: 600 dgement number: 600 dgement number: 600 dgement number: 600 dgement number: 600 state: 5782 ; 0x7c87 (correct (12 byter) state; taval 1032 tab)	34722) #6 3885444 888747275 .) 220570, taeor 190009774		
ternet Pri antenistic Source p Bestinat Sequence Next seq Acknowle Header 1 Flags: 0 Window z Obecksum Options: NOP Time a C10 bp	Lonirol Protoco ort; 1040 (1040) ion port; 34720 (1040) number: 8005844 uence number: 8005844 uence number: 16 anguht: 32 bytes to 0.7687 (correct (12 bytes) stamp: taval 1855 tes)	34722) #6 398464 888747275 :) :) 228579, tasor 130009774		
sernet Pri missission Source p Bestinat Sequence Next seq Acknowle Header 1 Flags: 0 Uptions: NCP NCP Time a 148 Epi	Lontrol Protoco ort; 1040 (1040) ion port; 34720 (number: 8005844 uence number: 66 degent number: 66 degent number: 6 degent	34722) #6 5807454 580747275 }]) 220579. tasor 190009774		
ternet Pri ansmission Source p Bestinat Sequence Next seq Acknowle Header 1 Brlags: 0 Window s Checksum Distors NCP NCP Time to (18 bp)	Lonfrol Protoco ort; 1040 (1040) ion port; 3472 (mader: 6005444 uence number: 6005444 uence number: 6 ength: 32 bytes dogsent number: 6 ength: 32 bytes io018 (PSH, ACK) ize: 5732 : 007687 (correct (12 bytes) stamp: taval 1033	34722) H6 309464 880747275 () 220570, taeor 130009774		
ternet Pri annaistio Source p Destinat Sequence Next seq Acknowle Header 1 B Flags: 0 Window s Direcksum D Options: NOP Time te (18 bp)	Lonfrol Protoco ort; 1040 (1040) ion port; 34722 (1050 mader: 8005444 uence number: 8005444 uence number: 80 degent number: 8 degent number: 8	34722) #8 080444 80747275)) 220570. tasor 130000774		
ternet Pri ansistor Source p Destinat Sequence Next seq Acknowle Header 1 B Flags: 0 Window s Checksum 0 ptions: NOP Time	Lontrol Protoco ort; 1040 (1040) ion port; 3472 (1050) mader: 8002844 uence number: 8002844 uence number: 80 degeent number: 6 ength: 32 bytes (12 bytes) ion (1978, 40%) ion	34722) # 803444 86847275 903444 903444 903444 90344 100000774 100000774		
ternet Pri annaistico Source p Jiest inat Sequence Next seq Acknowle Header 1 B Flags; 0 Uindou s Options; NCP Time tes (FB tep	Lontrol Protoco ort: 1040 (1040) ion port: 34722 (1040) mader: 8005444 wence number: 8005444 wence number: 8005444 wence number: 8005444 wence number: 10 wents 32 bytes 10 wence number: 10 10 wence 22 mere 10 wence 20	34722) 8 8 905447 90574725 1) 228578. taser 130000774 51 Ja 27 Jul et 8: 80 90 11	8	
ternet Pri annaistico Source p Bestinat Sequence Next seq Reknoule Header 1 B Flags: 0 Unidow s Checksus B Flags: 0 Uptions; NCP NCP Time to CFB by 0 0 02 04 16 a0 2c 83 ap 20 00 02 04	Lundrol Professor High Professor High (160) High Parts 2022 (High Parts 2022) High Parts 2022 (High Parts 2022) H	34722) # 803444 86647275)))) 200770, tasor 100000774 200770, tasor 100000774	a(. 0.*)	
ternet Pri amministico Source p Bestinat Sequence Next seq Acknowle Header 1 B Flags: 0 O Nindow s Checksum Options: NCP NCP Time ts (10 bpt 53 as 02 53 as 02 53 as 02	Landra) Professo Landra) Professo res (1946) (207) mabbert 580/5644 market 188/5644 market 188/5644 market 188/5644 187/572 (20 bytes) stampt table 1 stampt table 1	34722) 8 8 900747275 900747275 1) 200078, taser 130000774 51 1e 27 bd c1 8t 80 10 90 8t 6t 6t 6t 70 cc	·······(- 0.**	vinie v



Test Reference: TB_P3_FUNCT/CMES/3Test Purpose: Validate the REQ-REFERNCE-ID and PATH-COMPUTATION-ID

Procedure Same configure as for scenario 1 of TB P3 FUNCT/CMES/2 Expected result : REQ-REFERNCE-ID and PATH-COMPUTATION-ID that have been inserted in CANCEL, ACKNOWLEDGE and RESPONSE-PATH messages received from MESCAL21'PCS are identical to what have been inserted in REQUEST-PATH sent by MESCAL11'PCS. Execution date : 21/04/05 Result : The REQ-REFERNCE-ID and PATH-COMPUTATION-ID that have been inserted in CANCEL, ACKNOWLEDGE and RESPONSE-PATH messages received from MESCAL21'PCS are identical to what have been inserted in REOUEST-PATH sent by MESCAL11'PCS. : None. Failure level Remarks : None. Test Reference : TB P3 FUNCT/CMES/4 Test Purpose : Validate QoS information contained in REQUEST-PATH message Procedure Configure MESCAL11' PCS to send a REQUEST message to MESCAL21' PCS. Execute tests TB P2 FUNCT/CMES/5 until TB P2 FUNCT/CMES/10 Expected result : Same results as TB_P2_FUNCT/CMES/5 until TB_P2_FUNCT/CMES/10 Execution date : 21/04/05Result of TB P2 FUNCT/CMES/5 : Obtained results are those until TB P2 FUNCT/CMES/10. Failure level : None. Remarks : None. Test Reference : TB P3 FUNCT/CMES/5 Test Purpose : Validate QoS information contained in RESPONSE-PATH message Procedure Configure MESCAL11' PCS to send a REQUEST message to MESCAL21' PCS. Execute tests TB_P2_FUNCT/CMES/5 until TB_P2_FUNCT/CMES/10 Expected result : Same results as TB P2 FUNCT/CMES/5 until TB P2 FUNCT/CMES/10 Execution date : 21/04/05 Result Obtained results are those of _P2_FUNCT/CMES/5 until **TB P2 FUNCT/CMES/10** Failure level : None. Remarks : None. Test Reference : TB P3 FUNCT/CMES/6

Test Purpose : Check the format of PATH-ERROR and messages.

Procedure

Establish a PCP communication between MESCAL11'PSCA and MESCAL21'PCS.

Scenario 1: Force PATH-COMPUTATION-ID in REQUEST-PATH message to a value that already exists between the two peers.

Scenario 2: Force REF-COMPUTATION-ID in REQUEST-PATH message to a value already that has been handled.

Launch a traffic analyzer that captures MESCAL11' eth1 traffic.

Expected result: in both scenarios, PATH-ERROR message should be sent by MESCAL21'PCS to MESCAL11'PCS.

Execution date : 21/04/05

Result

: A PATH-ERROR message has been sent by MESCAL21 to MESCAL11. The figure below is a capture of the received message (Note that the reason code value is set to 4):

Text Source / Destination Product [min. 0.0001 secol111 BP STEMA 100 Kensage 0.0001 StEMA 100 Kensage StEMA 100 Kensage StEMA 100 Kensage 0.0001 StEMA 100 Kensage StEMA 100 Kensage StEMA 100 Kensage 0.0001 StEMA 100 Kensage StEMA 100 Kensage StEMA 100 Kensage 0.0001 StEMA 100 Kensage StEMA 100 Kensage StEMA 100 Kensage 0.0001 StEMA 100 Kensage StEMA 100 Kensage StEMA 100 Kensage 0.0001 StEMA 100 Kensage StEMA 100 Kensage StEMA 100 Kensage 0.0001 StEMA 100 Kensage StEMA 100 Kensage StEMA 100 Kensage 0.000	T	Tools			
0 0000 med 20 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Time Source	Destination	Protocol .	Info	
1.95839 metal22 metal21 metal111 metal22 mp 9 mp	0,609881 mescal111	mescal212	BGP	KEEPALIVE Hossage	
5.4000 mediality mercelality BP Million Name 5.4000 Million Name Million Name Million Name Million Name 1.45520 Million Name Million Name Million Name Million Name 1.45520 Million Name Million Name Million Name Million Name 7.45520 Million Name Million Name Million Name Million Name 7.45520 Million Name Million Name Million Name Million Name 7.45520 Million Name Million Name Million Name Million Name 7.45520 Million Name Million Name Million Name Million Name 7.45520 Million Name Million Name Million Name Million Name 7.45520 Million Name Million Name Million Name Million Name 7.45520 Million Name Million Name Million Name Million Name 7.45520 Million Name Million Name Million Name Million Name 7.45520 Million Name Million Name <td>1,953155 mescal212</td> <td>wescall11</td> <td>BCP</td> <td>KEEPALIVE Mossage</td> <td></td>	1,953155 mescal212	wescall11	BCP	KEEPALIVE Mossage	
b Scool mentalizy mentalizy mentalizy for the full for t	5,610082 mescal111	mescal212	BCP	KEEPK IVE Mensage	
1.9500 Hu.0.2 Bi.0.0.2 Bi.0.0.2 Bi.0.0.2 Bi.0.2 Bi	6,955207 mescal212	mescal111	BUP	R2EPE INE Ressage	
1.00021 10.0.0.2 10.0.0.2 10.0 10002 10.0002 10.0.0.2 10.0 10	1.953230 mescal111	mescal212	TCP	bgp / wooze (wck.) seq-passance.iz Hokings/sector with itraze Lemin 40022 - how facts' Sectors/SEAFS (inclusion/Sectors/Sectors) (inclusion)	
3.40000 44.00.02 1000 ≥ 2002 (PcH, NC) GerV2012000 Hext3000 Lend 3.40000 44.00.02 30.0.0.2 1000 ≥ 2002 (PcH, NC) GerV20120000 Hext30000 Hext300000 Hext30000 Hext3000Hext30000 Hext3000 Hext3000 Hext30000 Hext30000 Hext3000	3,937172 14,0,0,2	24.0.0.2	TCP	35952 > 1040 [PSH, ACK] Seg=2423219788 Ack=2441536882 Win=5840 Len=48	
3.40034 14.0.0.2 24.0.0.2 TCP 3552 > 100 (ROI) Sept242201085 Act-30435500 Unit-Sept2 5.00035 microllar mic	3,949604 24,0,0,2	14.0.0.2	TCP	1040 > 35952 (PSH, ACK) Seg=2441536002 Ack=2429210036 Win=5792 Len=10	
5,0005 mercal122 mercal22 mercal22 mercal 111 more bar > 4002 4001 See-3993643 41.4-5993664 MinuE3705 Lemo 6,5005 mercal111 mercal22 more bar (2012 see-3993643 41.4-5993664 MinuE3705 Lemo 1,00001 14.0.0.2 71,0.0.2 more series 2015 metration port: 555 9,18999 14.0.0.2 (14,0.0.2), bit A8: 14.0.0.2 (14,0.0.2) met Poteol. Ser Adv: 124.0.0.2 (14,0.0.2) hit A8: 14.0.0.2 (14,0.0.2) metration Control for Deals and Formation Control Series 2015 metration port: 555 Metration port: 1000 (100) Metration Control Series 2015	3.948834 14.0.0.2	24.0.0.2	TCP	35952 > 1040 [ACK] Seq=2429218836 Ack=2441535900 Win=5840 Len=0	
bettom interesting meterializity into a voice 2 beg iAU jeenergy meterializity with the state of the interest ison ison ison ison ison ison ison ison	5.610253 mescal212	mescal111	TCP	bgp > 40022 [ACK] Seq=993962431 Ack=997054594 Win=17220 Len=0	
1.5000 (14.0.0.2 11.0.0.2 100 p Surce port: 2005 Extinuition port: 500 5.16003 (14.0.0.2 71.0.0.2 100 p Surce port: 2016 Extination port: 500 5.16003 (14.0.0.2 71.0.0.2 100 p Surce port: 2016 Extination port: 500 5.16003 (14.0.0.2 71.0.0.2 100 p Surce port: 2016 Extination port: 500 met II. Set: 00500(4.95,555, Det: 00505)721:5004 met II. Set: 00500(4.95,550, Det: 00505) Set: 00500(4.95,650, Det: 005	6.955508 mescall11	Nescal212	TCP	40022 > bgp (HCK) 5eq=997854594 Ack=993962450 Win=16705 Len=0	
4.4644 14.00.2 71.0.0.2 (IBP Source port: 2384 Betination port: 5550 5.16859 14.0.0.2 71.0.0.2 (IBP Source port: 2384 Betination port: 5550 met. Pol. 2000 1000 (Pol. 1000 1000 Pol. 1000 (Pol. 1000 Pol. 1000 Pol. 1000 (Pol. 1000 Pol. 1000 (Pol. 1000 Pol. 1000 (Pol. 1000 Pol. 1000 Pol. 1000 (Pol. 1000 Pol. 1000 Pol. 1000 Pol. 1000 (Pol. 1000 Pol.	1 520021 14 0 0 2	71,0,0,0	ITP	Source port: Scold Bestination port: 5050	
9,180999 14,0,0,2 71,0,0,2 UEP Source port: 2014 Bestination port: 5050 e 8 (94 bytes on wire, 94 bytes centured) rest 11, 5er: 09:09:09:49:45:255, Det: 09:09:07:21:09:04 rest 11, 5er: 09:09:09:04:45:255, Det: 09:09:07:21:09:04 rest 14:00,0 for Put: 1040 (1040), Det Put: 30502 (30502), Seg: 2441538802, Rek: 242021808, Len: 18 Source port: 100 (1040) Bestination port: 5050 (2000) Bestination port: 2000 (2000) Bestination port: 2000 (2000) Bestination port: 5050 (2000) Bestination port: 2000 (6.349943 14.0.0.2	71.0.0.2	LIDP	Sarre port: 2016 Destination port: 5050	
# 8 (64 botes on vine, 64 botes contents) evel 11, 5x: 00000448,2535, Brt: 000000722:50004 evel 11, 5x: 000000448,2535, Brt: 00000000, Data 349: 14.0.0.2 (14.0.0.2) mestion Carteria States (3582) Source part: 1400 (1000) Bettinating ort: 245520000 Rets sequence mader: 2445520000 Rets sequence mader: 245520000 Rets sequence mader: 245520000 Ret sequence mader: 245520000	9,189999 14,0,0,2	71.0.0.2	UDP	Source port: 32814 Destination port: 5050	
Nindex 11:5752 (Strong 1:5156 (Sorrent)) NP NP NP NP NP NP NP NP NP NP	e 8 (84 bytes on wire, 8 rmet 11, 5rc: 00:59/04:44 smet Protocol, 5rc R&F met Protocol, 5rc R&F Search Protocol Search Protocol Search Protocol Search Protocol Search Research Amber: 24455500 Heat search Amber: 244555000 Heat Search Amber: 24455500 Heat Search Amber: 244555000 Heat Search Amber: 244555000 Heat Search Amber: 244555000 Heat Search Amber: 244555000000000000000000000000000000000	<pre>4 bytes captured) 1c2:565, Det: 00:d0:D7 1c2:565, Det: 00:d0:D7 1, Src Port: 1040 (1040 38562) 262 1536300 453210036</pre>	21:50:bd ht Add: 14.0.0.), Dat Port: 359	2 (14.0,0.2) 2 (35502), Seut 24415368882, Bokt 24230218838, Lent 18	
Deckson: 0.3585 (correct.) NGP The state: troil 10376677, teacr 105565004 NGP The state: troil 10376677, teacr 105565004 NGP NGP NGP NGP NGP NGP NGP NGP	Flags: 0x0018 (PSH, ACK) Window size: 5792				
Op/C NOP NOP	Checksum; 0x3F66 (correct.)			
NOP NOP The stare: tavel 1037/4677, teer 1995/5004 x (6) 52,ccs) 00 02 04 10 8c 70 51 86 +5 72 90 cs f0 14 80 18 15 ab 27 55 00 00 01 01 06 04 9c 44 06 c5 06 d5 31 de 8[00 001 100 000 114 0820 304 10 2100 00 x (9) [Perel] Apply Data (data), 18 bytes	And in the Add to A and the Address of the Address				
00 02 04 10 %; 70 51 85 +65 72 50 ca f0 14 50 18	obrious: (15 pBres)				
(1)(1)(2)(2)(2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2)	NOP	64677, taer 190545004			
00 02 04 10 월 70 51 월 87 72 50 ca 40 14 80 18	NOP NOP Time stamp: toyal 1937	A CALL & A C			
이 이 이 10 등 70 명 10 등 72 명 이 18 명 18	NOP NOP Time stamp: toval 1937 (48 Refera)				
00 02 04 10 원는 70 91 86 - 65 72 90 04 F0 14 80 18	NCP Time stamp: toval 1837 (18 bytes)				
00 02 04 10 急 70 11 回 47 72 20 04 10 15 回 18 15 47 27 25 00 02 01 01 04 05 05 14 05 16 05 16 14 45 17 55 00 02 18 05 16 05 15 15 15 15 15 15 15 15 15 15 15 15 15	NCP Time stamp: toval 1837 (18 bytes)				
00 02 04 10 8c 70 51 86 45 72 50 cs f0 14 80 18	NOP Time stamp: toval 1037 (10 Epter)				
	NOP NOP Time stamp: toval 1937 w(18 Rgtes)				
00 002 04 10 65 70 51 85 #5 72 50 04 #0 18 80	NGP NGP NGP Time stamp: toval 1937 (400 Rgtcs)				
00 00 00 10 00 00 73 16 00 - 47 7 28 00 c 4 70 14 00 15	NGP NGP NGP NGP NGP NGP NGP NGP NGP NGP				
15 að 25 fés 60 do ort út út de Los á fes fés 16 a de 14 að 15 fés 60 do andre 14 de 56 fés fés 16 a de tot at an de tot at an de (*) Restel [Apply] Data (data), 18 bytes	vop NOP NOP Time stamp: toval 1937 v(40 bytes)				
n: do general accessor in contract data accessor contract of the second data accessor in the second data accesso	00 02 04 10 5- 70 91 86 1	#5 72 90 ca 40 14 90 11	8 p. r.		
Reset Apply Data (data), 18 bytes	00 02 04 10 8c 70 51 66	#5 72 90 ca f0 14 80 11 66 0a 0a f4 05 c5 0b d	8Pr 5P.		
▼ Reset! Apply Data (data), 18 bytes	00 00 04 10 5c 70 51 66 54 06 00 Time stampt towal 1037 4 (00 kpccs) 00 02 04 10 5c 70 51 66 54 00 25 65 00 00 01 01	#5 72 90 cm f0 14 80 11 88 0a 0a f4 05 65 0b # 11 05 65 55 45 40 80 00 0	8PF 5PF		
	00 02 04 10 5c 70 51 55 15 4 0 25 50 00 01 00 05 16 4 0 20 04 10 5c 70 51 55 15 4 0 25 50 00 00 01 01 15 4 0 20 00 00	#5 72 90 ca f0 14 80 11 68 0a 0a f4 65 c5 bh d 11 03 65 c5 c4 0a 00 07	6 P F 5 P F 6 P F 7		

Failure level: None.Remarks: None

 Test Reference
 : TB_P3_FUNCT/CMES/7

 Test Purpose
 : Check the format of CANCEL and messages.

 Procedure
 :

 Establish a PCP communication between MESCAL11'PSC and MESCAL21'PCS.

Configuration is the same as for scenario 1 of TB_P3_FUNCT/CMES/2.

Configure MESCAL11'PCS to send a CANCEL message to MESCAL21'PCS during path negotiation.

Launch a traffic analyzer that captures MESCAL11' eth1 traffic.

Expected result : CANCEL message should be as specified in introduction if this test group. PATH-COMPUTATION-ID and REQ-REFERENCE-ID must be the same as what have been used in the REQUEST message.

Execution date : 21/04/05

Result

: The format of the CANCEL message is conform to D1.2 specifications. PATH-COMPUTATION-ID and REQ-REFERENCE-ID are the same as what have been used in the REQUEST message as illustrated by the figures below: the first one is for the REQUEST message, and the second one for the CANCEL message:

¥ <c< th=""><th>aptures =</th><th>Ethereal</th><th></th><th></th><th></th><th>- # X</th></c<>	aptures =	Ethereal				- # X
Elle	Edt Cat	pture Display	Tools			Help
No.	Time	Source	Destination	Protocol .	Info	
5	2,252468	24.0.0.2	14.0.0.2	TEP	1040 > 35058 [PSH, RDK] Seq=2320524176 Rok=2295500705 Win=5792 Len=4	
6	2,252544	14.0.0.2	24.0.0.2	TCP	35550 > 1040 [RDK] Seq=2259500705 Ack=2528524180 Win+5840 Len=0	
7	2,309911	14.0.0.2	24,0.0.2	109	3958 > 1040 [FSH, BK] Sep223550076 Rds:2525824180 Vin=5640 Lmv4	
8	5 745667	24.0.0.2	14.0.0.2	TCP	1040 > 50/56 (HK) 54/22/5524100 HKH/22/550/7/3 Min/5/32 Lan/0	
10	2,343003	24.0.0.2	14.0.0.2	102	Delto / Subo (FSR, MA) September Son Minute View State (Section Control of Co	
11	2.762249	14.0.0.2	24.0.0.2	TOP	100) 5099 [GC] Sec(3)47541 (e):23747834 [E:0579 Lex)]	
13	5,000233	nescal212	rescal111	TCP	bae > 40022 [ADK] Sev-983947117 Ack-987839299 Min-17220 Lev-0	
15	6,092265	mescal111	nescal212	TEP	40022 > bgs [ACK] Seg/997833299 Ack/993947136 Wini16705 Lev/0	
18	10,009918	nescal212	mescal111	TCP	bgp > 40022 [ACK] Seqi/93347136 Ack/997633318 Wini17220 Lenil0	
25	11,092334	nescal111	nescal252	TCP	40022 > bge [ADK] Seq1997033310 Ackr993347155 Wint16705 Lenn0	
25	\$5,020798	mescal212	mescal111	TCP	bge > 40022 [40K] Seqr993947155 Ackr997039357 Wins17220 Lenv0	
- 25	16,092365	mescal111	mescal212	TCP	40022 > bge [ADK] Seqr/997839337 Ack:993347174 Win=16705 Lenr0	
- 27	20,029924	mescal212	mescal111	1CP	bgp > 40022 [ADK] 5eqr/983947174 Ack:/987839856 Wins17220 Lenz0	
20	TO COMPANY	24.0.0.2	24.0.0.2	102	2000 2 1040 [P34: R51] 5eg223000/09 R6323200410 91m0040 Lem40	
20	20,001314	24.0.0.2	14.0.0.2	tre	1040 / SOUGE (MA) September 2000 March 2000 March 2000 March 2000 Land 2	
24	20.014601 30.01EE44	14002	24.0.0.2	trp	2040 / 30300 [F36, #A] 30[#22032430 H04223000090 [H142040] E-H142	
22	24 090474	associal tit	nenca1252	100	2009 > has [dfg] 5ac0909393956 intro2021/192 Unit5706 [acc0]	
34	21.616572	14.0.0.2	24.0.0.2	TCP	35058 > 1040 [PSH, ACK] Seu-2239500757 Ack-3328524222 Win-5840 Lev-18	
35	21,615745	24.0.0.2	14.0.0.2	TCP	1040 > 35058 [ACK] Sey=2328524222 Ack=2239500775 Win=5792 Len=0	
37	22,349651	14.0.0.2	24.0.0.2	TCP	1040 > 60959 [PSH, ACK] Seg=2314473641 Ack+2327478349 Win=5792 Len=4	
38	22,352313	24,0,0,2	14,0,0,2	TCP	60953 > 1040 [PSH, HCK] Seq#2327478348 Hok#2314473645 Win#5840 Len#4	
39	22,352409	14.0.0.2	24.0.0.2	TCP	1040 > 60959 [ACK] Seq12314473645 Ack12327478352 Vini5792 Lenit0	
-40	22,455/50	14.0.0.2	24,0,0,2	TOP	35058 > 1040 [PSH, ACK] Seq#2299500775 Ack#2329524222 Wie#5840 Levit8	
45	22,455236	24,0,0,2	14.0.0.2	TDP	1040 > 35050 [ADK] 5eqr2328524222 Adk:2239500793 Wind5792 Lenn0	
	lestinatio leguence n lert seque kchrouledg leader len lags: 0x0 lindow siz becksus: i becksus: i NOP NOP Time st	n port: 1040 (10 umber: 229950070 nce number: 22995 gsh: 32 bytes 020 (PSN, RC) e: 5840 0x8529 (correct) 12 bytes) amp; toval 19814 3	00) 900757 20524100 4340, taecr 18336131	0		
* 0040 0050 0050 0070	5 ⁶ 1e <mark>101</mark> 0 19 82 00 0 12 82 02 0 10 00	4 00 20 00 01 0 1 00 00 01 01 0 1 15 15 15 15 0	0 01 00 06 11 07 94 1 01 01 06 15 15 15 1 00 04 14 11 00 05			
Filter					🗢 Reset Appry Data (data), 48 bytes	
-		0000	2		III mescal #MES @ <capture> - Et #Network Config Path Computat Java</capture>	C D D Thu Apr 21, 15:21
		A. (1) 4				COLUMN STATE

	thuse Folial	and the second		
o. Time	Source	Destination	Protocol .	Info
8 2,310094	24.0.0.2	14.0.0.2	TCP	1040 > 35058 [ACK] Sey=2328524180 Ack=2239500708 Win=5732 Len=0
9 2,349883	14.0.0.2	24.0.0.2	TCP	1040 > 60959 [PSH, RCK] Seq=2314473637 Rek=2327478344 Win=5732 Len=4
10 2,362183	24,0,0,2	14,0,0,2	TCP	60953 > 1040 [PSH, RCK] Seq#2327478344 Rck#2314473641 Win#5840 Len#4
11 2,352249	14,0,0,2	24,0,0,2	TCP	1040 > 60959 [ACK] Seq#2314473641 Ack#2327478348 Win#5792 Len#0
13 5,000233	nescal212	nescal131	1CP	bgp > 40022 [ADK] Seq1993947117 Ack:1997839299 Wini17220 Leni0
15 6,092265	nescal111	Nescal212	TUP	40022 > bgp [ADK] Seqr997878299 Ackr997874136 Winn16705 Lenn0
18 10,009910	B mescal212	nescal111	TUP	bge > 40022 [ADX] Seqr993347136 Ackr997833318 Wiwt17220 Levr0
21 11,092334	4 mescal111	mescal212	TCP	40022 > bgp [ADK] Sep1997033318 Ack:193347155 Win:16705 Lenn0
23 15,020790	0 mescal212	mescal111	TCP	bgp > 40022 [ADX] 5eqr993947155 Ack:197039337 Winx17220 Lenn0
25 16,092308	5 mescall11	nescal212	TCP	40022 > bgp [ADK] Seqn997033837 Ackn993947174 Wint16705 Lenn0
27 20,02992	4 mescal212	mescal111	TOP	bgp > 40022 [40X] Sept80347174 Ack:00103056 Wint17220 Levn0
28 20,601745	5 14,0,0,2	24,0,0,2	TCP	SODE > 1040 [F3H, HKJ] Sept22550070 Hoks25252410 WintsHo Lens40
23 20,601914	26,0,0,2	14,0,0,2	109	Loss 2 Social (Inc.), sequencemental Relia 22/89/00/757 Winter/12/ Lenses
20.014601	4 44 0 0 2	24.0.0.2	TLP TCD	APPLY A SUMPLIFY THE THE STATE OF A SUPERCENT AND A SUPERCENT
35 2V.010044	1 24.V.V.Z	24.0.0.2	169	30/00 / 2/WV (HAR) SEPARATION THAT AND A THE AND A STATE AND A STA
24 21 61062	7 14 0 0 2	24.0.0.2	709	WARL / Up (Int.) 30(2000) 000 000 000 000 000 000 000 000 00
26 21.016074	5 24 0 0 2	14.0.0.2	709	100 NUMP (2011 Exec(PSI2022011) HERE2200422 HIRSDN Land
27 29 24966	1 14 0 0 9	24.0.0.2	trp	tan's pose texts and the second secon
38 29 36231	3 24 0 0 2	14.0.0.2	TEP	0969 > 1040 [PSi, 403] Sec(22727028) Bet/221427645 UnitS640 [cmit
25 22, 3524/4	9 14.0.0.2	24.0.0.2	TCP	1040 > 50959 [40X] Sext033423645 4e8+9327478382 Wint6739 Level
40 22,455/6/	14.0.0.2	24.0.0.2	TCP	25058 > 1040 [PSH, #01] 5epr2293500775 Ark=2339504222 Win=5940 Lev=18
41 22,49523	5 24,0.0.2	14.0.0.2	TEP	1040 > 35058 [ADX] Sem-2328524222 Adx+229550733 Kin+5792 Len+0
15 7,149932	14,0,0,2	71.0.0.2	UDP	Source port: 32808 Destination port: 5050
19 10,159993	2 14.0.0.2	71.0.0.2	UDP	Source port: 32806 Destination port: 5050
35 22,149910 Frame 40 (84 Ethernet II, Internet Prof Transission Source po Destinati Sequence Next sequ	5 14,0,0,2 bytes on wire, 1 Sre: 00:00:057:2 tocol, Src Mddr: Control Protoco rt: 35058 (35058) on port: 1040 (1 number: 2295007 ence number: 223	71,0,0,2 84 bytes captured) 1:502bd. Bet: 00:50:04:4 14,0,0,2 (14,0,0,2). Be 1, Src Part: 35058 (3505 0 000) 75 5900093	USP 8103163 t Addr: 24.0.0.1 8), Bst Port: 10	Sarce port 2000 Betination port2 5000 2 (24,0,0,2) 40 (1000), Seg 225950075, Hok: 253952422, Len: 18
35 22.149916 Frame 40 (84 Ethernet II, Internet Prof Transmission Source po Destinati Sequence Next sequence Next sequence Header 1e ⊞ Flags 0w Uindow si Checksung	5 14,0,0,2 bytes on wire, Sec: 00;00;27/2 tocol, Src Ndb7 Control Protoco rt: 30058 (3506) on port: 1040 (1 number: 2235500) erce number: 223 byment number: 2 byment number: 2 bytes number: 255500 ercs 5040 onfacb (correct	71,0,0,2 84 bytes captured) 1500td, Ibit: 00500444 14,0,0,2 (14,0,0,2), Ibi 1, Sre Part: 35958 (3595 0 000) 75 505034222)	UBP 80:03:63 t. Heddr 2 24,0,0,1 8), Dut Port: 10	Source ports 2000 Bentination ports 2000 2 (34,0,0,2) A0 (1000), Seg: 229590775, Hok: 2330524222, Lord 10
36 22.149916 Frame 40 (84 Ethernet II. Internet Prof Destinati Sequence Next sequ Acknowled Header In © Flags: 0x Undow si Checkwar © Options:	b 14.0.0.2 bytes on wire, 1 Src: 00(d)0572 Cool, Src 00(d)0572 Control Protoco resolution of the second reso number: 2295007 secon number: 22 syment number: 2 syment number: 3 syment number: 3	71,0,0,2 94 bytes captured) 150bd, hit: cor50;04:4 14,0,0,2 (14,0,0,2), hi 14,0,0,2 (14,0,0,2), hi 14,0,0,2 (14,0,0,2), hi 14,0,0,2 (14,0,0,2), hi 15,000;03 1500;04222 1500;04222	USP 82:63:63 4. Hoddr: 24.0.0.1 8), Bat Port: 20	Sorce perti 2000 Bentination perti 5000 2 (34,0,0,2) Au (1000), Segi 20050075, Aui 233052422, Lun; 18
36 22.149916 Frame 40 (84 Ethernet II., Internet Prof Destinati Sequence Next sequence Next sequence Header 1e @ Flags 0o Uindou si Checkung B Options: NOP	bito.co.2 bytes on wire, i Src: 00(d)057/2 Control Protoco ett: 3008 (3506 on port: 1040 (1 maber: 2295007 mean: namber: 229 jement namber: 229 je	71,0,0,2 84 bytes captured) 1500td, Int: 00500424 14,0,0,2 (14,0,0,2), In 1, Sre Part: 35968 (3505 0) 0(a0) 75 950093 120504222)	USP 9:c3:63 t Hode: 24,0,0,3 8), But Port: 10	Source ports 2000 Bestination ports 2000 2 (34,0,0,2) 40 (1000), Sey: 22950075, Rok: 2529524222. Lent 18
35 22.149916 Frame 40 (34 Ethernet II., Internet Prof Destinati Sequence Next sequence Header 1e B Flags: 0x Window B Options: NOP NOP	bit.o.o.2 bytes on wire. 1 Src: 00(d)05722 tocol. Src 00(d)05722 transformer state resolution of the state resolution of the state on port: 1040 [1 nuber: 2295000 state bytes out to the state state bytes out able (correct (12 bytes)	71,0,0,2 94 bytes captured) 1505db, Brit 2055(04:4 14,0,0,2 (14,0,0,2), br 1, Src Parts 35686 (3565 0) 0000) 75 95000933 1500524222	USP 80:53:53 1 Hoddr: 24,0,0,1 8), But Port: 10	Source port 12000 Settination port 2000 2 (34,0,0,2) 40 (1000), Set 22990075, Rok 25085422, Lent 18
36 22.149914 Frame 40 (34 Ethernet II., Internet Prof Destinati Sequence Next explosion Acknowled Header 16 E Flags: 0x Window 11 Oneckowg B Options: NOP Time 1	bit.ou.2 bytes on uire, i Seci 00:00:57:0 Control Protoco en port: 3008 (3508 on port: 3048 (3508 on port: 3048 (3508 maber: 2295007 ence nuber: 229 byment nuber: 2 bytes: 5040 0:nfa2b (correct (12 bytes) tame: truel 1985	71,0,0,2 94 bytes castured) 1505dd, Jmir 00;50(04;4 14,0,0,2) (14,0,0,2), Jm 14,0,0,2) (14,0,0,2), Jm 15 for first 305d6 (305d) 1000) 1000 100	00P 9:c3;63 t Hoddr: 24,0,0,0,0 8), But Port; 10	Sarce port 2000 Betlantin port 2000 2 (24,0.0.2) AD (1000), Sey 22950075, RG: 25352422, Lent 18
36 22.14991 Frame 40 (84 Ethernet II. Internet Froi Frameission Source po Destinati Sequence Net sequ Acknowled Header in © Flags: 0x Window © Options: NOP NOP NOP NOP	5 14.0.0.2 bytes on usres. Sec: 00:40/05/2 tocol, Sec: 00:40/05/2 tocol, Sec: 00:40/05/2 tocol, Sec: 00:40 control Protocol Control Protocol control 10:40 more number: 2205007 wroe number: 20050 wroe number: 200500 wroe numbe	71,0,0,2 71,0,0,2 1159104 [http://sof.sof. 1159104 [http://sof.sof. 14,0,0,2 (ltd.,0,0,2), http://sof. 14,0,0,2 (ltd.,0,0,2), http://sof. 14,0,0,2 (ltd.,0,0,2), http://sof.sof. 1000) 10000000000	UBP 92:53:63 t. Heddr: 24.0.0.1 8). Bet Port: 20	Source ports 2000 Bestination ports 2000 2 (34,0,0,2) AND (1000), Swg; 229950075, Hox; 230952422, Long 18
35 22.149914 Frame 40 (94 Exhernet 11, Internet Prof Destinati Source po Destinati Sequence Header 10 B Flags: 0x Window si Orekisway B Options: NOP Time 5 State (98 Ded	5 14.0.0.2 bytes on uire. See: 00;40;57;2 tocal, Srn Rddr: Control Protoco ret: 30508 (3506) secon suber: 203500 march rubber: 22 jament nubber: 2 jament nubber: 2 jament nubber: 2 jament subt set 5040 onfach (correct (12 bytes) stame: toval 1981 st)	74,0,0,2 15004, Brit Goldon, Goldon, Goldon, Goldon, Brit Goldon, Brit Goldon, Brit Goldon, Brit Goldon, Brit Goldon,	00P 92:53:63 t. Heddr: 24,0,0,1 8), But Port: 30	Sorce perti 2000 Settination perti 5000 2 (34,0,0,2) Au (1000), Set 2200500795, Roi EX0052422, Lon; 18
35 22,14991/ Frame 40 (34 Ethernet II, Internet Prei Source po Destination Source po Destination Source po Destination Source po Destination Heador II Birlags: 0x Window si Declaus Gatonics MP MP Dire I Stan Singd	5 14.0.0.2 bytes on wire. 5xc: 00:0015/2 tocol. Src: 00407 Control Protoco rt: 30008 (35008 more number: 2295007 more number: 2295007 more number: 2 nyth: 32 bytes 0048 (Correct (12 bytes) stamp: toval 1985 col	71,4,0,2 110924, Brit 095904 110924, Brit 095904 14,0,0,2 (14,0,0,2), L, Brit 0,0,0 14,0,0,2 (14,0,0,2), L, Brit 0,0 14,0,0,2 (14,0,0,2), L 8500513 130054222)) (44525, tarcr 10335341	UBP 92:53:63 t.Hodar: 24.0.0.3 8). But Port: 20	Source ports 2000 Bestination ports 2000 2 (34,0,0,2) A0 (1000), Seg: 22950075, Hok: 233052422, Lec: 18
36 22,14991/ Frame 40 (34 Ethernet 11,1 Internet Prof Destination Source of the Source of the Bestination of the Source of the Market seau Achroulde si Priagit on the Source of the MCP Time is standard	5 14.0.0.2 bytes on wire., 5rc: 00:0015/12 tocol, Src: 00:015/12 tocol, Src: 00:015/12 tocol, Src: 00:01 nuber: 120500 more nuber: 225 generi nuber: 225 gen	71,0,0,2 74,0,0,2 11000A, Bait 003004 11000A, Bait 003004 14,0,0,2 (14,0,0,2), b, b 14,0,0,2 (14,0,0,2), b 14,0,0,2 (14,0,0,2), b 100004 1000000000 10000000000	UBP S2c3:63 t. Hadr: 24,0,0,1 0). But Port: 10	Sores perti 2000 Settination perti 5000 2 (34,0,0,2) 40 (1000), Set 220050075, Roi: E20052422, Len: 18
35 22,14991 Frame 40 (34 Ethernet 11, Internet Prof Transmission Bentinati Sequence Mart sequence Mart sequence Mart sequence Bentinati Bendenas Bentinati Mart sequence Mart sequence	5 14.0.0.2 bytes on wire.1 Src: 00;40;57:2 tocol, Src: 00;40;57:2 tocol, Src: 00;40; runsber: 20;500;0 macher: 20;500;0 macher: 20;500;0 macher: 20;500;0 prot: 22;500;0 prot: 22;500;0 prot: 22;500;0 prot: 22;500;0 prot: 22;500;0 prot: 20;500;0 prot: 20;500;0 pr	71,0,0,2 74,0,0,2 150004, Mn1 005004 150004, Mn1 005004 1, 5er Port 20058 (386 000) 75 75 75 75 75 75 75 75 75 75	(BP 8v:5155 1 Hodr: 24.0,0,1 10), But Port: 10	Source porti 2000 Bestination porti 2000 2 (24,0,0,2) 40 (1000), Segi 225950075, Hok: 253952422, Lec: 18
36 22,14991 Frame 40 (34 Ethernet 11,1 Internet Prof Transmission Source po Destinati Sequence Net sequence Meta sequence Belags to Bilags to NOP Time Batan Source Source Source NOP Time Batan Source Sour	5 14,0,0,2 bytes on wire. 55:00(0)5075 bools. For Addr Control Protoco ext 30508 (3566 on sport 1004) maderi 225500 march auder: 12 255500 0016 (0,0) control transfer 25500 0016 (0,0) tage: tawal 1987 10 10 10 10 10 10 10 10 10 10	71,4,0,2 74,4,0,2 150004, Beit 0020040 150004, Beit 002004 1, 5c Port 2006 (500 000) 75 5005/02) 14605, taer 10330341	UBP 80:5163 # Addr 2 44,0,0,1 0), Bat Fort: 10	Source port 12000 Settination port 2000 2 (34,0,0,2) 40 (1000), Set 22990075, Rok: 250952622, Lon; 18
35 22,149914 Frame 40 (34 Ethernet 11,1 Internet Prof Transmission Bestinati Sequence Header Best Header Best Best Header Best Mor Mor State 400 pp Time p	5 14,0,0,2 bytes on virs. 55:00(3)(3)(2) tool. For Addr Control Protoco Art 3008 (3)(5)(3) mander 120000 mander 1200000 mander 120000 mander 1200000 mander 120000 mander 120000 mander 120000 mander 1200000 mander 1200	71,0,0,2 74,0,0,2 19004, Mn1 002004 19004, Mn1 002004 1,5c Port 2006 (200 000) 75 50050733 100054222))	UDP 95:5755 4 mdy: 24.0,0,0 80). But Port: 10	Source port 2000 Bestination port 2000 2 (24,0,0,2) 40 (1000), Seg: 22950075, RG: 250524222, Lord 18
36 22,14993 Frame 40 (34 Ethernet 11, Internet Prof Transmission Bentineti Segunce Hends resu Refs resu Berlineti B	5 14,0,0,2 bytes on wire, 55:00(0)502 toool, 5 vo Matri toool, 5 vo Matri toool, 1 vo toorkon f toool or spert 1000 (1 matrix 10000 (1 matrix 1000 (1	71,0,0,2 74,0,0,2 15000, Brit 002000,0,0,4 15000, Brit 00200,0,0,4 1, 5c Port 2006 (350 000) 000) 10054022) 144025, taser 10380344	υρρ Φεοδισδ Η επάφε 24.0,0,0 8), Det Port: 10	Source port 12000 Settination port 2000 2 (34,0,0,2) AN (1000), Set: 22090075, Hot: 20052422, Len: 18
36 22.149914 Frame 40 (84 Ethernet II.) Internet Prof Transmission Destinati Sequence Ment sea Ret se	5 14,0.0.2 bytes on wire. Sec: 000/00/201 bool. Sec: 000/00/201 bool. Sec: 000/00/201 bool. Sec: 000/00/201 bool. Sec: 000/00/201 bool. Document mader: 20500 (Tooler mader: 2050) bytes mader: 20500 (Tooler bool. Document output: table: tool. 1000 table: tool. 1000	71,4,0,2 74,0,0,2 150004, Bert 0025004 150004, Bert 0025004 150004, Bert 0025004 15000403 75 75 75 75 75 75 75 75 75 75	UDP 90:5:55 10:45:22 80), Dat Port: 10 80), Dat Port: 10	Source port 12000 Settination port 5000 2 (34,0,0,2) 40 (1000), Set 220000075, Roi 220052422, Len: 18
36 22,14993 Frame 40 (34 Ethernet 11, Internet Prof Transmission Bentinati Sequence Header to Beltinati Belgen to Unicke at Belgen to Unicke at Belgen to Unicke at Belgen to Decknag Options: State 4000 MCP	<pre>5 14,0.0.2 bytes on wire. 5ec: 00:00:00:12 toorls 7 book of Mart Control Protoco fits port 1 book of the formation of the mean maker 2 2008 person maker 2 ragbh: 2 bytes 0:midb (correct 12 bytes) 0:midb (correct 12 bytes) 0:midb (correct 12 bytes) 0:midb (correct 12 bytes) 0:midb (correct 12 bytes) 0:midb (correct 13 bytes)</pre>	71,0,0,2 74,0,0,2 15000, Brit 002000, 2 15000, Brit 00200, 2 1, 5c Fort, 2006 (385 000) 75 5000000 10000, 2 10000, 2	UPP 92:5355 84:95:24:0,0,0,0 8), Bet Port: 1	Source port 15000 Bestination port 2000 2 (34,0,0,2) W0 (1000), Swg 229950075, Wag 25095022, Long 18
36 22,14993 Frame 40 (34 Ethernet 11, Internet Prof Transistication Source po Destination Source po Destination Source po Destination Source po Destination Berlingtion Berlingtion NOP Times State (38 bod Source Source po Briageton NOP Times Source Source po Briageton NOP Times Source Source po Briageton NOP Times Source Source po Briageton NOP Times Source Source Source po Briageton NOP Times Source Sourc	5 14,0,0,2 bytes on wire, 55: 000(1):07: 000(1): bost: 000(1):07: 000(1): bost: 000(1):07: 000(1): bost: 000(1):000(1): meter subset: 2350(1): meter subset: 2350(1):	7,2,4,0,2 7,4,0,0,2 15000A, Brit 005000 15000A, Brit 005000 1, 500 Not 2008 (500 000) 75 50051422) 144525, taser 10536241 300 A Bac co 75 be 00 10 300 A Bac co 75 be 00 A Bac co 75 be 00 10 300 A Bac co 75 be 00 A Bac	UPP 92:53:53 8:65:23 (0,0) 0), Dat Port: 31	Source port 1 2000 Settination port 2000 2 (34,0,0,2) 40 (1000), Set 20000075, Not 20002022, Len; 18
36 22,14931 Frame 40 (34 Ethernet 11,2 Internet Prof Transmission Sector Prof Parametrision Sector Prof Bertaget Bertaget Bertaget Bertaget Bertaget Bertaget Bertaget Bertaget Star The Sector Star Secto	5 14,0,0,2 Bates on wire, 1 Sec1 (9,0,0,0)2 Sec1 (9,0,0,0)2 Control Protoco rest (9,0,0,0)2 Control Protoco rest (9,0,0)2 Control Protoco rest (9,0,0)2 Sec1 (9,0	7,4,0,02 7,4,0,02 10000, Brit 002004 10000, Brit 002004 10000, Brit 002004 10000, Brit 002004 10000 1000 10001 77 77 70 100000 100000 10000 100000 10000 10000 10000 1	UP 92:03:03 16:45:24,0,0,0 0), Bat Port: 31	Source port 1 2000 Settination port 5000 2 (34,0,0,2) Mo (1000), Set 20050075, Aug 22052422, Leg 18
36 22,14993 Frame 40 (34 Ethernet 11, Internet Prof Transistion Source po Destination Source po Destination Source po Destination Source po Destination Source po Destination Source po Program Nor These States Source po Nor These Source po Nor These Source Source po Source po Nor These Source Source po Source po Source po Source po Nor These Source Source po Source po Nor Source po Source po Nor Source po Source po Source po Source po Nor Source po Source po Nor Source po Source po Nor Source po Source po Nor Source po Source po Nor Source po Source po Nor Source po Nor Source po Source po Nor Source po Nor Nor Source po Nor Source po Nor Source po Nor Source po Nor Source po Nor Source po Nor Source po Nor Source po Nor Nor Source po Nor Source po Nor Nor Source po Nor Source po Nor Nor Nor Source po Nor Nor Nor Nor Nor Nor Nor No	5 14,0,0,2 5 14,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0	7,2,0,0,2 7,2,0,0,2 15000, Brit 00500,0,2,4, B 15004, Brit 00500,0,2,4, B 1, 5c, 1041, 2008 (1500 000) 75 5005422) 44525, there 103363341 90 -7 Ba c4 75 Be 00 11 90 -7 Ba c4 75 Be 00 11 91 -7 Ba c4 75 Ba c0 00 11 91 -7 Ba c4 75 Ba 00 11 91 -7 Ba 00 10 91 -7 Ba 00 10	UP 90:51-59 16:45:24,0,0,0 0), Bat Port2 30 00, Bat Port2 30	Source port 1 2000 Settination port 2000 2 (34,0,0,2) 40 (1000), Set: 22090075, Roi: 21002022, Lor: 18
36 22,14931 Frame 40 (34 Ethernet 11, Internet Prof Sarte 90 Bestinati Sarte 90 Bestinati	E 14.0.0.2 Bytes on version 22 Bytes on version 22 Bytes on version 22 Bytes on version 22 Control Protoco entry of the second second entry of the second second entry of the second second entry of the second second second entry of the second second second second entry of the second se	21,40,0.2 21,40,0.2 110004, Brit 00,50044,0.2,7, Brit 00,5004 110004, Brit 00,50044,0.2,7, Brit 00,000 10,000,1,000,0,000 20,000 20,000 21,40055, taeyr 10,5005042 20,000,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,0,	USP 92:03:03 16:45:24,0,0,0 0), Bat Port: 30 00:00 00	Source port 2000 Bestination port 2000 2 (24,0,0,2) 40 (1000), Seit 229950075, Hoi: 250954222, Leri: 19
36 22.14931 Prese 40 131 Prese 40 131 Pr	5 14.0.0.2 Set: 0.0.014 Set: 0.0.014 Set: 0.0.015 Set: 0.0.015 Set: 0.0.015 Set: 0.0.015 Set: 0.001 Set: 0	7,4,0,02 15000, Brit 005000, Brit 005000 15000, Brit 005000, Brit 00500 1, 50 Pril 2000 (1500 000) 75 5000 (202) 44525, taser 103352341 99 a7 Bacca 75 be 00 10 90 a7 Bacca 75 be 00 11 100 brit 00 75 brit 00 100 brit 00 75 brit 00 100 brit 00 50 brit 00 100 brit 00	USP 90:51:55 80:51:24,0,0,0 80: Bat Ports S	Source port 2 5000 Bettination port 2000 2 (34,0,0,2) 40 (1000), Set: 22090075, Rol: 220928222, Ler: 18

Failure level: None.Remarks: None.

Test Reference : TB P3 FUNCT/CMES/8 : Check operational behaviours when receiving REQUEST messages. Test Purpose Procedure Launch a traffic analyzer that captures MESCAL11' eth1 traffic. Establish a PCP session between MESCAL11'PCS and MESCAL21'PCS. Scenario 1: When PCP session is accepted by MESCAL21'PCS, configure MESCAL11'PCS to send a request for an LSP with a head-end-address 1.1.1.6 and tail-end-address in 2.2.2.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 0. Scenario 2: When PCP session is accepted by MESCAL21'PCS, configure MESCAL11'PCS to send a request for an LSP with a head-end-address 3.3.3.1 and tail-end-address in 2.2.2.1 in MC1 and a bandwidth of 1Mbit. Scenario 3: When PCP session is accepted by MESCAL21'PCS, configure MESCAL11'PCS to send a request for an LSP with a head-end-address 1.1.1.6 and tail-end-address in 2.2.2.1 in MC1 and a bandwidth of 5Mbit. Examine the messages exchanged between the two PCS. Expected result : Scenario 1: MESCAL21' PCS must send a ERROR-PATH message to MESCAL11' PCS with TTL expired error (value = 5)

 Scenario 2: MESCAL21'PCS must send a ERROR-PATH message to MESCAL11'PCS (value = 3) Scenario 3: MESCAL21's must send a ERROR-PATH message to MESCAL11'PCS (value = 1)

Execution date : 21/04/05

Result : Obtained results are as follows:

 Scenario 1: MESCAL21' PCS sent a ERROR-PATH message to MESCAL11'PCS with TTL expired error (value = 5)

«capture» - Ethereal				- 6
Edit Capture Display	Tools			E
Time Source	Destination	Protoco	· Into	
6 6,656535 mescal111	mescal212	TCP	40022 > bap [ACK] Seq1997853321 Ack:1993961177 Win:16705 Lent0	
3 10.020102 mescal212	mescal111	TOP	bgp > 40022 [ACK] Seq=993961177 Ack=997853340 Win=17220 Len=0	
1 11,656824 mescal111	mescal212	TOP	40022 > bap [ACK] Seq=997853340 Ack=993961196 Win=16705 Len=0	
5 15.031135 mescal212	mescal111	TOP	bgp > 40022 [ACK] Seq=993361196 Ack=997853359 Win=17220 Len=0	
5 16,666613 wescal111	mescal212	TOP	40022 > bgr [ACK] Seq=997953359 Ack=993961215 Win=16705 Len=0	
20,039803 mescal212	mescal111	TOP	bgp > 40022 [40K] Seq#993361215 Ack#997853378 Win#17220 Len#0	
21.866840 mescal111	nescal212	TOP	40022 > bgp [ACK] Seq=997853578 Ack=993961234 Win=16706 Len=0	
22.296672 24.0.0.2	14,0,0,2	TCP	1040 > 35863 [PSH, ACK] Seq:2049201298 Ack:2054260439 Win:5792 Len:4	
22.319662 14.0.0.2	24.0.0.2	TOP	35863 > 1040 [PSH, RCK] Seq=2054260439 Rck=2049201302 Min=5840 Len=4	
22,319837 24,0,0,2	14.0.0.2	TCP	1040 > 35863 [RCK] Seq=2045201302 Rot=2054260443 Win=5752 Len=0	
22.349695 14.0.0.2	24.0.0.2	TOP	1040 > 33509 [PSH, RCK] Seg=2047080726 Ack=2047051585 Win=5732 Len=4	
22.350047 24.0.0.2	14,0,0,2	TOP	33509 > 1040 [ACK] Seq=2047651585 Ack=2047396730 Vin=5840 Len=0	
22,406623 24,0.0,2	14,0,0,2	TOP	33509 > 1040 [PSH, ACK] Segr2047051585 Ack=2047986730 Win=5840 Len=4	
22,406812 14,0.0.2	24.0.0.2	TOP	1040 > 33509 (RCK) Seq=2047886730 Rck=2047651589 Win=5752 Len=0	
23,433936 14,0,0,2	24,0,0,2	TOP	35963 > 1040 [PSH, ACK] Sepr2054250443 Adx:2049201302 Mixr5840 Len:48	
23.434149 24.0.0.2	14.0.0.2	TCP	1040 > 35863 [ACK] Seq=2045001302 Ack=2054260491 Win=5792 Len=0	
23,438219,24,0,0,2	14,0,0,2	109	1040 > 37883 [PSH: HCK] Seg-2045201302 Rok:2054260491 Min:5792 Len:18	
23.463475 14.0.0.2	24.0.0.2	TCP	33865 > 1040 (HCK) Seq=2054250491 Hck=2049201520 Win=5840 Leri=0	
25.050625 mescal212	mescal111	TCP	bgp > 40022 [HEK] Seq4955561254 Rox495785337 Win47220 Len40	
26.666633 mescal111	nescal212	TOP	40022 > bgp [ACK] Seq=997053397 Ack=993961253 Win=16706 Len=0	
30.060453 mescal212	wescall11	TOP	bgn > 40022 [RDK] Seq=993361253 Ack=997853416 Win=17220 Len=0	
51,676755 sescal111	secol212	TCP	40022 > bgp [HDK] Seq199/053416 Ack/99/361272 Vin/16/06 Len/0	
35.070480 Nescal212	mescal111	TOP	bgp > 40022 [ADK] Seq=995961272 Ack=997653435 Win=17220 Len=0	
0,000000 mescal111	mesca1212	ETh.	R2294, IVE Pessage	
1,646738 Nescal212	mescal111	BGP	R2EPH_LVE_Pessage	
Here sequence number: 2003 Acknowledgement number: 2003 Header length: 32 bytes Flags: 0x0018 (PSH, ACK) Window size: 5752 Checksum: 0x80bc (correct) Options: (12 bytes) NOP	54260491			
NOP				
Time stamp: toval 18373	53123, tsecr 198514530			
ATTIC CONCELLE				
00 02 04 10 Bc 17 7a 24 5	4 96 7a 71 87 06 80 18		94	
15 a0 R0 hc 00 00 01 01 0 16 3a 01 06 00 10 00 06 1 30 05 00 00	8 0a 0a 73 85 83 05 df 1 03 65 23 28 56 00 01			
		33 32	V Reset Apply Data (data), 18 bytes	
	2	Manage al (24.0		The Arr 21 of
Con Car Child		annescal or his	source and a configuration and a construction and the source of the sour	

Scenario 2: MESCAL21'PCS sent a ERROR-PATH message to MESCAL11'PCS (value = 3)

Die Edit Capton Diegley Tool: Product P	- 5				capture» Ethereal
Imm Source Destination Produce1 Imp Source point 12 Source point 22	H			Tools	Edit Capture Display 1
7.000001 messili 7.000001 messili 6.40011 messili 6.40			Protocol *	Destination	Time Source
0.00000 metaliti metaliti 10 m be 34002 [RG] Ser 9100000 Aud 9200000 June 10 m be 3200 June 10 m be 32		tion port; 5050	UDP	71,0,0,2	7,038996 14,0,0,2
6.49831 Hersellil Hersellil Hersellil (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 3 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 4 kg/ [40] Jerselling And Starbing Munching Herselling (19) 4402 4 kg/ [40] Jerselling And Je		5430 Ack=397857653 Win=17220 Len=0	TCP	mescal111	0.000182 mescal212
2 4000 41000 2 5000 41000 2 1000 41000 2 1000 400 41000 200 5000 4000 4000 4000 4000 4000 40		7653 Ack=993969509 Win=16705 Len=0	TCP	mescal212	0.499511 mescal111
0.00000000000000000000000000000000000		25256712546 Ack#3557032439 W1/#5640 Lann48	TCP	24,0,0,2	2,302361 14,0,0,2
2.0002 44.002 2.40.002 107 5002 100 [00] Set22 1000 [00] Set25002045 44.000004545 Mundball Lundon 5.0007 excellation excellation in the set of		352455 H0475255712554 H1475752 L6970	TCP	14.0.0.2	2.502557 24.0.0.2
5.0002 menolitic menolitic menolitic produces and the second state of the second state	,	212934 Beka3257032457 Hind5040 Lend0	TCP	24.0.0.2	2 307527 14 0 0 2
5.0000 messili messili messili province setting settin		5509 Auk+997857672 Win+17220 Len+0	TEP	secol111	5,009017 aescal212
5.7055 14.0.02 44.0.02 14.0.02 TCP 100 3274 [Pis, K0] Sev23706828 And-32500858 Words United Several Se		7672 Ack=993965528 Win=16705 Len=0	TCP	nescal212	5,490679 mescal111
0.70000 4.0.0.2 14.0.0.2 TP TEX3 3 1000 [P0], 403 5ec250008140 Act-EX2000008 WurdBob Lend 0.70000 4.0.0.2 24.0.0.2 TP TEX3 3 1000 [P0], 403 5ec25008140 Act-EX2000088 WurdBob Lend 0.40001 4.0001111 weesall11 BP MERNAL DE Message 0.40001 4.0001111 weesall111 BP MERNAL DE Message 0.40001 4.00012 Wersage MERNAL DE Message		=3270309632 Ack=3260969409 Win=5792 Len=4	TCP	24.0.0.2	6.728971 14.0.0.2
5,2000 44:00.2 24:00.2 100 100:000 46:00111 46:000 40:000 40:00111 46:000 40:000 40:00111 46:000 40:000 40:00111 46:000 40:000 40:00111 46:000 40:000 40:001 40:0000 40:0000 40:0000 40:0000 40:0000 40:0000 40:0000 40:000		r3260969409 Ack::3270309636 Winr5840 Lenn4	TCP	14,0,0,2	6,730634 24,0,0,2
0.0000 mission in the second of the second o		303636 Ack=3260369413 Win=5792 Len=0	TCP	24,0.0.2	6.738764 14.0.0.2
<pre>i - 00001 + ===============================</pre>			10.0	nescal212	0,000000 mescal111
0.0000 H0011111 H001111 H00111 H00111 H001111 H00111 H001111 H00111 H00111 H001111 H001111 H001111 H001111 H001111 H00111 H0			207	Rescalill	0.450512 Mescal212
0.00000 Winshill WEXHIL WEXHILL WEXHILL 0.70000 Winshill WEXHILL WEXHILL WEXHILL			107	mercal212	5,008881 MMSCATTLA
en 7 (34 bytes en edre 54 bytes contered) west 11, 55: 0955045012632. Det 00000771150104 event 11, 55: 0955045012632. Det 0000071150104 matision Correl Protocol, see Parts 1040 (1940). Det Pert: 38121 (38121). Seg: 250702488. As: 256712584. Len; 18 Source sort: 30: 0(400) Detination part: 32:12 (3812) Detination part: 32:000000 Detination part: 32:0000000 Detination part: 32:000000 Detination part: 32:000000 Detination part: 32:000000 Detination part: 32:0000000 Detination part: 32:0000000 Detination part: 32:0000000 Detination part: 32:00000000 Detination part: 32:00000000 Detination part: 32:00000000000000000000000000000000000					
00 00 00 10 00 10 00 00 10		es: 205712594, Lan: 18	21:50:64 # Add: 14.0.0.2), Dat Port: 3612	bytes captured) 2345, Batt 00(20)27, 1 40,02 (24.0.02)7, 1 5 re Port; 1040 (1040 1221) 32257 47712534	me 7 (B4 bytes on wines B4 servet II. Scc 009504400 erret Fretzool. Src 1845-20 Source pert: 1040 (1040) Bestimation pert: 1040 (1040) Bestimation pert: 1041 (385 Segueno nuber: 325732203 Hadder lengths: 2 bytes Hadder lengths: 3 bytes Hadder len
이 약 책 10 분위 약 약 10 분위 약 위 10 명 15 약 책 10 분위 약 위 약 위 위 위 위 위 위 위 위 ··············					
00 (C (41 10 84 19 c) 22 54 77 c) 14 74 92 90 18					
				27	00 00 04 10 04 10 -0 00 F-
52 72 (14 (16 06 14 14 16 16 14 14 16 06 14 14 16 06 14 14 16 16 14 14 16 16 14 14 16 16 14 14 16 16 14 14 16 1 10 03 00 10 14 14 14 14 14 14 14 14 14 14 14 14 14				1 0a 0a 65 43 56 0b d	16 a0 95 0a 00 00 01 01 09
				03 65 34 12 02 00 0	ce 73 01 06 00 10 00 06 11
					0 03 00 00
			16-22		
Reset Apply Data (data), 16 bytes		Y Reset Apply Data (data), 16 bytes			
	II III III				a a ca ch a

 Scenario 3: MESCAL21' PCS sent a ERROR-PATH message to MESCAL11'PCS (value = 1)

Time So	re Direinu T	nolt			
Time S0	re Disbray T	January	laurun	los.	Ξe
4 5 555555 Ad	AURCE .	Destination	Protocol		
1 0,000000 14,	.0.0.2	71.0.0.2	LOP	Source port: 35303 Destination port: 5550	
2 6 259299 14	0.0.2	71.0.0.2	IDP	Source ports about addition ports (Source addition and the Source addition addita addition addition addition addition addition ad	
6 9,189783 14.	.0.0.2	71.0.0.2	LEP	Sauce port: 32014 Testination port: 5050	
1 14,999682 14.	.0.0.2	71.0.0.2	UDP	Source port: 32809 Jestination port: 5050	
4 16,529705 14.	.0.0.2	71.0.0.2	LOP	Source port; 32816 Iestination port; 5050	
\$ 1.019929 Her	scal212	mescal111	TOP	bgp > 40022 [#0X] Seq=993963210 Ack=997655373 Win=17220 Len=0	
6 2,147339 met	scal111	mescal212	TOP	40022 > bgp [RCK] Seq=937855373 Ack=993363229 Win=16706 Len=0	
7 4.383519 14.	.0.0.2	24.0.0.2	TOP	39952 > 1040 [PSH, RCK] Seg-24232210956 Rek-2441558300 Man-5040 Len-40	
8 4,385725 24,	.0.0.2	14.0,0,2	TUP	1040 > 35552 [RKK] Seq#2445555978 iick=2425219004 Win=5782 Len=0	
0 4 200245 14	0.0.2	24.0.0.2	TOP	1000 3 STORE [193], HAN] SEPTEMBERGER HAR ALTERNAL METTERS CONTRACT CONTRACT OF CONTRACT O	
2 6.005122 Hall	aca1912	assealitit	TOP	5 5 5 7 10 10 (10.1) September 200 (10.1) (10.1) (10.1) (10.1) (10.1)	
6 7.147414 mer	scal111	mennal212	TCP	00072 > hom [00X] Samm9903986880 Doi:::9903867288 Minet16306 (amo)	
8 11.029823 per	stca1212	nescal111	TOP	bap > 40022 [ADX] Sear993963248 Ack:997965411 Mini17220 Len:0	
0 12.147420 ser	scall11	nescal212	TOP	40022 > bgp [#0K] Seq=397055411 Ack=393365267 Win=16705 Len=0	
3 16,040802 mer	scal212	sescal111	TOP	bgs > 40022 [ACK] Sequ-933363267 Ack=997855430 Win=17220 Lev=0	
6 17.157499 ner	scalili	mescal212	TOP	40022 > bge [ADK] Seq=997855430 Ack=993963296 Win=16705 Len=0	
8 21,049743 Mps	sca1212	eccol111	TCP	bgp > 40022 [#KK] Set=953963266 Ack=997855449 Win=17220 Len=0	
2 1.019753 Met	scal111	mescal212	RP	PEEPALIVE Mossage	
5 2.14/242 Net	eca1212	Nescaill1	ECP.	FEETVALIVE ROSSAND	
1 6,024350 Met	ecal111	mescalill	8.8	Razing, Viz. Instance	
7 11 029672 mbi	1001212	mescaliti	DOP	KEEPA IVE Massage	
9 12.147330 per	scal212	secol111	ICP	REPAIR Parage	
2 16,040621 mm	titleon	mescal212	RCP	KEEPIL IVE Message	
Sequence number Next sequence Acknowledgemen Header length: B Flags: 0x0018 Mindow size:	per: 2441538070 e number: 244153 ent number: 2429 e: 82 bytes e: (PSH, ACK) 5732 15c7 (correct) bytes)	6996 219004			

Failure level	: None.
Remarks	: None.

10.3.2 TB_P3_FUNCT/QAGG

Only AS1, AS2 and AS3 are used to run this tests group. pSLSs are established between AS1 and AS2 and between AS1 and AS3 in order to extend Hard Solution Option of each domain.

- AS1offers a pSLS to AS2
 - MQC1: 0x68 bandwidth: 1Mbit
 - MQC0: 0x00 minimum bandwidth: 2Mbit maximum bandwidth: 3Mbit
- AS3 offers a pSLS to AS1
 - MQC1: 0x88 bandwidth: 1Mbit
 - MQC0: 0x00 minimum bandwidth: 2Mbit maximum bandwidth: 3Mbit

QoS capabilities of each AS are those described in 10.2.4

Test Reference	: TB_P3_FUNCT/QAGG/1
Test Purpose	: Check QoS aggregation operation
Procedure	: Launch a traffic analyzer that captures MESCAL11' eth1 and MESCAL21' eth1 traffic.
	Establish a PCP session between MESCAL11'PCS and MESCAL21'PCS. Establish a PCP session between MESCAL11'PCS and MESCAL31'PCS.

- Scenario 1: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 30. No QoS parameters are to be inserted in the request.
- Scenario 2: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 5Mbit. TTL is to be set to 30. No QoS parameters are to be inserted in the request.
- Scenario 3: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 30. Set minimum one-way delay to 45.
- Scenario 4: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 30. Set minimum one-way delay to 25.
- Scenario 5: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 30. Set maximum one-way delay to 80.
- Scenario 6: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 30. Set maximum one-way delay to 60.
- Scenario 7: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 30. Set average one-way delay to 40.
- Scenario 8: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 30. Set average one-way delay to 50.
- Scenario 9: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 30. Set jitter to 20.
- Scenario 10: When PCP session is established between MESCAL11'PCS and MESCAL21'PCS, configure MESCAL21'PCS to send a request for an LSP with a head-end-address 1.1.1.5 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 30. Set jitter one-way delay to 30.

Examine the messages exchanged between MESCAL11 and MESCAL21 and between MESCAL11 and MESCAL31.

Expected result :

Scenario 1: MESCAL11'PCS must forward this request to MESCAL31'PCS. This
request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1
in MC1 and a bandwidth of 1Mbit. TTL is to be set to 29. No QoS parameters are to
be inserted in the request.

- Scenario 2: MESCAL11's must send a ERROR-PATH message to MESCAL21'PCS (value = 1)
- Scenario 3: MESCAL11'PCS must forward this request to MESCAL31'PCS. This
 request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1
 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 29. Minimum one-way delay
 is set to a value less than 35.
- Scenario 4: MESCAL11's must send a ERROR-PATH message to MESCAL21'PCS (value = 1)
- Scenario 5: MESCAL11'PCS must forward this request to MESCAL31'PCS. This request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 29. Maximum one-way delay is set to a value less than 30.
- Scenario 6: MESCAL11's must send a ERROR-PATH message to MESCAL21'PCS (value = 1)
- Scenario 7: MESCAL11's must send a ERROR-PATH message to MESCAL21'PCS (value = 1)
- Scenario 8: MESCAL11'PCS must forward this request to MESCAL31'PCS. This
 request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1
 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 29. Average one-way delay is
 set to a value less than 20.
- Scenario 9: MESCAL11's must send a ERROR-PATH message to MESCAL21'PCS (value = 1)
- Scenario 10: MESCAL11'PCS must forward this request to MESCAL31'PCS. This request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is to be set to 29. Jitter is set to a value less than 10.

Execution date : 22/04/05

- Result
- : Obtained results are as follows:
- Scenario 1: MESCAL11'PCS forwards the request to MESCAL31'PCS. This request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is set to 29. No QoS parameters are inserted in the request.

Elle Edit Capture Display	1 Marcala			
	Taou			Het
No. Time Source	Destination	Protoco	ul * Indo	
90 70,022457 1,1,1,1	1,1,1,2	TCP	Byg 7 55256 (MCK) Septimized MCK/ID/052959 MIN/20560 Lenn	
94 75 022206 1 1 1 1	1.1.1.2	TCP	hup > 5256 (aCC) Sept101305288 Rev101805578 Micro20200 Level	
96 76,229859 1.1.1.2	1.1.1.1	TCP	53256 > bup [40X] Sep=1018025578 Ack=1011809368 Win=14265 Len=0	
97 76,239357 14,0,0,2	34,0,0,2	TOP	51571 > 1040 (P94, ACK) Segnill/760964 Acknill/2202204 Min/5040 Levn40	101
98 76,239507 34,0.0.2	14,0.0.2	TCP	1040 > 51571 (ADK) Seq=3122922394 Ack=3117750112 Win=5792 Lence	10
99 76,562162 34,0,0,2	14,0,0,2	TCP	1040 > 51571 (PSH, AOX) Seg=3122922394 Ack=3117761012 Wine5792 Len=42	
100 76,562528 14,0.0,2	34,0,0,2	TCP	51571 > 1040 [ROK] Seqr5117762012 Ackx5122822436 Winn5040 Lenn0	
101 77,296405 14,0,0,2	54,0,0,2	TCP	SISVI > 1040 [F94, HLK] Segesti //si012 HKRS12262(4) HINDER Levils	
104 80 022285 1 1 1 1	1112	TOP	http://www.com/article	
106 81,238807 1.1.1.2	1.1.1.1	TCP	51256 > bop [ACK] Sept1010025537 Ackx1011009387 Winst4265 Lenzt	
1 0,000000 1,1,1,2	1,1,1,1	RP	KEEPIK_IVE Nettage	
3 1.079763 1.1.1.1	1,1,1,2	30*	KEEPRLIVE Message	
5 5,000150 1.1.1.2	1.1.1.1	BCP	KEEPIK IVE Hessage	
7 6.090679 1.1.1.1	1,1,1,2	ICP	KEEPVILIVE Measage	
17 10,000336 1.1.1.2	1.1.1.1	102	KIIPR, IVE Records	
1				
				4
		, bear	Ivij Percej (zopij) Cota jatoj, 48 tojes	12
		l pina	(♥) [Rese](Apply)] Cuto (Solio), 48 5/64.	(P

 Scenario 2: MESCAL11' PCS sent a ERROR-PATH message to MESCAL21'PCS (value = 1)

Image: Service (prive) Toric Private (prive) Toric Private (prive) Toric Private (prive) Private (prive) <thp< th=""><th>- 6</th><th></th><th></th><th></th><th></th><th>Ethernal</th><th><captures +="" 1<="" th=""></captures></th></thp<>	- 6					Ethernal	<captures +="" 1<="" th=""></captures>
Image Description Protocol Unit 0. Links 14.0.0.2 100 4000 > 1000 000000000000000000000000000	Hel				ools	apture Display To	le Edt Cap
0 0		 Contract Contract Contract Contract		Protocol , Info	Destination	Source	o. Time
1 0 2010 2010 2010 2010 2010 2010 2010		 A DOUR ADD ALL STOLD AND A	the form and compared to the	TCD 2040 7 40	24,0,0,2	14.0.0.1	5 0.00000
11 125000 1.1.1.5 1.1.1.6 1.1.1.6 10 402.2 bg (0.3 bey0305544 8.6.9305555 0.0.001 and 0.0.001 bg (0.3 bey0305544 8.6.9305555 0.0.001 bg (0.3 bey0305545 0.0.001 bg (0.3 bey0305555 0.0.		2945599 Wind5792 Level	E0 [002] Seco3100585910 de	TCP 4040 > 45	24.0.0.2	14.0.0.2	7 0.210031
14 15898 11.1.6 1 11.1.8 TO 4 00 - 400 100 100 100 100 100 100 100 100 100		1347 Win=16705 Len=0	@ [AO(] Seg-998153443 Ack-	TOP 40022 > 8	1.1.1.5	1.1.1.5	11 1,890007
15 5.2003 14.0.0.2 34.0.0.2 14.0.0.2 10 40.0.2 10 10 10 10 10 10 50 500 [Pb].003 500 [Pb].003 500 [Pb].005 50		3452 Wins17220 Lens0	2 [ACK] Seu-994261347 Ack-	TCP bitp > 400	1.1.1.5	1.1.1.6	14 3,983089
16 1.2003 1.40.0.2 2.00.02 2.0		k=3120545512 Win=5840 Len=48	140 [PSH, RCK] Seg=30978466	TOP 49160 > 1	14.0.0.2	24,0,0,2	15 5,238367
19 5.570 19 5.072 19		7845736 Win=5792 Len=0	150 [ACK] Seq=3120545912 Ac	TOP 1040 > 45	24.0.0.2	14.0.0.2	16 5.238513
19 5.2079 54.0.0.2 14.0.0.2 17 41.0.0.2 17 41.0.0.2 17 40.00 24.00 24.00 10.01 Perf03104 First Res1500 94.00540 Level (1994) 14.00 1		\$=3097845735 Win=5792 Len=18	60 [PSH, ACK] Seq=31205455	TCP 1040 > 45	24,0,0,2	14.0.0.2	17 5,247264
00 General Lilio 1 Lilio 10 Hereita 10 Herei		0545930 Win=5840 Len=0	140 (ACK) Seq=3097946736 Ac	TCP 49160 > 1	14.0.0.2	24.0.0.2	18 5,289787
2 1993 0 1993 0 1115 1115 1115 0 109 0 199		1366 Win=16705 Len=0	₽ [ADK] 5eq=998153462 Ack=	TOP 40022 > t	1.1.1.6	1,1,1,5	20 6.890058
0 1 1.0000 1.1113 1.1116 0.000 1.000		3481 Win=17220 Len=0	12 [ACK] Seq=994261366 Ack=	TOP bgp > 400	1.1.1.5	1,1,1,6	22 8,993105
0 1 4 4 1 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		1385 Win=16705 Len=0	p [ACK] Seq=998153481 Ack=	TUP 40022 > 6	1.1.1.6	5 1.1.1.5	25 11,900235
14 15.15.25 11.11.6 11.11.5 107 50 4002 1002 1002 1002 1002 1002 1002 100		1404 Hug-12705 Land	ERCAJ 000-004201000 Hok- 10073 Chargedet2000 Ark-	TTP 40002 5 4	1.1.1.5	7 1.1.1.0 7 1 1 1 1 E	21 15 041570
0 0		2016 Mart 2000 Lenno	p [HUN] 500-550155000 HON-	TOP 40022 7 0	1.1.1.0	1.1.1.9	14 10.012072
26 20.0000 14.0.02 24.0.02 10.000000 10.000		10113 WIN-1/200 Len-0	2 [RUN] 000-004201404 RUN-	TTP 1040 \ 51	14.0.0.2	24.0.0.2	24 13,016303 25 30 000130
* 00.000 10101 10101 10100 10100 10100 10100 10100 101000 101000000		9039575 kini5840 Leni0	40 [B[K] Sec.3125652748 Br	TCP 51570.3.1	24.0.0.2	5 14 0.0.2	35 20.000316
Tex 12 (M 1946 on virs. H 344 de 2004) Herret II. Str (0005)215504. Holden (1940) Herret II. Str (0005)215504. Holden (1940) Herret II. Str (0005)21504. Holden (1940) Herret II. Str (0005) Herret II. Str (0		A TRADUTOR THE DESIGN A STORE	ten frent onu'l rain teneroom	TOD 0400071	Calotore	14000	
	I				70 50 45 53 40 00 10 0a 0c 45 95 00 05 03 50 47 44 03 00 00	10 c0 00 b0 ff c0 7 7a 00 00 01 01 00 0 00 00 10 00 c0 23 0	00 02 04 11 16 40 c2 7 05 40 20 0
e Reset Apply Data (data), 16 bytes		Apply Data (data), 10 bytes	✓ Re				e.
	E6 AN 22 143	continues . Ethernal	Meneral PMESCALLA	AND	1	ADD	100

Scenario 3: MESCAL11'PCS forwards the request to MESCAL31'PCS. This
request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1
in MC1 and a bandwidth of 1Mbit. TTL is set to 29. Minimum one-way delay is set
to a value less than 35.

Capture» - Ethereal					- T X
Ese Egt Cabona Disbrak	Toels				Heth
No. Time Source	Destination	Protocol	. Info		
1 0,000000 1.1.1.2	1.1.1.1	207	ALDHLINE Ressage		
7 5,000536 1,1,1,2	1.1.1.1	BCP	KEEPIK IVE Nessaue		
13 7,562935 1,1,1,1	1.1.1.2	IGP	KEEPILIVE Message		
15 10,010292 1,1,1,2	1.1.1.1	BCP	KEEPGLIVE Message		
2 0,000076 1.1.1.1	1.1.1.2	TOP	bgp > 53256 [4OK] Seqr1011012465 Ack=1010020634 Win=20960 Lenn0		
4 2,554095 1.1.1.2	1,1,1,1	TCP	53256 > bgp [40X] Seqn1010020634 Ack=1011812484 Winn14265 Lenn0		
5 4 017217 24 0 0 2	14.0.0.2	100	1040 N 51221 10221 Contractions and address of the State Sta		
8 5.000619 1.1.1.1	1.1.1.2	TOP	hop > 5256 [40X] Seption1012484 Acket010020215 Wing20960 Lengt		
9 5,405573 34,0,0,2	14,0,0,2	TOP	1040 > 51571 [PSH, ADK] Seq=3122922596 Ack=3117761256 Wine5792 Lene42		
10 5,495760 14,0,0,2	34,0,0,2	109	51571 > 1040 [ADK] Seqn3117761256 Ackn3122922638 Winn5840 Lenn0		
11 5,824772 14,0,0,2	34.0.0.2	TCP	51571 > 1040 [PSH, ADK] Seq=3117761256 Ack=3122922638 Win=5840 Len=18		
12 5.824952 34.0.0.2	14.0.0,2	1DP	1040 > 51571 [#CK] Seq=3122522538 Ack=3117761274 Win=5792 Len=0		
14 7.9638/9 1.1.1.2	1.1.1.1	10*	2728 > 88 (Hor) sedananceurs How narrans Annarges rewa		
E France 5 (118 butes on wire.	118 butes captured)		10000		
B Ethernet 11, Src; 00;90;27;5	7:58:52, Det; 00:60:00:1	OctOd:Fa			
E Internet Protocol, Src Addr:	14.0.0.2 (14.0.0.2), h	let Addr: 34.0.0	2 (34,0,0,2)		
S Transmission Control Protoco	1. See Port: 51571 (515)	71), But Port: :	1040 (1040), Seq: 3117761204, Ack: 3122322596, Len: 52		
Data (52 bytes)					
()					•
040 55 20 25 04 00 50 24 02	00 02 00 01 00 06 21 03	Y BOOM			
050 E9 da 56 5c 00 01 00 00	02 01 01 01 01 02 16 16	TWO IS NOT			
060 1F 1F 03 03 03 01 1F 1F	1F 1F 01 00 07 22 11 00				10
070 01 24 00 14 00 00		.s			
mer internet	a		[V] [Reset] [Apply] Data (data), 52 bytes		
🛎 🗖 🤪 🖗 🚳	e)	Capture .	> - Ethereal	0	Fri Apr 22, 14:28
	•	and a second con			A DE DE DE

 Scenario 4: MESCAL11' PCS sent a ERROR-PATH message to MESCAL21'PCS (value = 1)

in Park Combury Disaster				
a Ens Zohava Fishesh	Tools			Be
Time Source	Destination	Protocol *	infa	
5 0,210008 24,0,0,2	\$4,0,0,2	10	AMER 3 2040 DAME HOLI PERSONANANA MERUITAPANTE REVERSES FROM	
E 0.210454 14.0.0.2	24,0,0,2	TOP	1040 > 43160 [ACK] Seqr3129546116 Ackx3097846952 Mixx5792 Lexn0	
8 0,176767 1,1,1,6	1,1,1,5	107	bgp > 40122 (40X) Seq1294254054 90X/201256175 401417220 Level	
6 5 30500 11115	11110	100	http:// applications.com/application/appli	
6 5.747699 24.0.0.2	14.0.0.2	TOP	#9160 > tian (POH, aCVT Sear3057046942 availt20546115 MinefBat Land	
7 5,748235 14,0,0,2	24,0.0.2	10P	1040 > 49160 [ADX] Seu-3120546116 Rek-3097847000 Win-5792 Leni0	
8 5,886424 14,0,0,2	24,0.0,2	10*	1048 3 49160 [PSH, 40X] Seq=3120546116 Auk=3097647000 Mp=5782 Lanc18	
19 5.919817 24.0.0.2	14.0.0.2	TOP	49160 > 1040 [ACK] Seg-3097847000 Auk+3120546134 Wire-5840 Levr0	
1 7.310079 1.1.1.5	1.1.1.6	TOP	40022 > top [HOV] Segr998156199 Ack=994264102 Win=16705 Len=0	
4 10,096539 1,1,1,6	1.1.1.5	TOP	bgp > 40022 [403] begrepsedating Hokrepsedating Howel/220 Level	
1 15 ANDRE 1, 1 4 5	2 4 5 8	TOP	http://www.com/com/com/com/com/com/com/com/com/com/	
3 17,310081 1.1.1.5	1.1.1.6	TOP	40022 > hop [403] Sect000562% Ack:094264140 Min:1676 Level	
20.000227 24.0.0.2	14.0.0.2	TCP	1040 > 52570 [PSH, ACK] Sept1109030815 Acks1125052492 Minu5792 Lanu4	
5 20,020217 14,0,0,2	24,0.0.2	TCP	51570 > 5040 [PSH, 40X] Sep-3125652432 Ack+3109030029 Min=5040 Lenn4	
6 20,029421 24,0,0,2	\$4,0,0,2	TOP	1040 > 50570 [ACK] Segr8109030819 Ack+8125052496 Wine5792 Level	
	1.4.4 (0) (0) 44 (0) (0) 44 (0) (0) 44 (0) (0) 44 (0) (0) 44 (0) (0) 44 (0) (0) 44 (0) (0) 44 (0) (0) (0) (0) (0) (0) (0) (0) (0) (0)			 1
			المعدور (مومور) المعدي (معدور - 10 topics) المعدور (مومور) المعدي (معدور - 10 topics)	 1

Scenario 5: MESCAL11'PCS forwards the request to MESCAL31'PCS. This
request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1
in MC1 and a bandwidth of 1Mbit. TTL is set to 29. Maximum one-way delay is set
to a value less than 30.

Help
12
1
P
-
1
E 12 D D Fri Apr 22, 14:34
Sector Manager

 Scenario 6: MESCAL11' PCS sent a ERROR-PATH message to MESCAL21'PCS (value = 1)

The second se				
Time Source	Destination	Protocol	Info	
6 2,553024 14,0,0,2	24,0,0,2	TOP	51570 > 1040 (PSH, ACK) Seg-3129652544 Ack-3109033871 Win=5940 Len=4	
2,553184 24,0.0,2	14.0.0.2	TOP	1040 > 51570 [ACK] Seq=3109038871 Ack=3125652548 Win=5792 Len=0	
2,703209 14,0,0,2	24,0,0,2	TOP	1040 > 49160 [PSH, ACK] Seqt3120546228 Ack::3097047124 Win::5792 Lenn4	
2,703401 24.0.0.2	14.0.0.2	TCP	49160 > 1040 [AEK] Seq=3097847124 Ack=3120546232 Win=5840 Len=0	
2,750078 24.0.0.2	14.0.0.2	TOP	49160 > 1040 [PSH, RCK] Seq=8097847124 Rck=8120646232 Vin=8840 Len=4	
2.730256 14.0.0.2	24.0.0.2	TCP	1040 > 49160 [ACK] 5eq=3120546232 Ack=3097847128 Win=5792 Len=0	
5,455/65 1,1,1,6	1.1.1.5	102	bgp > 60022 [HLX] 5eqT356260128 HCKT358157285 Wint17220 Lento 60000 5 T000 [BD9] 0002 Comp2002042020 044-2000642220 Mine6040 Lento	
1.506240 24.0.0.2	24.0.0.2	TOP	43160 / 1040 [P34, H04] Segrad/0/04/120 H04/312/046222 MIN5040 Lenn40	
815300 14.0.0.2	24.0.0.2	TOP	1040 3 49160 [PSH, 60X] Security/646732 8ct/307047176 819-0722 Length	
.949995 24.0.0.2	14.0.0.2	TOP	49160 > 1040 [AEK] 5ept3097847176 Ack+3120546250 Wint5840 Lent0	
,000089 1.1.1.5	1.1.1.6	TOP	40022 > bop [ADK] Segr398157243 Apt=394265147 Win=16705 Len=0	
443021 1.1.1.6	1.1.1.5	TOP	bgp > 40022 [ACK] Seq=994265147 Ack=998157252 Win=17220 Len=0	
0.000215 1.1.1.5	1.1.1.6	TCP	40022 > bgp [ACK] Seq=998157262 Ack=994265166 Win=16705 Len=0	
2,495030 14,0,0,2	71,0,0,2	UIP	Source port: 32824 Destination port: 5050	
0.062918 14.0.0.2	71.0.0.2	UIP	Source port: 32822 Destination port: 5050	
1,292906 14.0.0.2	71.0.0.2	UDP	Source port: 32822 Destination port: 9050	
	111 111 111 111 115 55 回 100 11 11 回 助 100 11 11 15 75 105 55 10 11 10 10 10 10 10 10 10 10 10 10 10 1			

 Scenario 7: MESCAL11'PCS sent a ERROR-PATH message to MESCAL21'PCS (value = 1)

Edit Canture Display				
En Tobuc Fobul	Destruction	Destacat		10
n to 020040 1 1 1 6	1115	N.P.	INFORT DF Rename	
2 0,000058 1.1.1.5	1.1.1.6	TOP	40022 > top [ACX] Seqr998257585 Ack:994265489 Wunr16705 Lenn0	
6 3,541202 1.1.1.6	1.1.1.5	TOP	bgp > 40022 [AOK] Seqr994265489 Ack=998157604 Wire=17220 Lev=0	
9 5,010263 1.1.1.5	1.1.1.6	109	40022 > bge [#OK] Seqr998157604 #ck+994265508 Wprx15705 Lenv0	
0 5,055748 24,0,0,2	14,0,0,2	109	49160 > 1040 [PSH, ACK] Sep-3097847100 Ack=3120546282 Win=5040 Lenn48	
1 5.055944 14.0.0.2	24,0,0,2	TOP	1040 > 49160 [ADK] Seq=3120546262 Ack=3097847236 Vin=5732 Lam0	
5 5, 720125 24, 0, 0, 2	14.0.0.2	112	2010 2 5040 [202] Secon00704728; Ack-312054520; March10 Land	
4 7.440121 24.0.0.2	14.0.0.2	TOP	1040 > 51520 [PSH, 40(1 Senal2090/20027 0rks/20/965/964 Minu5792 Lenal	
5 7,440308 14,0,0,2	24.0.0.2	TOP	51570 > 1040 [ACK] Segr3125052564 Ack+3109030891 Winn5040 Lenvo	
6 7,470995 14,0,0,2	24,0,0,2	TOP	51570 > 1040 [P9H, ACK] Sequ3125652564 Ack=3109038891 W1rrs5840 Lenn4	
7 7,471390 24,0,0,2	14,0,0,2	TOP	1040 > 51570 [ACK] Seq=3109038891 Ack×3125652568 Win=5792 Lenv0	
9 8,551014 1.1.1.6	1.1.1.5	TOP	bgp > 40022 [4OK] Seqr994295508 Ack:r998157623 Worr17220 Lenr0	
10,020097 1,1,1,5	1.1.1.6	TOP	40022 > bgp [ACK] Seq=998057623 Ack=994265527 Wire=16705 Len=0	
\$ 2,981161 14.0.0.2	71,0,0,2	LOP	Source port: X322 Destination port: 5050	
3,211019 14,0,0,2	71,0,0,2	100	Source port: 52822 Destination port: 5050	
*108888# 14101012	1844444		mark hard more more and and 2000	
			2011	
			nterse (♥) [Reced](Apply)] Critis (Statio), 18 bytes	

Scenario 8: MESCAL11'PCS forwards the request to MESCAL31'PCS. This request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is set to 29. Average one-way delay is set to a value less than 20.

Edit Canhina Divelou	Tools			
se Eon Capture Display	Tools			Ust
0. Time Source	Destination	Protocol .	no remo har Maaaaa	
3 0.792624 1.1.1.2	1.1.1.1	BCP	EPHLIVE Neurope	
11 5.010087 1.1.1.1	1,1,1,2	BOP	EEPVLIVE Message	
13 5.802325 1.1.1.2	1,1,1,1	BGP	EEPKLIVE Hessage	
2 0,000179 1,1,1,2	1,1,1,1	TOP	3256 > bgp [ACK] Seq11018032874 Ack:1011816664 Win:14265 Len:0	
5 2 374252 14 0 0 2	1,1,1,2	102	p 2 55256 [HUK] 50041011816664 HOL41018052855 Minw20050 Lenv0 1571 h 1040 [Poil: 402] Securit17781594 Ackes1222222292 Minw2040 Lenv52	
6 2,374435 34,0,0,2	14,0,0,2	TOP	040 > 51571 [ADX] Seq#3122922892 Ack#3117761636 Win#5792 Len#0	
7 2,734559 34,0,0,2	14,0,0,2	TOP	040 > 51571 [PSH, MCK] Seq#3122922892 Ack#3117761636 Win#5732 Len#42	
8 2.734753 14.0.0.2	34.0.0.2	TCP	1571 > 1040 (ACK) Seq=3117761636 Ack=3122922934 Win=5840 Lev=0	
10 7 201721 74 0 0 2	34.0.0.2	TCP	15/1 / 1040 (P3H, HUK) SEGTELL/161636 HOKTEL22222034 VINTOHV LENTID	
12 5.010262 1.1.1.2	1.1.1.1	TOP	3256 > bup [ACK] Seg=1019032993 Ack=1011816683 Win=14265 Len=0	
14 5.002335 1.1.1.1	1,1,1,2	TOP	30 > 53256 [ACK] Seq=1011816883 Ack=1018032912 Win=20960 Len=0	
Frame 5 (118 bytes on wire, Ethernet II, Sret 0098/274 Internet Protocol, Sro Rider Transatision Control Protoco 3558 (Sologies)	118 bytes captured) 57:50:52, Det: 00:50:00 1 4.0,0,2 (14.0,0,2) 1, See Poet: 51571 (52	0:0c:0d:F& Bat Add-: 34,0.0.; 571). Dat Port: 10	34,0,0,2) (1940). Seq: 3117781584, Rei: 3122822892, Len: 52	
040 06 cc 01 04 00 50 15 02 050 59 46 90 62 00 01 00 00 050 17 17 05 03 03 01 17 17 070 01 26 00 14 00 00	00 02 00 01 00 06 21 0 04 01 01 01 02 17 1 1F 1F 01 00 07 22 11 0	03 14 00		•
140 06 cc 01 04 00 50 15 02 150 29 eb 90 22 00 01 00 00 150 27 17 03 05 03 00 01 14 17 170 01 26 00 14 00 00 Terr	00 02 00 01 00 06 21 0 04 01 01 01 01 02 17 1 17 17 01 00 07 22 11 0	030	VVVVV	•
040 06 cc 01 04 00 30 10 02 50 32 46 50 42 00 01 00 01 50 17 17 6 03 60 10 10 10 50 17 17 6 03 60 11 10 10 50 17 17 6 03 60 11 10 10 50 17 17 6 00 14 00 cc 18er			Reset Apply Data (data), 52 bytes	•

 Scenario 9: MESCAL11' PCS send a ERROR-PATH message to MESCAL21'PCS (value = 1)

a second to account to account of				
East National Fisherst	- Incomentary	Inches	1.2	-
12.7.562419 . 34.0.0.2	teasados	Protector	alter a state form, while the initiation and involutional investigation and	
13 7.542614 14.0.0.2	24.0.0.2	TOP	1040 > 49560 [602] Secrit20546404 9ckx3097047468 Minr6792 Lecco	
14 9,472754 14,0,0,2	24.0.0.2	109	1040 > 49550 (PSH, ADK) Seg-3120545404 Ack-3097547468 Mini-5792 Lengt8	
15 8,505462 24,0,0,2	14.0.0.2	109	43160 3 1040 [ADC] Seq=3097847468 Ack+3120546422 Mine5640 Lenvi0	
17 10,010614 1.1.1.6	1.1.1.5	TOP	bgp > 40022 [ADX] Seqr/994266762 Ack:/990150077 Word27220 Levr/0	
19 11.065768 1.1.1.5	1.1.1.6	TOP	40022 > bge [ADX] Seq=998158877 Ack=994266781 Win=16705 Len=0	
22 15,019900 1.1.1.6	1.1.1.5	109	bgp > 40022 [ACK] Seqr994256781 Ackr996158896 Wurd7220 Levr0	
4 16.075795 1.1.1.5	1.1.1.6	TOP	40022 > bge [ADX] Seq=990150096 Ack=994266000 Waxd6705 Levr0	
5 18,275911 24.0.0.2	14.0.0.2	1CP	1040 > 51570 (PSH, ACK) Seq=3109038955 Ack=3125652632 Min=5792 Len=4	
6 10,299054 14,0,0,2	24,0,0,2	109	51570 > 1040 [PSH, 4CK] Sept127802032 Acks100003999 Minr5040 Lenne	
7 18, 900040 24, 0, 0, 2	14.0.0.2	102	1/40 / 101/0 (MA) September / August /	
1 20,023000 1.1.1.5	1.1.1.0	11.7	big / which has been set as a set of the set	
5 75 OPSTET 1 1 1 4 6	1 4 1 5	179	how a suppression of the second secon	
8 26.096839 1.1.1.5	1.1.1.6	TOP	40022 > hop TAXX1 Sept/99059854 doku/942962888 Wow15705 Level	
1 10.049806 1.1.1.6	1.1.1.5	TOP	bap > 40027 [AIX] Segr@94288888 Ack:090150953 Word2220 Levr0	
5 31,095675 1,1,1,5	1.1.1.6	109	40022 > bije [ACK] Seg=99E158951 Ack=994256857 Wux=16705 Level	
	2 (4 10) 45 (6 × 10 1) 2 (4 10) 45 (6 × 10 1) 4 (6 × 11 - 2)	.d., f		
			···· ↓ Precel (Apply) Delta (cotto), 18 trytes	

Scenario 10: MESCAL11'PCS forwards this request to MESCAL31'PCS. This request is for an LSP with a head-end-address 1.1.1.2 and tail-end-address in 3.3.3.1 in MC1 and a bandwidth of 1Mbit. TTL is set to 29. Jitter is set to a value less than 10.



Failure level: None.Remarks: None.

10.3.3TB_P3_FUNCT/RESAV

Sub-group preamble

The established pSLSs between ASs concern only MC1 and Best-effort planes.

QoS guarantees of 1-QC implementing MC1 in each AS are configured as follows:
Attributes	Priority	I-QC11	l-QC21	l-QC31	I-QC41	I-QC51	I-QC61	I-QC71	I-QC81
Average-owd	1	10	13	12	15	8	9	10	12
Jitter	2	1	2	3	4	2	3	2	1
Max-owd	3	15	19	18	23	13	14	15	18
Loss-rate	4	70	100	110	185	80	90	80	75
Reserved-rate	5	18	22	23	35	24	33	24	21
Available- rate	6	20	28		39		35		23
Min-owd	7	7		9	8				

Table 38: Local QoS Class Characteristics

Psls	PCE1	PCE2	PCE3	PCE4	PCE5	PCE6	PCE7	PCE8
PCE1		15	15					
PCE2	15			20				
PCE3	15			10+10				
PCE4		20	10+10		20	15+15		
PCE5				20			18	
PCE6				15+15			20	20
PCE7					18	20		
PCE8						20		

Table 39: M	laximum	bandwidth	allowed	for	MC1
-------------	---------	-----------	---------	-----	-----

The PCSID of AS8's PCE is set to 81.81.81.81.

Test Reference : TB_P3_FUNCT/RESAV/1

Test Purpose : Check resource reservation

Procedure : Configure three path computation orders to be sent by PCE1 (AS1). The tail-end PCSID of these requests is set to 81.81.81(AS8) and the QoS constraints are: loss-rate=550, max-owd=90, average-owd=60, jitter=13. The requested bandwidths are 8, 8 and 6 Mb/s.

Expected result : The result of each request must be as follows:

- The first path computation order must be successfully achieved. The computed path must contain MESCAL11, MESCAL31, MESCAL41, MESCAL61 and MESCAL81.
- The second path computation order must be successfully achieved. The computed path must contain MESCAL11, MESCAL21, MESCAL42, MESCAL43, MESCAL61 and MESCAL81.
- The third path computation order must fail because of a lack or resources available in the inter-domain link between MESCAL61 and MESCAL81. AS6 must receive 4 orders from AS4 and cancel all of them because of the lack of resources available (REASON-CODE=1). No orders must be sent to MESCAL51 because of the requested QoS constraints.

Execution date : 27/05/05

Result

- : Obtained results are as follows:
 - The first path computation order is successfully achieved and the computed path contains MESCAL11, MESCAL31, MESCAL41, MESCAL61 and MESCAL81.



• The second path computation order is successfully achieved. The computed path contains MESCAL11, MESCAL21, MESCAL42, MESCAL43, MESCAL61 and MESCAL81.



• The third path computation order has failed because of a lack or resources available in the inter-domain link between MESCAL61 and MESCAL81. AS6 has

received 4 orders from AS4 and has canceled all of them because of the lack of resources available (REASON-CODE=1). No order has been sent to MESCAL51.



Failure level : None. *Remarks* : For th

: For the third order, there is no resource available between AS6 and AS8, and no suborder with the requested bandwidth could be satisfied. It could be useful to have a more precise description of the error in the path-error message and to use this information to exclude for instance AS6 in the next requests. In this case AS4 would have sent only one request to AS6 and AS1 would not have sent any request to AS2 because whatever the path is, the order would be rejected by AS6.

Test Reference : TB_P3_FUNCT/RESAV/2

Test Purpose : Check Resource release when an order is canceled by a requestor.

- *Procedure* : The following tasks must be done:
 - A first computation order must be sent by PCE1 with the tail-end PCSID set to 81.81.81.81. The requested bandwidth is 10 Mb/s.
 - A second computation order must be sent by PCE1 with the tail-end PCSID set to 81.81.81.81. The requested bandwidth is 15 Mb/s.
 - The first order must be canceled by PCE1.
 - An order with the same characteristics as the second one must be sent.

Expected result : The result of each task must be:

- The first order must be successfully computed and resources pre-reserved along the computed path.
- The second order must fail because there is no resource available between AS6 and AS8.
- Cancel Messages must be propagated along the path and resources must be released.
- The last order must be successfully computed and resources pre-reserved along the computed path.

Execution date : 30/05/05

	Result	: Obtained results are	as follows:
--	--------	------------------------	-------------

• The first order has been successfully computed and resources have been prereserved along the computed path.



• The second order has failed because there is no resource available between AS6 and AS8.



 Cancel Messages have been propagated along the path and resources have been released.



• The last order has been successfully computed and resources have been prereserved along the computed path.



Failure level	: None.
Kemarks	: None
Test Reference	: TB_P3_FUNCT/RESAV/3
Test Purpose	: Check if an expired order is correctly canceled along the path.
Procedure	: The PCE of the AS4 must be configured to set the validy date of computed order to 1 minute after the computation date. The other PCEs are configured to set the validity date to 1 hour after the computation date. An order must be sent by PCE1 with the tail-

end PCSID set to 81.81.81.81 and the requested bandwidth set to 1 Mb/s. Wait until the validity date received in the path-response message expired.

Expected result : The order must be successfully computed and the validity date must be set to 1 minute after the computation date. When this validity date expires, the order must be canceled in each PCE. The order could be canceled by the local PCE because the validity date expired or by a neighbor PCE.

Execution date : 30/05/05

Result : The order has been successfully computed and the validity date was set to 1 minute after the computation date. 1 minute later, the validity date expired and the AS4 sent cancel messages which have been then propagated along the computed path.

Failure level : None.

Remarks : If the AS4 had not sent cancel messages to its neighbours concerned by the order, each PCE would have canceled locally the order because each PCE has a validity date associated to this order.